Stability of Nonlinear Systems with Parameter Uncertainty

by

Michael Walter Fisher

A dissertation submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy (Electrical Engineering: Systems) in The University of Michigan 2020

Doctoral Committee:

Professor Ian Hiskens, Chair Professor Christopher DeMarco Associate Professor Shai Revzen Professor Ralf Spatzier Michael W. Fisher fishermw@umich.edu ORCID iD: 0000-0002-4640-5738

© Michael W. Fisher 2020 All Rights Reserved

DEDICATION

To my family.

ACKNOWLEDGEMENTS

I would like to begin by thanking my family, to whom this thesis is dedicated, for always being there for me, encouraging me when progress was slow, constantly supporting me, and giving me much-needed time to recharge and have some fun during my brief vacations. More specifically, I would like to thank my wife Navdeep for putting up with my bad jokes, always brightening my mood, endless optimism, excellent Indian food, and so much love and happiness. I want to thank my parents Carol and Dan, and my sisters Alie and Emily, for making me feel so at home no matter where we see each other, always believing in me, baking the best gluten free banana bread, unlimited love and affection, and countless hikes, bikes, movies, and adventures. This thesis would not have been possible without all of their love and support.

From the professional side, first and foremost I would like to thank my PhD adviser Professor Hiskens. He taught me how to develop theoretical motivation and algorithms in generality, and then apply them to solve practical problems in power systems - a central theme of this thesis. Even when progress was initially slow, he gave me tremendous flexibility to pursue the research in the directions I found most promising, ultimately resulting in the theory that led to the design of the central algorithms of this thesis. Both the breadth of the research - from technical mathematical theory to efficient numerical algorithms to practical power system applications - and the flexibility I was afforded are incredibly rare to receive from a PhD adviser, and I am tremendously grateful to Professor Hiskens for providing them. On a more personal note, Professor Hiskens taught me never to take life too seriously, and his constant jokes and childish humor never failed to keep me in good spirits. I would also like to acknowledge the members of my thesis committee for their contributions. Professor Spatzier generously devoted a lot of his time towards helping me with some of the more technical proofs in the thesis. Without his kindness and encouragement, the theoretical work on region of attraction continuity, which forms the technical backbone of this thesis, would not have been possible. I would also like to thank Professor Wouter van Limbeek for his help in a similar vein on the most technical proofs in this work. Professor Revzen helped

me extend this theory to a class of discontinuous vector fields: those with event-selected discontinuities. Furthermore, I appreciate his many questions and suggestions throughout this process, which led to the proof of differentiability of the inverse trajectory sensitivity function, among other results. Professor DeMarco provided insightful feedback that helped me to broaden the appeal of this work, better motivate its value, discuss important implementation considerations such as diverging runtimes, and consider future extensions to multiple faults. I am particularly thankful to him for helping me to properly situate the work in the context of related work in academia, as well as to highlight its improvements over current industry practice.

I also want to thank my colleagues and peers at the University of Michigan. Thank you to Ian Beil, Jonathan Martin, Jenny Felder, and Jonas Kersulis for welcoming me to the Hiskens lab and providing me with lots of help and encouragement, especially as I was getting started with my PhD. I am very grateful to have had the opportunity to interact with and get to know Salman Nazir and Sijia Geng. Overall, I found this lab group to be incredibly friendly, warm, and caring, and this was always a source of support for me during my PhD. I also want to thank the other members of my PhD cohort for their friendship and help through this process together: Yunus Emre Sahin, Greg Ledva, Steph Crocker, Brian Gonzalez, David Hiskens, Arvind Prasadan, and Kaushik Kannan. There are too many others to thank to include everyone in this brief acknowledgement, but overall I am tremendously fortunate to have had so many helpful, encouraging, kind, and caring people in my life, and to have stood on the shoulders of giants during the course of my PhD. I am so grateful to have had this opportunity, and I will always look back fondly on my time in Ann Arbor in the years to come.

TABLE OF CONTENTS

| DEDICATION | | ii |
|------------------|---|-----|
| ACKNOWLEDGEM | ENTS | iii |
| LIST OF FIGURES | | iii |
| LIST OF TABLES . | | iii |
| LIST OF APPENDIC | CES | iv |
| LIST OF ABBREVIA | ATIONS | xv |
| LIST OF SYMBOLS | | vi |
| ABSTRACT | | xx |
| CHAPTER | | |
| I. Introduction | | 1 |
| II. Theory | | 13 |
| 2.1 Counter | rexample to Historical Work | 14 |
| 2.1.1 | Introduction | 14 |
| 2.1.2 | Background | 14 |
| 2.1.3 | Example | 16 |
| 2.1.4 | Conclusion | 19 |
| 2.2 Continu | uity of Region of Attraction Boundary Under Perturbation \therefore | 20 |
| 2.2.1 | Introduction | 20 |
| 2.2.2 | Notation and Definitions | 21 |
| 2.2.3 | Motivating Example | 27 |
| 2.2.4 | Main Results | 30 |

| | 2.2.5 | Proof of Theorem $2.2.12 \ldots 36$ |
|------------|---------------------------------------|---|
| | 2.2.6 | Proof of Theorem 2.2.23 and Corollaries |
| | 2.2.7 | Conclusion |
| 2.3 | Ω -Stabil | ity of Region of Attraction Boundary |
| | 2.3.1 | Introduction |
| | 2.3.2 | Background |
| | 2.3.3 | Examples |
| | 2.3.4 | Main Results |
| | 2.3.5 | Proofs |
| | 2.3.6 | Conclusion |
| 2.4 | Extensio | ons to Vector Fields with Event-Selected Discontinuities 90 |
| | 2.4.1 | Introduction |
| | 2.4.2 | Example |
| | 2.4.3 | Background |
| | 2.4.4 | Results |
| | 2.4.5 | Proofs |
| | 2.4.6 | Conclusion |
| 2.5 | Theoreti | ical Motivation for Maximizing Time in Ball Algorithm 113 |
| | 2.5.1 | Introduction |
| | 2.5.2 | Results |
| | 2.5.3 | Example |
| | 2.5.4 | Proofs |
| | 2.5.5 | Conclusion |
| 2.6 | Theoreti | ical Motivation for Maximizing Trajectory Sensitivities Algo- |
| | rithm . | |
| | 2.6.1 | Introduction $\ldots \ldots 127$ |
| | 2.6.2 | Results |
| | 2.6.3 | Proofs |
| | 2.6.4 | Conclusions |
| | | |
| III. Algor | ithms . | |
| 0.1 | | |
| 3.1 | Emclent | Computation of Nonsmooth, Higher Order Trajectory Sensi- |
| | 1 1 1 1 1 1 1 1 1 1 | 144 |
| | 3.1.1 2.1.0 | Introduction |
| | 5.1.2 2.1.2 | Dackground |
| | う.1.う う 1 オ | 1 line Derivatives 140 Numerical Internation 140 |
| | 5.1.4 2.1.5 | Numerical Integration |
| | 5.1.5 2.1.0 | Initial Conditions |
| | J.1.0 | Discrete Events |
| | 5.1.7 2.1.0 | Emclent Computation and Storage |
| | J.1.8 | \cup onciusion \ldots \ldots \ldots \ldots \ldots \ldots \ldots 155 |

| 3.2 | Maximiz | ing Time in Ball Around Controlling Equilibrium Point Al- |
|-----------|-----------|---|
| | gorithms | 3 |
| | 3.2.1 | Introduction |
| | 3.2.2 | One Dimensional Parameter Space |
| | 3.2.3 | Two Dimensional Parameter Space |
| | 3.2.4 | Higher Dimensional Parameter Space |
| | 3.2.5 | Choice of Norms for Measuring Distance |
| | 3.2.6 | Example: Three Synchronous Machines |
| | 3.2.7 | Conclusion |
| 3.3 | Maximiz | ring Trajectory Sensitivities Algorithms |
| | 3.3.1 | Introduction |
| | 3.3.2 | One Dimensional Parameter Space |
| | 3.3.3 | Two Dimensional Parameter Space |
| | 3.3.4 | Higher Dimensional Parameter Space: Newton's Method . 1 |
| | 3.3.5 | Higher Dimensional Parameter Space: Alternate Method . 1 |
| | 3.3.6 | Example: Two Classical Machines |
| | 3.3.7 | Algorithm Adjustments Based on Linearization Theory 1 |
| | 3.3.8 | Conclusion |
| IV. Appli | cation to | Power Systems |
| 4.1 | Modified | I IEEE 37-Bus Feeder |
| | 4.1.1 | Introduction |
| | 4.1.2 | Numerical Experiments |
| | 4.1.3 | Results |
| | 4.1.4 | Conclusion |
| 4.2 | IEEE 39 | Bus New England Power System |
| | 4.2.1 | Introduction |
| | 4.2.2 | Numerical Experiments |
| | 4.2.3 | Results |
| | 4.2.4 | Conclusion |
| V. Conc | lusions a | nd Future Work |
| APPENDICE | S | 2 |
| | 2N | |
| BIBLIOGRA | PHY | |

LIST OF FIGURES

Figure

| 2.1 | This figure originally appeared in [GH83, p. 208] and was computed numer- ically by Y. Ueda. It shows the stable and unstable manifolds of the first return map of a cross section of the Duffing equation for parameter values $\epsilon \delta = 0.25$ and $\epsilon \gamma = 0.30$. Note that the stable and unstable manifolds have nonempty, transversal intersection. | 17 |
|-----|--|----|
| 2.2 | The graph of V_p for $p = 0.1$ (green dashed), $p = 0$ (blue solid), and $p = -0.1$ | |
| | (red dot-dashed). This figure originally appeared in [FH17] | 30 |
| 2.3 | The disjoint union of the regions of attraction of the family of stable equilib- | |
| | ria of the vector fields $\{V_p\}$ over $p \in [-0.1, 0.1]$ (blue). The region of attrac- | |
| | tion of the stable equilibrium point of V_0 is shown in red. Then $\partial W^s(X_0^s)$ | |
| | consists of the two points on the boundary of this red line segment, while | |
| | $\partial W^s(X^s_J) \cap (\mathbb{R} \times \{0\})$ is equal to the union of the cyan line segments. One | |
| | family of initial conditions (yellow) begins inside the regions of attraction | |
| | and passes through one of their boundaries at p is increased. Another fam- | |
| | ily of initial conditions (green) begins inside the regions of attraction and | |
| | passes outside without passing through one of their boundaries. This oc- | |
| | curs because the boundaries of the regions of attraction fail to be Hausdorff | |
| | continuous at $p = 0$. This figure originally appeared in [FH17] | 31 |
| 2.4 | The region of attraction boundary $\partial W^s(X^s_{p_0})$ (black line) of the stable equi- | |
| | librium point $X_{p_0}^s$ (black star) of Eqs. 2.8-2.9 is shown. It is equal to $W^s(X_{p_0}^1)$ | |
| | where $X_{p_0}^1$ (black triangle) is the unstable equilibrium point. The orbit | |
| | (dashed black line) from the initial condition y_{p_0} (black circle) is shown | 37 |
| 2.5 | The region of attraction boundaries $\partial W^s(X^s(p))$ of the stable equilibrium | |
| | points $X^{s}(p)$ (stars) for parameter values $p = 1.3$ (green dashed), $p = 1.568$ | |
| | (solid blue), and $p = 1.9$ (red dot dashed) are shown. Each boundary | |
| | is equal to $W^s(X_p^1)$ where X_p^1 (triangle) is the unstable equilibrium point | |
| | corresponding to parameter value p . The initial conditions (circles) are shown. | 37 |

2.6The compact set D (black line segments) and the neighborhood N (yellow shapes) mentioned in Lemma 2.2.28 for an equilibrium point (red star). The set D is contained in the unstable manifold (red line). As the neighborhood N is propagated backwards in time (first to the green shapes then to the cyan), it approaches the stable manifold (blue line) of the equilibrium point. From the figure, it appears that the union of the backward flows of N over all negative times, together with the stable manifold, will contain a neighborhood of the equilibrium point, which is the content of Lemma 2.2.28. 2.7Example 2.3.1. A stable hyperbolic equilibrium point (blue dot), saddle hyperbolic equilibrium points (red dots), sinks (black dots), nonhyperbolic equilibrium points (green dots), and a point at infinity (white dot) are shown. A sequence of periodic orbits (red for unstable and blue for stable) accumulates along the boundary of the region of attraction of the stable hyperbolic equilibrium point (blue dot). Black lines indicate stable and unstable manifolds of the saddle equilibrium points, with the arrows indicating direction 562.8Example 2.3.2. A stable hyperbolic equilibrium point (blue dot), saddle hyperbolic equilibrium point (red dot), continuum of nonhyperbolic equilibrium points which are unstable in the transverse direction (red line), a stable sink (black dot), and a point at infinity (white dot) are shown. Black lines indicate stable and unstable manifolds of the saddle equilibrium point, with the arrows indicating direction of flow. The left figure shows the vector field before perturbation and the right figure shows the vector field after a perturbation in a compact neighborhood of its saddle equilibrium point (red dot). A segment of the continuum of nonhyperbolic equilibrium points (red line) enters the boundary of the region of attraction of the stable equilibrium 572.9Example 2.3.3. A stable hyperbolic equilibrium point (blue dot), saddle hyperbolic equilibrium point (red dot), source and sink (black dots), and a point at infinity (white dot) are shown. Black lines indicate stable and unstable manifolds of the saddle equilibrium point, with the arrows indicating direction of flow. The left figure shows the vector field before perturbation and the right figure shows the vector field after a perturbation in a compact neighborhood of its saddle equilibrium point (red dot). A source (black dot) enters the boundary of the stable equilibrium point (blue dot) arbitrarily far from the boundary before perturbation and from infinity. 58

39

| 2.10 | Illustration of the key ideas behind the proof of Lemma 2.3.24. The light | |
|-------|--|----|
| | blue rectangle Y represents the choice of local coordinates, the orange circles | |
| | compose the neighborhood N_i , the yellow curves represent the disk family | |
| | centered along $W_{loc}(x_0)$ that is constructed in the proof, the purple heighbor- | |
| | nood U is contained in this disk family, the turquoise heighborhood $U_i \subset U$ | |
| | is contained in the disk family corresponding to perturbations to the vector field, and the graving pairbharhood $\hat{U} \subset U'$ is such that under parturbations to | |
| | the vector field every beckwards orbit originating in \hat{U} remains in U' until | |
| | the vector here every backwards orbit originating in U remains in U_i untili- it passes through N_i | 60 |
| 2 11 | Example of the construction of the directed graphs in the proof of Lemma 2.3.25 | 03 |
| 2.11 | A vector field on the sphere \mathbb{S}^2 (left) and the corresponding directed graph | |
| | (right) for the closure of the region of attraction of its stable hyperbolic equi- | |
| | librium point (blue dot) are shown. A saddle hyperbolic equilibrium point | |
| | (red dot), along with a sink and a source (black dots) are also shown. Note | |
| | that equilibrium point four does not appear in the directed graph since it is | |
| | not contained in the closure of the region of attraction. | 70 |
| 2.12 | The vector field V_p for $p = 0.3$. The red and green lines show two sets | |
| | of parameter dependent initial conditions, with the initial conditions for | |
| | p = 0.3 shown as red and green circles | 93 |
| 2.13 | The vector field V_p for $p = 0$. The red and green lines show two sets of | |
| | parameter dependent initial conditions, with the initial conditions for $p = 0$ | |
| 0.1.1 | shown as red and green circles | 94 |
| 2.14 | The vector field V_p for $p = -0.3$. The red and green lines show two sets | |
| | of parameter dependent initial conditions, with the initial conditions for | 04 |
| 9.15 | p = -0.5 shown as red and green circles | 94 |
| 2.10 | with C^{∞} event-selected discontinuities can become no longer event-selected | |
| | discontinuous under strong C^{∞} perturbations due to the presence of two | |
| | nontransversal switching surfaces (black curves). | 95 |
| 2.16 | The transversal intersection of several orbits with the ball N containing | 00 |
| | the unstable equilibria (red stars). Orbits are shown for parameter values | |
| | (driving torques) of 1.5 (cyan), 1.516 (yellow), 1.532 (green), 1.55 (magenta), | |
| | 1.568 (blue), and 1.57 (red), where only the initial condition corresponding | |
| | to the final parameter value lies outside the regions of attraction of the stable | |
| | equilibria. | 17 |

- 3.1 A continuation method for iteratively tracing a sequence of points along a curve defined by a set of algebraic equations. At each step a prediction is made by moving along the tangent to the curve. Then the predicted point is projected back onto the curve. This process repeats until the curve is traced. 159
- 3.3 The moment of inertia scaling factor is varied to maximize the time spent by the trajectory inside the ball around the controlling unstable equilibrium point. As this time increases, the moment of inertia scaling factor converges to its critical value.
 3.4 The recovery boundary for the moments of inertia of generators one and two. All parameter values above the boundary lead to fault recovery, whereas all
- - (a) the initial value $P_3 = 0.5$ (blue line) and (b) the critical value $P_3 = 0.7309$ (red line).

| 3.7 | Computation of critical parameter value for P_3 using the one-dimensional parameter space algorithm of Section 3.3.2. Iterations begin in the top left and proceed towards the bottom right. Rapid convergence to the critical | |
|------|---|-----|
| | parameter value is observed. | 175 |
| 4.1 | IEEE 37-bus test feeder. | 184 |
| 4.2 | Distance from the trajectory to the unstable equilibrium point as a function of time, for a trajectory that has been driven inside a ball of radius 190 for | |
| | a time $t_{ball} = 0.2$ s. The radius 190 is depicted by the dotted line | 185 |
| 4.3 | Time t_{ball} spent within a fixed radius of the unstable equilibrium point versus | |
| | moment of inertia scaling factor SF | 186 |
| 4.4 | Recovery boundary for the moments of inertia at buses 740 and 741 | 186 |
| 4.5 | Recovery boundary for the moments of inertia at buses 740 and 725 | 187 |
| 4.6 | Optimization algorithm to find the nearest point on the recovery boundary. The specified value p_0 , first iteration, second iteration, and third iteration are given by the black star green plus, evan circle, and red star, respectively | |
| | The black dashed line connects p_0 with the final solution and is orthogonal | |
| | to the recovery boundary | 187 |
| 47 | IEEE 39-bus New England power system | 189 |
| 4.8 | G as a function of background load scaling factor (blue line). The iterations | 100 |
| 1.0 | of the one dimensional parameter space algorithm are labeled in order (circles) | 190 |
| 4.9 | The angle of generator 5 relative to the angle of generator 2 as a function of time for the values of the background load scaling factor that correspond | 100 |
| | to the iterations of the one dimensional algorithm of Fig. 4.8 | 101 |
| 4 10 | The recovery boundary and the set of recovery values (recovery region) in the | 191 |
| 4.10 | two dimensional parameter space of AVR gain SF and reactive load voltage exponent. Colored circles indicate the parameter values whose corresponding | |
| | dynamic behavior is shown in Fig. 4.11. | 192 |
| 4.11 | The angle of generator five relative to the angle of generator two as a function | 10- |
| | of time for an AVR gain SF of 1.054 and a range of reactive load voltage | |
| | dynamics exponents. | 192 |
| 4.12 | The angle of generator five relative to the angle of generator two as a func- tion of time for the solutions of the optimization algorithm applied to the | |
| | parameter sets S_1 , S_2 , and S_3 , as well as for the nominal parameter values S_0 . | 193 |
| | | |

LIST OF TABLES

<u>Table</u>

| 3.1 | Synchronous machine parameters | 162 |
|-----|--|-----|
| 3.2 | Test Case Parameter Values | 172 |
| 3.3 | Values of the objective function of (3.51) for the critical parameter values | |
| | obtained by the higher dimensional optimization algorithm for the parameter | |
| | sets P_4 , P_6 , and P_8 | 176 |
| 4.1 | Values of the objective function of Eq. 3.64 for the parameter values on | |
| | the recovery boundary obtained as the optimal solutions to the higher di- | |
| | mensional parameter space optimization algorithm of Section 3.3.7.3 for the | |
| | parameter sets S_1 , S_2 , and S_3 | 194 |
| | | |

LIST OF APPENDICES

Appendix

| А. | Proof of Lemma 2.2.28 | 201 |
|----|---|-----|
| В. | Proof of Lemma 2.2.37 | 204 |
| С. | Proof of Lemma 2.3.24 | 206 |
| D. | Derivation of Trajectory Sensitivities at Discrete Events | 207 |

LIST OF ABBREVIATIONS

IC initial condition
SEP stable equilibrium point
recovery value recovery value of the parameters
recovery boundary recovery boundary in parameter space
RoA region of attraction
UEP unstable equilibrium point

LIST OF SYMBOLS

 Ω nonwandering set

- ${\cal M}\,$ smooth manifold
- C^1 continuously differentiable
- C^{∞} smooth
- \mathbb{R}^n Euclidean space
- $V\,$ vector field
- x_e equilibrium point
- $W^s(\gamma)$ stable manifold of γ
- $W^u(\gamma)$ unstable manifold of γ
- \hat{A} differentiable submanifold
- \hat{B} differentiable submanifold
- $T_x(\hat{A})$ tangent space of \hat{A} at x
- $\hat{\tau}$ Poincaré first return map
- $\overline{E}\,$ topological closure of $E\,$
- $\partial E \,$ topological boundary of E
- int E topological interior of E
- $\tilde{f}|_E\,$ restriction of the function \tilde{f} to E
- $D_{\epsilon} \epsilon$ -neighborhood of D
- $\mathcal{C}(K)$ the set of nonempty closed subsets of K

 d_h Hausdorff distance

 d_S set distance

 $\phi~{\rm flow}$

- ϕ_t time-*t* flow
- \tilde{M} smooth manifold
- d_{C^1} C^1 distance
- $d\tilde{f}_x$ differential of \tilde{f} at x
- $V_p\,$ vector field corresponding to parameter value p
- S Poincaré cross section
- X critical element
- $W^s_{\mathbf{loc}}(X)$ local stable manifold of X
- $W^u_{\mathbf{loc}}(X)$ local unstable manifold of X
- $W^{s}(X)$ stable manifold of X
- $W^u(X)$ unstable manifold of X
- $\Omega(V)$ nonwandering set under V
- $\omega(y, V)$ ω -limit set of y under V
- J connected smooth manifold which represents parameter values

 X_p critical element corresponding to parameter value p

- $X_Q \bigcup_{p \in Q} X_p$
- $W^s(X_Q) \sqcup_{p \in Q} W^s(X_p)$
- $W^u(X_Q) \sqcup_{p \in Q} W^u(X_p)$
- $X^{s}(p)$ perturbation of $X^{s}(p_{0})$ corresponding to parameter value p
- $X^{i}(p)$ perturbation of $X^{i}(p_{0})$ corresponding to parameter value p
- $W^{s}(X^{s}(p))$ stable manifold of $X^{s}(p)$
- $W^{s}(X^{i}(p))$ stable manifold of $X^{i}(p)$

$$\begin{split} W^{u}(X^{i}(p)) & \text{unstable manifold of } X^{i}(p) \\ X^{s}_{J} \sqcup_{p \in J} X^{s}(p) \\ W^{s}(X^{s}_{J}) \sqcup_{p \in J} W^{s}(X^{s}(p)) \\ X^{i}_{J} \sqcup_{p \in J} X^{i}(p) \\ W^{s}(X^{i}_{J}) \sqcup_{p \in J} W^{s}(X^{i}(p)) \\ W^{u}(X^{i}_{J}) \sqcup_{p \in J} W^{u}(X^{i}(p)) \\ S^{s} W^{s}_{\text{loc}}(X^{s}(p_{0})) \\ S^{i} X^{i}(p_{0}) \\ S^{i}_{u} W^{u}_{\text{loc}}(X^{i}(p_{0})) \\ F^{i} \text{ maps } p \text{ onto } X^{i}(p) \\ F^{i}_{s} \text{ maps } p \text{ onto } W^{s}_{\text{loc}}(X^{i}(p)) \\ F^{i}_{u} \text{ maps } p \text{ onto } W^{s}_{\text{loc}}(X^{i}(p)) \\ F^{s} \text{ maps } p \text{ onto } W^{s}_{\text{loc}}(X^{s}(p)) \\ F^{s} \text{ maps } p \text{ onto } W^{s}_{\text{loc}}(X^{s}(p)) \\ s \text{ event or switching function} \end{split}$$

 $\sigma\,$ function that gives the switching state status

 \boldsymbol{s}_p^j event or switching function j corresponding to parameter value p

 $Z_{j}^{p}\,$ switching surface j corresponding to parameter value p

- $Z^p \bigcup_{j=1}^m Z^p_j$
- $Z_j \sqcup_{p \in J} Z_j^p$
- $Z^J \sqcup_{p \in J} Z^p$
- $C^J (M \times J) \backslash Z^J$
- $S_s^i(p) \ W^s_{\text{loc}}(X^i(p))$
- $S^j_u(p) \ W^u_{\text{loc}}(X^j(p))$

 \hat{y}_p initial condition corresponding to parameter value p

- ${\cal R}\,$ set of recovery parameter values
- $\partial R\,$ recovery parameter boundary
- ${\cal C}\,$ set of critical parameter values

ABSTRACT

A novel approach is introduced to assess stability of nonlinear systems in the presence of parameter uncertainty. The idea is to consider the deterministic dynamics of the system as a function of parameter values, where the parameter-dependent initial condition may for example be the output of a particular finite-time disturbance. The goal is to numerically determine the boundary in parameter space, referred to as the *recovery boundary*, between parameter values which lead to recovery and those which lead to a failure to recover to an initial stable equilibrium point. *Critical parameter values*, which are defined to be those parameter values whose corresponding initial conditions lie on the boundary of the region of attraction of their corresponding stable equilibrium points, have the potential to provide an explicit connection between the recovery boundary in parameter space and the region of attraction boundary in state space that can be exploited for algorithm design.

However, examples are provided to illustrate that the recovery boundary may not contain critical parameter values when the boundary of the region of attraction of the stable equilibrium point varies discontinuously with parameter. Fortunately, it is shown that, for a large class of vector fields possessing stable equilibrium points, the boundaries of the regions of attraction of these equilibrium points vary continuously with respect to small variations in parameter values. This region of attraction boundary continuity ensures that the recovery boundary consists entirely of critical parameter values and that the nearest critical parameter value to any non-critical parameter value lies on the recovery boundary.

Two classes of theoretically motivated algorithms are developed to compute critical parameter values by exploiting the structure and behavior of the region of attraction boundary under parameter variation. The system trajectory corresponding to a critical parameter value converges to an invariant set and, therefore, spends an infinite amount of time in any neighborhood of that invariant set. A first class of algorithms proceed by varying parameter values so as to maximize the time in a neighborhood of the invariant set. Under reasonable assumptions, the system trajectory corresponding to a critical parameter value becomes infinitely sensitive to small changes in parameter value. A second class of algorithms proceed by varying parameters.

Theoretical motivation is provided for both classes of algorithms, and shows that under reasonable assumptions they will drive parameter values to their critical values. Both of these approaches transform the abstract problem of finding critical parameter values into concrete numerical optimization problems. Based on these approaches, algorithms are developed to find the closest parameter value on the recovery boundary in the case of one-dimensional parameter space, trace the recovery boundary in two-dimensional parameter space, and find the nearest point on the recovery boundary in parameter space of arbitrary dimension.

The algorithms are applied to assess fault vulnerability in power systems. Results from the test cases of a modified IEEE 37-bus feeder and a modified IEEE 39-bus system illustrate the algorithm performance. The emphases in these test cases are, respectively, to explore the onset of induction motor stalling during Fault Induced Delayed Voltage Recovery, and to analyse stability of a system of synchronous machines under high levels of load uncertainty.

CHAPTER I

Introduction

Engineered systems experience disturbances which have the potential to disrupt desired operation, such as a lightning strike on a particular transmission line in a power system. From a systems perspective, the disturbance can be thought of as a parameter dependent initial condition (IC) to the post-disturbance dynamical system (which itself is parameter dependent). The desired operating point is typically modeled as a stable equilibrium point (SEP) of the post-disturbance system. Ability of the system to recover from a particular finite-time disturbance to the desired operating point depends on the system parameters. Define a recovery value of the parameters (recovery value) to be a value for which the system is able to recover from the disturbance to the desired SEP. It is an important and challenging problem to determine the set of recovery values for a particular disturbance and parameter space. Solving this problem is of value for many applications, for example assessing fault vulnerability in power systems (a fault is a short circuit that causes instability in power systems and in some instances may lead to blackout conditions).

As parameter values are generally time-varying and uncertain in practice, knowledge of the recovery values provides system operators and engineers with a quantitative measure of the margins for safe operation. In order to determine the set of recovery values, the approach taken here focuses on determining their (topological) boundary in parameter space, which we term the recovery boundary in parameter space (recovery boundary). Variants of this problem have existed for a long time, but it remains of major importance to many application areas today. For example, it has existed since the first large scale power systems were developed about a century ago, but is as important as ever in power systems as increased penetration of renewable energy generation increases parameter uncertainty, and increased demand coupled with aging infrastructure cause the grid to operate closer to its limits. We note that the discussion here applies equally well to systems with uncertain, parameter dependent initial conditions; that the initial condition is the output of another dynamical system (such as a finite-time disturbance) is not required.

In power systems, current industry practice involves the construction of inner approximations of the recovery boundary known as nomograms [MWTP97]. These nomograms are typically piecewise linear, consisting of a polytope defined by bounding hyperplanes. Nomograms are constructed by a variety of heuristic methods. One such technique involves initially fixing the values of all parameters other than one, varying that last parameter value and running simulations until the recovery boundary is encountered, changing the other parameter values, and repeating. Another method is to start from an initial set of parameter values, choose a "stress" direction in parameter space, vary the distance traveled along this stress direction and run simulations until the recovery boundary is encountered, change the stress direction, and repeat. Every simulation is run for a time interval determined to be sufficient to assess whether or not the system has recovered. The recovery boundary is encountered numerically when parameter values are sampled which lead to recovery and failure to recover (ie. parameter values both within and outside of the recovery boundary) that are sufficiently close in parameter space.

Each of these approaches requires a large number of computationally intensive time domain simulations and, in particular, the number of required simulations grows rapidly with the dimension of parameter space. Hence, to avoid computational intractability, in practice power system operators typically limit parameter space to two or three important parameters of interest - often selected based on operator experience and intuition - and perform the construction of nomograms only during offline planning studies. Due to the low dimensionsionality, the nomograms can be visualized as a piecewise linear curve in the case of two parameter dimensions. In three dimensional parameter space, they can be viewed as a contour plot consisting of a piecewise linear curve for each value of a third parameter. Recent work has extended the techniques to parameter space with higher dimension than three, such as in $[MDL^+12]$, but nomogram construction remains a method for low dimensional parameter space. Overall, major limitations of current industry practice include:

- 1. The small number of parameters of interest are chosen based on operator intuition and experience, so many parameters which could be critical to system recovery may be overlooked.
- 2. The methods are inherently approximate and tend to be overly conservative or just inaccurate (especially as parameter values vary during real-time operation).

3. Nomogram construction is limited to offline planning studies, and they are not updated during real-time operation.

These drawbacks of standard industry practice have long motivated a search for alternative approaches to accurately and efficiently determine or approximate the recovery boundary.

Historical work focused on parameter independent post-disturbance dynamics, and the main parameter of interest was the duration of the finite-time disturbance. In this simpler setting, the post-disturbance system has a unique (parameter independent) region of attraction (RoA) of its unique SEP. Assuming (at least local) continuity of the IC with respect to disturbance duration, a disturbance duration value lies on the recovery boundary precisely when its corresponding IC lies on the boundary of the RoA of the post-disturbance SEP. More generally, we coin the term *critical parameter value* to refer to a parameter value whose corresponding IC lies on the boundary of the RoA of the corresponding SEP.

Classical algorithms relied primarily on the use of energy functions to estimate the critical disturbance duration, known in the power systems literature as critical fault clearing time. These methods were initially introduced by [Mag47, Ayl58, Gle66, EAN66] and subsequently extended in several directions. The common steps behind the algorithms were as follows:

- 1. Identify a real-valued function on state space, called the energy function, for the system of interest such that the SEP is a local minimum of the energy function, near the SEP this function is positive definite, and near the SEP its time derivative along system trajectories is negative definite.
- 2. For a particular disturbance, identify a critical energy such that if the energy of the IC is not larger than this value, then the IC lies inside the RoA of the SEP.
- 3. Numerically integrate the disturbance dynamics (no longer enforcing the finite-time limit) until the critical energy is encountered. The time at which this occurs is used to approximate the critical disturbance duration.

The intuition behind this technique is that since the SEP is a local minimum of the energy function, it is positive definite near the SEP, and its time derivative along system trajectories is negative definite near the SEP, there exists a sublevel set of the energy function in which all trajectories converge to the SEP in forwards time. Hence, this sublevel set is contained in the RoA of the SEP. Therefore, the disturbance duration is guaranteed to be at a recovery value if its corresponding IC lies in this sublevel set or, equivalently, if the energy of its corresponding IC lies beneath some critical level. Numerically integrating the disturbance dynamics until this critical energy is encountered therefore finds the disturbance time which puts its corresponding IC at the boundary of this sublevel set, thereby providing a conservative estimate of the critical disturbance duration.

The main differences in the various energy function approaches consist primarily of the choice of energy function and of the method to determine the critical energy. Many modified energy functions were proposed in early attempts to apply the methods to more accurate power systems models [BH81, BHD86, NM84, TAV85], but energy functions for more realistic, industry standard power system models have yet to be developed. To determine the critical energy, the initial focus was on the so-called closest unstable equilibrium point (UEP) method. The idea was to set the critical energy to be the minimum of the energy function over the boundary of the RoA of the SEP. Then, whenever the IC has energy less than this critical energy, it must lie inside the RoA. There was an implicit assumption in the closest UEP approach that every trajectory in the boundary of the RoA would converge to an equilibrium point in forwards time (it should be noted that this assumption is not true in general for more realistic power system models).

Define the *stable manifold* (resp. *unstable manifold*) of an equilibrium point to be the set of initial conditions that converge to it in forwards (resp. backwards) time. Then the assumption can be restated as: the RoA boundary is equal to the union of the stable manifolds of the equilibria it contains. As any energy function decreases in time along system trajectories, it follows that it achieves a minimum over the stable manifold of an equilibrium point at that equilibrium point. Therefore, the minimum of the energy function over the RoA boundary would be equal to the minimum of the energy function over the equilibria in the RoA boundary. Hence, methods were devised to determine the equilibrium point on the RoA boundary of minimum energy [TS72, PEA75], called the closest UEP.

It was quickly realized that the closest UEP method for determining critical energy yielded highly conservative estimates. One alternative approach, known as the potential energy boundary surface (PEBS) method [KOH78, APV79], was developed to estimate the critical energy by estimating the energy at the point when the disturbance system exits the post-disturbance RoA boundary (this exit point is difficult to estimate as well). If the IC has energy less than this for the given disturbance, then the system cannot escape the postdisturbance RoA. The PEBS technique estimated the exit point energy by first estimating the exit point energy of the simpler system given by the gradient of the potential energy (which was typically found by integrating the disturbance trajectory until a maximum in potential energy was encountered). Another approach, which came to be known as the controlling UEP method, was initially alluded to by [KOH78]. The key idea is that the trajectory of the disturbance system must cross the post-disturbance RoA boundary. So, by the decomposition above, it must cross the stable manifold of some UEP, which is called the controlling UEP. The controlling UEP method sets the critical energy to be the energy at the controlling UEP. So long as the IC has energy less than at the controlling UEP, the disturbance trajectory could not have crossed the stable manifold of the controlling UEP (since this stable manifold has higher energy than the controlling UEP). So, the system could not have escaped the post-disturbance RoA, which implies that the IC lies inside this RoA.

Several methods for identifying the controlling UEP were developed, including the mode of disturbance (MOD) method [FVO84], the exit point method (also known as the BCU method) [CWV88], and the shadowing method [TVK96], which ultimately rely on using numerical integration of the system dynamics to attempt to find a point near the controlling UEP which can be used as an initial guess to solve for the actual controlling UEP using Newton's method. Power system models contain many equilibria, and these methods are not guaranteed to converge to the single desired controlling UEP among the large number of possible equilibria.

Limitations for practical application of these energy methods were identified early on [BPRP+84]. These include:

- 1. Lack of suitable energy functions for more realistic generator, load, and network power system models.
- 2. Inherent conservativeness of the energy function approach.
- 3. Inability to handle hybrid dynamics including switching behavior and controller limits that are characteristic of modern power systems.
- 4. Methods are only able to determine the critical clearing time (as opposed to other critical parameter values), which does not adequately measure system stability.

To this list, we add:

5. The energy function methods are not able to satisfactorily handle the case when the exit point of the disturbance trajectory converges to an invariant set other than an equilibrium point.

6. Less conservative energy function methods require knowledge of the controlling UEP, which is challenging to find even for relatively small power system models.

Despite decades of research since their initial introduction in the 1960's, these fundamental challenges to energy function methods have yet to be overcome. However, the development of the energy function methods stimulated research into the topological structure of the RoA boundary of nonlinear systems, and its application to estimation of critical disturbance time. The key papers in this area were [CHW88, ZHZL88], which showed that for a large class of continuously differentiable (C^1) vector fields possessing a SEP, the boundary of its RoA is equal to the union of the stable manifolds of the equilibrium points and periodic orbits it contains. These results were of limited practical value because the assumptions under whch they were proved include that the RoA boundary is contained in the union of stable manifolds. Establishing that the RoA boundary is not just contained in but is actually equal to this union has not provided additional motivation or intuition for the development of algorithms for computing critical disturbance times. The foundation for this theoretical work came from developments in the theory of hyperbolic dynamical systems.

One such area involved the study of *generic* properties of differentiable dynamical systems. These were properties or behaviors that one might expect of a typical vector field, such as if a vector field could be chosen at random, but could still exclude certain pathological cases. The notion of generic in a topological space is similar to the concept of probability one in a probability space in that both are meant to capture the idea of being true almost always. Before discussing some generic properties, a few definitions are required. An equilibrium point (resp. periodic orbit) is *hyperbolic* if the linearization of the vector field at any point in it possesses exactly zero (resp. one) eigenvalues with zero real part. Two differentiable manifolds \hat{A} and \hat{B} inside some smooth manifold (M) are transverse if at every point x in their intersection, the tangent spaces to \hat{A} and \hat{B} at x together span the tangent space of M at x. A point x in a differentiable manifold M is said to be nonwandering for a vector field, which possesses an associated flow, if for every open neighborhood U of x and every T > 0, there exists t > T such that the flow of U forward or backward by time t has nonempty intersection with U. Let nonwandering set (Ω) be the set of all nonwandering points in M. Then Ω includes all the equilibrium points, periodic orbits, and nontrivial recurrence (including chaotic behavior) of the flow. Furthermore, every trajectory that does not diverge to infinity must converge to a subset of Ω .

The study of generic properties made major progress when it was shown that for a generic vector field, all its equilibrium points and periodic orbits are hyperbolic, and their stable and

unstable manifolds are pairwise transverse [Sma62, Kup63, Pei66]. Around the same time, a class of vector fields, which came to be known as Morse-Smale vector fields, was introduced [Sma60] which satisfied the following assumptions:

- 1. The nonwandering set consists of a finite union of equilibrium points and periodic orbits.
- 2. All equilibrium points and periodic orbits are hyperbolic.
- 3. The stable and unstable manifolds of the equilibrium points and periodic orbits are pairwise transverse.

It was initially conjectured that Morse-Smale vector fields could be generic [Sma60], although the same author later showed that this is not the case. It was also conjectured [Sma60] that a C^1 vector field was structurally stable, meaning informally that its orbit structure would be preserved under perturbations, if and only if it was a Morse-Smale vector field. Soon afterwards it was shown that Morse-Smale vector fields are open in the topological space of C^1 vector fields [Pal69], that they are structurally stable [PS68], and that there exist structurally stable C^1 vector fields which are not Morse-Smale [Sma63].

In [CHW88, ZHZL88], the authors make assumptions that are slightly more general than that the vector field is Morse-Smale along the RoA boundary of the SEP in order to arrive at their decomposition of the RoA boundary into a union of stable manifolds. However, these more general assumptions lead them to require a lemma [CHW88, Lemma 3-5], which is disproven via a counterexample in Section 2.1, to complete the proof of their main theorem. Therefore, to avoid the use of this flawed lemma, we will assume here that the vector field is Morse-Smale along the RoA boundary. More generally, the techniques used by [CHW88, ZHZL88] in their proofs were inherited from the study of Morse-Smale systems, especially [Pal69]. Since that initial work in 1988, there has not been significant progress towards obtaining new results regarding the structure and behavior of the RoA boundary.

Chiang's work related to the study of systems exhibiting Morse-Smale-like properties along their RoA boundaries has received criticism in the past. In particular, in [CC95, Theorem 5-2] Chiang makes an assumption that the stable and unstable manifolds of the equilibria in the RoA boundary are transversal over an entire one-parameter family (as a parameter is varied from zero to one). This assumption was challenged, both from the construction of simple counterexamples [LLM+95] and an illustration that the assumption is not generic [PL99]. Hence, from both practical and theoretical perspectives, this one-parameter family transversality assumption was not a good assumption. Fortunately, the transversality assumption used in [CHW88] and employed in this work requires only transversality of stable and unstable manifolds at one particular parameter value, which is a generic assumption, rather than over an entire one-parameter family. Hence, the transversality assumption used here was not the subject of this earlier criticism, and is generically true for C^1 vector fields.

To address the problem of determining the recovery boundary, it is beneficial to extend the historical work to incorporate behavior of the RoA boundary under parameter variation. Recall that a critical parameter value refers to a parameter value whose corresponding IC lies on the boundary of the RoA of the corresponding SEP. The recovery boundary is defined to be the (topological) boundary in parameter space of the set of recovery values. The definition of recovery boundary does not a priori provide a link between the parameter values it contains and their corresponding state space behavior. On the other hand, critical parameter values provide an explicit connection to the RoA boundary, which introduces an opportunity to develop algorithms that can exploit the structure and behavior of the RoA boundary. Therefore, from the perspective of recovery boundary estimation or computation, it would be highly desirable if the recovery boundary consisted entirely of critical parameter values.

By definition, every recovery value has a corresponding IC which lies inside the RoA of the corresponding SEP. Therefore, it seems natural to expect that every parameter value in the recovery boundary would have a corresponding IC which lies in the boundary of the RoA of the corresponding SEP. This would imply that every recover parameter value would be a critical parameter value. If the IC is (at least locally) continuous with respect to parameter, and if the post-disturbance vector field is parameter independent, then this is always true. This motivates the classical approach of designing algorithms to estimate the critical disturbance time in order to estimate the disturbance time recovery boundary (in fact, no distinction is made between these terms in the literature).

Unfortunately, the situation becomes more complex in the case where the post-disturbance vector field is parameter dependent. We will see in Example 2.2.1 that even if the IC is continuous with respect to parameter and the vector field is perturbed smoothly in a compact region, it is possible that the recovery boundary contains no critical parameter values at all. The reason for this is that, even though the IC is continuous and the vector field varies smoothly, the RoA boundary varies discontinuously and "jumps" over the IC so that they do not intersect. Furthermore, this behavior does not just hold for contrived, pathological cases: Example 2.3.3 shows such behavior for a (globally) Morse-Smale vector field over Eu-

clidean space. Therefore, it is important to study conditions under which continuity of the RoA boundary can be guaranteed. This will help ensure both that the recovery boundary consists entirely of critical parameter values and that the closest critical parameter value to any recovery value lies on the recovery boundary. In turn, these will facilitate the design of practical algorithms for computing the recovery boundary,

The main results of this effort are that for C^1 vector fields which are Morse-Smale along the RoA boundary of a SEP in a compact Riemannian manifold, or in Euclidean space with some additional generic assumptions and control near infinity, under sufficiently small variations in parameter values the RoA boundary will vary continuously. The proof proceeds in two main steps:

- 1. Under the additional assumption that the nonwandering set in the RoA boundary does not grow under parameter perturbations, prove continuity of the RoA boundary.
- 2. Show that the above assumptions imply that the nonwandering set in the RoA boundary does not grow under parameter perturbations.

As part of the continuity proof, it is also shown that the decomposition of the RoA boundary into the union of the stable manifolds of the equilibrium points and periodic orbits it contains persists under parameter perturbations. Afterwards, the RoA boundary continuity results are extended to a class of hybrid dynamical systems which exhibit restricted switching and limit behavior. The RoA boundary continuity results are then used to establish that the recovery boundary consists entirely of critical parameter values. Finally, it is shown that the closest critical parameter value to any recovery value lies on the recovery boundary.

The next step is to exploit the structure of the RoA boundary to motivate algorithms for numerically computing critical parameter values, thereby obtaining the desired recovery boundary. Theoretical motivation is provided for two such algorithms. The first algorithm builds upon the historical notion of the controlling UEP. It exploits the property that, at a critical parameter value, the corresponding system trajectory converges to an invariant set (such as an equilibrium point or periodic orbit) in the RoA boundary, and therefore spends an infinite amount of time in any neighborhood of this so-called controlling invariant set. It is shown that, assuming the system trajectory intersects a neighborhood of the controlling invariant set transversely, the time the system trajectory spends inside this neighborhood is continuous over the set of recovery values, and diverges to infinity as any critical parameter value is approached from within the set of recovery values. This motivates an algorithm to compute critical parameter values by varying parameter values so as to maximize the time the system trajectory spends inside a neighborhood (typically chosen to be a ball for convenience) of the controlling invariant set, thereby driving the parameter value to its critical value.

To motivate the second algorithm, consider first a hyperbolic linear system with a saddle equilibrium point. For any trajectory in the stable eigenspace of the equilibrium point, a slight perturbation would push it off the stable eigenspace and cause it to diverge to infinity along the unstable direction in forwards time. In particular, the system trajectory along the stable manifold of the equilibrium point (which is equal to its stable eigenspace) becomes infinitely sensitive to small perturbations as time goes to infinity. More generally, at a critical parameter value assume the corresponding system trajectory is in the stable manifold of a hyperbolic (or normally hyperbolic) invariant set. Then the sensitivity of the system trajectory to small changes in parameter value (ie. small perturbations) diverges to infinity as time goes to infinity. It is shown that the supremum over time of the sensitivity of the trajectory with respect to parameter is continuous over the set of recovery values, and diverges to infinity as any critical parameter value is approached from within the set of recovery values. This motivates an algorithm to numerically compute critical parameter values by varying parameter values so as to maximize the supremum over time of the sensitivity of the trajectory with respect to parameter, thereby driving the parameter value to its critical value. It is shown that, as recovery values approach the recovery boundary, the infimum over time of the inverse trajectory sensitivities approaches a dot product. This motivates the design of algorithms for computing critical parameter values by exploiting this linearity. In both cases discussed above, properties of the RoA boundary are exploited to convert the abstract problem of computing critical parameter values into a concrete optimization or algebraic problem which can be solved numerically.

Motivated by the theory discussed above, several algorithm variations are developed to numerically compute critical parameter values. In the case of the choice of a onedimensional parameter space, methods are proposed to find the unique parameter value on the recovery boundary. If a two-dimensional parameter space is chosen, techniques are developed to numerically trace the locally one-dimensional manifold recovery boundary using continuation methods. However, it is important to note that the recovery boundary can be nonsmooth in general. Nevertheless, empirical observations indicate that, at least on the test cases considered so far, the recovery boundary is smooth for practical power system models. For the case of arbitrary parameter space dimension, multiple approaches are devised to find the nearest set of parameter values on the recovery boundary from some initial set of parameter values. These techniques address many of the limitations of current industry practice:

- 1. The parameter space of interest can be chosen to be high dimensional, so that parameter values which could be critical to system recovery but might otherwise have been overlooked could be identified.
- 2. The methods determine the recovery boundary to arbitrary precision, and are not conservative or inaccurate.
- 3. Convergence is typically obtained in a small number of iterations, so that application of the algorithms during real-time operation may be feasible.

Furthermore, these techniques address many of the limitations of the classical algorithms:

- 1. They do not require the existence of suitable energy functions.
- 2. They compute the actual recovery boundary instead of conservative estimates of it.
- 3. They work for a class of hybrid systems exhibiting restricted switching and limit behavior.
- 4. They can consider parameter recovery boundaries for parameters which influence postdisturbance dynamics, thereby obtaining more useful measures of system stability than critical disturbance times.
- 5. They can still work even if trajectories corresponding to critical parameter values converge to invariant sets which are not equilibrium points.

Additionally, the second class of algorithms, related to minimizing the infimum over time of the inverse trajectory sensitivities, has the following additional advantages:

- 6. It does not require prior knowledge of the controlling invariant set.
- 7. For recovery values near the recovery boundary, it approaches a dot product, giving it a linear structure.

In sum, the algorithms address most of the shortcomings of current industry practice and of historical work related to energy function methods. For implementation, the algorithms both require computation of trajectory sensitivities. A modular scheme is introduced for trajectory sensitivity integration which exploits sparse network structure to improve computational efficiency and reduce storage requirements. This enables efficient application of the algorithms developed above.

The algorithms are applied to assess fault vulnerability in two power system test cases: the IEEE 37-bus feeder modified to include induction motor loads, and the IEEE 39-bus system, which includes controller limits. In the former, the algorithms determine the onset of induction motor stalling, representing an important step towards characterizing behavior of Fault Induced Delayed Voltage Recovery (FIDVR). In the latter, the algorithms analyze system stability in the presence of background load with high levels of uncertainty and variable voltage dynamics.

The thesis proposal is organized as follows. Chapter II presents the theoretical development, which includes the counterexample to the historical lemma in Section 2.1. Also included this chapter are the two parts of the proof of RoA continuity for the class of vector fields discussed in Sections 2.2-2.3, along with their extensions to a class of hybrid systems with restricted switching and limit behavior in Section 2.4. Theoretical justification of the maximizing time in the ball around the controlling invariant set algorithm is provided in Section 2.5 and theoretical justification of the minimizing infimum over time of the inverse trajectory sensitivities algorithm in Section 2.6. This will conclude Chapter II Chapter III then presents the numerical algorithms, which includes efficient methods for integration of trajectory sensitivities in Section 3.1. It then presents the class of algorithms corresponding to maximizing time in the ball around the controlling invariant set in Section 3.2, and the class of algorithms corresponding to maximizing the supremum over time of the trajectory sensitivities in Section 3.3. Chapter IV presents the results of application of the above algorithms to assess fault vulnerability in power systems for a modified IEEE 37-bus feeder in Section 4.1 and the IEEE 39-bus system in Section 4.2. Chapter V offers some concluding remarks and future directions.

CHAPTER II

Theory

This chapter is devoted to the theoretical motivation of the algorithms described in Chapter III for numerically computing the recovery boundary, and for finding the closest point on the recovery boundary to some initial set of recovery values. The first step towards providing justification for the algorithms is to show that the recovery boundary consists entirely of critical parameter values and that the closest critical parameter value to any recovery value lies on the recovery boundary. This provides the essential link between the recovery boundary in parameter space and the RoA boundary in state space. To ensure this (assuming the IC is locally continuous) it suffices to show that the RoA varies continuously with small changes in parameter values.

Before turning to the main proofs regarding RoA boundary continuity, Section 2.1 provides a counterexample to a historical lemma which was instrumental in the proof that the RoA boundary is equal to the union of the stable manifolds of the critical elements it contains [CHW88]. To avoid the use of this flawed lemma, and to enable extensions to parameter variation, the assumptions of [CHW88] are strengthened slightly to consider C^1 vector fields which are Morse-Smale along the RoA boundary (with some additional assumptions for the noncompact case).

It is shown that, for this class of vector fields, under small variations in parameter values the RoA boundary varies continuously. The notion of Ω -stable means informally that the structure of the nonwandering set remains the same under perturbations. The continuity proof proceeds in two steps. First, in Section 2.2 the main result is proved under the additional assumption of Ω -stability within the RoA boundary. Then, in Section 2.3 it is shown that Ω -stability within the RoA boundary follows from the original assumptions (plus some additional assumptions in the noncompact case). These RoA boundary continuity results are extended in Section 2.4 to a class of hybrid systems with restricted switching and limit behavior.

In Sections 2.5-2.6, the RoA continuity results are first used to show that the recovery boundary consists entirely of critical parameter values and that the closest critical parameter value to any recovery value lies on the recovery boundary. Then, Section 2.5 develops theoretical motivation for a first class of algorithms by showing that, under suitable assumptions, the time the system trajectory spends in a neighborhood around the controlling invariant set is continuous in parameter and diverges to infinity as any critical parameter value is approached from within the set of recovery values.

Finally, Section 2.6 develops theoretical motivation for a second class of algorithms by showing that, under reasonable assumptions, the supremum over time of the trajectory sensitivity with respect to parameters is continuous over the set of recovery values and diverges to infinity as any critical parameter value is approached from within the set of recovery values.

2.1 Counterexample to Historical Work

2.1.1 Introduction

In the proofs of their superb and groundbreaking theorems, the authors of [CHW88] rely on a lemma which states that if the stable manifold of a first hyperbolic closed orbit intersects transversely the unstable manifold of a second (possibly the same) hyperbolic closed orbit, then the dimension of the unstable manifold of the first is strictly less than the dimension of the unstable manifold of the second. However, we exhibit an example meeting the conditions of the lemma where the dimensions of the unstable manifolds are equal, thereby disproving the lemma. In particular, we present a hyperbolic closed orbit of a smooth (C^{∞}) vector field over \mathbb{R}^3 whose stable and unstable manifolds have nonempty, transversal intersection.

2.1.2 Background

Consider the system:

$$\dot{x} = V(x) \tag{2.1}$$

of a C^1 vector field (V) over Euclidean space (\mathbb{R}^n) for some n > 0. In [CHW88, Theorem 4.2], the authors claim that if a stable equilibrium point of Eq. 2.1 satisfies their assumptions, then the boundary of its region of attraction (RoA) is equal to the union of the stable manifolds of
the equilibrium points and periodic orbits it contains. The main technical result behind their proof of this impressive theorem is [CHW88, Theorem 3-8]. In turn, [CHW88, Lemma 3-5] is crucial in their proof of [CHW88, Theorem 3-8]. This note is devoted to the construction of a counterexample to [CHW88, Lemma 3-5].

The authors of [CHW88] attribute their Lemma 3-5 to a survey paper by Smale [Sma67] and do not provide a proof of the lemma. However, we have not found the lemma in [Sma67]. Furthermore, in another paper by Smale [Sma60] he considers a set of assumptions in which his conditions 1-4 include that all equilibrium points and periodic orbits are hyperbolic, and their stable and unstable manifolds have transversal intersection. Yet Smale also includes condition 5, which states that for any closed orbit γ there does not exist $x \notin \gamma$ such that x converges to γ in both forwards and backwards time. In other words, condition 5 states for every closed orbit γ , its stable and unstable manifolds have empty intersection outside of γ . In addition, Smale writes: "It is true that conditions (1)-(5) are independent." If [CHW88, Lemma 3-5] was correct, then conditions 1-4 would imply condition 5 (because as a consequence of [CHW88, Lemma 3-5] hyperbolic closed orbits could not have nonempty, transversal intersections between their stable and unstable manifolds), so Smale's statement that conditions (1)-(5) are independent would necessarily be incorrect. It is therefore strange that [CHW88, Lemma 3-5] is attributed to Smale, since it would invalidate his own claims in [Sma60]. Recently, [CHW88, Lemma 3-5] was reprinted by one of the original authors [CA15, Lemma 4-4] without modification and without proof, and no reference was provided in this reprinting.

It should be noted that the example presented here does not contradict [CHW88, Theorem 4.2], and that of course the theorem may be correct as stated. However, we have not been able to supply a complete proof. We have recently shown that the result of Theorem 4.2 can be proved analogously to the original proof under a slightly stronger assumption than the original theorem: that the nonwandering set along the RoA boundary consists entirely of critical elements.

Furthermore, it is possible that under the additional assumptions of Theorem 4.2, which were not assumed in the statement of Lemma 3-5, the conclusion of Lemma 3-5 may still be true. However, we have not been able to prove this either and wonder whether, for example, it might be possible for the suspended horseshoe in the transversal intersection of the stable and unstable manifolds of a hyperbolic closed orbit to intersect the RoA boundary only in orbits which converge to critical elements (since a suspended horseshoe has many orbits which could not be present in the RoA boundary under the assumptions of Theorem 4.2). If such an example exists, then even under the additional assumptions of Theorem 4.2, the conclusion of Lemma 3-5 would be incorrect.

Before presenting [CHW88, Lemma 3-5] in detail, we provide some preliminary definitions. Let a critical element refer to either an equilibrium point or a closed orbit of Eq. 2.1. An equilibrium point (x_e) of Eq. 2.1 is hyperbolic if dV_{x_e} has no imaginary eigenvalues. A periodic orbit γ of Eq. 2.1 is hyperbolic if there exists $x \in \gamma$, a smooth cross section Scentered at x, and a C^1 first return map $\hat{\tau} : S \to S$ such that $d\hat{\tau}_x$ has no eigenvalues of norm one. Let γ be a hyperbolic critical element and define the stable manifold of γ $(W^s(\gamma))$ to be the set of all initial conditions which converge to γ in forwards time. Similarly, define the unstable manifold of γ $(W^u(\gamma))$ to be the set of all initial conditions which converge to γ in backwards time. Note that both $W^s(\gamma)$ and $W^u(\gamma)$ are invariant under the flow of V. Consider a differentiable submanifold (\hat{A}) of M, let $x \in \hat{A}$, and define the tangent space of \hat{A} at x $(T_x(\hat{A}))$. Consider another differentiable submanifold (\hat{B}) of M. Then \hat{A} and \hat{B} satisfy the transversality condition, or have transversal intersection, if either they are disjoint or for every $x \in \hat{A} \cap \hat{B}$, $T_x \hat{A} + T_x \hat{B} = T_x M$.

The exact statement of [CHW88, Lemma 3-5] is as follows:

Lemma 1. Let x_i and x_j be hyperbolic critical elements of Eq. 2.1. Suppose that the intersection of stable and unstable manifolds of x_i , x_j satisfy the transversality condition and $\{W^u(x_i) - x_i\} \cap \{W^s(x_j) - x_j\} \neq \emptyset$. Then dim $W^u(x_i) \geq \dim W^u(x_j)$, where the equality sign is true only when x_i is an equilibrium point and x_j is a closed orbit.

2.1.3 Example

To disprove Lemma 1, we will provide an example in which dim $W^u(x_i) = \dim W^u(x_j)$ but both x_i and x_j are closed (periodic) orbits. In particular, we will give an example where $x_i = x_j$ is a hyperbolic periodic orbit whose stable and unstable manifolds are transverse and $\{W^u(x_i) - x_i\} \cap \{W^s(x_j) - x_j\} \neq \emptyset$. Then dim $W^u(x_i) = \dim W^u(x_j)$ because $x_i = x_j$.

So, consider the classic example of the Duffing equation with negative linear stiffness, weak damping, and weak periodic forcing, which is given by the following C^{∞} nonautonomous vector field on \mathbb{R}^2 [GH83, p. 191]:

$$\dot{u} = v$$

 $\dot{v} = u - u^3 + \epsilon(\gamma \cos t - \delta v)$

where $\epsilon, \delta, \gamma \geq 0$ are parameters and $u, v \in \mathbb{R}$. At $\epsilon = 0$ this system possesses a hyperbolic



Figure 2.1: This figure originally appeared in [GH83, p. 208] and was computed numerically by Y. Ueda. It shows the stable and unstable manifolds of the first return map of a cross section of the Duffing equation for parameter values $\epsilon \delta = 0.25$ and $\epsilon \gamma = 0.30$. Note that the stable and unstable manifolds have nonempty, transversal intersection.

saddle equilibrium point at (0,0) whose stable manifold and unstable manifold are equal (they consist of homoclinic orbits). We can rewrite this system as an autonomous vector field on $\mathbb{R}^2 \times \mathbb{S}^1$ by introducing a time coordinate \mathcal{T} :

$$\dot{u} = v$$

 $\dot{v} = u - u^3 + \epsilon(\gamma \cos T - \delta v)$
 $\dot{T} = 1.$

Letting $w = [u \ v \ \mathcal{T}]^T$, we can write the above vector field as:

$$\dot{w} = V(w). \tag{2.2}$$

Note that V is a C^{∞} vector field. For $\epsilon = 0$, this system has a hyperbolic periodic orbit given by $\Gamma_0 = \{(0, 0, \mathcal{T}) : \mathcal{T} \in \mathbb{S}^1\}$ whose stable and unstable manifolds are each two-dimensional. For $\epsilon > 0$ sufficiently small, there exists a unique hyperbolic periodic orbit Γ_{ϵ} which is C^1 close to Γ_0 . Furthermore, there exists a closed curve $\Lambda \subset \mathbb{R}^2$ such that for any $\epsilon \ge 0$ small, the vector field is inward pointing on Λ for all times $\mathcal{T} \in \mathbb{S}^1$ [GH83, Exercise 1.5.5] and $\pi_{\mathbb{R}^2}(\Gamma_{\epsilon})$ is contained in the interior of the region bounded by Λ , where $\pi_{\mathbb{R}^2}$ is the projection map onto \mathbb{R}^2 . This implies that $\pi_{\mathbb{R}^2}(W^u(\Gamma_{\epsilon}))$ is contained in the interior of the region bounded by Λ .

For any $\mathcal{T} \in \mathbb{S}^1$, let $S_{\mathcal{T}} = \mathbb{R}^2 \times \{\mathcal{T}\}$ be a smooth cross section. Define the Poincaré first return map $(\hat{\tau})$, where $\hat{\tau} : S_{\mathcal{T}} \to S_{\mathcal{T}}$, and note that it is well-defined and C^{∞} . Let $p_{\epsilon}^{\mathcal{T}} = \Gamma_{\epsilon} \cap S_{\mathcal{T}}$, which is a single point. Then it is straightforward to see that $p_{\epsilon}^{\mathcal{T}}$ is a hyperbolic fixed point of V. Let $W^s(p_{\epsilon}^{\mathcal{T}})$ and $W^u(p_{\epsilon}^{\mathcal{T}})$ denote its stable and unstable manifolds, respectively, in $S_{\mathcal{T}}$. Using Melnikov's method [GH83, Theorem 4.5.3], it can be shown [GH83, p. 193] that for any $\mathcal{T} \in \mathbb{S}^1$, $\epsilon > 0$ sufficiently small, and $\frac{\gamma}{\delta}$ sufficiently large, $W^s(p_{\epsilon}^{\mathcal{T}})$ and $W^u(p_{\epsilon}^{\mathcal{T}})$ are transverse and $\{W^s(p_{\epsilon}^{\mathcal{T}}) - p_{\epsilon}^{\mathcal{T}}\} \cap \{W^u(p_{\epsilon}^{\mathcal{T}}) - p_{\epsilon}^{\mathcal{T}}\} \neq \emptyset$. Fig. 2.1 (originally appearing in [GH83, p. 208], which was computed numerically by Y. Ueda) shows $W^s(p_{\epsilon}^{\mathcal{T}})$ and $W^u(p_{\epsilon}^{\mathcal{T}}) - p_{\epsilon}^{\mathcal{T}}\}$ and $\{W^u(p_{\epsilon}^{\mathcal{T}}) - p_{\epsilon}^{\mathcal{T}}\}$ have nonempty, transversal intersection. As $\mathcal{T} \in \mathbb{S}^1$ was arbitrary, it follows that $W^s(\Gamma_{\epsilon})$ and $W^u(\Gamma_{\epsilon})$ are transverse and $\{W^s(\Gamma_{\epsilon}) - \Gamma_{\epsilon}\} \cap \{W^u(\Gamma_{\epsilon}) - \Gamma_{\epsilon}\} \neq \emptyset$. Hence, we may fix $\epsilon, \gamma, \delta > 0$ such that $W^s(\Gamma_{\epsilon})$ and $W^u(\Gamma_{\epsilon})$ are transverse and $\{W^s(\Gamma_{\epsilon}) - \Gamma_{\epsilon}\} \neq \emptyset$.

However, the vector field above is over $\mathbb{R}^2 \times \mathbb{S}^1$, whereas Lemma 1 is stated for a vector field over Euclidean space. So, we will modify the example to obtain a similar vector field over Euclidean space. As $W^s(\Gamma_{\epsilon})$ and $W^u(\Gamma_{\epsilon})$ are transverse, $W^s_{\text{loc}}(\Gamma_{\epsilon})$ and $W^u(\Gamma_{\epsilon})$ are transverse, where $W^s_{\text{loc}}(\Gamma_{\epsilon})$ is the local stable manifold of Γ_{ϵ} . As $\{W^s(\Gamma_{\epsilon}) - \Gamma_{\epsilon}\} \cap \{W^u(\Gamma_{\epsilon}) - \Gamma_{\epsilon}\} \neq \emptyset$, by invariance $\{W^s_{\text{loc}}(\Gamma_{\epsilon}) - \Gamma_{\epsilon}\} \cap \{W^u(\Gamma_{\epsilon}) - \Gamma_{\epsilon}\} \neq \emptyset$. For any s > 0 let B(s) denote the open ball of radius s centered at the origin in \mathbb{R}^2 . As Λ is a closed curve in \mathbb{R}^2 , choose $\mu_1 > \mu_2 > 0$ such that $\Lambda \subset B(\mu_1 - \mu_2)$. As $\pi_{\mathbb{R}^2}(W^s_{\text{loc}}(\Gamma_{\epsilon}))$ and $\pi_{\mathbb{R}^2}(W^u(\Gamma_{\epsilon}))$ are contained in the interior of the region bounded by Λ , they are contained in $B(\mu_1 - \mu_2)$ as well. For any b > a > 0, let $\mathcal{S}_{(a,b)} : [0, \infty) \to [0, 1]$ denote a C^{∞} bump function with $\mathcal{S}^{-1}_{(a,b)}(1) = [0, a]$ and $\mathcal{S}^{-1}_{(a,b)}(0) = [b, \infty)$. Define the following vector field for $w \in \mathbb{R}^2 \times \mathbb{S}^1$:

$$\dot{w} = \mathcal{S}_{(\mu_1 - \mu_2, \mu_1)} \left(\sqrt{u^2 + v^2} \right) V(w).$$
(2.3)

As Eq. 2.3 and Eq. 2.2 agree for all points (u, v, t) such that $\pi_{\mathbb{R}^2}(u, v, t) \in B(\mu_1 - \mu_2)$, and since $\pi_{\mathbb{R}^2}(W^s_{\text{loc}}(\Gamma_{\epsilon}))$ and $\pi_{\mathbb{R}^2}(W^u(\Gamma_{\epsilon}))$ corresponding to the vector field of Eq. 2.2 are contained in $B(\mu_1 - \mu_2)$, $W^s_{\text{loc}}(\Gamma_{\epsilon})$ and $W^u(\Gamma_{\epsilon})$ corresponding to the vector field of Eq. 2.3 are equal to those of Eq. 2.2. In particular, this implies that $W^s_{\text{loc}}(\Gamma_{\epsilon})$ and $W^u(\Gamma_{\epsilon})$ of Eq. 2.3 are transverse, and also that they satisfy $\{W^s_{\text{loc}}(\Gamma_{\epsilon}) - \Gamma_{\epsilon}\} \cap \{W^u(\Gamma_{\epsilon}) - \Gamma_{\epsilon}\} \neq \emptyset$. Hence, for Eq. 2.3, $W^s(\Gamma_{\epsilon})$ and $W^u(\Gamma_{\epsilon})$ are transverse and $\{W^s(\Gamma_{\epsilon}) - \Gamma_{\epsilon}\} \cap \{W^u(\Gamma_{\epsilon}) - \Gamma_{\epsilon}\} \neq \emptyset$.

Now we construct the desired vector field over \mathbb{R}^3 . The key idea is to use Eq. 2.3 to construct a vector field over an embedding of $B(\mu_1) \times \mathbb{S}^1$ into \mathbb{R}^3 , and then to extend it to a C^{∞} vector field over \mathbb{R}^3 by defining it to be zero on the complement of the embedding of $B(\mu_1) \times \mathbb{S}^1$. This vector field will then be the desired counterexample to Lemma 1. We will use standard cylindrical coordinates $x := (r, \theta, z)$ to construct this vector field. Let

$$r = u + \mu_1 \tag{2.4}$$

$$\theta = \mathcal{T} \tag{2.5}$$

$$z = v. (2.6)$$

Then we define

$$\dot{x} = \mathcal{S}_{(\mu_1 - \mu_2, \mu_1)} \left(\sqrt{u^2 + v^2} \right) V(w) = \mathcal{S}_{(\mu_1 - \mu_2, \mu_1)} \left(\sqrt{(r - \mu_1)^2 + z^2} \right) V(x)$$
(2.7)

where

$$V(x) = \begin{bmatrix} \dot{r} \\ \dot{\theta} \\ \dot{z} \end{bmatrix} = \begin{bmatrix} z \\ 1 \\ (r - \mu_1) - (r - \mu_1)^3 + \epsilon(\gamma \cos \theta - \delta z) \end{bmatrix}$$

where V(x) was obtained from V(w) using Eqs. 2.4-2.6. Note that Eq. 2.7 defines a C^{∞} vector field on \mathbb{R}^3 . As the supports of Eq. 2.7 and Eq. 2.3 are diffeomorphic, and the two vector fields agree on their supports up to this diffeomorphism, Eq. 2.7 possesses a hyperbolic periodic orbit Γ such that $W^s(\Gamma)$ and $W^u(\Gamma)$ are transverse and $\{W^s(\Gamma) - \Gamma\} \cap \{W^u(\Gamma) - \Gamma\} \neq \emptyset$. Therefore, Eq. 2.7 is a counterexample to [CHW88, Lemma 3-5].

2.1.4 Conclusion

This brief section was devoted to the construction of a counterexample to [CHW88, Lemma 3-5], which was necessary in the proof of [CHW88, Theorem 4.2] regarding a decomposition of the region of attraction boundary into a union of stable manifolds. As one goal of the following sections will be to show that this decomposition persists under parameter perturbations, it will first be necessary to complete the original proof of [CHW88, Theorem 4.2]. To do so, the assumptions of [CHW88] will be slightly modified; this will ensure both that the original result of Theorem 4.2 holds, and that it persists under suitable parameter perturbation.

2.2 Continuity of Region of Attraction Boundary Under Perturbation

2.2.1 Introduction

The main technical challenge behind the theoretical justification of the desired algorithms is to show that the RoA boundary varies continuously in an appropriate sense under small changes in parameter values (see Corollaries 2.2.24-2.2.26). For, Example 2.2.1 shows that when the boundary of the RoA is discontinuous at a particular parameter value, then it is possible for the ICs to "jump" over the RoA boundary. In this case, the recovery boundary may not consist of critical parameter values and, in fact, its possible that no critical parameter values even exist. The former implies that computation of critical parameter values may not provide an accurate estimate of the recovery boundary, and hence of the set of recovery values, and the latter implies that any attempt to compute critical parameters must fail since they don't exist; both are problematic for the algorithms which will be developed below. Furthermore, discontinuity of the RoA boundary implies that there may not exist a controlling invariant set in the RoA boundary with the property that as parameter values approach critical values, the time the trajectory spends in a ball around the controlling invariant set diverges to infinity. Similarly, discontinuity of the RoA boundary may imply that as parameter values approach critical values, the supremum over time of the trajectory sensitivites may not diverge to infinity. Hence, if the recovery boundary does not consist of critical parameter values, the central strategy of the algorithms developed below may be unable to compute them. Therefore, establishing continuity of the RoA boundary for a large class of parameter dependent vector fields is crucial for motivating those algorithms.

This section is devoted to part one of the proof of continuity of RoA boundary: assuming the RoA boundary is Ω -stable (and the vector field is Morse-Smale along it), we will show in this section that it varies continuously with respect to small changes in parameter values. Section 2.3 shows that if the vector field is Morse-Smale along the RoA boundary then the RoA boundary is Ω -stable, thereby completing the proof. The stable manifold of a critical element is the set of initial conditions in state space which converge to that critical element in forwards time. The approach that is used to establish continuity of the RoA boundary is to show that at a fixed parameter value the RoA boundary is equal to the union of the stable manifolds of the critical elements it contains, and that this decomposition persists for small changes in parameter values, for a large class of parameter dependent vector fields (see Theorem 2.2.23). Earlier work [CHW88] reported this decomposition result for a large class of fixed parameter C^1 vector fields on Euclidean space. However, their proof relied on a Lemma [CHW88, Lemma 3-5] which has been disproven (see Section 2.1). Therefore, we begin by providing a complete proof for a fixed parameter RoA boundary decomposition result, and then focus on our main goal of extending this work to a continuous (in an appropriate sense - see below) parameterized family of C^1 vector fields on either Euclidean space or a compact Riemannian manifold.

The section is organized as follows. Section 2.2.2 presents relevant background and notation conventions. Section 2.2.3 provides a motivating example for discontinuity of the RoA boundary and the negative implications this can have. Section 2.2.4 gives the main results, focusing on parameter dependent vector fields, although results for parameter independent vector fields are also included. A simple example is provided to illustrate the main theorems. Section 2.2.5 proves the boundary decomposition results for the case where the vector field is parameter indepedent. Section 2.2.6 builds on the foundation of Section 2.2.5 to prove persistence of the boundary decomposition and continuity of the RoA boundaries for a large class of parameter dependent vector fields. Finally, Section 2.2.7 offers some concluding thoughts and future directions.

2.2.2 Notation and Definitions

Let E be a subset of a topological space. Then we define the topological closure of E(E), the topological boundary of $E(\partial E)$, and the topological interior of $E(\operatorname{int} E)$. If $\tilde{f}: A \to B$ is any function and $E \subset A$ is any subset of A, define the restriction of the function \tilde{f} to $E(\tilde{f}|_E)$ to be $\tilde{f}': E \to B$ where $\tilde{f}'(e) = \tilde{f}(e)$ for all $e \in E$. For a set D contained in a metric space K, define the ϵ -neighborhood of $D(D_{\epsilon})$ to be the set of $x \in K$ such that for each x there exists $y \in E$ with $d(x, y) < \epsilon$. Let $\{x_n\}_{n=1}^{\infty}$ be a sequence. Let $\{n_m\}_{m=1}^{\infty}$ be any collection of positive integers where we require that m' > m implies that $n_{m'} > n_m$ to ensure the ordering is preserved. Then any subsequence of $\{x_n\}_{n=1}^{\infty}$ can be written as $\{x_{n_m}\}_{m=1}^{\infty}$ for some choice of $\{n_m\}_{m=1}^{\infty}$. Let K be a nonempty, compact metric space. Note that compact Riemmanian manifolds are compact metric spaces, so the discussion below will apply to them. Define the set of nonempty closed subsets of $K(\mathcal{C}(K))$. Let $K_1, K_2 \in \mathcal{C}(K)$. We define the Hausdorff distance (d_h) by

$$d_h(K_1, K_2) = \inf\{r \ge 0 : K_1 \subset (K_2)_r, K_2 \subset (K_1)_r\}.$$

Then d_h is a well-defined metric on $\mathcal{C}(K)$ [Hau57, Section 28] and we say a sequence of sets $A_n \in \mathcal{C}(K)$ converges to $A \in \mathcal{C}(K)$, denoted $A_n \to A$, if $\lim_{n\to\infty} d_h(A_n, A) = 0$. For A, B subsets of a metric space with metric d, define a set distance (d_S) by $d_S(A, B) = \inf\{d(a, b) : a \in A, b \in B\}$. Then if A is compact, B is closed, and d(A, B) = 0, A and B must have nonempty intersection. Note that any Riemannian manifold is also a metric space, so this set distance is well-defined on Riemannian manifolds.

Let J be a topological space. For $p \in J$, we say that p has a countable neighborhood basis if there exists a countable collection $\{U_n\}_{n=1}^{\infty}$ of open sets in J that contain p such that for any open set \hat{U} which contains p, there exists n such that $U_n \subset \hat{U}$. Then we say that Jis first countable if every point $p \in J$ possesses a countable neighborhood basis. If J is first countable, E is any topological space, and $\tilde{f}: J \to E$ is a function, then \tilde{f} is continuous if and only if for every convergent sequence $\{p_n\}_{n=1}^{\infty}$ in J, say $p_n \to p$, $\tilde{f}(p_n) \to \tilde{f}(p)$. Let J be a first countable topological space and let $\tilde{F}: J \to \mathcal{C}(K)$. We say that the family $\{A_p\}_{p\in J}$ is a Hausdorff continuous family of subsets of K if there exists $\tilde{F}: J \to \mathcal{C}(K)$ such that $\tilde{F}(p) = A_p$ for $p \in J$ and \tilde{F} is continuous. Since J is first countable, \tilde{F} is continuous if and only if for every $p \in J$ and every sequence $p_n \in J$ with $p_n \to p$, $\tilde{F}(p_n) \to \tilde{F}(p)$.

We consider another notion of convergence on $\mathcal{C}(K)$. Let $A_n \in \mathcal{C}(K)$ be a sequence of sets. Define $\liminf_{n\to\infty} A_n$ to be the set of points $x \in K$ such that there exists a sequence $\{a_n\}$, with $a_n \in A_n$ for all n, such that $a_n \to x$. Define $\limsup_{n\to\infty} A_n$ to be the set of points $x \in K$ such that there exist $\{a_{n_m}\}$ with $a_{n_m} \in A_{n_m}$ a subsequence of $\{A_n\}$, such that $a_{n_m} \to x$. Both $\limsup_{n\to\infty} A_n$ and $\liminf_{n\to\infty} A_n$ are closed [Hau57, Section 28] and $\limsup_{n\to\infty} A_n$ is nonempty since K is sequentially compact, so if $\liminf_{n\to\infty} A_n$ is nonempty then both are elements of $\mathcal{C}(K)$. By definition, $\liminf_{n\to\infty} A_n \subset \limsup_{n\to\infty} A_n$. If $\limsup_{n\to\infty} A_n \subset$ $\liminf_{n\to\infty} A_n$ then we say the limit exists and $\lim_{n\to\infty} A_n = \limsup_{n\to\infty} A_n = \liminf_{n\to\infty} A_n$. By statement V of [Hau57, Section 28], since K is compact, if

 $\limsup_{n\to\infty} A_n = \liminf_{n\to\infty} A_n = \lim_{n\to\infty} A_n = \lim_{n\to\infty} A_n =: A \text{ then } d_h(A_n, A) \to 0. \text{ Thus, if there exists } \tilde{F} : J \to \mathcal{C}(K) \text{ such that for every } p \in J \text{ and every } p_n \to p, \limsup_{n\to\infty} \tilde{F}(p_n) = \lim_{n\to\infty} \tilde{F}(p_n) = \tilde{F}(p), \text{ then } d_h(\tilde{F}(p_n), \tilde{F}(p)) \to 0 \text{ so } \{\tilde{F}(p)\}_{p\in J} \text{ is a Hausdorff continuous family of subsets of } K.$

Let $M = \mathbb{R}^n$, and let $\mathcal{C}(M)$ be the closed, nonempty subsets of M. The standard Hausdorff distance is not well-defined for unbounded sets, so instead consider the one-point compactification of M, $\mathbb{R}^n \cup \infty \cong \mathbb{S}^n$, where \mathbb{S}^n is the *n*-sphere. Equip \mathbb{S}^n with the induced Riemannian metric from its inclusion into \mathbb{R}^{n+1} , and let its associated distance function be the desired metric on $M \cup \infty$. Then, since \mathbb{S}^n is a compact, nonempty metric space, the Hausdorff distance is well-defined for all closed, nonempty subsets of \mathbb{S}^n . Let $\overline{\mathcal{C}}(M) = \{A \cup \{\infty\} : A \in \mathcal{C}(M)\}$. Then all sets in $\overline{\mathcal{C}}(M)$ are closed and nonempty, so the Hausdorff distance is well-defined on $\overline{\mathcal{C}}(M)$, and the metric topology it induces on $\overline{\mathcal{C}}(M)$ is called the Chabauty topology. From the discussion above regarding Hausdorff continuity, it follows that if there exists $\tilde{F} : J \to \overline{\mathcal{C}}(M)$ such that for every $p \in J$ and every $p_n \to p$, $\limsup_{n \to \infty} \tilde{F}(p_n) = \liminf_{n \to \infty} \tilde{F}(p_n) = \tilde{F}(p)$, then $d_h(\tilde{F}(p_n), \tilde{F}(p)) \to 0$ so $\{\tilde{F}(p)\}_{p \in J}$ is a Chabauty continuous family of subsets of M.

Let V be a C^1 vector field on a Riemannian manifold M. An integral curve γ of V is a map from an open subset $U \subset \mathbb{R}$ to M such that for every $t \in U$, $\frac{d}{dt}\gamma(t) = V_{\gamma(t)}$. A flow (ϕ) is a map $\phi : U \times M \to M$, where $U \subset \mathbb{R}$ is open, such that for any $x \in M$, $\phi(\cdot, x)$ is an integral curve of V. For any C^1 vector field V on a Riemannian manifold M, there exists a C^1 flow ϕ [Lee13, Theorem 9.12]. We say that V is complete if possesses a flow ϕ defined on $\mathbb{R} \times M$. For $t \in \mathbb{R}$, we define the time-t flow (ϕ_t) , where $\phi_t : M \to M$, by $\phi_t(x) = \phi(t, x)$. Then acphit is a C^1 diffeomorphism of M for any t since ϕ is C^1 and $\phi_t^{-1} = \phi_{-t}$.

Consider a smooth manifold (\tilde{M}) . If M and \tilde{M} are Riemannian manifolds, let $C^1(M, \tilde{M})$ denote the set of C^1 maps from M to \tilde{M} . There are two common topologies that $C^1(M, \tilde{M})$ can be equipped with: the strong and weak C^1 topologies. Full definitions of these are available in [Hir76, Chapter 2], but the properties of these topologies which are most important for this work are summarized below. The purpose of introducing these topologies is to provide a framework for careful consideration of perturbations to vector fields, and to be able to define a continuous family of vector fields in a suitable way. We will typically equip $C^1(M, \tilde{M})$ with the weak topology, denoted $C^1_W(M, \tilde{M})$. A major benefit of the weak topology is that it has a complete metric, the C^1 distance (d_{C^1}) . If V is a C^1 vector field on M then $V \in C^1(M, TM)$. A pair of C^1 vector fields V, \tilde{V} on M are ϵC^1 -close if $d_{C^1}(V, \tilde{V}) < \epsilon$, where d_{C^1} is the C^1 distance on $C^1(M, TM)$. A (weak) C^1 perturbation to the vector field V is a vector field \tilde{V} such that V, \tilde{V} are ϵC^1 -close for sufficiently small $\epsilon > 0$. A parameterized family of C^1 vector fields $\{V_p\}_{p\in J}$ on M is (weakly) C^1 continuous if the induced map $J \to C^1_W(M, TM)$ that sends p to the vector field corresponding to parameter value $p(V_p)$ is continuous. Note that this implies that $V: M \times J \to TM \times TJ$ defined by $V(x,p) = (V_p(x),0)$ is a C^1 vector field on $M \times J$. In the case of M compact, the strong and weak topologies on $C^1(M, \tilde{M})$ coincide. For M noncompact, a strong C^1 perturbation to a vector field only involves changes to that vector field on a compact set, whereas a weak C^1 perturbation to that vector field could have changes that are unbounded. Hence, weak continuity of vector fields is a weaker assumption than strong continuity.

Let \hat{A} be a Riemannian manifold. Then for each $x \in \hat{A}$, let $T_x \hat{A}$ denote the tangent space to \hat{A} at x. Let $T\hat{A} = \bigsqcup_{x \in \hat{A}} T_x \hat{A}$ denote the tangent bundle, and note that it is naturally a manifold with dimension twice that of \hat{A} . Let the zero section be the subspace of $T\hat{A}$ consisting of the zero vector from each tangent space $T_x \hat{A}$ over $x \in \hat{A}$ (note that it is naturally diffeomorphic to \hat{A} itself). Let $\hat{B} \subset \hat{A}$. For any function $\tilde{f} : \hat{B} \to \hat{A}$ and $x \in \hat{B}$, define the differential of \tilde{f} at x $(d\tilde{f}_x)$. Let $i:\hat{B}\to \hat{A}$ by i(x)=x be the inclusion map. We say that i is a C^1 immersion if it is C^1 and for every $y \in \hat{B}$, di_y is injective. Note that a function $\tilde{f}: \hat{A} \to \hat{B}$, where \hat{A} and \hat{B} are C^1 manifolds, is a submersion if $d\tilde{f}_y$ is surjective for every $y \in M$. Then we say \hat{B} is an immersed submanifold if its inclusion map $i: \hat{B} \to \hat{A}$ is a C^1 immersion. Let $T_{\hat{B}}M = \bigsqcup_{x \in \hat{B}}T_xM$ denote the tangent bundle of M over \hat{B} . Consider a pair of C^1 immersed submanifolds \hat{A} and \hat{B} . We say that \hat{A} and \hat{B} are transverse at a point $x \in \hat{A} \cap \hat{B}$ if $T_x \hat{A} \oplus T_x \hat{B}$ spans $T_x M$. Then we say that \hat{A} and \hat{B} are transverse if for every $x \in \hat{A} \cap \hat{B}$, \hat{A} and \hat{B} are transverse at x. Note that if \hat{A} and \hat{B} are disjoint, they are vacuously transverse. A C^1 disk is the image of $i: B \to M$ where $B \subset \mathbb{R}^m$ is a closed ball around the origin in some Euclidean space \mathbb{R}^m , and i is a C^1 immersion. A continuous family of C^1 disks is a parameterized family $\{D_x\}_{x\in E}$ where E is a topological space, D_x is a C^1 disk for each x, and $\{D_x\}_{x\in S}$ is a Hausdorff continuous family. Suppose that \hat{A} is a C^1 immersed submanifold of \hat{B} , which is a C^1 immersed submanifold of M. By the tubular neighborhood theorem [Lee13, Theorem 6.24], there exists a C^1 continuous family of pairwise disjoint disks $\{D(x)\}_{x\in\hat{B}}$ in M centered along \hat{B} and transverse to it such that their union is an open neighborhood of \hat{B} in M. Taking the restriction $\{D(x)\}_{x\in \hat{A}}$ gives a C^1 continuous family of pairwise disjoint disks in M centered along \hat{A} and transverse to \hat{B} . If $\tilde{F}: \hat{A} \to \hat{B}$ is a continuous and injective map between manifolds \hat{A} and \hat{B} of the same dimension, then by invariance of domain [Hat01, Theorem 2B.3], \tilde{F} is an open map, which means the image under \tilde{F} of every open set is open.

There is a notion of a generic C^1 vector field, which is meant to represent typical behavior, similar to the idea of probability one in a probability space. If a property holds for a generic class of C^1 vector fields, it is therefore considered to be typical or usual behavior. As there exist many pathological C^1 vector fields, it is often advantageous to restrict attention to certain classes of generic C^1 vector fields when possible, and to prove results for generic vector fields that often would not hold for arbitrary vector fields. We follow this approach here. In a topological space E, a Baire set is a countable intersection of open, dense subsets of E. A topological space E is metrizable if there exists a metric on E whose metric topology corresponds with the original topology on E. It is completely metrizable if the resultant metric space is complete. The Baire category theorem states that if the topology on E is completely metrizable, then every Baire set in E is dense. By the discussion above, $C^1(M, \tilde{M})$ is a complete metric space for M and \tilde{M} under consideration here, so every Baire set will be dense. Suppose \mathcal{P} is a property that may be possessed by elements of a topological space E. Then \mathcal{P} is called a generic property if the set of elements in M which possess the property \mathcal{P} contains a Baire set in M. So, a property of vector fields is generic (with respect to the weak topology) if the subset of vector fields in $C^1_W(M, TM)$ that possess this property contains a Baire set.

An equilibrium point $x_e \in M$ is a singularity of the vector field, i.e. $V(x_e) = 0$. A periodic orbit $\gamma \subset M$ is an integral curve of V such that there exists T > 0 such that each point of γ is a fixed point of ϕ_T . For each point $x \in \gamma$, there exists a codimension-one embedded submanifold transverse to the flow called a Poincaré cross section (S) and a neighborhood \hat{U} of x in S such that the Poincaré first return map $\hat{\tau} : \hat{U} \to S$ is well-defined and C^1 [HS74, Page 281]. Define a critical element (X) in M to be either an equilibrium point or a periodic orbit of V.

A set $E \subset M$ is forward invariant if $\phi_t(E) \subset E$ for all t > 0. It is backward invariant if $\phi_t(E) \subset E$ for all t < 0, and invariant if it is both forward and backward invariant. Note that critical elements are invariant.

Let x_e be an equilibrium point. Then x_e is hyperbolic if $d(\phi_1)_{x_e}$ is a hyperbolic linear map, i.e. if it has no eigenvalues of modulus one. It is stable if every eigenvalue of $d(\phi_1)_{x_e}$ has modulus less than one. If γ is a periodic orbit then let $x \in \gamma$, S a cross section centered at x, \hat{U} a neighborhood of x in S, and $\hat{\tau} : \hat{U} \to S$ the C^1 first return map. Then γ is hyperbolic if $d\hat{\tau}_x$ is a hyperbolic linear map.

If $X \,\subset M$ is a hyperbolic critical element then there exist a local stable manifold of X $(W^s_{\text{loc}}(X))$ and a local unstable manifold of X $(W^u_{\text{loc}}(X))$ [KH99, Chapter 6] such that $\phi_t(W^s_{\text{loc}}(X)) \,\subset W^s_{\text{loc}}(X)$ and $\phi_{-t}(W^u_{\text{loc}}(X)) \,\subset W^u_{\text{loc}}(X)$ for all t > 0. Furthermore, the local stable and unstable manifolds are chosen to be compact. The global stable manifold of X $(W^s(X))$ and unstable manifold of X $(W^u(X))$ are then defined as $W^s(X) = \bigcup_{t \leq 0} \phi_t(W^s_{\text{loc}}(X))$ and $W^u(X) = \bigcup_{t \geq 0} \phi_t(W^u_{\text{loc}}(X))$, respectively. By [KH99, Chapter 6], $W^s(X)$ consists of the set of $x \in M$ such that the forward time orbit of x converges to X, and $W^u(X)$ consists of the set of $x \in M$ such that the backward time orbit of x converges to X. Note that they are invariant under the flow. If X is a hyperbolic periodic orbit and S is a cross section of X with C^1 first return map $\hat{\tau}$, it is often convenient to consider $W^s_{\text{loc}}(X) \cap S$ and $W^u_{\text{loc}}(X) \cap S$. Then $\hat{\tau}(W^s_{\text{loc}}(X) \cap S) \subset W^s_{\text{loc}}(X) \cap S$ and $\hat{\tau}^{-1}(W^u_{\text{loc}}(X) \cap S) \subset W^u_{\text{loc}}(X) \cap S$.

Hence, we abuse notation and let $W^s_{\text{loc}}(X)$ refer either to $W^s_{\text{loc}}(X)$ as defined above or to $W^s_{\text{loc}}(X) \cap S$ for some cross section S of X. The distinction should be clear from context. Define the notation $W^s_{\text{loc}}(X)$ analogously.

Let X be a hyperbolic critical element with $(W^s(X) - X) \cap (W^u(X) - X) \neq \emptyset$. Then the orbit of each $x \in (W^s(X) - X) \cap (W^u(X) - X)$ is called a homoclinic orbit. If, in addition, $W^s(X)$ and $W^u(X)$ have nonempty, transversal intersection, then the orbit of each $x \in (W^s(X) - X) \cap (W^u(X) - X)$ is called a transverse homoclinic orbit. Let X, Y be hyperbolic critical elements with $(W^s(X) - X) \cap (W^u(Y) - Y) \neq \emptyset$. Then the orbit of each $x \in (W^s(X) - X) \cap (W^u(Y) - Y)$ is called a heteroclinic orbit, and it is called a transverse heteroclinic orbit if the intersection is transverse. Let $X^1, ..., X^n$ be a finite set of hyperbolic critical elements with $X^n = X^1$. If $(W^s(X^i) - X^i) \cap (W^u(X^{i+1}) - X^{i+1})$ is nonempty and transverse for each $i \in \{1, ..., n-1\}$, then we call $\{X^i\}_{i=1}^n$ a heteroclinic cycle. If $X^1, X^2, ...$ is a sequence of hyperbolic critical elements with $(W^s(X^i) - X^i) \cap (W^u(X^{i+1}) - X^{i+1})$ nonempty and transverse for all i, then we call $\{X^i\}_{i=1}^\infty$ a heteroclinic sequence.

Let X be a hyperbolic critical element. If X is an equilibrium point, let $B = W_{loc}^u(X)$, let D be a C^1 disk in M such that D has nonempty, transversal intersection with $W^s(X)$, and let $\hat{f} = \phi_1$ be the time-one flow. If X is a periodic orbit, let $B = W_{loc}^u(X) \cap S$, where S is a cross section of X, let D be a C^1 disk in S such that D has nonempty, transversal intersection in S with $W^s(X) \cap S$, and let \hat{f} be the C^1 first return map defined on an open subset of S. Suppose dim $D \ge \dim B$. Let $q \in D \cap W^s(X)$. Then the Inclination Lemma, otherwise known as the Lambda Lemma, states [Pal69] that for every $\epsilon > 0$ there exists $n_0 > 0$ such that $n \ge n_0$ implies a submanifold of $\hat{f}^n(D)$ containing $\hat{f}^n(q)$ is ϵC^1 -close to B. For convenience, we often omit the submanifold qualifier and implicitly redefine (shrink) D so that $\hat{f}^n(D)$ itself is ϵC^1 -close to B.

Let V be a C^1 vector field on a Riemannian manifold M with corresponding flow ϕ . A point $x \in M$ is nonwandering for V if for every open neighborhood U of x and every T > 0, there exists t > T such that $\phi_t(U) \cap U \neq \emptyset$. Let the nonwandering set under $V(\Omega(V))$ be the set of nonwandering points for V in M where, for brevity, we often omit V and write Ω if the vector field is understood from context. If $y \in M$ and V is a vector field on M, define the ω -limit set of y under $V(\omega(y, V))$ to be the set of points $x \in M$ such that there exists a sequence $t_i \to \infty$ with $\phi_{t_i}(y) \to x$. Let $\omega(V) = \bigcup_{y \in M} \omega(y, V)$. For convenience, we often suppress the dependence on the vector field V when it is clear from context, and write $\omega(x)$ instead. If $\gamma \subset M$ is an orbit, define its ω -limit set to be the ω -limit set of any $y \in \gamma$, and note that this is well-defined because all points on an orbit share the same ω -limit set.

Define the α -limit set of an orbit analogously, for $t_i \to -\infty$. Write $\omega(\gamma)$ and $\alpha(\gamma)$ for the ω -limit set and α -limit set, respectively, of the orbit γ . V is a Morse-Smale vector field if it satisfies:

- 1. $\Omega(V)$ is a finite union of critical elements.
- 2. Every critical element is hyperbolic.
- 3. The stable and unstable manifolds of each individual and all pairs of critical elements have transversal intersection.

By [Sma62, Kup63], Assumptions 2 and 3 are generic, whereas Assumption 1 is not. Note that Morse-Smale vector fields were defined for compact Riemannian manifolds [Sma60]. We will see below that an additional assumption is necessary for Euclidean space.

Define a connected smooth manifold which represents parameter values (J) and fix $p_0 \in J$. Let $J_r = \{p \in J : |p - p_0| < r\}$ and $J_{\overline{r}} = \{p \in J : |p - p_0| \le r\}$. For $Q \subset J$, let $M_Q = M \times Q$ and let $M_p = M_{\{p\}}$. Let $\{V_p\}_{p \in J}$ be a C^1 continuous family of vector fields on M. Suppose X_{p_0} is a hyperbolic critical element of V_{p_0} for some $p_0 \in J$. Then for J sufficiently small, $p \in J$ implies that there exists a unique hyperbolic critical element corresponding to parameter value $p(X_p)$ of V_p which is C^1 -close to X_{p_0} . This defines the family $\{X_p\}_{p\in J}$ of a critical element of the vector fields $\{V_p\}_{p\in J}$. To avoid ambiguity, we reserve the phrase "family of a critical element" to refer to the family obtained from a single critical element as the parameter value p varies over J. In particular, this implies that for each fixed parameter value p, the family of a critical element will possess exactly one critical element of V_p . Throughout the paper, for a fixed parameter value $p \in J$, it will sometimes be convenient to think of a critical element X_p as being a subset of M, and sometimes as a subset of $M \times J$. Therefore, we abuse notation and let X_p denote a critical element of V_p , where sometimes we consider $X_p \subset M$ and sometimes we consider $X_p \subset M \times \{p\} \subset M \times J$. The distinction should be clear from context. For $Q \subset J$, we write $\bigcup_{p \in Q} X_p(X_Q) \subset M \times J$. We write $\sqcup_{p \in Q} W^s(X_p)$ $(W^s(X_Q)) \subset M \times J$, and $\sqcup_{p \in Q} W^u(X_p)$ $(W^u(X_Q)) \subset M \times J$.

2.2.3 Motivating Example

Example 2.2.1 (Lack of Hausdorff Continuity of Boundaries). We show that every smooth manifold M possesses a family of smooth vector fields which is continuous with respect to the strong C^{∞} topology and such that the vector fields have a family of stable equilibria whose boundaries of their regions of attraction are not Hausdorff continuous. As the strong

 C^{∞} topology is the most restrictive of the standard C^r topologies, this implies that even such a high degree of regularity is not sufficient to prevent a lack of Hausdorff continuity of the boundaries. We define the family of vector fields such that they are supported within a single chart, and then extend them trivially to the entire manifold M by declaring them to be zero outside this chart. So, it suffices to consider $M = \mathbb{R}^n$. Let $\mathcal{S}_{(a,b)} : [0,\infty) \to [0,1]$ be a smooth bump function with $\mathcal{S}_{(a,b)}^{-1}(1) = [0,a]$ and $\mathcal{S}_{(a,b)}^{-1}(0) = [b,\infty)$. Let $p \in \mathbb{R}$, e_1 denote the first standard basis vector, and define, for $x \in \mathbb{R}^n$,

$$V_p(x) = -x\mathcal{S}_{(0,1.5)}(|x|) + p\mathcal{S}_{(2,3)}(|x|)e_1.$$

Then $\{V_p\}_{p\in(-0.2,0.2)}$ is continuous with respect to the weak C^1 topology. In fact, $\{V_p\}_{p\in(-0.2,0.2)}$ is also continuous with respect to the strong C^{∞} topology, the most restrictive of the standard smooth vector field topologies, so $\{V_p\}_{p\in(-0.2,0.2)}$ varies as smoothly as might be desired. Furthermore, for $p \in (-0.2, 0.2)$, the vector field V_p is unchanged outside a fixed compact set. Nevertheless, despite the smoothness of $\{V_p\}_{p\in(-0.2,0.2)}$ and the fact that variations are restricted to a fixed compact set, this family of vector fields exhibits discontinuity in the boundaries of the regions of attraction of a family of stable equilibria.

For each p, V_p has a stable equilibrium point near the origin, call them $\{X^s(p)\}_{p \in (-0.2, 0.2)}$. The case of n = 1 is illustrated in Fig. 2.2, which shows the vector field V_p for a few values of p. For p = 0.1, the vector field is positive for $x \in (-3, 0)$, driving initial conditions in this range towards $X_{0.1}^s$, and is negative for x greater than $X_{0.1}^s$ but less than about 1.1, driving these initial conditions towards $X_{0.1}^s$ as well. So, $W^s(X_{0.1}^s) \approx (-3, 1.1)$ consists of a line segment and $\partial W^s(X_{0,1}^s) \approx \{-3, 1.1\}$ consists of the two points on the boundary of the line segment. In fact, for any $p \in (0, 0.2)$, $W^s(X^s(p))$ will be a line segment that includes (-3,0), and $\partial W^s(X^s(p))$ will consist of the two points on its boundary, one of which is $\{-3\}$. For p = -0.1, the vector field is negative for $x \in (0,3)$, driving initial conditions in this range towards $X_{-0.1}^s$, and is positive for x less than $X_{-0.1}^s$ but greater than about -1.1, driving these initial conditions towards $X_{-0.1}^s$ as well. So, $W^s(X_{-0.1}^s) \approx (-1.1, 3)$ consists of a line segment and $\partial W^s(X^s_{-0,1}) \approx \{-1,1,3\}$ consists of the two points on the boundary of the line segment. In fact, for any $p \in (-0.2, 0)$, $W^s(X^s(p))$ will be a line segment that includes (0,3), and $\partial W^s(X^s(p))$ will consist of the two points on its boundary, one of which is $\{3\}$. Now consider the case where p = 0. By analogous reasoning to the above, based on the sign of the vector field, $W^s(X_0^s) = (-1.5, 1.5)$. So, $\partial W^s(X_0^s) = \{-1.5, 1.5\}$. But, we saw that as p approaches zero from above, $\partial W^s(X^s(p))$ contains the point $\{-3\}$, and as p approaches zero from below, $\partial W^s(X^s(p))$ contains the point {3}, neither of which are contained in $\partial W^s(X_0^s) = \{-1.5, 1.5\}$. Hence, the family $\{\partial W^s(X^s(p))\}_{p \in (-0.2, 0.2)}$ is Hausdorff discontinuous at p = 0 from both above and below.

Fig. 2.3 illustrates this more clearly, by showing $W^s(X^s(p))$ and $\partial W^s(X^s(p))$ for $p \in$ (-0.1, 0.1), as well as $\partial W^s(X^s_{(-0.1, 0.1)})$. Let $M = \mathbb{R}$ and let J = (-0.1, 0.1). Then for $p \in J$ with $p \neq 0$, Fig. 2.3 plots $W^s(X^s(p)) \subset M \times \{p\} \subset M \times J$ in blue. And at $p = 0, W^s(X_0^s) \subset M \times \{0\}$ is shown in red. For $p > 0, W^s(X^s(p))$ includes (-3, 0), so $\partial W^s(X^s(p))$ contains $\{-3\}$. For p < 0, $W^s(X^s(p))$ includes (0,3), so $\partial W^s(X^s(p))$ contains {3}. However, for p = 0, $W^s(X_0^s) = (-1.5, 1.5)$, so $\partial W^s(X_0^s) = \{-1.5, 1.5\}$, which does not contain {3} or $\{-3\}$. Thus, as discussed above, $\{\partial W^s(X^s(p))\}_{p\in J}$ is Hausdorff discontinuous at p = 0 from both above and below. Now consider $\partial W^s(X_J^s)$. First note that $\partial W^s(X^s_J)$ contains $\sqcup_{p \in J} \partial W^s(X^s(p))$, so for each $p \in J$ it contains the two points of $\partial W^s(X^s(p))$. However, as $\partial W^s(X^s_J)$ is obtained by taking the topological boundary of $W^{s}(X_{J}^{s})$ in $M \times J$, it also contains the cyan line segments shown at p = 0, which are [-3, -1.5]and [1.5, 3]. Hence, $\partial W^s(X_J^s) \cap (M \times \{0\})$ contains the line segments [-3, -1.5] and [1.5, 3], whereas $\partial W^s(X_0^s) = \{-1.5, 1.5\}$ consists only of two points. In particular, $\partial W^s(X_J^s)$ is strictly larger than $\sqcup_{p \in J} \partial W^s(X^s(p))$. For a large class of families of C^1 vector fields, Theorem 2.2.23 shows that $\partial W^s(X_J^s) = \bigsqcup_{p \in J} \partial W^s(X^s(p))$, and Corollaries 2.2.24-2.2.25 show that $\{\partial W^s(X^s(p))\}_{p\in J}$ varies Chabauty or Hausdorff continuously, respectively.

From a practical perspective, we consider an initial condition y_p which is a C^1 function of parameter p and represents the system state after a finite time, parameter-dependent disturbance. In order to provide theoretical motivation for the algorithms which find the recovery boundary by computing critical parameter values, it is essential that there exists a critical parameter value p^* such that $y_{p^*} \in \partial W^s(X_{p^*}^s)$. Suppose for some values of p that $y_p \in W^s(X_J^s)$, so the system recovers from the disturbance, and for other values of p that $y_p \notin$ $W^s(X_J^s)$, so the system does not recover from the disturbance. Then since y_p is continuous in p, y_J is connected, so there must exist at least one p^* such that $y_{p^*} \in \partial W^s(X_J^s)$. However, as this example shows, $y_{p^*} \in \partial W^s(X_J^s)$ does not necessarily imply that $y_{p^*} \in \partial W^s(X_{p^*}^s)$ as is required for the proof of Theorem 2.5.9. In particular, Fig. 2.3 shows two families of initial conditions y_J : a yellow family of initial conditions which does pass through $\partial W^s(X_{p^*}^s)$ for some parameter value p^* , and a green family of initial conditions which does not pass through $\partial W^s(X^s(p))$ for any $p \in J$ but passes through $\partial W^s(X_J^s)$ through one of the cyan line segments. Hence, this example shows that the conclusions of Theorem 2.2.23 are not always satisfied and, if not, then the conclusions of Theorem 2.5.9 may not hold either.



Figure 2.2: The graph of V_p for p = 0.1 (green dashed), p = 0 (blue solid), and p = -0.1 (red dot-dashed). This figure originally appeared in [FH17].

The discussion above generalizes to arbitrary dimension n. Example 2.3.3 of Section 2.3 shows that the assumptions of this paper are tight in a reasonable sense by exhibiting a strong C^1 continuous family of Morse-Smale vector fields on \mathbb{R}^2 that has a new equilibrium point entering the boundary of the RoA for p arbitrarily close to p_0 , and the boundary is Chabauty discontinuous at p_0 .

2.2.4 Main Results

2.2.4.1 Vector Field is Parameter Independent

The primary motivation for presenting the results of this Section for parameter independent vector fields is to provide a foundation for, and to improve the clarity of presentation of, the results for parameter dependent vector fields in Section 2.2.4.2. However, the main result here (see Theorem 2.2.12) may also be of some independent interest as it provides a complete proof for parameter independent vector fields of a result for which earlier proofs [CHW88] are incomplete.

Let V be a complete C^1 vector field on M, where M is either a compact Riemannian manifold or \mathbb{R}^n . Let X^s be a stable equilibrium point of V. We make the following assumptions.

Assumption 2.2.2. There exists a neighborhood N of $\partial W^s(X^s)$ such that $\Omega(V) \cap N$ consists of a finite union of critical elements; call them $\{X^i\}_{i \in I}$ where $I = \{1, ..., k\}$.

Assumption 2.2.3. For every $x \in \partial W^s(X^s)$, the forward orbit of x under V is bounded.



Figure 2.3: The disjoint union of the regions of attraction of the family of stable equilibria of the vector fields $\{V_p\}$ over $p \in [-0.1, 0.1]$ (blue). The region of attraction of the stable equilibrium point of V_0 is shown in red. Then $\partial W^s(X_0^s)$ consists of the two points on the boundary of this red line segment, while $\partial W^s(X_J^s) \cap (\mathbb{R} \times \{0\})$ is equal to the union of the cyan line segments. One family of initial conditions (yellow) begins inside the regions of attraction and passes through one of their boundaries at p is increased. Another family of initial conditions (green) begins inside the regions of attraction and passes outside without passing through one of their boundaries. This occurs because the boundaries of the regions of attraction fail to be Hausdorff continuous at p = 0. This figure originally appeared in [FH17].

Assumption 2.2.4. Every critical element in $\partial W^s(X^s)$ is hyperbolic.

Assumption 2.2.5. For each pair of critical elements in $\partial W^s(X^s)$, say X^i and X^j , $W^s(X^i)$ and $W^u(X^j)$ are transversal.

Remark 2.2.6. Assumptions 2.2.2,2.2.4, and 2.2.5 ensure that V is Morse-Smale along $\partial W^s(X^s)$.

Remark 2.2.7. Assumption 2.2.3 is necessary in the case $M = \mathbb{R}^n$ since Morse-Smale vector fields were defined on compact manifolds [Sma60], whereas for $M = \mathbb{R}^n$ it is necessary to prohibit orbits in $\partial W^s(X^s)$ from diverging to infinity in forwards time.

Remark 2.2.8. By the Kupka-Smale Theorem for M compact [Sma60, Kup63], and its generalization for M σ -compact [KH99, Page 294], Assumptions 2.2.4 and 2.2.5 are generic with respect to the weak C^1 topology.

Remark 2.2.9. By Remark 2.3.7 and Lemma 2.3.8 from Section 2.3, Assumption 2.2.2 can be relaxed to the assumption that there exists a neighborhood of $\partial W^s(X^s)$ in which the number of equilibrium points and periodic orbits is finite, together with an additional assumption that is generic with respect to the strong C^1 topology.

Remark 2.2.10. By Assumption 2.2.4, hyperbolicity of the critical elements implies that their stable and unstable manifolds exist.

Remark 2.2.11. Assumptions 2.2.2-2.2.3 together imply that for any orbit $\gamma \subset \partial W^s(X^s)$, $\omega(\gamma) = X^i$ for some $i \in I = \{1, ..., k\}$.

Theorem 2.2.12 gives a decomposition of the boundary of the region of attraction for a parameter independent vector field as a union of the stable manifolds of the critical elements it contains.

Theorem 2.2.12. Let M be either a compact Riemannian manifold or Euclidean space, and suppose V is a C^1 vector field on M satisfying Assumptions 2.2.2-2.2.5 Let $\{X^i\}_{i\in I}$ be the critical elements contained in $\partial W^s(X^s)$. Then $\partial W^s(X^s) = \bigcup_{i\in I} W^s(X^i)$.

Remark 2.2.13. Theorem 2.2.12 was originally reported in [CHW88, Theorem 4-2] under slightly more general assumptions. Namely, our Assumption 2.2.2 was replaced by the assumption that for every $x \in \partial W^s(X^s)$, the trajectory of x converges to a critical element in forwards time. Hence, the number of critical elements in $\partial W^s(X^s)$ was not assumed to be finite, and the set of ω limit points in $\partial W^s(X^s)$, rather than the nonwandering set on a neighborhood of $\partial W^s(X^s)$, was assumed to consist solely of critical elements (in general the nonwandering set may be larger than the closure of the set of ω limit points). The main purpose for presenting Theorem 2.2.12 under these weaker assumptions is that its treatment more closely parallels the results and proofs of Theorem 2.2.23 for the case of parameter dependent vector fields below. For example, a finite number of critical elements is necessary to ensure that all critical elements persist under small perturbations to the vector field. However, the proof of [CHW88, Theorem 4-2] relies on [CHW88, Lemma 3-5], which has been disproven (see Section 2.1).

Therefore, the proof of [CHW88, Theorem 4-2] is incomplete, so the proof of Theorem 2.2.12 presented here represents the first complete proof of this result, and therefore may be of independent interest.

2.2.4.2 Vector Field is Parameter Dependent

Next we generalize the above results to the case where the vector field is parameter dependent. Let J be a connected smooth manifold representing a family of parameters, and let $\{V_p\}_{p\in J}$ be a weak C^1 continuous family of complete C^1 vector fields on M. Let V be the complete C^1 vector field on $M \times J$ defined by $V(x, p) = (V_p(x), 0) \subset T_x M \times T_p J$. Let ϕ be the C^1 flow of V, ϕ^p be the flow of V_p , and note that $\phi(t, x, p) = (\phi^p(t, x), p)$ for $x \in M$ and $t \in \mathbb{R}$. For fixed t, we often write $\phi_t : M \times J \to M \times J$ by $\phi_t(x, p) = \phi(t, x, p)$ and note that ϕ_t is a C^1 diffeomorphism for each t.

Let $\{X^s(p)\}_{p\in J}$ be a C^1 continuous family of stable equilibria of the vector fields $\{V_p\}_{p\in J}$. Let $X_J^s = \bigcup_{p\in J} X^s(p)$ and let $W^s(X_J^s) = \sqcup_{p\in J} W^s(X^s(p))$. In this setting, there are potentially two different boundaries of regions of attraction to consider. First, for a fixed parameter value $p_0 \in J$ we have $\partial W^s(X_{p_0}^s)$, where the topological boundary operation is taken in M. Second, we have $\partial W^s(X_J^s)$, where the topological boundary operation is taken in $M \times J$. In general, $\sqcup_{p\in J}\partial W^s(X_{p_0}^s) \subset \partial W^s(X_J^s)$. Therefore, we make assumptions regarding the behavior of V along $\partial W^s(X_J^s)$ rather than along $\sqcup_{p\in J}\partial W^s(X_{p_0}^s)$. For some fixed $p_0 \in J$ we make the following assumptions.

Assumption 2.2.14. There exists a neighborhood N of $\partial W^s(X_J^s) \cap M_{p_0}$ in M_{p_0} such that $\Omega(V) \cap N$ consists of a finite union of critical elements of V_{p_0} ; call them $\{X_{p_0}^i\}_{i \in I}$ where $I = \{1, ..., k\}$ and $k \geq 1$.

Assumption 2.2.15. Every critical element in $\partial W^s(X_J^s) \cap M_{p_0}$ is hyperbolic in M with respect to V_{p_0} .

Remark 2.2.16. By Assumption 2.2.15, the critical elements $\{X_{p_0}^i\}_{i\in I}$ in $\pi_M(\partial W^s(X_J^s) \cap M_{p_0})$ are hyperbolic so, since I is finite, they and their stable and unstable manifolds persist for Jsufficiently small. For $p \in J$, define the perturbation of $X^s(p_0)$ corresponding to parameter value $p(X^s(p))$. For $p \in J$ and $i \in I$, define the perturbation of $X^i(p_0)$ corresponding to parameter value $p(X^i(p))$. In M, define the stable manifold of $X^s(p)(W^s(X^s(p)))$, and for $i \in I$ define the stable manifold of $X^i(p)(W^s(X^i(p)))$ and the unstable manifold of $X^i(p)(W^u(X^i(p)))$. In $M \times J$, define $\sqcup_{p \in J} X^s(p)(X_J^s)$, $\sqcup_{p \in J} W^s(X^s(p))(W^s(X_J^s))$, and for $i \in I$ define $\sqcup_{p \in J} X^i(p)(X_J^i)$, $\sqcup_{p \in J} W^s(X^i(p))(W^s(X_J^i))$, and $\sqcup_{p \in J} W^u(X^i(p))(W^u(X_J^i))$

Assumption 2.2.17. For each $p \in J$, $\Omega(V_p) \cap (\partial W^s(X_J^s) \cap M_p) = \bigcup_{i \in I} X^i(p)$ and for every $x \in \partial W^s(X_J^s) \cap M_p$ its forward orbit under V_p is bounded.

Assumption 2.2.18. For each pair of critical elements that are contained in $\partial W^s(X_J^s) \cap M_{p_0}$, say $X_{p_0}^i$ and $X_{p_0}^j$, $W^s(X_{p_0}^i)$ and $W^u(X_{p_0}^j)$ are transversal in M.

Remark 2.2.19. Assumptions 2.2.14,2.2.15, and 2.2.18 are straightforward generalizations of Assumptions 2.2.2,2.2.4, and 2.2.5. They ensure that V_{p_0} is Morse-Smale along $\pi_M (\partial W^s(X_J^s) \cap M_{p_0}).$

Remark 2.2.20. Assumption 2.2.17 generalizes Assumption 2.2.3 by ensuring that, for every $p \in J$, every orbit in $\partial W^s(X_J^s) \cap M_p$ converges to $X^i(p)$ for some $i \in I$. So, no new critical elements can enter $\partial W^s(X_J^s) \cap M_p$ for $p \in J$, $p \neq p_0$.

Remark 2.2.21. Using Remark 2.3.7 and Lemma 2.3.8 from Section 2.3, Assumption 2.2.14 can be partially relaxed as in Remark 2.2.9.

Remark 2.2.22. In the case of M a compact Riemannian manifold, by Theorem 2.3.9 in Section 2.3, Assumption 2.2.17 is not necessary. In the case of M Euclidean, by Theorem 2.3.20 in Section 2.3, Assumption 2.2.17 can be relaxed, for $\{V_p\}_{p\in J}$ a strong C^1 continuous family of vector fields, to the assumptions that for every $x \in \partial W^s(X_{p_0}^s)$ the forward orbit of x is bounded, and there exists a neighborhood N of infinity such that $\Omega(V_{p_0}) \cap N = \emptyset$ and no orbit under V_{p_0} entirely contained in N has empty ω and α limit sets under V_{p_0} .

Theorem 2.2.23 gives a decomposition of $\partial W^s(X_J^s)$ as a disjoint union over parameter values in J of a union of the stable manifolds of its critical elements. Furthermore, it shows that the topological boundary in $M \times J$, $\partial W^s(X_J^s)$, is equal to the disjoint union over $p \in J$ of the topological boundaries in M of the stable manifolds of the stable equilibria. Using Theorem 2.2.23, it is straightforward to then show that $\{\partial W^s(X^s(p))\}_{p \in J}$ is a continuous family of subsets of M (Corollary 2.2.24). Hence, if M is a compact Riemannian manifold, this implies that $\{\partial W^s(X^s(p))\}_{p \in J}$ is a Hausdorff continuous family of subsets of M (Corollary 2.2.25). Finally, if V_{p_0} is Morse-Smale on M a compact Riemannian manifold, using persistence of the so-called phase diagram of Morse-Smale vector fields under perturbation [Pal69], one can show that for any C^1 continuous family of vector fields containing V_{p_0} , and for J sufficiently small, $\{\partial W^s(X^s(p))\}_{p \in J}$ is a Hausdorff continuous family of subsets of M(Corollary 2.2.26).

Theorem 2.2.23. Let M be either a compact Riemannian manifold or Euclidean space, and let $\{V_p\}_{p\in J}$ be a family of vector fields on M continuous with respect to the weak C^1 topology and satisfying Assumptions 2.2.14-2.2.18. Let $\{X_{p_0}^i\}_{i\in I}$ denote the critical elements of V_{p_0} in $\partial W^s(X_J^s) \cap M_{p_0}$. Then in $M \times J$ for sufficiently small J, $\partial W^s(X_J^s) = \sqcup_{p\in J} \partial W^s(X^s(p)) =$ $\bigcup_{i\in I} W^s(X_J^i)$.

Corollary 2.2.24. Let $M = \mathbb{R}^n$ and let $\{V_p\}_{p \in J}$ be a weak C^1 continuous family of vector fields on M satisfying Assumptions 2.2.14-2.2.18. Then $\{\partial W^s(X^s(p))\}_{p \in J}$ is a Chabauty continuous family of subsets of M.

Corollary 2.2.25. Let M be a compact Riemannian manifold and let $\{V_p\}_{p\in J}$ be a C^1 continuous family of vector fields on M satisfying Assumptions 2.2.14-2.2.18. Then $\{\partial W^s(X^s(p))\}_{p\in J}$ is a Hausdorff continuous family of subsets of M.

Corollary 2.2.26. Let M be a compact Riemannian manifold and let V_{p_0} be a Morse-Smale vector field on M. Then for any C^1 continuous family of vector fields $\{V_p\}_{p \in J}$ on M with $p_0 \in J$, for sufficiently small J, $\{\partial W^s(X^s(p))\}_{p \in J}$ is a Hausdorff continuous family of subsets of M.

2.2.4.3 Illustrative Example

Example 2.2.27 (Illustration of Main Theorems). To illustrate the results of

Theorems 2.2.12, 2.2.23, 2.5.9 and Corollary 2.2.24 we consider the simple example of a damped, driven nonlinear pendulum with constant driving force. The dynamics are given by

$$\dot{x}_1 = x_2 \tag{2.8}$$

$$\dot{x}_2 = -c_1 \sin(x_1) - c_2 x_2 + c_3 \tag{2.9}$$

where $c_1, c_2, c_3 > 0$ are real parameters and $x = (x_1, x_2) \subset \mathbb{R}^2$. Physically, x_1 represents the angle of the pendulum, x_2 its angular velocity, c_1 the natural frequency of the pendulum, c_2 a damping coefficient due to air drag, and c_3 the constant driving torque. Eqs. 2.8-2.9 can also be interpreted as an electrical generator with (x_1, x_2) the angle and angular velocity of the turbine, c_1 a constant determining the electrical torque supplied by the generator, c_2 a damping coefficient due to friction, and c_3 the constant driving mechanical torque. For the demonstration below, we set $c = (c_1, c_2, c_3) = (2, 0.5, 1.5)$ and we restrict x_1 to a single interval of length 2π since x_1 is defined modulo 2π . Although c_3 is initially given the fixed value of 1.5, we let $p = c_3$ and will subsequently treat it as a free parameter, setting $p_0 = 1.5$. At p_0 , this system possesses one stable equilibrium point, $X_{p_0}^s$, at (0.848,0), one unstable equilibrium point, $X_{p_0}^1$, at (2.294, 0), and no other nonwandering elements. Variation of the value of p over a range J that contains p_0 then generates a C^1 continuous family of vector fields, as well as families of equilibria $\{X^s(p)\}_{p\in J}$ and $\{X_p^1\}_{p\in J}$. We define an initial condition to Eqs. 2.8-2.9 as the output of the following related system.

$$\dot{z}_1 = z_2 \tag{2.10}$$

$$\dot{z}_2 = -c_2 z_2 + c_3 \tag{2.11}$$

Let ϕ_d denote the flow of Eqs. 2.10-2.11 and let J = (1.3, 2). Then the initial condition of Eqs. 2.8-2.9 is given by $y : J \to \mathbb{R}^2$ by $y_p = \phi_d(c_4, X^s(p), p)$, where we set $c_4 = 0.8$ to be the length of time the disturbance is active. If Eqs. 2.8-2.9 are interpreted as an electrical generator, then Eqs. 2.10-2.11 represent a short circuit across the terminals of the generator so it no longer supplies any electrical torque. This is modeled by setting c_1 to zero in Eqs. 2.8-2.9, which then gives Eqs. 2.10-2.11.

Fig. 2.4 shows $\partial W^s(X_{p_0}^s)$. Note that the intersection of $\partial W^s(X_{p_0}^s)$ with the nonwandering set is X^1 , every orbit $\gamma \subset \partial W^s(X_{p_0}^s)$ has $\omega(\gamma) = X_{p_0}^1, X_{p_0}^1$ is hyperbolic, and the transversality assumption is vacuously true since $X_{p_0}^1$ is the only critical element in $\partial W^s(X_{p_0}^s)$. Therefore, the system satisfies Assumptions 2.2.2-2.2.5, so by Theorem 2.2.12 we must have $\partial W^s(X_{p_0}^s) = W^s(X_{p_0}^1)$, as can be seen in Fig. 2.4.

Fig. 2.5 shows the boundaries of the regions of attraction of the family of vector fields for several values of the parameter $p = c_3$. For p > 2 the stable and unstable equilibria $X^s(p)$ and X_p^1 collide in a saddle-node bifurcation and annihilate each other, so we must restrict to sufficiently small J = (1.3, 2), as in Theorem 2.2.23. Fix $p_0 = 1.5$ as above. Then the intersection of the nonwandering set with $\partial W^s(X_J^s) \cap M_{p_0}$ is $X_{p_0}^1$, for every orbit $\gamma \subset \partial W^s(X_J^s)$ we have $\omega(\gamma) \subset X_J^1$, $X_{p_0}^1$ is hyperbolic, and the transversality condition for $\partial W^s(X_J^s) \cap M_{p_0}$ is vacuously satisfied since the only critical element in $\partial W^s(X_J^s) \cap M_{p_0}$ is $X_{p_0}^1$. Therefore, the system satisfies Assumptions 2.2.14-2.2.18, so by Theorem 2.2.23 we must have $\partial W^s(X_J^s) = \bigsqcup_{p \in J} \partial W^s(X^s(p)) = W^s(X_J^1)$, and by Corollary 2.2.24, $\{\partial W^s(X^s(p))\}_{p \in J}$ is a Chabauty continuous family of subsets of M.

2.2.5 **Proof of Theorem 2.2.12**

This Section is devoted to the proof of Theorem 2.2.12. Many of the results and proofs of this Section will be recycled for additional use for the parameter dependent vector field results and proofs in Section 2.2.6. Most of the lemmas presented in this Section are similar to results given elsewhere, especially for diffeomorphisms of compact Riemannian manifolds, but the presentation and proofs here are our own unless otherwise specified.

We let M be either a compact Riemannian manifold or Euclidean space unless stated



Figure 2.4: The region of attraction boundary $\partial W^s(X_{p_0}^s)$ (black line) of the stable equilibrium point $X_{p_0}^s$ (black star) of Eqs. 2.8-2.9 is shown. It is equal to $W^s(X_{p_0}^1)$ where $X_{p_0}^1$ (black triangle) is the unstable equilibrium point. The orbit (dashed black line) from the initial condition y_{p_0} (black circle) is shown.



Figure 2.5: The region of attraction boundaries $\partial W^s(X^s(p))$ of the stable equilibrium points $X^s(p)$ (stars) for parameter values p = 1.3 (green dashed), p = 1.568 (solid blue), and p = 1.9 (red dot dashed) are shown. Each boundary is equal to $W^s(X_p^1)$ where X_p^1 (triangle) is the unstable equilibrium point corresponding to parameter value p. The initial conditions (circles) are shown.

otherwise.

Since $W^s(X^s)$ is invariant, its topological closure $\overline{W^s}(X^s)$ is invariant. For, if $x \in \overline{W^s}(X^s)$ and $t \in \mathbb{R}$ then there exists a sequence $\{x_n\}_{n=1}^{\infty} \subset W^s(X^s)$ such that $x_n \to x$. By invariance of $W^s(X^s), \phi_t(x_n) \in W^s(X^s)$ for all n. By continuity of $\phi_t, \phi_t(x_n) \to \phi_t(x)$, so $\phi_t(x) \in \overline{W^s}(X^s)$. Hence, $\overline{W^s}(X^s)$ and $W^s(X^s)$ are invariant, so $\partial W^s(X^s) = \overline{W^s}(X^s) - W^s(X^s)$ is invariant.

Let $\{X^i\}_{i\in I}$ denote the critical elements in $\partial W^s(X^s)$. Then $W^u_{\text{loc}}(X^i)$ and $W^s_{\text{loc}}(X^i)$ are well-defined local unstable and stable manifolds for X^i for all $i \in I$. Lemma 2.2.28 provides a technical construction, for any critical element, of a compact set contained in its unstable manifold such that for any sufficiently small neighborhood N of this compact set in M, the following holds. The union over time of the time-t flow ϕ_t of N over all negative times t, together with the stable manifold of the critical element, contains an open neighborhood of the critical element in M. This result will be instrumental in making the claim below that if a critical element is contained in $\partial W^s(X^s)$ then its unstable manifold intersects $\overline{W^s}(X^s)$. Lemma 2.2.28 is analogous to [Pal69, Corollary 1.2], which states the corresponding result for diffeomorphisms without proof, whereas here the result is shown for vector fields. Fig. 2.6 illustrates the content of Lemma 2.2.28. Recall that if D is a subset of a metric space and $\epsilon > 0$, the notation D_{ϵ} refers to the subset of the metric space such that for each $x \in D_{\epsilon}$ there exists $y \in D$ with $d(x, y) < \epsilon$.

Lemma 2.2.28. For any $i \in I$ and any $\epsilon > 0$ there exists a compact set $D \subset W^u_{loc}(X^i) - X^i$ and an open neighborhood N of D in M disjoint from X^i such that $N \subset D_{\epsilon}$ and $\bigcup_{t < 0} \phi_t(N) \bigcup W^s(X^i)$ contains an open neighborhood of X^i in M.

Proof of Lemma 2.2.28. If X^i is an equilibrium point let $\hat{f} = \phi_1$ be the time-one flow. If X^i is a periodic orbit, let $\hat{f} = \hat{\tau}$ be the first return map of a cross section S of X^i . Let $D' = W^u_{\text{loc}}(X^i)$ and let D be the topological closure of $D' - (\hat{f})^{-1}(D')$ in M. In order to show the existence of the desired open neighborhood of X^i , the first step will be constructing a C^1 continuous disk family centered along D and contained in an open neighborhood $N \subset D_{\epsilon}$. Then, this C^1 disk family is extended to a C^1 disk family centered along $W^u_{\text{loc}}(X^i)$ by backward iteration and the inclusion of the disk $W^s_{\text{loc}}(X^i)$. It is shown that this family is in fact C^1 continuous using the Inclination Lemma. Finally, once the C^1 continuous and backward invariant disk family has been constructed, invariance of domain [Hat01, Theorem 2B.3] is applied to conclude that the disk family contains an open neighborhood of X^i . By construction, this implies that $\bigcup_{t\leq 0} \phi_t(N) \bigcup W^s(X^i)$ contains an open neighborhood of X^i . The full proof is provided in Appendix A.



Figure 2.6: The compact set D (black line segments) and the neighborhood N (yellow shapes) mentioned in Lemma 2.2.28 for an equilibrium point (red star). The set D is contained in the unstable manifold (red line). As the neighborhood N is propagated backwards in time (first to the green shapes then to the cyan), it approaches the stable manifold (blue line) of the equilibrium point. From the figure, it appears that the union of the backward flows of N over all negative times, together with the stable manifold, will contain a neighborhood of the equilibrium point, which is the content of Lemma 2.2.28. This figure originally appeared in [FH17].

We will use the technical result of Lemma 2.2.28 to show that the unstable manifold of a critical element in the boundary of the region of attraction must have nonempty intersection with $\overline{W^s}(X^s)$. The following lemma is analogous to the combination of [CHW88, Theorem 3-3] (for equilibrium points in $\partial W^s(X^s)$) and [CHW88, Corollary 3-4] (for periodic orbits in $\partial W^s(X^s)$), although [CHW88, Corollary 3-4] was unproven. Our proof is similar to the proof of [CHW88, Theorem 3-3], although we have explicitly proved Lemma 2.2.28 whereas [CHW88] states a similar technical result without proof, and we also explicitly prove the contents of [CHW88, Corollary 3-4].

Lemma 2.2.29. If $X^i \subset \partial W^s(X^s)$ then $\{W^u(X^i) - X^i\} \cap \overline{W^s}(X^s) \neq \emptyset$.

Proof of Lemma 2.2.29. Using Lemma 2.2.28 we will produce a neighborhood of X^i from its stable and unstable manifolds. Since X^i is in the topological boundary, this neighborhood must intersect $W^s(X^s)$. Then since stable manifolds cannot intersect, by invariance, and by sending ϵ in the statement of Lemma 2.2.28 to zero we will obtain the result.

Let $x \in X^i$ and let $\epsilon > 0$. By Lemma 2.2.28, there exists a compact set $D \subset W^u_{\text{loc}}(X^i) - X^i$ and an open neighborhood N of D in M disjoint from X^i such that $N \subset D_{\epsilon}$ and $\bigcup_{t \leq 0} \phi_t(N) \bigcup W^s(X^i)$ contains a neighborhood of X^i in M, call it U_{ϵ} . Then U_{ϵ} is a neighborhood of $x \in \partial W^s(X^s)$, so $U_{\epsilon} \cap W^s(X^s) \neq \emptyset$. Since $W^s(X^i) \cap W^s(X^s) = \emptyset$, there must exist

some $T \leq 0$ such that $\phi_T(N) \cap W^s(X^s) \neq \emptyset$. Since $W^s(X^s)$ is invariant, this implies that $N \cap W^s(X^s) \neq \emptyset$. Since $N \subset D_{\epsilon}$, letting d_S be the set distance on the Riemannian manifold M, we have

$$\epsilon \ge d_S(D, W^s(X^s)) = d_S(D, \overline{W^s}(X^s))$$

holds for all $\epsilon > 0$, so $d_S(D, \overline{W^s}(X^s)) = 0$. Since D is compact and $\overline{W^s}(X^s)$ is closed, this implies that $D \cap \overline{W^s}(X^s) \neq \emptyset$. Hence, since $D \subset W^u_{\text{loc}}(X^i) - X^i$, $\{W^u(X^i) - X^i\} \cap \overline{W^s}(X^s) \neq \emptyset$. \Box

For X a critical element, let $n^t(X) = 0$ if X is an equilibrium point and let $n^t(X) = 1$ if X is a periodic orbit. Let $n^u(X) = \dim W^u(X) - n^t(X)$ and let $n^s(X) = \dim W^s(X) - n^t(X)$. Lemma 2.2.30 was proven in [Sma60, Lemma 3.1]. It is reproduced here for clarity of presentation. A slightly different result, that was reported in [CHW88, Lemma 3-5] and was fundamental in the proof of [CHW88, Theorem 4-2], has been disproven (see Section 2.1).

Lemma 2.2.30. If $W^{s}(X^{i}) \cap W^{u}(X^{j}) \neq \emptyset$ then $n^{u}(X^{i}) \leq n^{u}(X^{j}) + n^{t}(X^{j}) - 1$ and dim $W^{u}(X^{i}) \leq \dim W^{u}(X^{j}) + n^{t}(X^{i}) - 1$.

Proof of Lemma 2.2.30. Since $W^{s}(X^{i})$ and $W^{u}(X^{j})$ have a point of transversal intersection and are invariant under the flow, they have an orbit δ of transversal intersection. Then $\dot{\delta}(0) \in T_{\delta(0)}W^{s}(X^{i}) \cap T_{\delta(0)}W^{u}(X^{j})$. By transversality, this implies that $(\dim W^{s}(X^{i}) - 1) + (\dim W^{u}(X^{j}) - 1) \geq n - 1$. Hence, $(n^{s}(X^{i}) + n^{t}(X^{i}) - 1) + (n^{u}(X^{j}) + n^{t}(X^{j}) - 1) \geq n - 1$. Since $n^{s}(X^{i}) + n^{t}(X^{i}) + n^{u}(X^{i}) = n$, this implies that $(n - n^{u}(X^{i}) - 1) + (n^{u}(X^{j}) + n^{t}(X^{j}) - 1) \geq n - 1$ so $n^{u}(X^{i}) \leq n^{u}(X^{j}) + n^{t}(X^{j}) - 1$. Hence, dim $W^{u}(X^{i}) = n^{u}(X^{i}) + n^{t}(X^{i}) \leq \dim W^{u}(X^{j}) + n^{t}(X^{i}) - 1$.

As defined in Section 2.2.2, a heteroclinic sequence is a sequence of hyperbolic critical elements such that the stable manifold of each critical element intersects the unstable manifold of the next element of the sequence. A heteroclinic cycle is a finite heteroclinic sequence where the first and last critical elements are the same. Lemmas 2.2.31-2.2.33 show that Assumptions 2.2.2,2.2.4, and 2.2.5 imply that there are no heteroclinic cycles and, therefore, that all heteroclinic sequences are finite. These are analogous to several Lemmas in [Pal69] for diffeomorphisms, but are proved here for vector fields. Lemma 2.2.31 shows that the intersection of stable and unstable manifolds of critical elements satisfies the transitive property. It was shown in [Pal69, Corollary 1.3] for diffeomorphisms, and is proven here for vector fields.

Lemma 2.2.31. If $(W^s(X^i) - X^i) \cap (W^u(X^j) - X^j) \neq \emptyset$ and $(W^s(X^j) - X^j) \cap (W^u(X^k) - X^k) \neq \emptyset$ then $(W^s(X^i) - X^i) \cap (W^u(X^k) - X^k) \neq \emptyset$.

Proof of Lemma 2.2.31. The proof revolves around the openness of transversal intersection of compact submanifolds which are C^1 close, and the use of the Inclination Lemma to guarantee that the submanifolds are C^1 close.

If X^j is an equilibrium point, let $B = W^u_{loc}(X^j)$. If X^j is a periodic orbit, let $B = W^u_{loc}(X^j) \cap S$, where S is any cross section of X^j . By invariance of $W^s(X^i)$ and the assumptions of the Lemma, we have that $W^s(X^i) \cap B \neq \emptyset$. We claim that B is transverse to $W^s(X^i)$. This follows trivially if X^j is an equilibrium point, so suppose it is a periodic orbit. For any $x \in W^s(X^i) \cap W^u(X^j)$, $T_x W^s(X^i)$ and $T_x W^u(X^j)$ together span $T_x M$ since the intersection is transverse. Then B is obtained by intersecting $W^u_{loc}(X^j)$ with S, so $T_x W^u(X^j)$ is equal to the span of $T_x B$ and the flow direction V(x). However, as $W^s(X^j)$ is invariant under $V, V(x) \in T_x W^s(X^j)$. Therefore, $T_x W^s(X^i)$ and $T_x B$ together have the same span as $T_x W^s(X^i)$ and $T_x W^u(X^j)$, which implies that $W^s(X^i)$ and B are transverse at x. As x was arbitrary, the claim follows.

By the definition of $W^s(X^i)$, there exists T < 0 such that $\phi_T W^s_{\text{loc}}(X^i) \cap W^u_{\text{loc}}(X^j) \neq \emptyset$. Note that B is a compact embedded submanifold, and that it is transverse to $\phi_T W^s_{\text{loc}}(X^i)$ since it is transverse to $W^s(X^i)$. Since $\phi_T W^s_{\text{loc}}(X^i)$ and B are compact submanifolds with transversal intersection, by [KH99, Corollary A.3.18] there exists $\epsilon > 0$ such that if D is a compact submanifold which is ϵC^1 -close to B then it has nonempty, transversal intersection with $\phi_T W^s_{\text{loc}}(X^i)$, and hence with $W^s(X^i)$.

Let $y \in (W^s(X^j) - X^j) \cap (W^u(X^k) - X^k)$. Since the intersection is transverse, if X^j is an equilibrium point there exists a compact submanifold $D \subset W^u(X^k)$, which we choose to be a C^1 disk centered at y for the purpose of applying the Inclination Lemma, such that D is transverse to $W^s(X^j)$. Similarly, if X^j is a periodic orbit, there exists a C^1 disk $D \subset$ $W^u(X^k) \cap S$ centered at y such that D is transverse to $W^s(X^j) \cap S$ in S. By Lemma 2.2.30, dim $W^u(X^k) \ge \dim W^u(X^j)$, so we may choose D such that dim $D = \dim B$. Let $\hat{f} = \phi_1$ if X^j is an equilibrium point, and let \hat{f} be a C^1 first return map on S if X^j is a periodic orbit. Then, by the Inclination Lemma for equilibria or periodic orbits, there exists $n_0 > 0$ such that $n \ge n_0$ implies $\hat{f}^n(D)$ is ϵC^1 -close to B. By the choice of ϵ , $\hat{f}^n(D) \cap W^s(X^i) \neq \emptyset$. Since $D \subset W^u(X^k)$ invariant, this implies that $W^s(X^i) \cap W^u(X^k) \neq \emptyset$.

Lemma 2.2.32 shows that there are no homoclinic orbits in $\partial W^s(X^s)$. A similar claim was shown for diffeomorphisms in [Pal69, Corollary 1.4], but the result here is proven for

vector fields.

Lemma 2.2.32. For any X^i , $W^s(X^i) \cap W^u(X^i) = X^i$.

Proof of Lemma 2.2.32. Using transversality and the Inclination Lemma we show that $W^s(X^i) \cap W^u(X^i)$ is nonwandering. By Assumption 2.2.2, this will imply that $W^s(X^i) \cap W^u(X^i) = X^i$.

Clearly $X^i \,\subset W^s(X^i) \cap W^u(X^i)$. Assume towards a contraction that $(W^s(X^i) - X^i) \cap (W^u(X^i) - X^i) \neq \emptyset$. If X^i is an equilibrium point, then by Lemma 2.2.30 this implies that dim $W^u(X^i) \leq \dim W^u(X^i) - 1 < \dim W^u(X^i)$, which is a contradiction. So, suppose X^i is a periodic orbit, let S be a C^1 cross section of X^i , and let $B = W^u_{\text{loc}}(X^i) \cap S$. By an analogous argument as in the proof of Lemma 2.2.31, B is transverse to $W^s(X^i)$. By invariance of $W^s(X^i)$, there exists a point $q \in (W^s(X^i) - X^i) \cap (B - X^i)$. We claim that qis nonwandering. For, let U be any neighborhood of q in M. Let $\epsilon > 0$ such that the ball of radius ϵ centered at q is contained in U. As B is transverse to $W^s(X^i)$, let $D \subset B$ be a C^1 disk centered at q of the same dimension as B such that $D \subset U$ and D is transverse to $W^s(X^i)$. Note that D is transverse to $W^s(X^i) \cap S$ in S as well. Let $\hat{\tau}$ be a C^1 first return map on S. Then, by the Inclination Lemma there exists $n_0 > 0$ such that $n \ge n_0$ implies that $\tau^n(D)$ is ϵC^1 -close to B. As $q \in B$ and U contains the ball of radius ϵ centered at q, this implies that $\tau^n(D) \cap U \neq \emptyset$. Hence, as $D \subset U$, we have that $\tau^n(U) \cap U \neq \emptyset$ for $n \ge n_0$, so q is nonwandering.

By Assumption 2.2.2, there exists a neighborhood N of $\partial W^s(X^s)$ such that $\Omega(V) \cap N = \bigcup_{j \in I} X^j$. As $\omega(q) = \alpha(q) = X^i$ and $q \notin X^i$, $q \neq X^j$ for any $j \in I$. But, since $X^i \subset N$ open and $\omega(q) = X^i$, there exists T > 0 such that $\phi_T(q) \in N$. As the nonwandering set is invariant, $\phi_t(q)$ is nonwandering in N. As X^j is invariant for each $j \in I$, $\phi_t(q) \notin \bigcup_{j \in I} X^j$, which is a contradiction to the choice of N.

Lemma 2.2.33 now shows that every heteroclinic sequence has finite length.

Lemma 2.2.33. There do not exist any heteroclinic cycles. Hence, every heteroclinic sequence has finite length.

Proof of Lemma 2.2.33. Assume towards a contradiction that $\{X^j\}_{j=1}^m$ is a heteroclinic cycle. By transitivity (Lemma 2.2.31), since $X^n = X^1$ this implies that $(W^s(X^1) - X^1) \cap (W^u(X^1) - X^1) \neq \emptyset$. This contradicts Lemma 2.2.32.

Since $\Omega(V) \cap \partial W^s(X^s)$ consists of a finite number of critical elements, and since there are no heteroclinic cycles, every heteroclinic sequence must be finite.

Lemma 2.2.34 will be used to complete the proof of Lemma 2.2.35. It is analogous to [Sma67, Lemma 7.1.b.], but for vector fields instead of diffeomorphisms.

Lemma 2.2.34. Suppose that $W^u(X^i) \cap W^s(X^s) \neq \emptyset$ and $W^s(X^i) \cap W^u(X^j) \neq \emptyset$. Then $W^u(X^j) \cap W^s(X^s) \neq \emptyset$.

Proof of Lemma 2.2.34. The proof uses the fact that $W^s(X^s)$ is open and, by invariance, intersects $W^u_{\text{loc}}(X^i)$, so any submanifold K which is C^1 close to $W^u_{\text{loc}}(X^i)$ also intersects $W^s(X^s)$. The Inclination Lemma then guarantees that a disk in $W^u(X^j)$ is C^1 close to $W^u_{\text{loc}}(X^i)$.

Since $W^s(X^s)$ is invariant and intersects $W^u(X^i)$, $W^s(X^s) \cap W^u_{\text{loc}}(X^i) \neq \emptyset$. So, let $q \in W^s(X^s) \cap W^u_{\text{loc}}(X^i)$. By the definition of $W^s(X^s)$, there exists T < 0 such that $q \in W^u_{\text{loc}}(X^i) \cap \phi_T W^s_{\text{loc}}(X^s)$. If X^i is an equilibrium point, let $B = W^u_{\text{loc}}(X^i)$. If X^i is a periodic orbit, let S be a cross section containing q and let $B = W^u_{\text{loc}}(X^i) \cap S$. Then it can be shown that B is transverse to $\phi_T W^s_{\text{loc}}(X^s)$ (in S if X^i is a periodic orbit) by an analogous argument as in the proof of Lemma 2.2.31. Since $\phi_T W^s_{\text{loc}}(X^s)$ and B are compact submanifolds with transversal intersection, by [KH99, Proposition A.3.16,Corollary A.3.18] there exists $\epsilon > 0$ such that if D' is a compact submanifold which is ϵC^1 -close to B then it has a point of transversal intersection with $\phi_T W^s_{\text{loc}}(X^s)$, hence with $W^s(X^s)$.

Let $x \in W^s(X^i) \cap W^u(X^j)$. Since the intersection is transversal by Assumption 2.2.5, if X^i is an equilibrium point there exists a C^1 disk $D \subset W^u(X^j)$ centered at x with D transverse to $W^s(X^i)$. Similarly, if X^i is a periodic orbit there exists a disk $D \subset W^u(X^j) \cap S$ centered at x with D transverse to $W^s(X^i) \cap S$ in S. By Lemma 2.2.30, dim $W^u(X^j) \ge \dim W^u(X^i)$, so we may choose D such that dim $D = \dim B$. If X^i is an equilibrium point let $\hat{f} = \phi_1$, and if X^i is a periodic orbit, there exists $n_0 > 0$ such that $n \ge n_0$ implies $\hat{f}^n(D)$ is ϵC^1 -close to B. By the choice of ϵ , $\hat{f}^n(D) \cap W^s(X^s) \neq \emptyset$. Since $D \subset W^u(X^j)$ invariant, this implies that $W^u(X^j) \cap W^s(X^s) \neq \emptyset$.

Lemma 2.2.35 was reported as [CHW88, Theorem 3-8], where our Assumption 2.2.2 was replaced by the weaker assumption that for every $x \in \partial W^s(X^s)$, the trajectory of x converges to a critical element in forwards time. However, the proof of [CHW88, Theorem 3-8] relies crucially on [CHW88, Lemma 3-5], which has been disproven (see Section 2.1) to show that a particular heteroclinic sequence has finite length. In contrast, the proof of Lemma 2.2.35 shows that an analogous heteroclinic sequence has finite length using Lemma 2.2.33, which relies on Assumption 2.2.2.

Lemma 2.2.35. If $X^i \subset \partial W^s(X^s)$ then $W^u(X^i) \cap W^s(X^s) \neq \emptyset$.

Proof of Lemma 2.2.35. We first construct a heteroclinic sequence of critical elements, which must be finite by Lemma 2.2.33. Then we show that the unstable manifold of the final critical element in the sequence intersects $W^s(X^s)$. Working backwards, we argue that the unstable manifold of every critical element in the sequence intersects $W^s(X^s)$ using Lemma 2.2.34, which implies the result.

The first step is the construction of the heteroclinic sequence $\{X^j\}_{j\in \tilde{I}}$ where \tilde{I} is an ordered and countable indexing set. As $X^j \subset \partial W^s(X^s)$, by Lemma 2.2.29 there exists $x_j \in (W^u(X^j) - X^j) \cap \overline{W^s}(X^s)$. If $x_j \in W^s(X^s)$ then we have finished constructing the heteroclinic sequence, so suppose $x_j \in \partial W^s(X^s)$. Then by Assumptions 2.2.2-2.2.3, $x_j \in W^s(X^{j+1})$ for some critical element $X^{j+1} \subset \partial W^s(X^s)$. Iterating this procedure yields a heteroclinic sequence $\{X^j\}_{j\in \tilde{I}}$. By Lemma 2.2.33 it has finite length. The final element of the sequence, call it X^m , must satisfy $W^u(X^m) \cap W^s(X^s) \neq \emptyset$, since otherwise there would be another element X^{m+1} that would be added to the heteroclinic sequence by the procedure above.

We conclude by showing that the unstable manifold of each critical element in the heteroclinic sequence must intersect $W^s(X^s)$, which implies the result. For any $j \in \tilde{I}$, suppose $W^u(X^j) \cap W^s(X^s) \neq \emptyset$. By recursion, it suffices to show that this implies $W^u(X^{j-1}) \cap W^s(X^s) \neq \emptyset$. However, by the construction of the sequence we have that $W^u(X^{j-1}) \cap W^s(X^j) \neq \emptyset$ is a transversal intersection. Hence, the result follows from Lemma 2.2.34.

Proof of Theorem 2.2.12. Fix $i \in I$. By Lemma 2.2.35, $W^u(X^i) \cap W^s(X^s) \neq \emptyset$. To show that $W^s(X^i) \subset \partial W^s(X^s)$, it suffices to show that $W^s_{\text{loc}}(X^i) \subset \partial W^s(X^s)$ since $\partial W^s(X^s)$ is invariant and by the definition of $W^s(X^i)$. Let $x \in W^s_{\text{loc}}(X^i)$. By the proof of Lemma 2.2.34, there exists a disk D centered at x, contained in the ϵ -neighborhood of x in M, and transverse to $W^s_{\text{loc}}(X^i)$, such that $\phi_t(D) \cap W^s(X^s) \neq \emptyset$ for some t > 0. By invariance, $D \cap W^s(X^s) \neq \emptyset$. Since D is contained in the ϵ -neighborhood of x in M, $d_S(x, \overline{W^s}(X^s)) = d_S(x, W^s(X^s)) \leq \epsilon$. As this holds for all $\epsilon > 0$, $d_S(x, \overline{W^s}(X^s)) = 0$. Since $\{x\}$ is compact and $\overline{W^s}(X^s)$ is closed, this implies that $x \in \overline{W^s}(X^s)$. However, $x \in W^s_{\text{loc}}(X^i)$ implies that $x \in \partial W^s(X^s)$. Thus, $W^s_{\text{loc}}(X^i) \subset \partial W^s(X^s)$, so $W^s(X^i) \subset \partial W^s(X^s)$. Hence $\bigcup_{i \in I} W^s(X^i) \subset \partial W^s(X^s)$.

By Assumption 2.2.3, if $\gamma \subset \partial W^s(X^s)$ is an orbit then $\omega(\gamma) = X^j$ for some $j \in I$, which implies that $\gamma \subset W^s(X^j)$. Thus, $\partial W^s(X^s) \subset \bigcup_{i \in I} W^s(X^i)$.

2.2.6 Proof of Theorem 2.2.23 and Corollaries

The proof of Theorem 2.2.23 and its corollaries proceeds by paralleling the treatment of the fixed parameter case in Section 2.2.5. The recurring strategy of the proofs of this section will be to reduce to the fixed parameter case as much as possible, and then to rely on the results and proofs of Section 2.2.5 to complete the arguments.

We begin by defining functions whose images for fixed $p \in J$ are the critical elements and their local stable and unstable manifolds for the vector field V_p . As there are finitely many hyperbolic critical elements

 $\{X_{p_0}^i\}_{i\in I}$, we may assume J sufficiently small such that they and their local stable and unstable manifolds are well-defined and vary C^1 continuously with parameter over J. For brevity, define the notation: $W^s_{\text{loc}}(X^s(p_0))$ (S^s), and for $i \in I$ define $X^i(p_0)$ (S^i), $W^s_{\text{loc}}(X^i(p_0))$ (S_s^i) , and $W_{loc}^u(X^i(p_0))$ (S_u^i) . As the critical elements and their local stable and unstable manifolds vary C^1 continuously with parameter, for $i \in I$ there exist the following C^1 maps. One maps p onto $X^{i}(p)$ (F^{i}), where $F^{i}: S^{i} \times J \to M$, as the image of $F^{i}(\cdot, p)$. Another maps p onto $W^s_{\text{loc}}(X^i(p))$ (F^i_s) , where $F^i_s: S^i_s \times J \to M$, as the image of $F^i_s(\cdot, p)$. The next maps p onto $W^u_{\text{loc}}(X^i(p))$ (F^i_u) , $F^i_u: S^i_u \times J \to M$, as the image of $F^i_u(\cdot, p)$. The last one maps p onto $W^s_{\text{loc}}(X^s(p))$ (F^s), where $F^s: S^s \times J \to M$, as the image of $F^s(\cdot, p)$. Furthermore, for any $p \in J, F^i(\cdot, p), F^i_s(\cdot, p), F^i_u(\cdot, p), \text{ and } F^s(\cdot, p) \text{ are } C^1 \text{ diffeomorphisms onto their images. In$ other words, F^i , F^i_s , F^i_u , and F^s describe quantitatively how the critical elements and their local stable and unstable manifolds vary C^1 with parameter. Let π_J be the projection onto parameter space: $\pi_{J}(x,p) = p$. The function above have codomain M, but it will sometimes be convenient to have the codomain be $M \times J$. To this end, let $G^i = (F^i, \pi_J), G^i_s = (F^i_s, \pi_J),$ $G_u^i = (F_u^i, \pi_J)$, and $G^s = (F^s, \pi_J)$, and note that these functions are C^1 injections because for fixed $p \in J$ the functions above are C^1 diffeomorphisms onto their images.

Lemma 2.2.36 shows that $W^s(X_I^s)$ is open and invariant in $M \times J$.

Lemma 2.2.36. $W^{s}(X_{J}^{s})$ is open and invariant in $M \times J$.

Proof of Lemma 2.2.36. Since S^s is C^1 diffeomorphic to $W^s_{\text{loc}}(X^s_{p_0})$, a codimension-zero embedded submanifold with boundary in M, $G^s|_{\text{int } S^s \times J}$ is a continuous injection between manifolds of the same dimension so, by invariance of domain [Hat01, Theorem 2B.3], an open map. Thus, $G^s(\text{int } S^s \times J)$ is an open set in $M \times J$. Hence, by definition of the local stable manifold, $W^s(X^s_J) = \bigcup_{t \leq 0} \phi_t(G^s(\text{int } S^s \times J))$ is a union of open sets since ϕ_t is a C^1 diffeomorphism for each t, hence open. Since $W^s(X^s_J) = \bigcup_{p \in J} W^s(X^s(p))$ is a union of invariant sets, it is invariant.

Let $p_0 \in J$ be a fixed parameter value such that Assumptions 2.2.14-2.2.18 hold. Recall from Section 2.2.2 that the family of a critical element refers here to the family obtained from a single critical element as the parameter value is varied over $p \in J$. Similar to Lemma 2.2.28, Lemma 2.2.37 provides a technical construction, for any critical element contained in $\partial W^s(X_J^s)$, of a compact set contained in its family of unstable manifolds such that for any sufficiently small neighborhood N of this compact set in $M \times J$, the following holds. The union over time of the time-t flow ϕ_t of N over all negative times t, together with the family of stable manifolds of the critical element, contains an open neighborhood of the critical element in $M \times J$. The key difference from the fixed parameter case Lemma 2.2.28 is that here the open neighborhood that is contained in the union is open in $M \times J$, whereas for Lemma 2.2.28 it was open in M alone. This is important because for a critical element contained in $\partial W^s(X_J^s)$, an open neighborhood in $M \times J$ of that critical element is required to guarantee it intersects $W^{s}(X_{J}^{s})$. This result will be fundamental in proving the claim below that if a critical element in M_{p_0} is contained in $\partial W^s(X_J^s)$ then its unstable manifold intersects $\overline{W^s}(X_J^s)$ in M_{p_0} . Recall that if D is a subset of a metric space and $\epsilon > 0$, the notation D_{ϵ} refers to the subset of the metric space such that for each $x \in D_{\epsilon}$ there exists $y \in D$ with $d(x, y) < \epsilon$.

Lemma 2.2.37. For any $i \in I$ and any $\epsilon > 0$ sufficiently small, there exists a compact set $D \subset W^u_{loc}(X^i_J) - X^i_J$ and an open neighborhood N of D in $M \times J$ such that $N \subset D_{\epsilon}$, $D_{\epsilon} \cap X^i_J = \emptyset$, and $\bigcup_{t \leq 0} \phi_t(N) \bigcup W^s(X^i_J)$ contains an open neighborhood of $X^i_{p_0}$ in $M \times J$.

Proof of Lemma 2.2.37. Let $\epsilon > 0$. If $X_{p_0}^i$ is an equilibrium point, let $\hat{f} = \phi_1$ be the time-one flow. If $X_{p_0}^i$ is a periodic orbit, let $\hat{f} = \hat{\tau}$ be the first return map of a Poincaré cross section S (note that this map is well-defined and C^1 with respect to parameter value $p \in J$). Let $D'_p = G^i_u(S^i_u \times \{p\})$ for any $p \in J$. Let D_p be the topological closure of $D'_p - (\hat{f})^{-1}(D'_p)$ in M. We will prove the following claim: there exists an open neighborhood N' of D_{p_0} in M and an open neighborhood \hat{U} of $X^i_{p_0}$ in M such that for J sufficiently small, $p \in J$ implies that $D_p \subset N' \subset (D_p)_{\epsilon}, X^i(p) \subset \hat{U}$, and the forward orbit of any point $x \in \hat{U} - W^s(X^i(p))$ under V_p enters N' in finite time. For fixed $p \in J$, this claim is illustrated by Fig. 2.6. From this claim, the main result is straightforward as follows. Choose a subset $J' \subset J$ compact and connected with $p_0 \in$ int J'. Let $D' = G^i_u(S^i_u \times J')$ the continuous image of a compact set, hence compact in $M \times J$. Let D be the topological closure of $D' - (\hat{f})^{-1}(D')$ in $M \times J$. Since D' is contained in the local unstable manifold, $(\hat{f})^{-1}|_{D'}$ is contracting. Hence, $(\hat{f})^{-1}(D') \subset D'$, so $D \subset D'$. As D is closed in D' compact, D is compact. By the claim above, for J sufficiently small, $D \subset N' \times J \subset D_{\epsilon}$, D is disjoint from X_J^i , and $\bigcup_{t \leq 0} \phi_t(N' \times J) \bigcup W^s(X_J^i)$ contains $\hat{U} \times J'$ where \hat{U} is an open neighborhood in M containing $X_{p_0}^i$. Letting $N = N' \times J$ completes the proof.

So, it suffices to prove the claim above. We begin with the construction of the C^1 disk family for \hat{f}_{p_0} exactly as in the proof of Lemma 2.2.28. Then it is shown using the Inclination Lemma that for a C^1 perturbation of the diffeomorphism \hat{f}_{p_0} , constructing the C^1 disk family for the perturbed diffeomorphism gives a C^1 continuous disk family that is uniformly C^1 - close to the original C^1 continuous disk family. Consequently, it is possible to choose \hat{U} an open neighborhood of $W^u_{\text{loc}}(X^i_{p_0})$ sufficiently small such that it is contained in the perturbed disk family and, therefore, the forwards orbit of each point in \hat{U} under the perturbed diffeomorphism either converges to the perturbation of $X^i_{p_0}$ or enters N' in finite time. The full proof is provided in Appendix B.

The technical construction of Lemma 2.2.37 is used to show that the unstable manifold of any critical element in $\partial W^s(X_J^s)$ must have nonempty intersection with $\overline{W^s}(X_J^s) \cap M_{p_0}$. By requiring that the intersection occurs in M_{p_0} , we will be able to reduce to the fixed parameter case of Lemma 2.2.35, which will ensure that the unstable manifold actually intersects $W^s(X_J^s) \cap M_{p_0}$ (see Lemma 2.2.39). Although Lemma 2.2.29 and Lemma 2.2.38 both show the intersection of the unstable manifold with the closure of a stable manifold, there is a crucial difference: for Lemma 2.2.29 this closure is taken in M for a fixed parameter, whereas for Lemma 2.2.38 the closure is taken in $M \times J$. As Example 2.2.1 showed, taking the closure in $M \times J$, namely $\overline{W^s}(X_J^s)$, will in general give a larger set than taking the closure in M, namely $\sqcup_{p \in J} \overline{W^s}(X^s(p))$. This motivates the need for Lemma 2.2.37 and Lemma 2.2.38 to explicitly treat the more difficult case where the closure is taken in $M \times J$.

Lemma 2.2.38. For any $i \in I$, $\{W^u(X^i_{p_0}) - X^i_{p_0}\} \cap \overline{W^s}(X^s_J) \neq \emptyset$.

Proof of Lemma 2.2.38. The proof is similar to that of Lemma 2.2.29, which relied on the technical result of Lemma 2.2.28 to show that the distance between an annulus in $W_{\text{loc}}^u(X^i)$ (denoted by D in that proof) and $W^s(X^s)$ was less than ϵ for any $\epsilon > 0$, and then sent $\epsilon \to 0$ to establish the desired intersection. Here, the goal is to use Lemma 2.2.37 in a similar fashion. The key difference is that, since the critical element $X_{p_0}^i$ lies in $\partial W^s(X_J^s)$, and not necessarily in $\partial W^s(X_{p_0}^s)$, it is necessary to consider distances in parameter space J as well. In particular, Lemma 2.2.37 is used to establish that there exists a point $(x_r, p_r) \in W^s(X_J^s)$ such that the distance from an annulus in $W_{\text{loc}}^u(X_J^i)$ (denoted by D in this proof) to (x_r, p_r) is less then ϵ and the distance from p_0 to p_r is less than r for any $\epsilon > 0$ and r > 0. Then,

sending $r \to 0$ reduces to the fixed parameter argument from the proof of Lemma 2.2.29 (which only requires sending $\epsilon \to 0$). Let $B(p_0, r)$ be the ball of radius r centered at p_0 in J.

Let $\epsilon > 0$. Shrinking ϵ if necessary, by Lemma 2.2.37 there exists a compact set $D \subset W^u(X^i_J) - X^i_J$ and an open neighborhood N of D in $M \times J$ such that $N \subset D_{\epsilon}$, $D_{\epsilon} \cap X_J^i = \emptyset$, and $\bigcup_{t \leq 0} \phi_t(N) \bigcup W^s(X_J^i)$ contains an open neighborhood of $X_{p_0}^i$ in $M \times J$ - call this open neighborhood N'. Let $N_r = N \cap (M \times B(p_0, r))$ and $N'_r = N' \cap (M \times B(p_0, r))$ be the intersections of the above neighborhoods with $M \times B(p_0, r)$. Since N'_r is an open neighborhood of $X_{p_0}^i \subset \partial W^s(X_J^s), N_r' \cap W^s(X_J^s) \neq \emptyset$. Since $\bigcup_{t \leq 0} \phi_t(N_r) \bigcup W^s(X_J^i)$ contains N'_r , $N'_r \cap W^s(X^s_J) \neq \emptyset$, and $W^s(X^i_J) \cap W^s(X^s_J) = \emptyset$, there exists T > 0 such that $\phi_{-T}(N_r) \cap W^s(X_J^s) \neq \emptyset$. By invariance of $W^s(X_J^s)$, this implies that $N_r \cap W^s(X_J^s) \neq \emptyset$. So, let $(x_r, p_r) \in N_r \cap W^s(X_J^s)$ and send r to zero. As $N_r \subset D_\epsilon$ and $W^s(X_J^s) \subset \overline{W^s}(X_J^s)$, $N_r \cap W^s(X_J^s) \subset D_\epsilon \cap \overline{W^s}(X_J^s)$ compact. Hence, passing to a subsequence if necessary we have that $(x_r, p_r) \to (x, p) \in D_{\epsilon} \cap \overline{W^s}(X_J^s)$. By definition of N_r , since $r \to 0$ we must have $p = p_0$. In the proof of Lemma 2.2.37, D is constructed as the disjoint union of a C^1 continuous family of embedded submanifolds in M. Therefore, $(x, p_0) \in D_{\epsilon}$ implies that $d_S((x, p_0), D \cap M_{p_0}) \leq \epsilon$. Thus, $d_S(\overline{W^s}(X_J^s), D \cap M_{p_0}) \leq \epsilon$ for all $\epsilon > 0$. Now we have reduced to the fixed parameter case, and the rest of the proof proceeds as in the proof of Lemma 2.2.29 by noting that the above implies (sending $\epsilon \to 0$) that $d_S(\overline{W^s}(X_J^s), D \cap M_{p_0}) = 0$, so that $\overline{W^s}(X_J^s) \cap (D \cap M_{p_0}) \neq \emptyset$, which implies the result.

Thanks to the work of Lemma 2.2.37 and Lemma 2.2.38, the varying parameter case treated in this section was essentially reduced to the fixed parameter case of Section 2.2.5. Hence, Lemma 2.2.39 is exactly analogous to its fixed parameter counterpart Lemma 2.2.35 in both its statement and its proof.

Lemma 2.2.39. For any $i \in I$, $W^u(X^i_{p_0}) \cap W^s(X^s_{p_0}) \neq \emptyset$.

Proof of Lemma 2.2.39. The proof is identical to the proof of Lemma 2.2.35, substituting Lemma 2.2.38 for Lemma 2.2.29. \Box

Proof of Theorem 2.2.23. For any $p \in J$ and any $x \in \partial W^s(X^s(p))$, there exist $x_n \in W^s(X^s(p))$ with $x_n \to x$. Hence, $(x_n, p) \in W^s(X^s_J)$ with $(x_n, p) \to (x, p)$, so $(x, p) \in \overline{W^s}(X^s_J)$ closed. As $x \notin W^s(X^s(p))$, this implies that $(x, p) \in \partial W^s(X^s_J)$. Hence,

$$\sqcup_{p \in J} \partial W^s(X^s(p)) \subset \partial W^s(X^s_J). \tag{2.12}$$

We claim that for J sufficiently small, for any $p \in J$ and $i \in I$, $W^u(X^i(p)) \cap W^s(X^s(p)) \neq \emptyset$. For, let $i \in I$. Then by Lemma 2.2.39, we have that $W^u(X_{p_0}^i) \cap W^s(X_{p_0}^s) \neq \emptyset$. This implies that there exists T > 0 such that $\phi_T(W_{\text{loc}}^u(X_{p_0}^i)) \cap W_{\text{loc}}^s(X_{p_0}^s) \neq \emptyset$. This intersection is trivially transverse since $W^s(X_{p_0}^s)$ is an open set in M_{p_0} . Since $\{\phi_T(W_{\text{loc}}^u(X^i(p)))\}_{p\in J}$ and $\{W_{\text{loc}}^s(X^s(p))\}_{p\in J}$ are two C^1 continuous families over J of compact embedded submanifolds with boundary, and since they have a point of transversal intersection at $p = p_0$, for J sufficiently small $p \in J$ implies [KH99, Proposition A.3.16,Corollary A.3.18] that $\phi_T(W_{\text{loc}}^u(X^i(p))) \cap W_{\text{loc}}^s(X^s(p)) \neq \emptyset$. Hence, for sufficiently small J, since I is finite, $W^u(X^i(p)) \cap W^s(X^s(p)) \neq \emptyset$ for all $i \in I$ and $p \in J$, so the claim above follows. Let $(x,p) \in \partial W^s(X_j^s)$. By Assumption 2.2.17, $x \in W^s(X_p^j)$ for some $j \in I$. By the claim above applied to the particular j here, $W^u(X_p^j) \cap W^s(X^s(p)) \neq \emptyset$. Now the argument reduces to the fixed parameter case from earlier, and we can use the proof of Theorem 2.2.12 to show that $W^s(X_p^j) \subset \partial W^s(X^s(p))$. As $(x,p) \in \partial W^s(X_j^s)$ was arbitary, we have

$$\partial W^s(X_J^s) \subset \bigcup_{j \in I} W^s(X_J^i) \subset \sqcup_{p \in J} \partial W^s(X^s(p)).$$
(2.13)

Then Eqs. 2.12-2.13 imply the result.

Proof of Corollary 2.2.24. Recall the definitions of S_s^i and F_s^i from the beginning of Section 2.2.6. Let $p_n \in J$ with $p_n \to p'$ for some $p' \in J$. First, let $x \in \partial W^s(X_{p'}^s)$. Then $x \in W^s(X_{p'}^i)$ for some $i \in I$. So, there exists T > 0 such that $\phi_T(x, p') \in W_{loc}^s(X_{p'}^i)$ and $y \in S_s^i$ such that $F_s^i(y, p') = \phi_T(x, p')$. Let $x_n = \phi_{-T}(F_s^i(y, p_n), p_n) \in W^s(X_{p_n}^i)$ by invariance of $W^s(X_{p_n}^i)$. Thus, $x_n \in W^s(X_{p_n}^i) \subset \partial W^s(X_{p_n}^s)$ by Theorem 2.2.23. Furthermore, $x_n \to x$ since ϕ_{-T} and F_s^i are C^1 . Hence, $x \in \liminf_{n \to \infty} \partial W^s(X_{p_n}^s)$, so $\partial W^s(X_{p'}^s) \subset \liminf_{n \to \infty} \partial W^s(X_{p_n}^s)$.

Next, let $x \in \limsup_{n \to \infty} \partial W^s(X_{p_n}^s)$. Then there exist a subsequence $\{p_{n_m}\}_{m=1}^{\infty}$ of $\{p_n\}_{n=1}^{\infty}$ and a sequence $\{x_m\}_{m=1}^{\infty}$ such that $x_m \in \partial W^s(X_{p_{n_m}}^s)$ for all m and $x_m \to x$. By Theorem 2.2.23, $\partial W^s(X_{p_{n_m}}^s) \subset \partial W^s(X_J^s)$, so that $(x_m, p_{n_m}) \in \partial W^s(X_J^s)$. As $\partial W^s(X_J^s)$ is closed, $\lim_{m\to\infty}(x_m, p_{n_m}) = (x, p') \in \partial W^s(X_J^s)$. By Theorem 2.2.23, $\partial W^s(X_J^s) = \sqcup_{p \in J} \partial W^s(X^s(p))$, so intersecting both sides with $M_{p'}$ implies that $\partial W^s(X_J^s) \cap M_{p'}$ $= \partial W^s(X_{p'}^s) \times \{p'\}$. Hence, $(x, p') \in \partial W^s(X_J^s)$ implies that $x \in \partial W^s(X_{p'}^s)$. Thus, $\limsup_{n\to\infty} \partial W^s(X_{p_n}^s) \subset \partial W^s(X_{p'}^s)$. Together, these imply that $\lim_{n\to\infty} \partial W^s(X_{p_n}^s) = \partial W^s(X_{p'}^s)$. As $p_n \to p'$ were arbitrary, this implies that $\{\partial W^s(X^s(p))\}_{p\in J}$ is a Chabauty continuous family of subsets of M.

Proof of Corollary 2.2.25. By Corollary 2.2.24, we have that $\{\partial W^s(X^s(p))\}_{p\in J}$ is a Chabauty

continuous family of subsets of M. Since M is compact, Hausdorff continuity is equivalent to Chabauty continuity. Hence, $\{\partial W^s(X^s(p))\}_{p\in J}$ is a Hausdorff continuous family of subsets of M.

Proof of Corollary 2.2.26. Let $\Omega(V_{p_0}) = \{X_{p_0}^i\}_{i=1}^n$ a finite union of hyperbolic critical elements since V_{p_0} is Morse-Smale. Palis showed [Pal69, Theorem 3.5] that for any sufficiently small C^1 perturbation to V_{p_0} , so for J sufficiently small, $p \in J$ implies that V_p is still Morse-Smale with $\Omega(V_p) = \{X^i(p)\}_{i=1}^n$. Let $\{X_{p_0}^i\}_{i=1}^k = \Omega(V_{p_0}) \cap (\partial W^s(X_J^s) \cap M_{p_0})$ which is a finite union of critical elements of V_{p_0} since $\Omega(V_{p_0})$ is finite, and k < n. Note that $\{V_p\}_{p \in J}$ satisfies Assumption 2.2.15 and Assumption 2.2.18 for J sufficiently small since V_{p_0} is Morse-Smale. Note that both $\bigcup_{i>k} X_{p_0}^i$ and $\partial W^s(X_J^s) \cap M_{p_0}$ are compact, so since M is a normal space there exists an open set N such that $\partial W^s(X^s_J) \cap M_{p_0} \subset N$ and $N \cap \bigcup_{i>k} X^i_{p_0} = \emptyset$. As $\Omega(V_{p_0}) = \bigcup_{i=1}^n X_{p_0}^i, \text{ this implies that } \Omega(V_{p_0}) \cap N = \bigcup_{i=1}^k X_{p_0}^i = \Omega(V_{p_0}) \cap (\partial W^s(X_J^s) \cap M_{p_0}).$ Hence, Assumption 2.2.14 is satisfied. So, it suffices to show that $\{V_p\}_{p\in J}$ satisfies Assumption 2.2.17 as well. By Lemma 2.2.39, for any $i \in \{1, .., n\}$ and any $p \in J, X^i(p) \subset \partial W^s(X^s_J)$ implies that $W^u(X^i_{p_0}) \cap W^s(X^s_{p_0}) \neq \emptyset$. So, let $x \in W^u(X^i_{p_0}) \cap W^s(X^s_{p_0})$. As $x \in \overline{W^s}(X^s_{p_0})$ closed and invariant, the closure of the orbit of x is contained in $\overline{W^s}(X^s_{p_0})$. Since $\alpha(x) = X^i_{p_0}$ is contained in the closure of the orbit of x, $X_{p_0}^i \subset \overline{W^s}(X_{p_0}^s)$. Since $X_{p_0}^i$ does not intersect $W^s(X^s_{p_0}), X^i_{p_0} \subset \partial W^s(X^s_{p_0})$. By Theorem 2.2.23, $\partial W^s(X^s_J) = \sqcup_{p \in J} \partial W^s(X^s(p))$, so $X_{p_0}^i \subset \partial W^s(X_{p_0}^s) \subset \partial W^s(X^s(p)) \cap M_{p_0}$. Therefore, by definition of k above, we must have $i \in \{1, ..., k\}$. Hence, for any $i \in \{1, ..., n\}$ and $p \in J, X^i(p) \subset \partial W^s(X^s_J)$ implies that $i \in \{1, ..., k\}$, so that $X_{p_0}^i \subset \partial W^s(X_J^s)$. As $\Omega(V_p) = \{X^i(p)\}_{i \in \{1, ..., n\}}$, this implies that no new nonwandering elements can enter $\partial W^s(X_J^s)$ for $p \neq p_0$, so Assumption 2.2.17 is satisfied. Thus, $\{V_p\}_{p\in J}$ satisfy Assumptions 2.2.14-2.2.18. Therefore, by Corollary 2.2.25, $\{\partial W^s(X^s(p))\}_{p\in J}$ is a Hausdorff continuous family of subsets of M.

2.2.7 Conclusion

This work provides theoretical motivation for the development of algorithms which estimate the set of recovery values by estimating its boundary via the computation of critical parameter values. The main technical result required for the algorithms' motivation was that, for the family of vector fields discussed above, which are in essense Morse-Smale along the RoA boundary, the RoA boundary is Hausdorff (for a compact Riemannian manifold) or Chabauty (for Euclidean space) continuous with respect to parameter. This result was established together with a decomposition of the RoA boundary into the union of the stable
manifolds of the critical elements it contains, and it was shown that this decomposition persists under small variations in parameter values. Section 2.3 shows that Assumption 2.2.17, which represents Ω -stability of the RoA boundary, is a consequence of the other assumptions. Hence, this will complete the proof of RoA boundary continuity under small parameter variation for vector fields which are in essence Morse-Smale along the RoA boundary.

2.3 Ω -Stability of Region of Attraction Boundary

2.3.1 Introduction

The purpose of this section is to show the second part of the proof of continuity of the RoA boundary under small parameter variation. The goal is to show that if the vector field is Morse-Smale along the RoA boundary, then the RoA boundary is Ω -stable, so the additional assumptions required for Corollaries 2.2.24-2.2.25 beyond the Morse-Smale assumptions at a fixed parameter value are not required. Unfortunately, Example 2.3.3 illustrates a smooth vector field in Euclidean space which is Morse-Smale, but whose RoA boundary varies discontinuously under arbitrarily small changes in parameter value due to the entrance of an additional equilibrium point into the RoA boundary. This is a counterexample to Theorem 5.4 of [CC95], which claimed that under weaker assumptions than the above, for strong C^1 perturbations to the vector field no new nonwandering points could enter the RoA boundary. Hence, while the RoA boundary continuity result holds when the vector field is Morse-Smale along the RoA boundary for compact Riemannian manifolds, as shown in Theorem 2.3.9, the corresponding result for Euclidean space requires additional generic assumptions as well as an assumption regarding behavior near infinity, as given by Theorem 2.3.20.

The section is organized as follows. Section 2.3.2 provides relevant background and notation. Section 2.3.3 introduces several examples in Euclidean space. Section 2.3.4 describes the main results and Section 2.3.5 provides their proofs for each of compact Riemannian manifolds and Euclidean space. Finally, Section 2.3.6 offers some conclusions.

2.3.2 Background

This section collects definitions and notation that will be used throughout the paper. Let M be either a compact Riemannian manifold or Euclidean space. Let V_0 be a C^1 vector field on M. We let ϕ denote its flow, and write $\phi_t(x, V_0)$ for the flow under V_0 at time t starting from initial condition x. For a set $E \subset M$, let $\phi_t(E, V_0) = \bigcup_{x \in E} \phi_t(x, V_0)$. We say that a set

 $E \subset M$ is forward (backward) invariant under V_0 if $\phi_t(E, V_0) \subset E$ for all t > 0 (t < 0). E is invariant if it is both forward and backward invariant. The nonwandering set of V_0 , denoted $\Omega(V_0)$, is the set of points $x \in M$ such that for any open neighborhood U of x in M and any T > 0, there exists $t \in \mathbb{R}$ with |t| > T such that $\phi_t(U, V_0) \cap U \neq \emptyset$. $\Omega(V_0)$ is closed in M and invariant under V_0 . A point $x \in M$ is wandering under V_0 if $x \notin \Omega(V_0)$. For any $x \in M$, its ω limit set under V_0 , denoted $\omega(x, V_0)$, is the set of points $y \in M$ such that there exists a sequence of times $t_n \to \infty$ such that $\phi_{t_n}(x, V_0) \to y$. For brevity, and when the vector field is clear from context, we omit it and write simply $\omega(x)$. Similarly, for any $x \in M$, its α limit set under V_0 , denoted $\alpha(x, V_0)$, is the set of points $y \in M$ such that there exists a sequence of times $t_n \to -\infty$ such that $\phi_{t_n}(x, V_0) \to y$. Note that for any $x \in M$, $\omega(x)$ and $\alpha(x)$ are closed in M, invariant under V_0 , and contained in $\Omega(V_0)$. Furthermore, if $x, y \in M$ are part of the same orbit under V_0 , then their ω and α limit sets under V_0 are equal.

An equilibrium point x_e of V_0 is hyperbolic if for every $t \in \mathbb{R}$, the differential $d(\phi_t)_{x_e}$ has no eigenvalues with absolute value equal to one. A periodic orbit γ of V_0 is hyperbolic if there exists $x \in \gamma$, a smooth cross section S of γ centered at x, and a C^1 first return map $\hat{\tau}: S \to S$ such that $d\hat{\tau}_x$ has no eigenvalues with absolute value equal to one. Let critical elements refer to the union of equilibrium points and periodic orbits of V_0 . A sink denotes a stable hyperbolic equilibrium point, and a source denotes an unstable hyperbolic equilibrium point. By [HPS77], a hyperbolic critical element X of V_0 has local C^1 stable and unstable manifolds, denoted $W^s_{\text{loc}}(X)$ and $W^s_{\text{loc}}(X)$, respectively, and for every t > 0 $\phi_t(W^s_{\text{loc}}(X), V_0) \subset W^s_{\text{loc}}(X)$ and $\phi_{-t}(W^u_{\text{loc}}(X), V_0) \subset W^u_{\text{loc}}(X)$. In other words, the local stable and unstable manifolds are contracting under the time-t flow ϕ_t in forwards and backwards time, respectively, and therefore are forward and backward invariant, respectively. The global stable manifold $W^{s}(X)$ and unstable manifold $W^{u}(X)$ are obtained as $W^{s}(X) =$ $\bigcup_{t \leq 0} \phi_t(W^s_{\text{loc}}(X), V_0)$ and $W^u(X) = \bigcup_{t \geq 0} \phi_t(W^u_{\text{loc}}(X), V_0)$, respectively. For $\hat{A} \subset M$ a C^1 submanifold, let $T_x \hat{A}$ denote the tangent space to \hat{A} at x. A pair \hat{A} and \hat{B} of C^1 submanifolds of M are transverse if for every $x \in \hat{A} \cap \hat{B}$, $T_x \hat{A} \oplus T_x \hat{B} = T_x M$. A vector field V_0 is Morse-Smale [Sma60] if $\Omega(V_0)$ consists of a finite union of critical elements, each critical element is hyperbolic, and all stable and unstable manifolds are transverse.

If M is Euclidean, there are two standard topologies for the set of C^1 vector fields on M: the strong and weak C^1 topologies [Hir76, p. 34-35]. For M a compact Riemannian manifold, the strong and weak C^1 topologies are the same, and we can unambiguously refer to merely the C^1 topology in this case. If a sequence $\{V_n\}_{n=1}^{\infty}$ of C^1 vector fields on M converges in the strong C^1 topology to a C^1 vector field V_0 , then there exists a compact set

 $K \subset M$ such that for n sufficiently large, V_n and V_0 agree outside of K. Both the strong and weak C^1 topologies are Baire spaces, meaning that countable intersections of open, dense sets are themselves dense. A residual set in a topological space is a set containing a countable intersection of open, dense sets. A property is called generic in either the strong or weak C^1 topology if the set of vector fields satisfying this property contains a residual set. By the above, this implies that generic properties are satisfied by a dense set in either the strong or weak C^1 topologies. Hyperbolicity of all critical elements and transversality of their stable and unstable manifolds are generic in these topologies [KH99, Ch. 7], although the assumption that $\Omega(V_0)$ consists of a finite union of critical elements is not. A topological space E is first countable if for every $x \in E$ there exists a countable collection $\{N_i^x\}_{i=1}^{\infty}$ of open neighborhoods of x such that for every open neighborhood N of x, there exists i such that $N_i^x \subset N$. Note that every manifold is first countable. If E is first countable, $x \in E$, and there does not exist an open set N containing E such that every point $y \in N$ possesses some property \mathcal{P} , then there must exist a sequence $\{x_n\}_{n=1}^{\infty}$ with $x_n \to x$ such that x_n does not possess property \mathcal{P} for all n. This fact will be used repeatedly to prove, by means of contradiction, the existence of an open set in parameter space (which will be a manifold) whose corresponding vector fields possess some desired property.

Morse-Smale vector fields on compact Riemannian manifolds were shown to be open in the C^1 topology [Pal69]. Crucial tools in these proofs were fundamental domains and neighborhoods. Let X = x be a hyperbolic equilibrium point with $\hat{f} = \phi_1(\cdot, V_0) : M \to M$ or let X be a hyperbolic periodic orbit with smooth cross section S centered at $x, \hat{\tau} : S \to S$ a C^1 first return map, and let $\hat{f} = \hat{\tau}$. A fundamental domain D of the stable manifold of \hat{f} is given by taking the closure of $W^s_{loc}(x) - \hat{f}(W^s_{loc}(x))$. It is straightforward to see that $\bigcup_{n \in \mathbb{Z}} \hat{f}^n(D) = W^s(x)$. Let N be an open neighborhood of D which is disjoint from $W^u(x)$. Then N is called a fundamental neighborhood of the stable manifold of \hat{f} . It is shown in Corollary 2 [JPdM82, p. 86] that $\bigcup_{n \geq 0} \hat{f}^n(N) \cup W^u_{loc}(x)$ contains an open neighborhood of x.

Let M be a compact Riemannian manifold and let \mathcal{M} be the set of closed subsets of M. For $A, B \in \mathcal{M}$ nonempty, let r(A, B) be the infimum over $\epsilon \geq 0$ such that the ϵ neighborhood of A, the set of points within a distance ϵ from A, contains B. For $A, B \in \mathcal{M}$ nonempty, define the Hausdorff distance $d_H(A, B)$ to be the maximum of r(A, B) and r(B, A). Then d_h is a metric [Hau57, Section 28] and its corresponding metric topology is called the Hausdorff topology. Furthermore, letting \mathcal{M} denote the set of closed subsets of M equipped with the Hausdorff metric, \mathcal{M} is compact [Hau57, Section 28]. If M is Euclidean, then the Chabauty topology is obtained by first considering the compact Riemannian manifold $M \cup \{\infty\}$. For every $A \in \mathcal{M}$, let $\overline{A} = A \cup \{\infty\}$. Then we can define a metric d_C on \mathcal{M} such that $d_C(A, B) = d_H(\overline{A}, \overline{B})$ where the Hausdorff distance is taken in the compact manifold $M \cup \{\infty\}$. Then the metric topology induced by d_C is called the Chabauty topology on \mathcal{M} .

If $\{E_n\}_{n=1}^{\infty}$ is a sequence of subsets of M, define $\liminf_{n\to\infty} E_n$ to be the set of points $x \in M$ such that there exists a sequence $x_n \to x$ with $x_n \in E_n$ for all n. Similarly, define $\limsup_{n\to\infty} E_n$ to be the set of points $x \in M$ such that there exists a subsequence $x_{n_m} \to x$ with $n_m \to \infty$ and $x_{n_m} \in E_{n_m}$ for all m. Then $E_n \to E_0$ in the Hausdorff topology (Chabauty topology if M is Euclidean) if and only if $\liminf_{n\to\infty} E_n = \limsup_{n\to\infty} E_n = E_0$. A parameterized family of subsets of M, $\{E_p\}_{p\in J}$, is Hausdorff (Chabauty) continuous if for every $p_0 \in J$, $p_n \to p_0$ implies that $E_{p_n} \to E_{p_0}$. For M a compact Riemannian manifold, a function $\tilde{f}: \tilde{E} \to \mathcal{M}$, where \tilde{E} is a topological space, is lower semi-continuous if for every convergent sequence $E_n \to E_0$ in \tilde{E} , $\liminf_{n\to\infty} \tilde{f}(E_n) \supset \tilde{f}(E_0)$. The set of points of continuity of such a lower semi-continuous function are residual in \tilde{E} by the Remark following Corollary 1 of [Kur68]. This property was used to prove that for generic V_0 in the strong C^1 topology, the closure of the union of equilibrium points and periodic orbits of V_0 contains all points $x \in \Omega(V_0)$ such that $\omega(x, V_0) \cup \alpha(x, V_0) \neq \emptyset$ [PR83]. The key technical element of this proof is the C^1 closing lemma for noncompact manifolds, which states that for any $x \in \Omega(V_0)$ such that $\omega(x, V_0) \cup \alpha(x, V_0) \neq \emptyset$, there exists V arbitrarily close to V_0 in the strong C^1 topology and y arbitrarily close to x in M such that y is periodic under V [PR83].

For V_0 a vector field on M, write $\{X^i(V_0)\}_{i\in I}$ for a collection of critical elements of V_0 . For V near V_0 , let $\{X^i(V)\}_{i\in I}$ denote the perturbations of those critical elements. If $\{V_p\}_{p\in J}$ is a strong C^1 continuous family of vector fields on M, let $X^i_J = \bigsqcup_{p\in J} X^i(V_p) \subset M \times J$. Similarly, let $W^s(X^i_J) = \bigsqcup_{p\in J} W^s(X^i(V_p)) \subset M \times J$ and let $W^u(X^i_J) = \bigsqcup_{p\in J} W^u(X^i(V_p)) \subset M \times J$. Let $\partial W^s(X^i)$ denote the boundary of $W^s(X^i)$ in M and let $\overline{W}^s(X^i)$ denote its closure in M. More generally, for a set E let ∂E be its boundary and \overline{E} its closure.

2.3.3 Examples

All of the examples in this section are of C^1 vector fields over \mathbb{R}^2 . For purposes of illustration, we identify its one point compactification, $\mathbb{R}^2 \cup \{\infty\}$, with the sphere \mathbb{S}^2 via stereographic projection, where the point infinity is identified with the north pole of the sphere. Each example is shown as a view of the northern hemisphere of the sphere from above. The vector field then extends naturally to the full sphere, and the lower hemisphere

of the sphere does not contain any nonwandering points in any of the examples below.

Example 2.3.1 (Nonwandering Points Accumulate on Boundary). One consequence of Theorem 2.3.9 is that for a compact Riemannian manifold possessing a stable hyperbolic equilibrium point $X^s(V_0)$, and such that V_0 is Morse-Smale along $\partial W^s(X^s(V_0))$, there exists an open neighborhood N of $\overline{W}^s(X^s(V_0))$ such that $\Omega(V_0) \cap N = \Omega(V_0) \cap \overline{W}^s(X^s(V_0))$. In other words, there are no nonwandering points outside of $\overline{W}^s(X^s(V_0))$ that accumulate on $\partial W^s(X^s(V_0))$. The existence of this neighborhood is crucial to the proof of Theorem 2.3.9, which argues that no new nonwandering points can enter $\partial W^s(X^s(V))$ for V close to V_0 , and its Corollary 2.3.10, which says that this boundary varies Hausdorff continuously for a C^1 continuous family of vector field. However, this example shows that in the case of Euclidean space, if V_0 is Morse-Smale along $\partial W^s(X^s(V_0))$, there may not exist an open neighborhood N of $\overline{W}^s(X^s(V_0))$ such that $\Omega(V_0) \cap N = \Omega(V_0) \cap \overline{W}^s(X^s(V_0))$. In particular, Figure 2.7 shows an example in which V_0 is Morse-Smale along $\partial W^s(X^s(V_0))$ but there exists a sequence of periodic orbits outside of $\overline{W}^s(X^s(V_0))$ which accumulate along $\partial W^s(X^s(V_0))$, so that no such neighborhood N exists. Hence, the methods used to prove Theorem 2.3.9 and Corollary 2.3.10 for compact Riemannian manifolds do not apply for Euclidean space.

Example 2.3.2 (Nonwandering Points Enter Boundary Near Infinity). This example shows that even if V_0 is Morse-Smale along $\partial W^s(X^s(V_0))$ and there exists an open neighborhood Nof $\overline{W}^s(X^s(V_0))$ in \mathbb{R}^2 such that $\Omega(V_0) \cap N = \Omega(V_0) \cap \overline{W}^s(X^s(V_0))$, there exists V arbitrarily close to V_0 in the strong C^1 topology such that $\partial W^s(X^s(V))$ contains nonwandering points in addition to the perturbations of the nonwandering points in $\partial W^s(X^s(V_0))$. In other words, new nonwandering points enter the boundary under strong C^1 perturbations. By Theorem 2.3.9, this is not possible for compact Riemannian manifolds with V_0 Morse-Smale along $\partial W^s(X^s(V_0))$. Figure 2.8 shows an example satisfying the assumptions above, and such that under C^1 perturbations to V_0 in a compact neighborhood of a hyperbolic saddle equilibrium point, a continuum of nonhyperbolic equilibrium points enter $\partial W^s(X^s(V))$.

Example 2.3.3 (Explosion of Closure of Region of Attraction). This example show that even if V_0 is Morse-Smale along $\partial W^s(X^s(V_0))$, there exists an open neighborhood N' of $\overline{W}^s(X^s(V_0))$ in \mathbb{R}^2 such that $\Omega(V_0) \cap N' = \Omega(V_0) \cap \overline{W}^s(X^s(V_0))$, and there exists an open neighborhood N of ∞ in M such that $\Omega(V_0) \cap N = \emptyset$ and every $x \in N$ satisfies $\omega(x) \cup \alpha(x) \neq \emptyset$, there still exists V arbitrarily close to V_0 in the strong C^1 topology such that such that $\partial W^s(X^s(V))$ contains nonwandering points in addition to the perturbations of the nonwandering points in $\partial W^s(X^s(V_0))$. Furthermore, $\overline{W}^s(X^s(V))$ explodes in the sense that it grows



Figure 2.7: Example 2.3.1. A stable hyperbolic equilibrium point (blue dot), saddle hyperbolic equilibrium points (red dots), sinks (black dots), nonhyperbolic equilibrium points (green dots), and a point at infinity (white dot) are shown. A sequence of periodic orbits (red for unstable and blue for stable) accumulates along the boundary of the region of attraction of the stable hyperbolic equilibrium point (blue dot). Black lines indicate stable and unstable manifolds of the saddle equilibrium points, with the arrows indicating direction of flow.



Figure 2.8: Example 2.3.2. A stable hyperbolic equilibrium point (blue dot), saddle hyperbolic equilibrium point (red dot), continuum of nonhyperbolic equilibrium points which are unstable in the transverse direction (red line), a stable sink (black dot), and a point at infinity (white dot) are shown. Black lines indicate stable and unstable manifolds of the saddle equilibrium point, with the arrows indicating direction of flow. The left figure shows the vector field before perturbation and the right figure shows the vector field after a perturbation in a compact neighborhood of its saddle equilibrium point (red dot). A segment of the continuum of nonhyperbolic equilibrium points (red line) enters the boundary of the region of attraction of the stable equilibrium point (blue dot).



Figure 2.9: Example 2.3.3. A stable hyperbolic equilibrium point (blue dot), saddle hyperbolic equilibrium point (red dot), source and sink (black dots), and a point at infinity (white dot) are shown. Black lines indicate stable and unstable manifolds of the saddle equilibrium point, with the arrows indicating direction of flow. The left figure shows the vector field before perturbation and the right figure shows the vector field after a perturbation in a compact neighborhood of its saddle equilibrium point (red dot). A source (black dot) enters the boundary of the stable equilibrium point (blue dot) arbitrarily far from the boundary before perturbation and from infinity.

discontinuously under arbitarily small perturbations from V_0 . Hence, not only are new nonwandering points entering the boundary under perturbation, which is not possible for the compact Riemannian manifold case by Theorem 2.3.9, but also a C^1 continuous family of vector fields $\{V_p\}_{p\in J}$ containing V_0 would have $\partial W^s(X^s(V_p))$ discontinuous in p, which is also not possible for the compact Riemannian manifold case by Corollary 2.3.10. The purpose of Theorem 2.3.20 and Corollary 2.3.21 is to show that this behavior is not possible under the assumptions above for vector fields which are generic in a certain sense. Figure 2.9 shows an example which satisfies the assumptions above, is globally a Morse-Smale vector field on \mathbb{R}^2 , and such that, for a C^1 perturbation to V_0 in a compact neighborhood of a hyperbolic saddle equilibrium point, a new hyperbolic equilibrium point enters $\partial W^s(X^s(V))$ arbitrarily far away from $\partial W^s(X^s(V_0))$

Theorem 5-4 of [CC95] claims (among other things) that if V_0 is a C^1 vector field on Euclidean space such that $\Omega(V_0) \cap \partial W^s(X^s(V_0))$ is a finite union of hyperbolic equilibrium points $\{X^i(V_0)\}_{i \in I}, \partial W^s(X^s(V_0)) = \bigcup_{i \in I} W^s(X^i(V_0))$, and if the intersections of their stable and unstable manifolds are transverse, then V sufficiently close to V_0 in the weak C^1 topology implies that $\partial W^s(X^s(V_0)) = \bigcup_{i \in I} W^s(X^i(V))$. Recently, the authors appear to have recognized that their theorem was incorrect [CA15] via a counterexample that uses a weak C^1 perturbation, and have attempted to informally fix their theorem. They claim in this more recent setting that if $\{V_p\}_{p\in(a,b)}$ is a weakly C^1 continuous family of vector fields such that for every $p \in (a, b), \Omega(V_p) \cap \partial W^s(X^s(V_p))$ consists of a finite union of hyperbolic equilibrium points, every orbit in $\partial W^s(X^s(V_p))$ converges in forwards time to one of these equilibrium points, the stable and unstable manifolds of these equilibrium points are transverse, and if we call $\{X^i(V_{p_0})\}_{i\in I}$ the equilibrium points in $\partial W^s(X^s(V_{p_0}))$ for some fixed $p_0 \in (a, b)$, then shrinking (a, b) if necessary implies that $\partial W^s(X^s(V_p)) \subset \bigcup_{i \in I} W^s(X^i(V_p))$ unless there is an appearance of a new hyperbolic equilibrium point "at infinity." They go on to say that under these assumptions "and some control at infinity," the boundary "does not suffer from drastic changes with small parameter variation." If we let V_{p_0} be the unperturbed vector field of Figure 2.9, and parametrize its perturbation in that figure in a natural way, then we obtain a strong C^1 continuous family of vector fields $\{V_p\}_{p \in (a,b)}$ satisfying the assumptions above (and more, since they are globally Morse-Smale), and such that there is an open neighborhood of infinity which contains no equilibrium points or nonwandering points for all $p \in (a, b)$ and on which V_p agrees with V_{p_0} for all $p \in (a, b)$. Yet, for $p \in (a, b)$ arbitrarily close to p_0 , a new equilibrium point enters the boundary, and the boundary undergoes a drastic expansion. Thus, this example contradicts both the original Theorem 5-4 [CC95] and the attempted revision by the original lead author [CA15]. Furthermore, it illustrates that the conventional paradigm for studying the impact of parameter variations on the boundary by considering only loss of hyperbolicity or transversality overlooks crucial global bifurcations.

2.3.4 Main Results

2.3.4.1 Compact Riemannian Manifolds

For this section, let M be a compact Riemannian manifold and let V_0 be a C^1 vector field on M possessing a stable hyperbolic equilibrium point $X^s(V_0)$. We make the following assumptions.

Assumption 2.3.4. $\Omega(V_0) \cap \partial W^s(X^s(V_0))$ consists of an arbitrary union of critical elements; call them $\bigcup_{i \in I} X^i(V_0)$ where I is some indexing set.

Assumption 2.3.5. For $i \in I$, $X^i(V_0)$ is hyperbolic.

Assumption 2.3.6. For every $i, j \in I$, $W^s(X^i(V_0))$ and $W^u(X^j(V_0))$ are transverse.

Remark 2.3.7. Assumptions 2.3.4-2.3.6 ensure that the vector field V_0 is Morse-Smale along $\partial W^s(X^s(V_0))$ (in fact, it is not assumed initially that the index set I is finite, but this will follow as a consequence of the other assumptions by Lemma 2.3.22). Assumptions 2.3.5-2.3.6 are generic in the C^1 topology. By [Pug67], $\Omega(V_0)$ is generically (in the C^1 topology) equal to the closure of the union of equilibrium points and periodic orbits of V_0 . Hence, Assumption 2.3.4 could be relaxed to this generic assumption along with the assumption that there exists an open neighborhood of $\partial W^s(X^s(V_0))$ in which the number of equilibrium points and periodic orbits is finite.

Lemma 2.3.8 is a slight modification of the results in [CHW88] to compact Riemannian manifolds under the assumptions above. It shows that $\overline{W}^s(X^s(V_0))$ admits a decomposition into a union of the stable manifolds of the critical elements it contains, that it is contained in the union of their unstable manifolds, and that these critical elements do not possess any homoclinic orbits or heteroclinic cycles.

Lemma 2.3.8. Let V_0 be a C^1 vector field on a compact Riemannian manifold M which possesses a stable hyperbolic equilibrium point $X^s(V_0)$ and satisfies Assumptions 2.3.4-2.3.6. Then $\overline{W}^s(X^s(V_0)) = \bigcup_{i \in I} W^s(X^i(V_0)) \bigcup W^s(X^s(V_0)) \subset \bigcup_{i \in I} W^u(X^i(V_0)) \bigcup X^s(V_0)$, I and $\partial W^s(X^s(V_0))$ are nonempty, $W^u(X^i(V_0)) \cap W^s(X^s(V_0)) \neq \emptyset$ for all $i \in I$, and there are no homoclinic orbits or heteroclinic cycles among the $\{X^i(V_0)\}_{i \in I}$.

Theorem 2.3.9 is our main technical result for compact Riemannian manifolds. It states that, under the assumptions above, for V a vector field sufficiently C^1 close to V_0 , $\overline{W}^s(X^s(V))$ is contained in a fixed neighborhood of $\overline{W}^s(X^s(V_0))$, the only nonwandering points under V in this neighborhood are the perturbations of the critical elements in $\overline{W}^s(X^s(V_0))$, $\overline{W}^s(X^s(V))$ permits a decomposition into the union of stable manifolds of the perturbations of those critical elements, and $\overline{W}^s(X^s(V))$ is contained in the union of unstable manifolds of the perturbations of those critical elements. In particular, under small C^1 perturbations, $\overline{W}^s(X^s(V))$ cannot explode, no nonwandering points appear in $\overline{W}^s(X^s(V))$ other than the perturbations of those critical elements, and the decomposition into stable manifolds persists under perturbation.

Theorem 2.3.9. Let V_0 be a C^1 vector field on a compact Riemannian manifold M which possesses a stable hyperbolic equilibrium point $X^s(V_0)$ and satisfies Assumptions 2.3.4-2.3.6. Then there exists an open neighborhood W of V_0 in the C^1 topology and an open neighborhood $U \text{ of } \overline{W}^{s}(X^{s}(V_{0})) \text{ in } M \text{ such that } V \in W \text{ implies that}$ $\overline{W}^{s}(X^{s}(V)) = \bigcup_{i \in I} W^{s}(X^{i}(V)) \bigcup W^{s}(X^{s}(V)) \subset U \subset \bigcup_{i \in I} W^{u}(X^{i}(V)) \bigcup X^{s}(V)$ and $\Omega(V) \cap U = \bigcup_{i \in I} X^{i}(V) \bigcup X^{s}(V).$

Using Theorem 2.3.9 and Corollary 2.2.25, it can then be shown that given a C^1 continuous family of vector fields on M which satisfy the assumptions above, the boundaries of their regions of attraction vary Hausdorff continuously with parameter. This is encapsulated in Corollary 2.3.10.

Corollary 2.3.10. Let J be a smooth manifold and let $\{V_p\}_{p\in J}$ be a C^1 continuous family of vector fields on M possessing a family of stable hyperbolic equilibria $\{X^s(p)\}_{p\in J}$. Suppose there exists $p_0 \in J$ such that V_{p_0} satisfies Assumptions 2.3.4-2.3.6. Then for J sufficiently small such that $p_0 \in J$, $\partial W^s(X^s(V_J)) = \sqcup_{p\in J} \partial W^s(X^s(V_p))$, and $\{\partial W^s(X^s(V_p))\}_{p\in J}$ is Hausdorff continuous.

2.3.4.2 Euclidean Space

For this section, let M be any Euclidean space and let V_0 be a complete C^1 vector field on M possessing a stable hyperbolic equilibrium point $X^s(V_0)$. We augment the assumptions above for compact Riemannian manifolds with the following.

Assumption 2.3.11. $\partial W^s(X^s(V_0)) \neq \emptyset$. In particular, I is nonempty.

Assumption 2.3.12. For every $x \in \partial W^s(X^s(V_0))$, its forward orbit is bounded.

Assumption 2.3.13. There exists a neighborhood N of infinity such that $\Omega(V_0) \cap N = \emptyset$ and every orbit under V_0 contained in N has at least one of its ω and α limit sets under V_0 nonempty.

Remark 2.3.14. It is possible that $W^s(X^s(V_0)) = M$, such as if V_0 is the vector field corresponding to a stable hyperbolic linear system. Assumption 2.3.11 rules out this possibility.

Remark 2.3.15. If any forward orbit in $\partial W^s(X^s(V_0))$ is unbounded then it cannot be contained in $W^s(X^i(V_0))$ for any $i \in I$. Thus, $\partial W^s(X^s(V_0))$ would not admit the desired decomposition into a union of stable manifolds of critical elements. Assumption 2.3.12 rules out this possibility.

Remark 2.3.16. The first part of Assumption 2.3.13, regarding a neighborhood of infinity with no nonwandering points, is required so that Examples 2.3.1-2.3.2 cannot occur, in

which there is no neighborhood of $\overline{W}^{s}(X^{s}(V_{0}))$ in $M \cup \{\infty\}$ which does not contain additional nonwandering points other than the $X^{i}(V_{0})$ for $i \in I$. The second part of this assumption, that there is a neighborhood of infinity in which every orbit it contains has at least one of its ω and α limit sets nonempty, will be required to show that for V a vector field sufficiently close to V_{0} , every point in $\overline{W}^{s}(X^{s}(V))$ has a nonempty ω limit set under V.

Remark 2.3.17. Assumptions 2.3.5-2.3.6 are still generic in the strong C^1 topology. Assumption 2.3.4 follows from the assumptions that there exists an open neighborhood of $\partial W^s(X^s(V_0))$ which contains only finitely many equilibrium points and periodic orbits, and the strong C^1 generic assumption that, under Assumption 2.3.13, $\Omega(V_0)$ is contained in the closure of the union of the equilibrium points and periodic orbits of V_0 .

Lemma 2.3.18 shows that under the assumptions above, $\partial W^s(X^s(V_0))$ admits a decomposition into the union of stable manifolds of the critical elements it contains, those critical elements do not possess any homoclinic orbits or heteroclinic cycles, and that there exists an open neighborhood of $\overline{W}^s(X^s(V_0))$ containing no additional nonwandering points other than the critical elements in $\overline{W}^s(X^s(V_0))$ and such that the backward orbit of every point in this neighborhood converges to either one of those critical elements or to infinity.

Lemma 2.3.18. Let V_0 be a complete C^1 vector field on some Euclidean space M which possesses a stable hyperbolic equilibrium point $X^s(V_0)$ and satisfies Assumptions 2.3.4-2.3.12. Then $\partial W^s(X^s(V_0)) = \bigcup_{i \in I} W^s(X^i(V_0))$ and there are no homoclinic orbits or heteroclinic cycles among the $\{X^i(V_0)\}_{i \in I}$. If in addition $\overline{W}^s(X^s(V_0))$ is bounded or V_0 satisfies Assumption 2.3.13, then there exists an open neighborhood U of $\overline{W}^s(X^s(V_0))$ in $M \cup \{\infty\}$ such that $\Omega(V_0) \cap U = \bigcup_{i \in I} X^i(V_0) \cup X^s(V_0)$ and for every $x \in U$ its orbit in backwards time either enters $W^u_{loc}(X^j(V_0))$ for some j or the neighborhood of infinity of Assumption 2.3.13.

In case $\overline{W}^s(X^s(V_0))$ is bounded, by an analogous proof to the case of compact Riemmanian manifolds in Theorem 2.3.9, the results of that Theorem will hold for Euclidean space as well. In particular, Theorem 2.3.19 shows that, under the assumptions above - excluding Assumption 2.3.13 - for V a vector field sufficiently close to V_0 , $\overline{W}^s(X^s(V))$ is contained in a fixed neighborhood of $\overline{W}^s(X^s(V_0))$, the only nonwandering points under V in this neighborhood are the perturbations of the critical elements in $\overline{W}^s(X^s(V_0))$, and $\overline{W}^s(X^s(V))$ permits a decomposition into the union of stable manifolds of the perturbations of those critical elements. So, under small C^1 perturbations, $\overline{W}^s(X^s(V))$ cannot explode in size, no nonwandering points appear in $\overline{W}^s(X^s(V))$ other than the perturbations of those critical elements, and the decomposition into stable manifolds persists under perturbation. **Theorem 2.3.19.** Let V_0 be a C^1 vector field on some Euclidean space M which possesses a stable hyperbolic equilibrium point $X^s(V_0)$, satisfies Assumptions 2.3.4-2.3.12, and such that $\overline{W}^s(X^s(V_0))$ is bounded. Then there exists an open neighborhood W of V_0 in the strong C^1 topology and an open neighborhood U of $\overline{W}^s(X^s(V_0))$ in M such that U is bounded and $V \in W$ implies that $\overline{W}^s(X^s(V)) = \bigcup_{i \in I} W^s(X^i(V)) \bigcup W^s(X^s(V)) \subset U$ and $\Omega(V) \cap U = \bigcup_{i \in I} X^i(V) \bigcup X^s(V)$.

The most interesting cases for Euclidean space occur when $\overline{W}^s(X^s(V_0))$ is unbounded. Example 2.3.3 showed that for M Euclidean and V_0 satisfying all of the assumptions above, there may exist V arbitrarily C^1 close to V_0 such that $\overline{W}^s(X^s(V))$ contains nonwandering point in addition to the perturbations of the critical elements in $\overline{W}^s(X^s(V_0))$ and that $\overline{W}^s(X^s(V))$ may escape any open neighborhood of $\overline{W}^s(X^s(V_0))$. In other words, under strong C^1 perturbations new nonwandering points can enter the closure of the region of attraction, and the closure of the region of attraction can grow discontinuously. Theorem 2.3.20 shows that for a strong C^1 continuous family $\{V_p\}_{p\in J}$ of vector fields on M, if V_{p_0} satisfies the assumptions above as well as some additional strong C^1 generic assumptions, then J sufficiently small implies that for $p \in J$, $\overline{W}^s(X^s(V_p))$ is equal to the union of the stable manifolds of the perturbations of the critical elements in $\overline{W}^s(X^s(V_{p_0}))$ and no additional nonwandering points are in $\overline{W}^s(X^s(V_p))$ other than the perturbations of those critical elements.

Theorem 2.3.20. Let \hat{V} be a C^1 vector field on some Euclidean space M which possesses a stable hyperbolic equilibrium point $X^s(\hat{V})$. Let W' be the open neighborhood of \hat{V} in the strong C^1 topology such that $V \in W'$ implies that $X^s(V)$ exists and is a stable hyperbolic equilibrium point. Then for $V_{p_0} \in W'$ complete satisfying Assumptions 2.3.4-2.3.13, and satisfying some additional generic assumptions, J a sufficiently small open neighborhood of p_0 implies that for every $p \in J$, $\overline{W}^s(X^s(V_p)) = \bigcup_{i \in I} W^s(X^i(V_p)) \cup W^s(X^s(V_p))$ and $\Omega(V_p) \cap \overline{W}^s(X^s(V_p)) = \bigcup_{i \in I} X^i(V_p) \cup X^s(V_p)$.

Similarly to the case for compact Riemannian manifolds above, using Theorem 2.3.20 and Corollary 2.2.24, it can then be shown that given a strong C^1 continuous family of vector fields on M which satisfy the assumptions above, the boundaries of their regions of attraction vary Chabauty continuously with parameter. This is the content of Corollary 2.3.21.

Corollary 2.3.21. Let J be a smooth manifold and let $\{V_p\}_{p\in J}$ be a strong C^1 continuous family of vector fields on M possessing a family of stable hyperbolic equilibria $\{X^s(p)\}_{p\in J}$. Suppose there exists $p_0 \in J$ such that V_{p_0} is complete, $\partial W^s(X^s(V_{p_0})) \neq \emptyset$, V_{p_0} satisfies Assumptions 2.3.4-2.3.12, and V_{p_0} satisfies some additional generic assumptions. Then for J sufficiently small such that $p_0 \in J$, $\partial W^s(X^s(V_J)) = \sqcup_{p \in J} \partial W^s(X^s(V_p))$, and $\{\partial W^s(X^s(V_p))\}_{p \in J}$ is Chabauty continuous.

2.3.5 Proofs

Let $I^+ := I \cup \{s\}$ so that $\bigcup_{i \in I^+} X^i(V_0) = \bigcup_{i \in I} X^i(V_0) \cup X^s(V_0)$.

2.3.5.1 Compact Riemannian Manifolds

The purpose of this section is to prove Theorem 2.3.9 and Corollary 2.3.10. Several technical lemmas will be required.

Lemma 2.3.22 shows that Assumptions 2.3.4-2.3.6 imply that the intersection of the nonwandering set with the RoA boundary consists of a finite union of critical elements rather than a possibly infinite union. This is important because any finite collection of hyperbolic critical elements will persist under sufficiently small perturbations to the vector field, whereas for an infinite number of hyperbolic critical elements there may be no perturbation sufficiently small such that all of the critical elements persist.

Lemma 2.3.22. Under the conditions of Lemma 2.3.8 the number of critical elements of V_0 contained in $\partial W^s(X^s(V_0))$ is finite.

Proof of Lemma 2.3.22. Let $K = \partial W^s(X^s(V_0))$ for notational convenience. As K is closed in M compact, K is compact. By Assumption 2.3.5, all equilibrium points and periodic orbits of V_0 in K are hyperbolic. Since K is compact, this implies that there are at most countably many equilibrium points and periodic orbits of V_0 in K. So, assume towards a contradiction that there are infinitely many equilibrium points and periodic orbits $\{X^m\}_{m=1}^{\infty}$ in K. The proof proceeds by showing that $\{X^m\}_{m=1}^{\infty}$ converges along a subsequence to a limit X. It then shows that X is nonwandering and invariant under the flow, which implies it is equal to a union of critical elements since it is in K, and that X is compact, connected, and Hausdorff, so it is a continuum. A theorem of Sierpinski then gives that X must be equal to exactly one critical element, which then contradicts that $X^m \to X$ because hyperbolic critical elements are isolated.

As K is compact, \mathcal{K} is compact. Since each X^m is closed and contained in $K, X^m \in \mathcal{K}$ for all m. As \mathcal{K} is compact, passing to a subsequence if necessary implies that $X^m \to X \in \mathcal{K}$. So, $X \subset K$ is closed and nonempty. As K is compact, X is compact. As M is Hausdorff

and X is a subspace of M, X is Hausdorff. Let $x \in X$. We claim that x is nonwandering under V_0 . For, let U be any open neighborhood of x and let T > 0. As $X^m \to X$ and $x \in X$, there exist $x_m \in X^m$ such that $x_m \to x$. As $x_m \to x \in U$ and U is open, choosing M sufficiently large implies that $x_M \in U$. If x_M is an equilibrium point, then fix t > T. If x_M is contained in a periodic orbit then let t be a multiple of its period such that t > T. Then $\phi_t(U) \cap U \supset \{\phi_t(x_M)\} \cap \{x_M\} = \{x_M\} \neq \emptyset$. As x was arbitrary, this implies that X is nonwandering under V_0 . Since X is contained in K, by Assumption 2.3.4 this implies that $X \subset \bigcup_{m=1}^{\infty} X^m$. Let $x \in X$ and $t \in \mathbb{R}$. Then since $X^m \to X$ and $x \in X$, there exist $x_m \in X^m$ such that $x_m \to x$. As X^m is invariant under V_0 and $x_m \in X^m$ for each $m, \phi_t(x_m) \in X^m$ for each m. By continuity of the flow, $x_m \to x$ implies that $\phi_t(x_m) \to \phi_t(x)$. Then $\phi_t(x_m) \in X^m$ for all $m, X^m \to X$, and $\phi_t(x_m) \to \phi_t(x)$ implies that $\phi_t(x) \in X$. As x and t were arbitrary, this implies that X is invariant under V_0 . So, there must exist $E \subset \{1, 2, 3, ...\}$ such that $X = \bigcup_{m \in E} X^m$. We claim that X is connected. For, suppose not. Then there exist A and B nonempty, disjoint subsets of X which are open in X and such that $X = A \cup B$. Hence, A and B are both closed in X. As X is compact, A and B are both compact. As A and B are compact and disjoint in M Hausdorff, there exist U_A and U_B disjoint, open sets in M such that $A \subset U_A$, and $B \subset U_B$. Note that U_A and U_B disjoint and open implies that U_A and U_B are disjoint. Hence, ∂U_A is disjoint from U_A and U_B . In particular, this implies that ∂U_A is disjoint from A and B, so ∂U_A is disjoint from $X = A \cup B$. Let $y \in A$ and $z \in B$. As $y, z \in X$ and $X^m \to X$, there must exist $y_m, z_m \in X^m$ such that $y_m \to y$ and $z_m \to z$. Hence, since $y_m \to y \in U_A$ open and $z_m \to z \in U_B$ open, for m sufficiently large $y_m \in U_A$ and $z_m \in U_B \subset M - U_A$. As y_m and z_m lie in X^m connected, there must exist $x_m \in X^m$ such that $x_m \in \partial U_A$. As ∂U_A is compact, passing to a subsequence if necessary implies that $x_m \to x \in \partial U_A$. As $X^m \to X$ and $x_m \to x$, we also have that $x \in X$. So, $x \in \partial U_A \cap X$. But, this contradicts that ∂U_A is disjoint from X. So, X is connected. As X is connected, compact, and Hausdorff, it is a continuum. Since X is a continuum and $X = \bigcup_{m \in E} X^m$ is a countable union where $\{X^m\}_{m \in E}$ are pairwise disjoint and closed in X, by a theorem of Sierpinski [Eng89, Theorem 6.1.28] E contains at most one element. As X is nonempty, E contains exactly one element. So, there exists m^* such that $X = X^{m^*}$. Since X is a hyperbolic equilibrium point or periodic orbit, by the proofs of [JPdM82], Lemmas 2.1-2.2] there exists an open neighborhood \hat{N} of X such that \hat{N} does not entirely contain any equilibrium points or periodic orbits of V_0 other than X. As $X^m \to X$, m sufficiently large implies that $X^m \subset \hat{N}$. As X^m is either an equilibrium point or period orbit for all m, this contradicts the choice of \hat{N} . By Lemma 2.3.22, the number of critical elements in the RoA boundary is finite. Hence, the set indexing these critical elements, I, is also finite. Therefore, in the remainder of this section we will use that I is finite without explicitly referencing Lemma 2.3.22.

The following lemma is a technical result that states that the presence of a transverse heteroclinic cycle of hyperbolic critical elements implies that the interesections of their stable and unstable manifolds are nonwandering. This lemma appears in [Sma67, Lemma 7.2] for periodic points of diffeomorphisms. The adoption of the proof to critical elements of vector fields is straightforward and is presented below.

Lemma 2.3.23. Let V_0 be a C^1 vector field on a compact Riemannian manifold M. Let $(X^1, ..., X^m)$ be a transverse heteroclinic cycle of hyperbolic critical elements of V_0 and let $(x_1, ..., x_m)$ be such that $x_m = x_1$ and $x_i \in \{W^u(X^i) - X^i\} \cap \{W^s(X^{i+1}) - X^{i+1}\}$. for $i \in \{1, ..., m-1\}$. Then each x_i is nonwandering.

Proof of Lemma 2.3.23. The idea is to fix one x_k and argue directly that x_k is nonwandering by taking any open neighborhood U of x_k , any T > 0, and showing that some forward iterate of $\phi_T(U)$ must intersect U. To do so, first it is shown that $\bigcup_{t\geq T} \phi_t(U) \cap W^s(x_k) \neq \emptyset$ by repeated application of Lemma 2.2.34 using the fact that $(X^1, ..., X^m)$ is a transverse heteroclinic cycle. Then the Inclination Lemma [Pal69] is used to show that this implies $\bigcup_{t>T} \phi_t(U) \cap U \neq \emptyset$.

First we show that $\bigcup_{t\geq T} \phi_t(U) \cap W^s(x_k) \neq \emptyset$. Let $x_k \in (x_1, ..., x_m)$, let U be an open neighborhood of x_k , and let T > 0. As $x_k \in \{W^u(X^k) - X^k\} \cap \{W^s(X^{k+1}) - X^{k+1}\}$, by invariance of the stable and unstable manifolds, $x'_k = \phi_T(x_k) \in \phi_T(U) \cap \{W^u(X^k) - X^k\} \cap \{W^s(X^{k+1}) - X^{k+1}\}$. Note that $\bigcup_{t\geq T} \phi_t(U)$ is open and forwards invariant.

Lemma 2.2.34 states that, for X^i and X^j hyperbolic critical elements and X^s a hyperbolic stable equilibrium point, if $W^u(X^i) \cap W^s(X^s) \neq \emptyset$ and $W^s(X^i) \cap W^u(X^j) \neq \emptyset$ then $W^u(X^j) \cap W^s(X^s) \neq \emptyset$. In the proof of Lemma 2.2.34 the only properties of $W^s(X^s)$ that were used were that it was open and backwards invariant. Reversing time in that proof and replacing $W^s(X^s)$ with $\bigcup_{t\geq T} \phi_t(U)$ shows that if $W^s(X^i) \cap \bigcup_{t\geq T} \phi_t(U) \neq \emptyset$ and $W^u(X^i) \cap W^s(X^j) \neq \emptyset$ then $W^s(X^j) \cap \bigcup_{t\geq T} \phi_t(U) \neq \emptyset$. Letting $X^j = X^{i+1}$ and proceeding inductively implies that $W^s(X^k) \cap \bigcup_{t\geq T} \phi_t(U) \neq \emptyset$.

Next we show that $\bigcup_{t\geq T} \phi_t(U) \cap U \neq \emptyset$. If X^k is an equilibrium point let $B = W^u_{\text{loc}}(X^k)$ and if X^k is a periodic orbit let S be a cross section for a C^1 first return map and let $B = S \cap W^u_{\text{loc}}(X^k)$. Since $x_k \in W^u(X^k)$, there exist $y \in W^u_{\text{loc}}(X^k)$ and T' > 0 such that $x_k = \phi_{T'}(y)$. By continuity of $\phi_{T'}$ and since U is open, there exists $\epsilon > 0$ such that d(y, z) < 0 ϵ implies $\phi_{T'}(z) \in U$. By the Inclination Lemma, since $W^s(X^k) \cap \bigcup_{t \geq T} \phi_t(U) \neq \emptyset$ and $\bigcup_{t \geq T} \phi_t(U)$ is open and forward invariant, there exists a C^1 disk $D \subset \bigcup_{t \geq T} \phi_t(U)$ such that D is ϵ C^1 -close to B. Then $\phi_{T'}(D) \cap U \neq \emptyset$. By forward invariance, this implies that $\bigcup_{t \geq T} \phi_t(U) \cap U \neq \emptyset$ so x_k is nonwandering. \Box

Lemma 2.3.23 is used in the proof of Lemma 2.3.8 to show that the presence of transverse homoclinic orbits or heteroclinic cycles for the critical elements in $\partial W^s(X^s(V_0))$ would generate additional nonwandering points in $\partial W^s(X^s(V_0))$ and, therefore, would contradict Assumption 2.3.4. Hence, no homoclinic orbits or heteroclinic cycles will exist in $\partial W^s(X^s(V_0))$. This will be required in the proof of Lemma 2.3.25.

Proof of Lemma 2.3.8. That $\partial W^s(X^s(V_0))$ and I are nonempty is a straightforward consequence of α limit sets nonempty since M is compact. Similarly, $\overline{W}^s(X^s(V_0))$ contained in both $\bigcup_{i \in I^+} W^u(X^i(V_0))$ and $\bigcup_{i \in I^+} W^s(X^i(V_0))$ follows from similar reasoning. To show that $\bigcup_{i \in I^+} W^s(X^i(V_0)) \subset \overline{W}^s(X^s(V_0))$, a transverse heteroclinic sequence is constructed inductively using $\bigcup_{i \in I} X^i(V_0)$. Using Lemma 2.3.23 it is shown that this sequence cannot contain any cycles or homoclinic orbits so, as a consequence of I finite, it must terminite with a finite number of elements. By the construction of the sequence, and using Lemma 2.2.34, this will imply that the unstable manifold of each element of the sequence intersects $W^s(X^s(V_0))$. By the proof of Theorem 2.2.12, this is used to conclude that $W^s(X^i(V_0)) \subset \partial W^s(X^s(V_0))$ for all $i \in I$.

First we show that $\partial W^s(X^s(V_0))$ and I are nonempty since M compact implies that α limit sets are nonempty. As ϕ_t restricted to $W^s_{\text{loc}}(V_0)$ is a contraction for any t > 0, $\Omega(V_0) \cap W^s_{\text{loc}}(X^s(V_0)) = X^s(V_0)$. By invariance of the nonwandering set, and since $W^s(X^s(V_0)) = \bigcup_{t \leq 0} \phi_t(W^s_{\text{loc}}(X^s(V_0))), \Omega(V_0) \cap W^s(X^s(V_0)) = X^s(V_0)$. Hence,

 $\Omega(V_0) \cap \overline{W}^s(X^s(V_0)) = \bigcup_{i \in I^+} X^i(V_0)$. There exists $x \in W^s_{\text{loc}}(X^s(V_0))$ with $x \neq X^s(V_0)$. As $\alpha(x)$ is nonempty, nonwandering, and contained in $\overline{W}^s(X^s(V_0))$, and since $\alpha(x) \neq X^s(V_0)$ because ϕ_t restricted to $W^s_{\text{loc}}(V_0)$ is a contraction for any t > 0, $\alpha(x) \subset \partial W^s(X^s(V_0)) \cap \Omega(V_0) = \bigcup_{i \in I} X^i(V_0)$. This implies that both $\partial W^s(X^s(V_0))$ and I are nonempty.

Next we show that $\overline{W}^s(X^s(V_0))$ is contained in both $\bigcup_{i\in I^+} W^u(X^i(V_0))$ and $\bigcup_{i\in I^+} W^s(X^i(V_0))$, again using the fact that the α and ω limit sets are nonempty. Let $x \in \overline{W}^s(X^s(V_0))$. Then $\omega(x)$ and $\alpha(x)$ are closed, nonempty, and nonwandering. Hence, since $\overline{W}^s(X^s(V_0))$ is closed and invariant, $\omega(x), \alpha(x) \subset \Omega(V_0) \cap \overline{W}^s(X^s(V_0)) = \bigcup_{i\in I^+} X^i(V_0)$. As $\bigcup_{i\in I^+} X^i(V_0)$ is a finite union of hyperbolic critical elements, they are separable by open sets. Since $\omega(x), \alpha(x)$ are compact they are connected so, since they are also nonempty, $\omega(x) = X^{j}(V_{0})$ and $\alpha(x) = X^{k}(V_{0})$ for some $j, k \in I^{+}$. Furthermore, since $\omega(x) = X^{j}(V_{0})$ and $\alpha(x) = X^{k}(V_{0})$ are compact, the forward orbit of x must converge to $X^{j}(V_{0})$ and the backward orbit of x must converge to $X^{k}(V_{0})$. For, if V is any precompact open neighborhood of $X^{j}(V_{0})$ and the forward orbit of x does not converge to $X^{j}(V_{0})$ then it must intersect ∂V at arbitrarily large times, say $\{t_{n}\}$ with $t_{n} \to \infty$. As $\bigcup_{n} \phi_{t_{n}}(x) \subset \partial V$ compact, passing to a subsequence if necessary we have that $y = \lim_{n\to\infty} \phi_{t_{n}}(x) \in \omega(x) \cap \partial V$, which contradicts that $\omega(x) = X^{j}(V_{0})$. The backwards orbit case is proved analogously. Hence, $x \in W^{s}(X^{j}(V_{0})) \cap W^{u}(X^{k}(V_{0}))$ for some $j, k \in I^{+}$. Thus, $\overline{W}^{s}(X^{s}(V_{0})) \subset \bigcup_{i \in I^{+}} W^{u}(X^{i}(V_{0}))$.

Now we show that $\bigcup_{i \in I^+} W^s(X^i(V_0)) \subset \overline{W}^s(X^s(V_0))$. In particular, it suffices to show that $W^s(X^i(V_0)) \subset \partial W^s(X^s(V_0))$ for every $i \in I$. Let $i_0 \in I$. First we construct a sequence inductively. Given a sequence $(i_0, i_1, ..., i_k)$, since $X^{i_k}(V_0) \subset \partial W^s(X^s(V_0))$ is a hyperbolic critical element, by Lemma 2.2.29, either $W^u(X^{i_k}(V_0)) \cap W^s(X^s(V_0)) \neq \emptyset$, in which case we end the sequence, or $\{W^u(X^{i_k}(V_0)) - X^{i_k}(V_0)\} \cap \partial W^s(X^s(V_0)) \neq \emptyset$. In the latter case, since $\partial W^s(X^s(V_0)) \subset \bigcup_{i \in I} W^s(X^i(V_0))$, there exists i_{k+1} such that $(W^u(X^{i_k}(V_0)) - \bigcup_{i \in I} W^s(X^i(V_0)))$ $X^{i_k}(V_0) \cap (W^s(X^{i_{k+1}}(V_0)) - X^{i_{k+1}}(V_0)) \neq \emptyset$, thus yielding the new sequence $(i_0, i_1, ..., i_k, i_{k+1})$. We iterate this process to construct the sequence. Suppose that the sequence $(i_0, i_1, ...)$ never terminates. Since I is finite, this implies that there must exist a cycle $(i_l, i_{l+1}, ..., i_{l+m}) \subset$ $(i_0, i_1, ...)$ with $i_{l+m} = i_l$. By the construction of the sequence, this implies there there exist $(x_l, ..., x_{l+m})$ with $x_l = x_{l+m}$ such that $x_{l+j} \in (W^u(X^{i_{l+j}}(V_0)) - X^{i_{l+j}}(V_0)) \cap (W^s(X^{i_{l+i+j}}(V_0)) - X^{i_{l+j}}(V_0))$ $X^{i_{l+i+j}}(V_0)) \cap \partial W^s(X^s(V_0))$ for $j \in \{0, 1, ..., m-1\}$. Thus, by Lemma 2.3.23 each x_{l+j} for $j \in \{0, 1, ..., m-1\}$ is nonwandering. By construction, each such x_{l+j} is contained in $\partial W^s(X^s(V_0))$ and not contained in $\bigcup_{i \in I} X^i(V_0)$. However, this contradicts that $\Omega(V_0) \cap \partial W^s(X^s(V_0)) = \bigcup_{i \in I} X^i(V_0)$. So, the sequence $(i_0, i_1, ...)$ must terminate in a finite number of steps. Hence, suppose the sequence $(i_0, i_1, ...)$ terminates in a finite number of steps, say k. Then $W^u(X^{i_k}(V_0)) \cap W^s(X^s(V_0)) \neq \emptyset$. By repeated application of Lemma 2.2.34, this implies that $W^u(X^{i_l}(V_0)) \cap W^s(X^s(V_0)) \neq \emptyset$ for each $i_l \in (i_0, ..., i_k)$. In particular, $W^u(X^{i_0}(V_0)) \cap W^s(X^s(V_0)) \neq \emptyset$. By the proof of Theorem 2.2.12, this implies that $W^s(X^{i_0}(V_0)) \subset \partial W^s(X^s(V_0))$. Thus, $\bigcup_{i \in I} W^s(X^i(V_0)) \subset \partial W^s(X^s(V_0))$ so, combining with the above we have that $\overline{W}^s(X^s(V_0)) = \bigcup_{i \in I^+} W^s(X^i(V_0)) \subset \bigcup_{i \in I^+} W^u(X^i(V_0))$. As $\Omega(\overline{W}^s(X^s(V_0))) = \bigcup_{i \in I^+} X^i(V_0)$, there cannot be any homoclinic orbits or heteroclinic cycles among the $\{X_i(V_0)\}_{i \in I^+}$ since transverse homoclinic points are nonwandering (see for example the proof of Lemma 2.2.32) and, by the argument above regarding heteroclinic cycles, both would introduce additional nonwandering points into $\overline{W}^{s}(X^{s}(V_{0}))$.



Figure 2.10: Illustration of the key ideas behind the proof of Lemma 2.3.24. The light blue rectangle Y represents the choice of local coordinates, the orange circles compose the neighborhood N_i , the yellow curves represent the disk family centered along $W^s_{\text{loc}}(x_0)$ that is constructed in the proof, the purple neighborhood U' is contained in this disk family, the turquoise neighborhood $U'_i \subset U'$ is contained in the disk family corresponding to perturbations to the vector field, and the gray neighborhood $\hat{U} \subset U'_i$ is such that under perturbations to the vector field every backwards orbit originating in \hat{U} remains in U'_i until it passes through N_i .

Lemma 2.3.24 is a technical construction that is required in the proof of Lemma 2.3.25, which itself is the main result used to prove Theorem 2.3.9. The Lemma states that for any critical element with stable manifold having dimension at least one, for any fundamental neighborhood of that stable manifold there exists an open neighborhood of that critical element such that for any sufficiently small C^1 perturbation of the vector field, the backward orbit of every point in that open neighborhood either converges to the critical element or enters the fundamental neighborhood in finite negative time. A similar result was used in the proof of [Pal69, Theorem 1.9], but a complete proof was not provided.

Lemma 2.3.24. Let $i \in I$ such that the dimension of $W^s(X^i(V_0))$ is at least one. If $X^i(V_0)$ is an equilibrium point let $D_i(V)$ be the closure of $\{W^s_{loc}(X^i(V)) - \phi_1(W^s_{loc}(X^i(V)))\}$ in Mand let N_i be an open neighborhood of $D_i(V_0)$ in M. If $X^i(V_0)$ is a periodic orbit let S be a smooth cross section with $\hat{\tau} : S \to S$ the C^1 first return map, let $D_i(V)$ be the closure of $\{(W^s_{loc}(X^i(V)) \cap S) - \hat{\tau}(W^s_{loc}(X^i(V)) \cap S)\}$ in S, and let N_i be an open neighborhood of $D_i(V_0)$ in M. Then there exist an open neighborhood U_i of $W^s_{loc}(X^i(V_0))$ in M and an open



Figure 2.11: Example of the construction of the directed graphs in the proof of Lemma 2.3.25. A vector field on the sphere S^2 (left) and the corresponding directed graph (right) for the closure of the region of attraction of its stable hyperbolic equilibrium point (blue dot) are shown. A saddle hyperbolic equilibrium point (red dot), along with a sink and a source (black dots) are also shown. Note that equilibrium point four does not appear in the directed graph since it is not contained in the closure of the region of attraction.

neighborhood W_i of V_0 in the C^1 topology such that $V \in W_i$ implies that $W^s_{loc}(X^i(V)) \subset U_i$, $D_i(V) \subset N_i$, and for every $x \in U_i - W^u(X^i(V))$, the backwards orbit of x under V intersects N_i in finite negative time. Furthermore, there exists $\hat{U} \subset U_i$ open such that $V \in W$ implies $X^i(V) \subset \hat{U}$ and backwards orbit of any $x \in \hat{U}$ under V is contained in U_i at least until it enters N_i .

Proof of Lemma 2.3.24. The result nearly follows from Lemma 2.2.37, but a slight extension is required. The details are presented in Appendix C. \Box

Lemma 2.3.25 is the main technical result behind Theorem 2.3.9. It states that there exists an open neighborhood such that for any vector field sufficiently C^1 close to V_0 , this open neighborhood contains the union of the stable manifolds of the perturbations of all the critical elements in I^+ , and this open neighborhood is contained in the union of their unstable manifolds.

Lemma 2.3.25. Under the conditions of Lemma 2.3.8 there exists an open neighborhood U of $\overline{W}^s(X^s(V_0))$ in M and an open neighborhood W of V_0 in the C^1 topology such that $V \in W$ implies that $\bigcup_{i \in I^+} W^s(X^i(V)) \subset U \subset \bigcup_{i \in I^+} W^u(X^i(V))$.

Proof of Lemma 2.3.25. The first step in the proof is the construction of a graph whose vertices are the critical elements in I^+ and whose directed edges indicate the intersection of the stable and unstable manifolds of the vertices they connect. Figure 2.11 provides an example directed graph for a vector field on the sphere \mathbb{S}^2 . The goal is to construct an open neighborhood of the critical elements such that under the flow of any vector field sufficiently C^1 close to V_0 , the backwards orbits of every point in the open neighborhood eventually enter the local unstable manifold of some critical element. This is achieved by constructing the open neighborhood inductively by traversing the directed graph starting from the critical elements which have zero dimensional stable manifolds: call them sources. When each critical element is reached in the induction, an open neighborhood U_i is constructed containing that critical element and an open fundamental neighborhood of its stable manifold is constructed such that under the flow of any vector field V C^1 close to V_0 , the backward orbit of any point in U_i will enter either $W^u_{\text{loc}}(X^i(V))$ or the fundamental neighborhood, and the backward orbit of any point in the fundamental neighborhood will enter U_j for some element j which is strictly closer to the sources in the directed graph than i is. Lemma 2.3.24 is used to construct the desired U_i . This will then imply that the union of these neighborhoods is contained in the union of the unstable manifolds of the critical elements. To construct a neighborhood which contains their stable manifolds as well, it is necessary to construct $U'_i \subset U_i$ and $N'_i \subset N_i$ contained in the fundamental neighborhood such that the backward orbit of any point in U'_i does not escape U_i at least until it enters N'_i and the backward orbit of any point in N'_i does not escape N_i at least until it enters U_j for some $j \neq i$. As the local stable manifolds are contained in the union of the U_i by construction, the above construction will imply that the backwards orbits of any point in the fundamental domain of the stable manifold of $X^{i}(V)$ does not escape the union of the neighborhoods above. Consequently, the entire stable manifold is contained in this neighborhood.

First the directed graph of the critical elements is constructed. By Lemma 2.3.8, $\overline{W}^{s}(X^{s}(V_{0})) = \bigcup_{i \in I^{+}} W^{s}(X^{i}(V_{0})) \subset \bigcup_{i \in I^{+}} W^{u}(X^{i}(V_{0}))$. Hence, we can draw a directed graph with the elements of I^{+} as vertices such that there is an edge from i to j if and only if $W^{s}(X^{i}(V_{0})) \cap W^{u}(X^{j}(V_{0})) \neq \emptyset$ for any $i, j \in I^{+}$. By Lemma 2.3.8, I and $\partial W^{s}(X^{s}(V_{0}))$ are nonempty, so this graph contains at least two vertices and at least one edge. By Lemma 2.3.8 there are no homoclinic orbits, so no self edges in the graph, and no heteroclinic cycles, so no cycles in the graph. Thus, since I^{+} is finite, every directed path, hereafter referred to simply as paths, in the graph must be finite. We define a forward maximal path to be a path whose final vertex has no outgoing edges. Similarly, a backward maximal path is defined to be a path whose first vertex has no incoming edges. A maximal path is a path which is both forward and backward maximal. Let $E_0 \subset I^+$ be the collection of vertices such that for $i \in E_0$, the dimension of $W^u(X^i(V_0))$ is equal to the dimension of M. We claim that every backward maximal path starts with s and every forward maximal path ends with some element of E_0 . For, suppose $(i_1, ..., i_k)$ is backward maximal and assume towards a contradiction that $i_1 \neq s$. Then, by Lemma 2.3.8, $W^s(X^s(V_0)) \cap W^u(X^{i_1}(V_0)) \neq \emptyset$. Hence, $(s, i_1, ..., i_k)$ is a path, which contradicts that $(i_1, ..., i_k)$ was backward maximal. Now suppose $(i_1, ..., i_k)$ is forwards maximal and assume towards a contradiction that $i_k \notin E_0$. This implies that the dimension of $W^u(X^{i_k}(V_0))$ is less than the dimension of M. Since $X^{i_k}(V_0)$ is hyperbolic, $W^s(X^{i_k}(V_0))$ must have dimension at least one, so there exists $x \in \{W^s(X^{i_k}(V_0)) - X^{i_k}(V_0)\}$. Then $x \subset \partial W^s(X^s(V_0)) \subset \bigcup_{i \in I} W^u(X^i(V_0))$ so $x \in W^u(X^j(V_0))$ for some $j \in I$. Hence, $(i_1, ..., i_k, j)$ is a path, which contradicts that $(i_1, ..., i_k)$ is forwards maximal.

Next some notation is introduced which will be useful below. Let b(i, j) be the behavior of i to j [Pal69], which is defined to be the path of maximum length from i to j, where we define b(i, j) = 0 if there is no path from i to j. For example, in the directed graph of Figure 2.11, b(1, 2) = 1, b(2, 3) = 1, and b(1, 3) = 2. For any $E \subset I^+$, define b(i, E)to be the maximum of b(i, j) over $j \in E$. Let $s_k = \{i \in I^+ : b(i, E_0) = k\}$ and let $E_k = \bigcup_{l=0}^k s_l$. For each $i \in I^+ - E_0$, if $X^i(V_0)$ is an equilibrium point let $D_i(V_0)$ be the closure of $\{W_{loc}^s(X^i(V_0)) - \phi_1(W_{loc}^s(X^i(V_0))\}$, and if $X^i(V_0)$ is a periodic orbit let S be a smooth cross section of $X^i(V_0)$ with C^1 first return map $\hat{\tau}$ and let $D_i(V_0)$ be the closure of $\{W_{loc}^s(X^i(V_0)) \cap S - \hat{\tau}(W_{loc}^s(X^i(V_0)) \cap S)\}$. Then $D_i(V_0)$ is called a fundamental domain of $X^i(V_0)$ [Pal69] and satisfies $\bigcup_{t \in \mathbb{R}} \phi_t(D_i(V_0)) = W^s(X^i(V_0))$. Note that $D_i(V_0)$ is compact since it is closed in M compact. For each $i \in E_0$, let $U'_i \subset U_i$ be open neighborhoods of $X^i(V_0)$ in M and W_i an open neighborhood of V_0 in the C^1 topology such that $V \in W_i$ implies that $U_i \subset W_{loc}^u(X^i(V))$ and the backwards orbit of U'_i does not escape U_i under V. This is possible since $U'_i \subset U_i \subset W_{loc}^s(X^i(V))$ and the time-t flow ϕ_t of V restricted to $W_{loc}^s(X^i(V))$ is a contraction for any t > 0.

The main core of the argument involves proving the claim below, which constructs the pieces needed to give the neighborhood U in the statement of the Lemma with the desired properties. We claim that for every $i \in I^+ - E_0$, say $b(i, E_0) = l > 0$, there exist open neighborhoods $N'_i \subset N_i$ of $D_i(V_0)$ in M, open neighborhoods $U'_i \subset U_i$ of $W^s_{\text{loc}}(X^i(V_0))$ in M, $T_i > 0$, and an open neighborhood W_i of V_0 in the C^1 topology such that $V \in W_i$ implies that $W^s_{\text{loc}}(X^i(V)) \subset U'_i$, $D_i(V) \subset N'_i$, for every $x \in N_i$ there exists $t \in [0, T_i]$ such that $\phi_{-t}(x) \in \bigcup_{j \in E_{l-1}} U_j$, for every $x \in N'_i$ there exists $t \in [0, T_i]$ such that $\phi_{[-t,0]}(x, V) \subset N_i$ and

 $\phi_{-t}(x) \in \bigcup_{j \in E_{l-1}} U'_j$, every backward orbit of points in U_i is either in $W^u(X^i(V))$ or enters N_i in finite negative time, and every backward orbit of points in U'_i is contained in U_i at least until it enters N_i .

We proceed by induction in k on the sets E_k of the directed graph for V_0 . Suppose that for every $i \in E_k$, say $b(i, E_0) = l \leq k$, there exist open neighborhoods $N'_i \subset N_i$ of $D_i(V_0)$, open neighborhoods $U'_i \subset U_i$ of $W^s_{\text{loc}}(X^i(V_0)), T_i > 0$, and an open neighborhood W_i of V_0 such that $V \in W_i$ implies that $W^s_{\text{loc}}(X^i(V)) \subset U'_i, D_i(V) \subset N'_i$, for every $x \in N_i$ there exists $t \in [0, T_i]$ such that $\phi_{-t}(x) \in \bigcup_{j \in E_{l-1}} U_j$, for every $x \in N'_i$ there exists $t \in [0, T_i]$ such that $\phi_{[-t,0]}(x,V) \subset N_i$ and $\phi_{-t}(x) \in \bigcup_{j \in E_{l-1}} U_j$, every backward orbit of points in U_i is either in $W^{u}(X^{i}(V))$ or enters N_{i} in finite negative time, and every backward orbit of points in U'_i is contained in U_i at least until it enters N_i . Let $i \in s_{k+1}$. Note that this implies that every path starting from i consists only of vertices in E_k (other than i itself) since otherwise $b(i, E_0)$ would be greater than k+1. Hence, $W^s(X^i(V_0)) \cap W^u(X^j(V_0)) \neq \emptyset$ is possible only if $j \in E_k$. Consequently, $D_i(V_0) = D_i(V_0) \cap \bigcup_{j \in E_k} W^u(X^j(V_0))$ since $D_i(V_0) \subset W^s(X^i(V_0)) \subset W^s(X^i(V_0))$ $\partial W^s(X^s(V_0)) \subset \bigcup_{j \in I^+} W^u(X^j(V_0))$ and by the above remark. Let $x \in D_i(V_0) \cap W^u(X^j(V_0))$ for some $j \in E_k$. As U'_j is an open neighborhood of $X^j(V_0)$, there exists $T_x > 0$ such that $\phi_{-T_x}(x) \in U'_i$. By continuity of the flow, there exists an open neighborhood U_x of x such that $\phi_{-T_x}(U_x) \subset U'_j$. Repeating this for each $j \in E_k$, we have that the collection $\{U_x\}_{x \in D_i(V_0)}$ is an open cover of $D_i(V_0)$, so there exists a finite subcover, say $\{U_{x_n}\}$. Choose U'_{x_n} and U''_{x_n} open such that $\overline{U}''_{x_n} \subset U'_{x_n} \subset \overline{U}'_{x_n} \subset U_{x_n}$ and $D_i(V_0) \subset \bigcup_n U''_{x_n}$. Let $N'_i = \bigcup_n U''_{x_n}$ and let $T_i = \max_n T_{x_n}$. Then N'_i is an open neighborhood of $D_i(V_0)$ such that for each $x \in N'_i$ there exists $t \in [0, T_i]$ such that $\phi_{-t}(x) \in \bigcup_{j \in E_k} U'_j$. Since N'_i is an open neighborhood of $D_i(V_0)$, by Lemma 2.3.24 there exist an open neighborhood U_i of $W^s_{\rm loc}(X^i(V_0))$ in M, an open neighborhood W_i of V_0 in the C^1 topology, and an open neighborhood U'_i of $X^i(V_0)$ in M such that $V \in W_i$ implies that $W^s_{\text{loc}}(X^i(V)) \subset U_i, X^i(V) \in U'_i, D_i(V) \subset N'_i$, for every $x \in U_i - W^u(X^i(V))$, the backwards orbit of x under V intersects N'_i in finite negative time, and for every $x \in U'_i$ its backwards orbit is contained in U_i at least until it enters N'_i . Shrink W_i if necessary so that $V \in W_i$ implies that $\phi_{-T_{x_n}}(U_{x_n}, V) \subset \bigcup_{j \in E_k} U'_j$ for all n, which is possible by continuity of the flow and since n is finite. Let $\epsilon_i > 0$ such that the Hausdorff distance from $\partial \phi_{[-T_{x_n},0]}(U'_{x_n},V_0)$ to $\partial \phi_{[-T_{x_n},0]}(U_{x_n},V_0)$ is greater than ϵ_i , which is possible since the closure of the former is contained in the latter. Similarly, shrink ϵ_i if necessary so that the Hausdorff distance from $\partial \phi_{[-T_{x_n},0]}(U'_{x_n},V_0)$ to $\partial \phi_{[-T_{x_n},0]}(U''_{x_n},V_0)$ is greater than ϵ_i . Shrink W_i if necessary so that $V \in W_i$ implies that $\phi_{[-T_{x_n},0]}(U_{x_n},V)$ is $\epsilon_i \ C^1$ -close to $\phi_{[-T_{x_n},0]}(U_{x_n},V_0)$ and $\phi_{[-T_{x_n},0]}(U''_{x_n},V)$ is $\epsilon_i \ C^1$ -close to $\phi_{[-T_{x_n},0]}(U''_{x_n},V_0)$. Let $N_n := \phi_{[-T_{x_n},0]}(U'_{x_n},V_0)$. The above implies that for any $V \in W_i$, $\phi_{[-T_{x_n},0]}(U''_{x_n},V) \subset N_n \subset \phi_{[-T_{x_n},0]}(U_{x_n},V)$. Let $N_i = \bigcup_n N_n$. Then N_i is open and for every $V \in W_i$ and $x \in N_i$ there exists $t \in [0,T_i]$ such that $\phi_{-t}(x,V) \in \bigcup_{j \in E_k} U'_j$. Furthermore, for every $V \in W_i$ and $x \in D_i(V)$ there exists $t \in [0,T_i]$ such that $\phi_{-t}(x,V) \in \bigcup_{j \in E_k} U'_j$ and $\phi_{[-t,0]}(x,V) \subset N_i$. This satisfies the induction step for i. Repeating for all $i \in s_{k+1}$ completes the induction.

Now we prove the claim. Let $W = \bigcap_{i \in I^+} W_i$. Let $U = \bigcup_{i \in I^+} U_i \bigcup_{i \in I^+ - E_0} N_i$. First we show that U is contained in the union of unstable manifolds of the critical elements. Now let $V \in W$ and let $x \in U$. By construction, the backwards orbit of x passes from U_i to either $W^u(X^i(V))$ or to N'_i and then to U'_j with $b(j, E_0) < b(i, E_0)$ and repeats this process. Note that for any $x \in N_i$, its backwards orbit also passes to U_j with $b(j, E_0) < b(i, E_0)$. Since $b(i, E_0)$ is finite, repeating this process the orbit must ultimately enter $W^u(X^j(V))$ for some $j \in I^+ - E_0$ or must enter $U_k \subset W^u(X^k(V))$ for some $k \in E_0$. Hence, $U \subset \bigcup_{i \in I^+} W^u(X^i(V))$. Next we show that U contains the union of stable manifolds of the critical elements. This is trivially true for sources since for any $V \in W$ and $i \in E_0$, $W^s(X^i(V)) = X^i(V) \subset U_i \subset U$. So, let $i \in I^+ - E_0$. Then $W^s_{\text{loc}}(X^i(V)) \subset U_i \subset U$. Therefore, it suffices to show that for any $x \in D_i(V)$ the backwards orbit of x under V is contained in U. So, let $x \in D_i(V)$. Then $x \in N'_i$ so there exists $t \in [0, T_i]$ such that $\phi_{[-t,0]}(x, V) \subset N_i, \phi_{-t}(x, V) \in U'_i$ with $b(j, E_0) < 0$ $b(i, E_0)$, and either $\phi_{-t}(x, V) \in W^u_{\text{loc}}(X^j(V))$ or the backwards orbit of $\phi_{-t}(x, V)$ remains in U_j until it enters N'_j , say at y. Then there exists $t \in [0, T_j]$ such that $\phi_{[-t,0]}(y, V) \subset N_j$ and $\phi_{-t}(y,V) \in U'_k$ with $b(k,E_0) < b(j,E_0)$. As $b(i,E_0)$ is finite, this process must terminate in a finite number of steps with the backwards orbit of x entering $W^u_{\text{loc}}(X^j(V)) \cap U'_j$ for some $j \in I^+$, say at z. By construction, the backwards orbit of z under V does not escape U_j . As $W^s_{\text{loc}}(X^i(V)) \bigcup_{t < 0} \phi_t(D_i(V)) = W^s(X^i(V))$, this implies that $W^s(X^i(V)) \subset U$. Hence, $U \supset \bigcup_{i \in I^+} W^s(X^i(V))$. Thus, by Lemma 2.3.8, $U \supset \bigcup_{i \in I^+} W^s(X^i(V_0)) = \overline{W}^s(X^s(V_0))$.

Lemma 2.3.26 shows that under the conditions of Lemma 2.3.25, for any vector field sufficiently C^1 close to V_0 , the only nonwandering points in U for this vector field are the perturbations of the critical elements of V_0 in U.

Lemma 2.3.26. Under the conditions of Lemma 2.3.25, for $V \in W$, $\Omega(V) \cap U = \bigcup_{i \in I^+} X^i(V)$.

Proof of Lemma 2.3.26. Assuming the existence of a nonwandering point other than the perturbations of the critical elements in I^+ , the C^1 Closing Lemma implies that there is a nearby vector field and a nearby point in U which is periodic. However, by Lemma 2.3.25 this contradicts that U is contained in the union of unstable manifolds of the perturbed

critical elements. Let $V \in W$ and assume towards a contradiction that $x \in \Omega(V) \cap U$ such that $x \notin \bigcup_{i \in I^+} X^i(V)$. As $\bigcup_{i \in I^+} X^i(V)$ is a finite union of hyperbolic critical elements, they are separable by open sets, say $\bigcup_{i \in I^+} U'_i$ with $X^i(V) \subset U'_i$ for each $i \in I^+$ and the U'_i pairwise disjoint. Let U'_0 be an open neighborhood of x which, by shrinking the set of U'_i , can be made disjoint from $\bigcup_{i \in I^+} U'_i$. For each $i \in I^+$ let $U_i = U'_i \cap U$ and let $U_0 = U'_0 \cap U$. Let $W' \subset W$ such that $V \in W'$ and $V' \in W'$ implies $X^i(V') \subset U_i$ for all $i \in I^+$. By the C^1 Closing Lemma [Pug67] there exists $V' \in W'$ and $y \in U_0$ such that y is contained in a periodic orbit. By construction, $y \notin X^i(V')$ for any $i \in I^+$. As $V' \in W' \subset W$, $y \in U_0 \subset U \subset \bigcup_{i \in I^+} W^u(X^i(V'))$, which contradicts that y is periodic. \Box

Lemma 2.3.27 shows that the closure of the stable manifold of the perturbed stable equilibrium point of any vector field sufficiently close to V_0 is equal to the union of the stable manifolds of the perturbed critical elements in I^+ .

Lemma 2.3.27. Under the conditions of Lemma 2.3.25, shrinking W if necessary, for $V \in W \overline{W}^s(X^s(V)) = \bigcup_{i \in I^+} W^s(X^i(V)).$

Proof of Lemma 2.3.27. It suffices to show that $\partial W^s(X^s(V)) = \bigcup_{i \in I} W^s(X^i(V))$. That $\bigcup_{i \in I} W^s(X^i(V)) \subset \partial W^s(X^s(V))$ will follow from the proof of Theorem 2.2.23. For the other direction, choosing $U' \subset \overline{U}' \subset U$ from Lemma 2.3.25, we show that $\overline{W}^s(X^s(V)) \subset U$ which implies, by Lemma 2.3.26, that its nonwandering set is equal to $\bigcup_{i \in I^+} X^i(V)$. The result then follows.

First we show that $\bigcup_{i \in I} W^s(X^i(V)) \subset \partial W^s(X^s(V))$. By the argument in the proof of Theorem 2.2.23, shrinking W if necessary implies that for $V \in W$ and $i \in I$, $W^u(X^i(V)) \cap W^s(X^s(V)) \neq \emptyset$. By the proof of Theorem 2.2.12, this implies that $W^s(X^i(V)) \subset \partial W^s(X^s(V))$ for every $i \in I$. Hence, $\bigcup_{i \in I} W^s(X^i(V)) \subset \partial W^s(X^s(V))$.

Next we show that $\partial W^s(X^s(V)) \subset \bigcup_{i \in I} W^s(X^i(V))$. Observing the proof of Lemma 2.3.25, we may choose $U' \subset U$ open in M and $W' \subset W$ open in the C^1 topology such that $\overline{U}' \subset U$ and the conclusions of Lemma 2.3.25 hold for U' and W'. Hence, $V \in W'$ implies that $W^s(X^s(V)) \subset U'$, which implies that $\overline{W}^s(X^s(V)) \subset \overline{U}' \subset U$. So, $\partial W^s(X^s(V)) \subset U$. By Lemma 2.3.26, $\Omega(V) \cap U = \bigcup_{i \in I^+} X^i(V)$, so $\Omega(V) \cap \partial W^s(X^s(V)) = \bigcup_{i \in I^+} X^i(V)$. For $x \in \partial W^s(X^s(V)), \, \omega(x)$ is closed, invariant, and nonwandering, so $\omega(x) \subset \bigcup_{i \in I} X^i(V)$. As in the proof of Lemma 2.3.8, this implies that $\omega(x) = X^j(V)$ for some $j \in I$ and that x converges to $X^j(V)$ in forwards time. Hence, $x \in W^s(X^j(V))$. Thus, for $V \in W'$, $\partial W^s(X^s(V)) \subset \bigcup_{i \in I} W^s(X^i(V))$.

Proof of Theorem 2.3.9. Follows immediately from Lemmas 2.3.25-2.3.27.

Lemma 2.3.28 states that the nonwandering points in the boundary of the union of stable manifolds of the stable equilibrium point, where the boundary is taken in $M \times J'$, is equal to the union of the perturbations of the critical elements in I. In addition, this boundary in $M \times J'$ does not contain any additional nonwandering points beyond those contained in the union of boundaries of the stable manifolds taken in M.

Lemma 2.3.28. Under the conditions of Corollary 2.3.10, for J sufficiently small with $p_0 \in J$,

$$\Omega(V_J) \bigcap \partial W^s(X^s(V_J))$$

= $\sqcup_{p \in J} \Omega(V_p) \bigcap \partial W^s(X^s(V_p))$
= $\bigcup_{i \in I} X_J^i.$

Lemma 2.3.28. The proof uses Lemma 2.3.25 to get neighborhoods $U' \subset \overline{U}' \subset U$ such that $W^s(X^s(V_p)) \subset U'$ for all $p \in J$, so $W^s(X^s(V_J)) \subset U' \times J$. Then $\overline{W}^s(X^s(V_J)) \subset \overline{U}' \times J \subset U \times J$ and the result follows from Lemma 2.3.26.

By Lemma 2.3.25 and Lemma 2.3.27, there exist U an open neighborhood of $W^s(X^s(p_0))$ and J sufficiently small such that $p_0 \in J$ and for any $p \in J$, $\overline{W}^s(X^s(V_p)) = \bigcup_{i \in I^+} W^s(X^i(V_p)) \subset U \subset \bigcup_{i \in I^+} W^u(X^i(V_p))$. Observing the proof of Lemma 2.3.25, we may choose $U' \subset U$ open in M and shrink J further such that $\overline{U}' \subset U$ and the conclusions of Lemma 2.3.25 hold for U' and $p \in J$. In particular, for any $p \in J$, $W^s(X^s(V_p)) \subset U'$. Hence, $W^s(X^s(V_J)) \subset$ $U' \times J$, so $\overline{W}^s(X^s(V_J)) \subset \overline{U}' \times J$ where the closure is taken in $M \times J$. As $\overline{U}' \subset U$, this implies that $\overline{W}^s(X^s(V_J)) \subset U \times J$. By Lemma 2.3.26, for each $p \in J \ \Omega(V_p) \cap U =$ $\bigcup_{i \in I^+} X^i(V_p)$. Thus, $\Omega(V_J) \cap U \times J = \bigcup_{i \in I^+} X^i(V_J)$. So, $\overline{W}^s(X^s(V_J)) \subset U \times J$ implies that $\Omega(V_J) \cap \overline{W}^s(X^s(V_J)) \subset \bigcup_{i \in I^+} X^i(V_J)$. By Lemma 2.3.27, we have $\bigcup_{i \in I^+} X^i(V_J) \subset$ $\overline{W}^s(X^s(V_J))$, so $\Omega(V_J) \cap \overline{W}^s(X^s(V_J)) = \bigcup_{i \in I} X^i(V_J)$. Since $\bigcup_{i \in I} X^i(V_J) \subset \partial W^s(X^s(V_J))$, this implies that $\Omega(V_J) \cap \partial W^s(X^s(V_J)) = \bigcup_{i \in I} X^i(V_J)$. By Lemma 2.3.26, $\bigsqcup_{p \in J} \Omega(V_p) \cap \partial W^s(X^s(V_p)) = \bigsqcup_{p \in J} \bigcup_{i \in I} X^i(V_J)$. By Lemma 2.3.26,

Proof of Corollary 2.3.10. The conclusions of the corollary will follow immediately from Theorem 2.2.23 and Corollary 2.2.25 once the assumptions behind them are verified. In particular, Assumptions 2.2.14-2.2.15 and Assumptions 2.2.17-2.2.18 must be verified. Let U be the neighborhood of Lemma 2.3.26, so that $\Omega(V_J) \cap (U \times \{p_0\}) = \Omega(V_p) \cap U =$ $\bigcup_{i \in I} X^i(p_0)$. By the proof of Lemma 2.3.28, $\partial W^s(X^s(V_J)) \cap (M \times \{p_0\}) \subset U \times \{p_0\}$. In particular, U is an open neighborhood of $\partial W^s(X^s(V_J)) \cap (M \times \{p_0\})$ in $(M \times \{p_0\})$ such that $\begin{aligned} \Omega(V_J) \cap (U \times \{p_0\}) &= \bigcup_{i \in I} X^i(p_0) \text{ is a finite union of hyperbolic critical elements in } M, \text{ thus satisfying Assumptions 2.2.14-2.2.15.} The above implies that the critical elements contained in <math>\partial W^s(X^s(V_J)) \cap (M \times \{p_0\})$ are at most $\{X_{p_0}^i\}_{i \in I}$. By Assumption 2.3.6, the stable and unstable manifolds of each pair of critical elements in $\{X_{p_0}^i\}_{i \in I}$ are transverse, thus verifying Assumption 2.2.18. Let $p \in J$. For each orbit γ contained in $\partial W^s(X^s(V_J)) \cap (M \times \{p\})$, since M is compact, $\omega(\gamma)$ is nonempty, nonwandering, compact, and connected. By Lemma 2.3.28, $\Omega(V_J) \cap (\partial W^s(X^s(V_J)) \cap (M \times \{p\})) = \bigcup_{i \in I} X^i(p)$. Hence, $\omega(\gamma) \subset \bigcup_{i \in I} X^i(p)$ as it is nonwandering. As $\{X^i(p)\}_{i \in I}$ are a finite collection of hyperbolic critical elements, they are separable by open sets. As $\omega(\gamma)$ is connected, this implies that $\omega(\gamma) = X^i(p)$ for some $i \in I$. The above verifies Assumption 2.2.17.

2.3.5.2 Euclidean Space

Similar to Lemma 2.3.22, Lemma 2.3.29 shows that under Assumptions 2.3.4-2.3.13, the intersection of the nonwandering set with the RoA boundary consists of a finite union of critical elements, as opposed to an arbitrary union. This is important because a finite union of hyperbolic critical elements persists under perturbations to the vector field, whereas an arbitrary union may not.

Lemma 2.3.29. Under the conditions of Lemma 2.3.18 the number of critical elements of V_0 contained in $\partial W^s(X^s(V_0))$ is finite.

Proof of Lemma 2.3.29. By the assumptions of Lemma 2.3.18, either $\partial W^s(X^s(V_0))$ is bounded or by Assumption 2.3.13 there exists a neighborhood N of ∞ such that $\Omega(V_0) \cap N = \emptyset$. In either case, there exists a compact set $K \subset \partial W^s(X^s(V_0))$ such that $\Omega(V_0) \cap \partial W^s(X^s(V_0)) = \Omega(V_0) \cap K$. The rest of the proof proceeds exactly as in the proof of Lemma 2.3.22.

Because of Lemma 2.3.29, henceforth we assume that I, the set indexing the critical elements in the RoA boundary as in Assumption 2.3.4, is finite.

We first prove the main result in the case that the RoA is bounded. This reduces to the same proof as for compact Riemannian manifolds above.

Proof of Theorem 2.3.19. As $\overline{W}^{s}(X^{s}(V_{0}))$ is bounded, it is compact. Then the proofs of Lemma 2.3.8 and Lemmas 2.3.23-2.3.27 follow exactly as for compact Riemannian manifolds. The theorem then follows from Lemmas 2.3.25-2.3.27.

Proof of Lemma 2.3.18. That I is nonempty will follow from Assumptions 2.3.11-2.3.12. That $\overline{W}^s(X^s(V_0)) = \bigcup_{i \in I^+} W^s(X^i(V_0))$ follows from the same argument as in the proof of Lemma 2.3.8. If $\overline{W}^s(X^s(V_0))$ is bounded, then the proof of Lemma 2.3.25 can be used to find the desired neighborhood U. Otherwise, if Assumption 2.3.13 is satisfied, it can be shown there is an open, invariant, wandering set U_{∞} containing ∞ , and the proof of Lemma 2.3.25 can be modified by including ∞ in the directed graph of the critical elements in the natural way.

First we show that I is nonempty. By Assumption 2.3.11, $\partial W^s(X^s(V_0)) \neq \emptyset$, so there exists $x \in \partial W^s(X^s(V_0))$. By Assumption 2.3.12, the closure of the forward orbit of x is compact, so $\omega(x)$ is nonempty. As $\omega(x)$ is nonwandering, by Assumption 2.3.4 this implies that I must be nonempty.

Next we show that $\overline{W}^s(X^s(V_0)) = \bigcup_{i \in I^+} W^s(X^i(V_0))$. By Assumption 2.3.12, for every $x \in \partial W^s(X^s(V_0)), \omega(x)$ is nonempty and compact. Hence, the same argument as in the proof of Lemma 2.3.8 shows that $\overline{W}^s(X^s(V_0)) \subset \bigcup_{i \in I^+} W^s(X^i(V_0))$. Similarly, the same argument as in the proof of Lemma 2.3.8 shows that $\bigcup_{i \in I^+} W^s(X^i(V_0)) \subset \overline{W}^s(X^s(V_0))$ and that there are no homoclinic orbits or heteroclinic cycles among the critical elements in I^+ for V_0 .

If $\overline{W}^s(X^s(V_0))$ is bounded, then the construction in Lemma 2.3.25 can be applied to show, as in the proof of Lemma 2.3.26, that there exists an open neighborhood U of $\overline{W}^s(X^s(V_0))$ in M such that $\Omega(V_0) \cap U = \bigcup_{i \in I^+} X^i(V_0)$.

If Assumption 2.3.13 holds then let N be an open neighborhood of $\{\infty\}$ in $M \times \{\infty\}$ such that $\Omega(V_0) \cap N = \emptyset$. Define, for the sake of convenience, $W^u(\infty) := \{x \in M : \alpha(x) = \emptyset\}$. As in the proof of Lemma 2.3.25, define a directed graph of $I^+ \cup \{\infty\}$. As, by the above, there are no homoclinic orbits or cycles, and $I \cup \{\infty\}$ is finite, every maximal forward path must end in either some $i \in E_0$ or in ∞ . Let $U_\infty = \bigcup_{t \in \mathbb{R}} \phi_t(N, V)$. Then U_∞ is an open invariant neighborhood of ∞ which consists of wandering points since N does and the wandering set is invariant. Therefore, repeating the construction of U in Lemma 2.3.25 for the case of a fixed parameter vector field only yields a neighborhood U of $\overline{W}^s(X^s(V_0))$ such that $\Omega(V_0) \cap U = \bigcup_{i \in I^+} X^i(V_0)$. Furthermore, for every $x \in U$ its orbit in backwards time enters either $W^u_{\text{loc}}(X^j(V_0))$ for some $j \in I^+$ or the neighborhood N.

Remark 2.3.30. Despite the similarities between the proof of Lemma 2.3.25 and the proof of Lemma 2.3.18, even if there exists a neighborhood of ∞ containing no nonwandering points, the result of Lemma 2.3.25 does not hold for M Euclidean. The reason is that U_{∞} in the argument above can have nonwandering points in its closure, as well as nonwandering points which enter it under perturbation away from ∞ , so that perturbed orbits in $\partial W^s(X^s(V))$

can approach these nonwandering points through U_{∞} . Alternatively, setting $U_{\infty} = N'$ for N' a sufficiently small neighborhood of ∞ containing no nonwandering points implies that N' will not have nonwandering points in its closure or under perturbation, but N' is not backwards (or forwards) invariant, so the perturbed orbits in $\partial W^s(X^s(V))$ can exit N' and approach other nonwandering points.

Let \mathcal{M} be the set of closed subsets of M with the Chabauty metric. Let $\{K_n\}_{n=1}^{\infty}$ be an exhaustion of M by compact sets and let \mathcal{K}_n be the set of closed subsets of K_n with the Hausdorff metric. Define a family of functions $G_n^1 : W' \to \mathcal{K}_n$ by sending V to the closure of $\overline{W}^s(X^s(V)) \cap \operatorname{int} K_n$. Define another family of functions $G_n^2 : W' \to \mathcal{K}_n$ by sending V to the closure of the intersection of int K_n with the closure of the union of the equilibrium points and periodic orbits of V. Define the family of functions $G_n^3 : W' \to \mathcal{K}_n$ by sending V to the closure of the intersection of int K_n with the closure of the union of points in M whose ω limit sets under V are empty.

Lemma 2.3.31 shows that each of the functions defined above is lower semi-continuous over W'. As \mathcal{K}_n is a compact metric space for each n, and the set of continuity points of semi-continuous functions with codomain a compact metric space is residual by the Remark following Corollary 1 of [Kur68], these functions are continuous for generic vector fields in W'. This fact will be the basis of the proof of Theorem 2.3.20.

Lemma 2.3.31. For each n, the functions G_n^1 , G_n^2 , and G_n^3 are lower semi-continuous.

Proof of Lemma 2.3.31. Lower semi-continuity of G_n^2 will follow from the proof of [PR83, Theorem 11.3]. Lower semi-continuity of G_n^1 is shown by noting that int $W_{\text{loc}}^s(X^s(V))$ is C^1 continuous and open, and for any point in $W^s(X^s(V_0))$ there exists some time at which it enters int $W_{\text{loc}}^s(X^s(V))$. Lower semi-continuity of G_n^3 is proven by recognizing that convergence in the strong C^1 topology implies that vector fields agree outside some compact set, and that if a point has an empty ω limit set then its forward orbit must exit and remain outside this compact set for all future time.

Fix n > 0, let $V_0 \in W'$, and let $V_m \in W'$ with $V_m \to V_0$. First we show that G_n^1 is lower semi-continuous. It suffices to show that $\liminf_{m\to\infty} G_n^1(V_m) \supset G_n^1(V_0)$. First suppose there exists $x \in G_n^1(V_0) \cap W^s(X^s(V_0)) \cap \operatorname{int} K_n$. Then there exists T > 0 such that $\phi_T(x, V_0) \in \operatorname{int} W^s_{\operatorname{loc}}(X^s(V_0))$. As $V_m \to V_0$, $\operatorname{int} W^s_{\operatorname{loc}}(X^s(V))$ is C^1 continuous with respect to V, $\operatorname{int} W^s_{\operatorname{loc}}(X^s(V_0))$ is open, and the flow is C^1 continuous, there exists Zsuch that $m \geq Z$ implies that $\phi_T(x, V_m) \in \operatorname{int} W^s_{\operatorname{loc}}(X^s(V_0)) \subset W^s_{2\epsilon}(X^s(V_m))$. Thus, $x \in \operatorname{int} K_n \cap W^s(X^s(V_m)) \subset G_n^1(V_m)$ for $m \geq M$, so $x \in \liminf_{m\to\infty} G_n^1(V_m)$. Next suppose there exists $x \in G_n^1(V_0) \cap \partial W^s(X^s(V_0)) \cap \operatorname{int} K_n$. Since $x \in \partial W^s(X^s(V_0))$ and $\operatorname{int} K_n$ is open, there exist $\{x_k\}_{k=1}^{\infty} \subset \operatorname{int} K_n \cap W^s(X^s(V_0))$ such that $x_k \to x$. By the above, $\{x_k\}_{k=1}^{\infty} \subset \liminf_{m\to\infty} G_n^1(V_m)$. As $\liminf_{m\to\infty} G_n^1(V_m)$ is closed [Hau57, Section 28], this implies that $x \in \liminf_{m\to\infty} G_n^1(V_m)$. Now suppose there exists $x \in G_n^1(V_0) \cap \partial K_n$. Then by definition of G_n^1 , there exist $\{x_k\}_{k=1}^{\infty} \subset \operatorname{int} K_n \cap \overline{W}^s(X^s(V_0))$ such that $x_k \to x$. By the above, $\{x_k\}_{k=1}^{\infty} \subset \liminf_{m\to\infty} G_n^1(V_m)$. As $\liminf_{m\to\infty} G_n^1(V_m)$ is closed, this implies that $x \in \liminf_{m\to\infty} G_n^1(V_m)$. Hence, $\liminf_{m\to\infty} G_n^1(V_m) \supset G_n^1(V_0)$.

That G_n^2 is lower semi-continuous is shown in the proof of [PR83, Theorem 11.3] in which $\overline{\Gamma}_i$ plays the role of G_n^2 .

Next we show that G_n^3 is lower semi-continuous. Since $V_m \to V_0$ in the strong C^1 topology there exists a compact set K such that for m sufficiently large V_m and V_0 are equal outside of K. Suppose $x \in G_n^3(V_0) \cap \operatorname{int} K_n$. Since $\omega(x) = \emptyset$ and K is compact, there must exist T > 0such that $\phi_t(x, V_0) \notin K$ for $t \geq T$. Let $y = \phi_T(x, V_0)$ and for each m let $x_m = \phi_{-T}(y, V_m)$. Since V_m and V_0 agree outside of K, $\phi_t(x_m, V_m) \notin K$ for $t \geq T$ and the foward orbits of y under V_0 and $\phi_t(x_m, V_m)$ under V_m agree. Hence, their ω limit sets must be equal, so $\omega(x_m) = \omega(y) = \emptyset$. Then for m sufficiently large, $x_m \in \operatorname{int} K_n$ since int K_n is open. Clearly $x_m \to \phi_{-T}(y, V_0) = x$. Thus, $x \in \liminf_{m \to \infty} G_n^3(V_m)$. Now suppose there exists $x \in$ $G_n^3(V_0) \cap \partial K_n$. Then by definition of G_n^3 , there exist $\{x_k\}_{k=1}^{\infty} \subset \operatorname{int} K_n$ such that $x_k \to x$ and $\omega(x_k) = \emptyset$ for each k. By the above, $\{x_k\}_{k=1}^{\infty} \subset \liminf_{m \to \infty} G_n^3(V_m)$. As $\liminf_{m \to \infty} G_n^3(V_0)$. \Box

Lemma 2.3.32 shows that the assumption that a vector field in W' is a point of continuity of G_n^1 , G_n^2 , and G_n^3 for all n is generic.

Lemma 2.3.32. Let W be an open set in the strong C^1 topology. Then for generic $V_0 \in W$, V_0 is a point of continuity of G_n^1 , G_n^2 , and G_n^3 for all n.

Proof of Lemma 2.3.32. By the Remark following Corollary 1 of [Kur68], for fixed n, since K_n is compact and therefore \mathcal{K}_n is a compact metric space, the set of continuity points C_n of G_n^1 , G_n^2 , and G_n^3 is residual in W'. Let $\tilde{C} = \bigcap_{n=1}^{\infty} C_n$. Then \tilde{C} is residual in W' and, since the space of C^1 vector fields with the strong C^1 topology has the Baire property, \tilde{C} is dense in W'.

Lemma 2.3.33 shows that for p sufficiently close to p_0 , every point x in $\overline{W}^s(X^s(V_p))$ has $\omega(x, V_p)$ nonempty. This ensures that under small perturbation, no orbits are in $\overline{W}^s(X^s(V_p))$ which flow to infinity in forward time. This is one of the key lemmas behind the proof of

Theorem 2.3.20 because it will help ensure that every orbit in $\overline{W}^{s}(X^{s}(V_{p}))$ converges to a critical element.

Lemma 2.3.33. If $V_{p_0} \in W'$ complete, satisfies Assumptions 2.3.4-2.3.13, and V_{p_0} is a point of continuity of G_n^1 , G_n^2 , and G_n^3 for all n, then for J a sufficiently small open neighborhood of p_0 , $p \in J$ implies that for all $x \in \overline{W}^s(X^s(V_p))$, $\omega(x, V_p) \neq \emptyset$.

Proof of Lemma 2.3.33. We claim that there exists an open neighborhood of p_0 in J such that for every p in this neighborhood and every $x \in \overline{W}^s(X^s(V_p))$, $\omega(x, V_p)$ is nonempty. Assume towards a contradiction than no such open neighborhood exists. As J is a manifold it is first countable, so the above implies that there exists a sequence $\{p_m\}_{m=1}^{\infty}$ such that $p_m \to p_0$ and there exist $x_m \in \overline{W}^s(X^s(V_{p_m}))$ such that $\omega(x_m, V_{p_m}) = \emptyset$. As $x_m \in M \cup \{\infty\}$ compact, passing to a subsequence if necessary we have that $x_m \to x \in M \cup \{\infty\}$. We will see that if $x \in M$ we arrive at a contradiction via continuity of G_n^1 and G_n^3 at V_{p_0} . If $x \in \infty$, then either we are able to reduce to the case of $x \in M$ by following each x_n along its orbit or, as V_{p_0} agrees with V_{p_n} outside a compact set since $V_{p_n} \to V_{p_0}$ in the strong C^1 topology, we obtain a contradiction to Assumption 2.3.13.

First assume that $x \in M$. Then there exists n such that $x \in \operatorname{int} K_n$. By continuity of G_n^1 at V_{p_0} , $x \in \overline{W}^s(X^s(V_{p_0}))$. By continuity of G_n^3 at V_{p_0} , $\omega(x, V_{p_0}) = \emptyset$. By Lemma 2.3.18, $\overline{W}^s(X^s(V_{p_0})) = \bigcup_{i \in I^+} W^s(X^i(V_0))$ so that $x \in W^s(X^i(V_{p_0}))$ for some $i \in I^+$. Thus, $\omega(x, V_{p_0}) = X^i(V_{p_0})$ which contradicts that $\omega(x, V_{p_0}) = \emptyset$. Next assume that $x = \infty$. Let O_m denote the orbit in both forwards and backwards time of x_m under V_{p_m} . Assume towards a contradiction that there exists a subsequence $\{O_{m_k}\}_{k=1}^{\infty}$ and $K' \subset M$ compact such that $O_{m_k} \cap K' \neq \emptyset$ for all k. Letting $y_{m_k} \in O_{m_k} \cap K'$ for all k implies that, since K' is compact, passing to a subsequence if necessary $y_{m_k} \to y \in K' \subset M$. Hence, as the ω limit set is invariant along an orbit, this reduces to the case of $x \in M$ above, which leads to a contradiction. So, there must exist a subsequence $\{O_{m_k}\}_{k=1}^{\infty}$ such that $O_{m_k} \cap K_{m_k} = \emptyset$ for all k. Let N be the neighborhood of Assumption 2.3.13 so that $\Omega(V_{p_0}) \cap N = \emptyset$ and no orbit under V_{p_0} contained in N has both its ω and α limit sets under V_{p_0} empty. As $V_{p_{m_k}} \to V_{p_0}$, there exists $K \subset M$ compact such that $V_{p_{m_k}}$ and V_{p_0} agree outside K. Then for k sufficiently large, the closure of O_{m_k} is contained in $(M-K) \cap N$. As V_{p_0} and $V_{p_{m_k}}$ agree on M-K, this implies that O_{m_k} is also the orbit of x_{m_k} under V_{p_0} , so that $\omega(O_{m_k}, V_{p_0}) = \omega(O_{m_k}, V_{p_{m_k}}) = \emptyset$. As $O_{m_k} \subset N$, this implies that $\alpha(O_{m_k}, V_{p_0}) \neq \emptyset$. As the closure of O_{m_k} is contained in N, $\alpha(O_{m_k}, V_{p_0}) \subset N$. However, since $\alpha(O_{m_k}, V_{p_0})$ is nonempty and nonwandering under V_{p_0} , this contradicts that $\Omega(V_{p_0}) \cap N = \emptyset$. Hence, $x_m \to x = \infty$ results in a contradiction. As $x \notin M$ and $x \neq \infty$, we obtain a contradiction to the main assertion above. Lemma 2.3.34 is the main technical result behind Theorem 2.3.20. It states that under the conditions of the theorem, for p sufficiently close to p_0 , the only nonwandering points that are present in $\overline{W}^s(X^s(V_p))$ are the perturbations of the original critical elements contained in $\overline{W}^s(X^s(V_{p_0}))$.

Lemma 2.3.34. If $V_{p_0} \in W'$ complete, satisfies Assumptions 2.3.4-2.3.13, and V_{p_0} is a point of continuity of G_n^1 , G_n^2 , and G_n^3 for all n, then for J a sufficiently small open neighborhood of p_0 , $p \in J$ implies that $\Omega(V_p) \cap \overline{W}^s(X^s(V_p)) = \bigcup_{i \in I^+} X^i(V_p)$.

Proof of Lemma 2.3.34. In the case where $\overline{W}^s(X^s(V_{p_0}))$ is bounded, the result follows by Theorem 2.3.19 under weaker assumptions than stated in this lemma. So, we may assume that $\overline{W}^s(X^s(V_{p_0}))$ is unbounded. We claim that there exists an open neighborhood of p_0 in J such that for every p in this neighborhood, $\Omega(V_p) \cap \overline{W}^s(X^s(V_p)) = \bigcup_{i \in I^+} X^i(V_p)$. Assume towards a contradiction that no such open neighborhood exists. As J is a manifold it is first countable, so the above implies that there exists a sequence $\{p_m\}_{m=1}^{\infty}$ such that $p_m \to p_0$ and there exist $x_m \in \overline{W}^s(X^s(V_{p_m}))$ with $x_m \in \Omega(V_{p_m})$ such that $x_m \notin \bigcup_{i \in I^+} X^i(V_{p_m})$. By Lemma 2.3.33 we may shrink J if necessary so that $p \in J$ and $x \in \overline{W}^s(X^s(V_p))$ implies that $\omega(x, V_p) \neq \emptyset$. As $p_m \to p_0$, we may assume $p_m \in J$ for all m. In particular, $\omega(x_m, V_{p_m}) \neq \emptyset$ for all m. As $\{V_p\}_{p \in J}$ is strong C^1 continuous and $p_m \to p_0$, $V_{p_m} \to V_{p_0}$ in the strong C^1 toplogy. This implies that there exists K compact such that for m sufficiently large, V_{p_m} and V_{p_0} agree on M - K. By Assumption 2.3.13, there exists a neighborhood N of ∞ such that $\Omega(V_{p_0}) \cap N = \emptyset$.

The proof will proceed as follows. First we show that there exists a point y_m in the orbit of x_m such that $\omega(y_m, V_{p_m}) \cap (K \cup (M - N)) \neq \emptyset$ for all m and $\{y_m\}_{n=1}^{\infty}$ are contained in a compact set. Passing to a subsequence if necessary implies that $y_m \to y \in M$, and continuity of G_n^1 at V_{p_0} implies that $y \in \overline{W}^s(X^s(V_{p_0}))$. Then we will apply the C^1 Closing Lemma to obtain a sequence of vector fields $\{V'_m\}_{m=1}^{\infty}$ with $V'_m \to V_{p_0}$ in the strong C^1 topology such that V'_m and V_{p_0} agree outside the compact set $K \cup (M - N)$ for all m, and there exist points z_m near y_m with $z_m \to y$ and z_m periodic under V'_m . Continuity of G_n^2 at V_{p_0} then implies that y is periodic. Hence, $y \in X^j(V_{p_0})$ for some j. It is then shown that the orbits of z_m under V'_m converge with respect to the Hausdorff metric to $X^j(V_{p_0})$. As $X^j(V'_m) \to X^j(V_{p_0})$ with respect to the Hausdorff metric as well, and $X^j(V'_m)$ is disjoint from the orbit of z_m under V'_m , by hyperbolicity of $X^j(V_{p_0})$ we will obtain a contradiction.

First, assume towards a contradiction that there exists m such that $\omega(x_m, V_{p_m}) \subset (M - K) \cap N$. Since $(M - K) \cap N$ is open, this implies that, after some time $T_m > 0$, the forward

orbit of x_m under V_{p_m} remains in $(M - K) \cap N$ for all future time. Let $y'_m = \phi_{T_m}(x_m, V_{p_m})$. Then the above implies that the forward orbit of y'_m under V_{p_m} agrees with the forward orbit of y'_m under V_{p_0} . So, $\omega(y'_m, V_{p_0}) = \omega(y'_m, V_{p_m}) = \omega(x_m, V_{p_m}) \subset (M - K) \cap N$ is nonempty since $\omega(x_m, V_{p_m})$ is. So, $\omega(y'_m, V_{p_0})$ is nonempty, nonwandering, and contained in N. This contradicts that $\Omega(V_{p_0}) \cap N = \emptyset$. Therefore, we must have $\omega(x_m, V_{p_m}) \cap (K \cup (M - N)) \neq \emptyset$ for all m. As $K \cup (M-N)$ is compact, there exists n such that $K \cup (M-N) \subset \operatorname{int} K_{n-1}$. As $\omega(x_m, V_{p_m}) \cap \text{int } K_{n-1} \neq \emptyset$, int K_{n-1} is open, and $\omega(x_m, V_{p_m})$ is in the closure of the forward orbit of x_m under V_{p_m} , there exists y_m a point in the forward orbit of x_m under V_{p_m} such that $y_m \in \text{int } K_{n-1}$ for each m. Note that $y_m \in \overline{W}^s(X^s(V_{p_m}))$ and is nonwandering since x_m is and both $\overline{W}^s(X^s(V_{p_m}))$ and $\Omega(V_{p_m})$ are invariant. Furthermore, $y_m \notin \bigcup_{i \in I^+} X^i(V_{p_m})$ as x_m is not and these sets are also invariant. As K_{n-1} is compact and $y_m \in K_{n-1}$ for all m, passing to a subsequence if necessary we have that $y_m \to y \in K_{n-1} \subset \text{int } K_n$. Note that $y_m \in G_n^1(V_{p_m}) \cap \operatorname{int} K_n$ for each m since $y_m \in \overline{W}^s(X^s(V_{p_m})) \cap \operatorname{int} K_n$ for each m. As G_n^1 is continuous at V_{p_0} , $V_{p_m} \to V_{p_0}$, $y_m \to y$, and $y_m \in G_n^1(V_{p_m})$ for all $m, y \in \overline{W}^s(X^s(V_{p_0}))$. Furthermore, $\omega(y_m, V_{p_m}) = \omega(x_m, V_{p_m})$, so, by the above, $\omega(y_m, V_{p_m}) \cap (K \cup (M - N)) \neq \emptyset$ for all m. Then by the C^1 Closing Lemma for noncompact manifolds [PR83, p. 311] there exists a sequence $\{V'_m\}_{m=1}^{\infty}$ of vector fields on M such that V'_m and V_{p_m} agree on $M - (K \cup (M - N))$ and are $(1/m) C^1$ -close on $(K \cup (M-N))$, and such that there exists a point z_m with distance less than (1/m) from y_m such that z_m is periodic under V'_m . As $y_m \notin \bigcup_{i \in I^+} X^i(V_{p_m})$, each $X^i(V_{p_m})$ is compact for $i \in I^+$, and I^+ is finite, y_m and $\bigcup_{i \in I^+} X^i(V_{p_m})$ are separable by open sets. As z_m can be chosen arbitrarily close to y_m and each $X^i(V_p)$ varies C^1 continuously with respect to p, for m sufficiently large we may choose z_m such that $z_m \notin \bigcup_{i \in I^+} X^i(V'_m)$. As V'_m and V_{p_m} agree outside the compact set $K \cup (M-N)$ for all m, and as their C^1 distance on $K \cup (M-N)$ tends to zero, $V'_m \to V_{p_0}$ in the strong C^1 topology. Furthermore, $z_m \to y$, and z_m periodic with distance less than (1/m) from y_m implies that for m sufficiently large, $z_m \in \text{int } K_n \text{ and } z_m \in G_n^2(V'_m)$. Since G_n^2 is continuous at $V_{p_0}, z_m \in G_n^2(V'_m)$ for m sufficiently large, $V'_m \to V_{p_0}$, and $z_m \to y, y \in G^2_n(V_{p_0})$. As $y \in \text{int } K_n$ as well, this implies that y is contained in the union of equilibria and periodic orbits of V_{p_0} . As $y \in \overline{W}^s(X^s(V_{p_0}))$ by the argument above, and by Lemma 2.3.18 $\Omega(V_{p_0}) \cap \overline{W}^s(X^s(V_{p_0})) = \bigcup_{i \in I^+} X^i(V_{p_0})$, this implies that $y \in X^j(V_{p_0})$ for some $j \in I^+$.

Let O_m denote the orbit of z_m under V'_m for each m. Let \mathcal{M}^{∞} denote the metric space of closed subsets of $M \cup \{\infty\}$ equipped with the Hausdorff metric. As $M \cup \{\infty\}$ is compact, \mathcal{M}^{∞} is compact Then $\{O_m\}_{m=1}^{\infty}$ is a sequence in \mathcal{M}^{∞} compact, so passing to a subsequence if necessary implies that $O_m \to O \subset \mathcal{M}^{\infty}$. Note that since $y \in X^j(V_{p_0})$ periodic, the orbit of y under V_{p_0} is $X^j(V_{p_0})$. Let $\phi_t(y, V_{p_0}) \in X^j(V_{p_0})$ for any $t \in \mathbb{R}$. Then, since $z_m \to y$ and $V'_m \to V_{p_0}, \phi_t(z_m, V'_m) \to \phi_t(y, V_{p_0})$ by continuity of the flow. As $\phi_t(z_m, V'_m) \in O_m$ and $O_m \to O$, this implies that $\phi_t(y, V_{p_0}) \in O$ for all $t \in \mathbb{R}$, so $X^j(V_{p_0}) \subset O$. Assume towards a contradiction that there exists $x \in O - X^j(V_{p_0})$. As $\{x\}$ and $X^j(V_{p_0})$ are compact and disjoint, there exists an open neighborhood U' containing $X^{j}(V_{p_{0}})$ such that \overline{U}' is disjoint from x. By Lemma 2.3.18, there exists an open neighborhood U of $\overline{W}^s(X^s(V_{p_0})) \supset X^j(V_{p_0})$ such that $\Omega(V_{p_0}) \cap U = \bigcup_{i \in I^+} X^i(V_{p_0})$. Shrinking U' if necessary, we may have that $\overline{U}' \subset U$ and, since I^+ is finite and each $X^i(V_{p_0})$ is compact, \overline{U}' is disjoint from $X^i(V_{p_0})$ for each $i \neq j$. The above implies that $\partial U' \cap \Omega(V_{p_0}) = \emptyset$. As $x \in O$ there exists a sequence $x'_m \in O_m$ such that $x'_m \to x$. So, m sufficiently large implies that $x'_m \in O_m \cap (M - \overline{U}')$ and $z_m \in O_m \cap U'$ since $z_m \to y \in X^j(V_{p_0}) \subset U'$. Then O_m periodic implies it is connected, so there exists $y'_m \in O_m \cap \partial U'$. As $\partial U'$ is compact, passing to a subsequence if necessary we have that $y'_m \to y' \in \partial U' \cap O$. Choose n sufficiently large such that $y' \in \operatorname{int} K_n$. Then m sufficiently large implies that $y'_m \in \operatorname{int} K_n$ periodic, so $y'_m \in G^2_n(V'_m)$ for m sufficiently large. By continuity of G_n^2 at V_{p_0} , since $y'_m \to y'$, $V'_m \to V_{p_0}$, and $y'_m \in G_n^2(V'_m)$ for m sufficiently large, $y' \in G_n^2(V'_m) \cap \operatorname{int} K_n$. So, y' is periodic under V_{p_0} , which implies that $y' \in \Omega(V_{p_0}) \cap \partial U'$. But $\Omega(V_{p_0}) \cap \partial U' = \emptyset$, which is a contradiction. So, $O = X^j(V_{p_0})$. By the proofs of [JPdM82, Lemmas 2.1-2.2] there exists an open neighborhood N_j of $X^j(V_{p_0})$ and an open neighborhood \hat{W} of V_{p_0} in the strong C^1 topology such that $N_j \subset U'$ and for every $V \in \hat{W}$, N_i does not entirely contain any equilibrium points or periodic orbits of V other than $X^{j}(V)$. As $V'_{m} \to V_{0}$ in the strong C^{1} topology, m sufficiently large implies that $V'_m \in \hat{W}$. As $O_m \to O = X^j(V_{p_0}) \subset N_j$, m sufficiently large implies that $O_m \subset N_j$. As O_m is periodic and disjoint from $\bigcup_{i \in I^+} X^i(V'_m)$, this contradicts the choice of \hat{W} and N_j .

Lemma 2.3.35 shows that there exists an open neighborhood U of $\overline{W}^s(X^s(V_{p_0}))$ in $M \cup \{\infty\}$ such that for p sufficiently close to p_0 , $\overline{W}^s(X^s(V_p))$ is contained in U. This will be used in the proof of Corollary 2.3.21.

Lemma 2.3.35. If $V_{p_0} \in W'$ complete, satisfies Assumptions 2.3.4-2.3.13, and V_{p_0} is a point of continuity of G_n^1 , G_n^2 , and G_n^3 for all n, then for J a sufficiently small open neighborhood of p_0 , $p \in J$ implies that $\overline{W}^s(X^s(V_p)) \subset U$, where U is the open neighborhood of $\overline{W}^s(X^s(V_{p_0}))$ constructed in Lemma 2.3.18.

Proof of Lemma 2.3.35. In the case where $\overline{W}^s(X^s(V_{p_0}))$ is bounded, the result follows by Theorem 2.3.19 under weaker assumptions than stated in this lemma. So, we may assume that $\overline{W}^s(X^s(V_{p_0}))$ is unbounded. Let U be the neighborhood of Lemma 2.3.18. Then U is an open neighborhood of $\overline{W}^s(X^s(V_{p_0}))$ in $M \cup \{\infty\}$. We claim that there exists an open neighborhood of p_0 in J such that for every p in this neighborhood, $\overline{W}^s(X^s(V_p)) \subset U$. Assume towards a contradiction that no such open neighborhood exists. As J is a manifold it is first countable, so the above implies that there exists a sequence $\{p_m\}_{m=1}^{\infty}$ such that $p_m \to p_0$ and there exist $x_m \in \overline{W}^s(X^s(V_{p_m}))$ with $x_m \notin U$. As $M \cup \{\infty\}$ is compact, passing to a subsequence if necessary we have that $x_m \to x \in M \cup \{\infty\}$. First suppose $x = \infty$. As $\overline{W}^s(X^s(V_{p_0}))$ is unbounded and U is an open neighborhood of $\overline{W}^s(X^s(V_{p_0}))$ in $M \cup \{\infty\}$, $\infty \in U$. As U is open and $x_m \to \infty \in U$, m sufficiently large implies that $x_m \in U$, which is a contradiction. So, we must have $x \in M$. Then there exists n such that $x \in \operatorname{int} K_n$. As $x_m \to x$, m sufficiently large implies that $x_m \in \overline{W}^s(X^s(V_{p_m})) \cap \operatorname{int} K_n \subset G_n^1(V_{p_m})$. By continuity of G_n^1 at V_{p_0} , since $x_m \to x$, $V_{p_m} \to V_{p_0}$, and $x_m \in G_n^1(V_{p_m})$ for m sufficiently large, $x \in G_n^1(V_{p_0}) \cap \operatorname{int} K_n$. So, $x \in \overline{W}^s(X^s(V_{p_0})) \subset U$. As U is open and $x_m \to x \in U$, msufficiently large implies that $x_m \in U$, which is a contradiction. \Box

Proof of Theorem 2.3.20. First it is shown using Lemma 2.3.33 and Lemma 2.3.34 that for J sufficiently small, $p \in J$ implies that $\overline{W}^s(X^s(V_p)) \subset \bigcup_{i \in I^+} W^s(X^i(V_p))$. Then by Lemma 2.3.18, $W^u(X^i(V_{p_0})) \cap W^s(X^s(V_{p_0})) \neq \emptyset$ is a transversal intersection, which persists by openness of transversality for J sufficiently small, and this implies by the proof of Theorem 2.2.12 that for $p \in J$, $\bigcup_{i \in I^+} W^s(X^i(V_p)) \subset \overline{W}^s(X^s(V_p))$.

By Lemma 2.3.32 the additional assumptions present in the statement of Lemma 2.3.34 that do not appear in the statement of the theorem hold for generic $V_{p_0} \in W'$. By Lemma 2.3.34, shrinking J if necessary implies it is an open neighborhood of p_0 such that $p \in J$ implies that $\Omega(V_p) \cap \overline{W}^s(X^s(V_p)) = \bigcup_{i \in I^+} X^i(V_p)$. By Lemma 2.3.33, shrinking J further if necessary implies it is an open neighborhood of p_0 such that $p \in J$ and $x \in \overline{W}^{s}(X^{s}(V_{p}))$ implies that $\omega(x, V_{p}) \neq \emptyset$. So, let $p \in J$ and $x \in \overline{W}^{s}(X^{s}(V_{p}))$. As $\overline{W}^{s}(X^{s}(V_{p}))$ is closed and invariant under V_{p} , and since $\omega(x, V_{p})$ is contained in the closure of the orbit of x under V_p , $\omega(x, V_p) \subset \overline{W}^s(X^s(V_p))$. Then $\omega(x, V_p)$ is nonempty, nonwandering, closed, and contained in $\overline{W}^{s}(X^{s}(V_{p}))$, so by the above this implies that $\omega(x, V_p) \subset \bigcup_{i \in I^+} X^i(V_p)$. As $\bigcup_{i \in I^+} X^i(V_p)$ is compact and $\omega(x, V_p)$ is closed, $\omega(x, V_p)$ is compact. Since $\omega(x, V_p)$ is compact, it is connected. Hence, $\omega(x, V_p) = X^j(V_p)$ for some $j \in I^+$. As $\omega(x, V_p)$ is compact, the forward orbit of x must converge to $X^j(V_p) = \omega(x, V_p)$. Thus, $x \in W^s(X^j(V_p))$. Therefore, $\overline{W}^s(X^s(V_p)) \subset \bigcup_{i \in I^+} W^s(X^i(V_p))$. By the proof of Lemma 2.3.18, $W^u(X^i(V_{p_0})) \cap W^s(X^s(V_{p_0})) \neq \emptyset$ for all $i \in I$. So, for each *i* there exists $T_i > 0$ such that $\phi_{T_i}(W^u_{\text{loc}}(X^i(V_{p_0}))) \cap W^s_{\text{loc}}(X^s(V_{p_0})) \neq \emptyset$. This intersection is trivially transverse since $W^s_{\text{loc}}(X^s(V_{p_0}))$ is codimension zero in M. Thus, $\{\phi_{T_i}(W^u_{\text{loc}}(X^i(V_p)))\}_{p \in J}$

and $\{W^s_{\text{loc}}(X^s(V_p))\}_{p\in J}$ are C^1 continuous families over J of compact embedded submanifolds with a point of transversal intersection, so shrinking J if necessary implies that $\phi_{T_i}(W^u_{\text{loc}}(X^i(V_p))) \cap W^s_{\text{loc}}(X^s(V_p)) \neq \emptyset$ for all $p \in J$ [KH99, Proposition A.3.16, Corollary A.3.18]. Hence, as I is finite, shrinking J if necessary gives that for all $i \in I$ and $p \in J$, $W^u(X^i(V_p)) \cap W^s(X^s(V_p)) \neq \emptyset$. By the proof of Theorem 2.2.12, this implies that for all $p \in J, \bigcup_{i \in I^+} W^s(X^i(V_p)) \subset \overline{W}^s(X^s(V_p))$.

Theorem 2.3.20 showed results for the family $\{\overline{W}^s(X^s(V_p))\}_{p\in J}$ in M. In contrast, Corollary 2.3.21 and the lemmas below will derive results for $\overline{W}^s(X_J^s)$ in $M \times J$, which a priori is potentially larger than $\sqcup_{p\in J}\overline{W}^s(X^s(V_p))$. It is $\overline{W}^s(X_J^s)$ that turns out to be relevant in the applications of interest, although the Corollary will show that under the assumptions of Theorem 2.3.20 they are equal. Lemma 2.3.36 shows that if $\overline{W}^s(X^s(V_{p_0}))$ is bounded, then for J sufficiently small, $p \in J$ and $x \in \overline{W}^s(X_J^s)$ implies that $\omega(x, V_p)$ is nonempty.

Lemma 2.3.36. Let J be a smooth manifold and let $\{V_p\}_{p\in J}$ be a strong C^1 continuous family of vector fields on M possessing a family of stable hyperbolic equilibria $\{X^s(p)\}_{p\in J}$. Suppose there exists $p_0 \in J$ such that V_{p_0} is complete, V_{p_0} satisfies Assumptions 2.3.4-2.3.12, and $\overline{W}^s(X^s(V_{p_0}))$ is bounded. Then for J open and sufficiently small, and such that $p_0 \in J$, for every $(x, p) \in \overline{W}^s(X^s_J)$, $\omega(x, V_p) \neq \emptyset$.

Proof of Lemma 2.3.36. By Theorem 2.3.19, for every $p \in J$ and every $x \in \overline{W}^s(X^s(V_p))$, $\omega(x, V_p) \neq \emptyset$. Assume towards a contradiction that there exists $(x, p) \in \overline{W}^s(X_J^s)$ with $\omega(x, V_p) = \emptyset$. By Theorem 2.3.19, there exists an open neighborhood U of $\overline{W}^s(X^s(V_{p_0}))$ in M such that U is bounded and, for J sufficiently small, $q \in J$ implies that $\overline{W}^s(X^s(V_q)) \subset U$. Hence, there exists $K \subset M$ compact such that $U \subset K$. As $\omega(x, V_p) = \emptyset$, there exists yin the forward orbit of x under V_p such that $y \in M - K$. As $\overline{W}^s(X^s(V_p))$ is V_p -invariant, $y \in \overline{W}^s(X^s(V_p))$. However, this contradicts that $\overline{W}^s(X^s(V_p)) \subset U \subset K$.

Lemma 2.3.37 extends the results of Lemma 2.3.33 to show that for J sufficiently small, $p \in J$ and $(x, p) \in \overline{W}^s(X_J^s)$ implies that $\omega(x, V_p)$ is nonempty.

Lemma 2.3.37. Let J be a smooth manifold and let $\{V_p\}_{p\in J}$ be a strong C^1 continuous family of vector fields on M possessing a family of stable hyperbolic equilibria $\{X^s(p)\}_{p\in J}$. Suppose there exists $p_0 \in J$ such that V_{p_0} is complete, V_{p_0} satisfies Assumptions 2.3.4-2.3.13, and V_{p_0} is a point of continuity of G_n^1 , G_n^2 , and G_n^3 for all n. Then for J open and sufficiently small, and such that $p_0 \in J$, for every $(x, p) \in \overline{W}^s(X_J^s)$, $\omega(x, V_p) \neq \emptyset$.
Proof of Lemma 2.3.37. First it is shown that for every $x \in \overline{W}^s(X_J^s) \cap (M \times \{p_0\}), \omega(x, V_{p_0}) \neq \emptyset$. This is proven by contradiction, which first assumes the existence of $(x, p_0) \in \overline{W}^s(X_J^s)$ such that $\omega(x, V_{p_0})$ is empty. Then continuity of G_n^1 at V_{p_0} will imply that $x \in \overline{W}^s(X^s(V_{p_0}))$, but this contradicts Lemma 2.3.33. Next, it is shown that for J sufficiently small, for every $(x, p) \in \overline{W}^s(X_J^s), \ \omega(x, V_p) \neq \emptyset$. This is again proven by obtaining a contradiction using the first claim above and Assumption 2.3.13, using an argument similar to the proof of Lemma 2.3.33.

First we claim that for every $x \in \overline{W}^s(X_J^s) \cap (M \times \{p_0\}), \ \omega(x, V_{p_0}) \neq \emptyset$. For, assume towards a contradiction that there exists $x \in \overline{W}^s(X_J^s) \cap (M \times \{p_0\})$ with $\omega(x, V_{p_0}) = \emptyset$. Then there exists n such that $x \in \operatorname{int} K_n$. Furthermore, by Lemma 2.3.33, $x \notin \overline{W}^s(X^s(V_{p_0}))$ since for every $y \in \overline{W}^s(X^s(V_{p_0})), \ \omega(y, V_{p_0}) \neq \emptyset$. As $(x, p_0) \in \overline{W}^s(X_J^s)$, there exist $(x_m, p_m) \in$ $W^s(X_J^s)$ with $(x_m, p_m) \to (x, p_0)$. As int K_n is open and $x_m \to x, m$ sufficiently large implies that $x_m \in \operatorname{int} K_n$. By continuity of G_n^1 at $V_{p_0}, \ x_m \in W^s(X^s(V_{p_m})) \cap \operatorname{int} K_n \subset G_n^1(V_{p_m})$ implies that $x \in G_n^1(V_{p_0})$. As $x \in \operatorname{int} K_n$, this implies that $x \in \overline{W}^s(X^s(V_{p_0}))$, which is a contradiction. So, we must have $\omega(x, V_{p_0}) \neq \emptyset$.

Next we claim that for J sufficiently small, for every $(x,p) \in \overline{W}^s(X_J^s), \ \omega(x,V_p) \neq \emptyset$. Assume towards a contradiction that no such open J containing p_0 exists. Then there must exist sequences $p_m \to p_0$ and $(x_m, p_m) \in \overline{W}^s(X_J^s)$ such that $\omega(x_m, V_{p_m}) = \emptyset$ for all m. As $M \cup \{\infty\}$ is compact, passing to a subsequence if necessary we have that $x_m \to x \in$ $M \cup \{\infty\}$. First suppose $x \in M$. As $(x_m, p_m) \in \overline{W}^s(X_J^s)$ closed and $(x_m, p_m) \to (x, p_0)$, $(x, p_0) \in \overline{W}^s(X_J^s)$. Choose n such that $x \in \operatorname{int} K_n$ and note that for m sufficiently large, $x_m \in \text{int } K_n$. By continuity of G_n^3 at V_{p_0} , $\omega(x_m, V_{p_m}) = \emptyset$ implies that $\omega(x, V_{p_0}) = \emptyset$. But, this contradicts the claim proved above. So, suppose $x = \infty$. Let O_m denote the orbit in both forwards and backwards time of x_m under V_{p_m} . Assume towards a contradiction that there exists a subsequence $\{O_{m_k}\}_{k=1}^{\infty}$ and $K' \subset M$ compact such that $O_{m_k} \cap K' \neq \emptyset$ for all k. Letting $y_{m_k} \in O_{m_k} \cap K'$ for all k implies that, since K' is compact, passing to a subsequence if necessary $y_{m_k} \to y \in K' \subset M$. Hence, as the ω limit set is invariant along an orbit, this reduces to the case of $x \in M$ above, which leads to a contradiction. So, there must exist a subsequence $\{O_{m_k}\}_{k=1}^{\infty}$ such that $O_{m_k} \cap K_{m_k} = \emptyset$ for all k. Let N be the neighborhood of Assumption 2.3.13 so that $\Omega(V_0) \cap N = \emptyset$ and no orbit under V_0 contained in N has both its ω and α limit sets under V_0 empty. As $V_{m_k} \to V_0$ in the strong C^1 topology, there exists $K \subset M$ compact such that V_{m_k} and V_0 agree outside K. Then for k sufficiently large, the closure of O_{m_k} is contained in $(M-K) \cap N$. As V_0 and V_{m_k} agree on M-K, this implies that O_{m_k} is also the orbit of x_{m_k} under V_0 , so that $\omega(O_{m_k}, V_0) = \omega(O_{m_k}, V_{m_k}) = \emptyset$. As $O_{m_k} \subset N$, this implies that $\alpha(O_{m_k}, V_0) \neq \emptyset$. As the closure of O_{m_k} is contained in N, $\alpha(O_{m_k}, V_0) \subset N$. However, since $\alpha(O_{m_k}, V_0)$ is nonempty and nonwandering under V_0 , this contradicts that $\Omega(V_0) \cap N = \emptyset$. Hence, $x_m \to x = \infty$ results in a contradiction. As $x \notin M$ and $x \neq \infty$, we obtain a contradiction to the main assertion above.

Lemma 2.3.38 extends the results of Lemma 2.3.34 to show that for J sufficiently small, $p \in J$ implies that the nonwandering points under V_p in $M \times \{p\}$ that are contained in $\overline{W}^s(X_J^s)$ are precisely $\bigcup_{i \in I^+} X^i(V_p)$.

Lemma 2.3.38. Let J be a smooth manifold and let $\{V_p\}_{p\in J}$ be a strong C^1 continuous family of vector fields on M possessing a family of stable hyperbolic equilibria $\{X^s(p)\}_{p\in J}$. Suppose there exists $p_0 \in J$ such that V_{p_0} is complete, V_{p_0} satisfies Assumptions 2.3.4-2.3.13, and V_{p_0} is a point of continuity of G_n^1 , G_n^2 , and G_n^3 for all n. Then for J open and sufficiently small, and such that $p_0 \in J$, for every $p \in J$, $\Omega(V_p) \cap (\overline{W}^s(X_J^s) \cap (M \times \{p\})) = \bigcup_{i \in I^+} X^i(V_p)$.

Proof of Lemma 2.3.38. First it is shown that $\Omega(V_{p_0}) \cap (\overline{W}^s(X_J^s) \cap (M \times \{p_0\})) = \bigcup_{i \in I^+} X^i(V_{p_0})$. This is proven by contradiction, which first assumes the existence of $(x, p_0) \in \overline{W}^s(X_J^s)$ such that x is nonwandering under V_{p_0} and x is not in $\bigcup_{i \in I^+} X^i(V_{p_0})$. Then continuity of G_n^1 at V_{p_0} will imply that $x \in \overline{W}^s(X^s(V_{p_0}))$, but this contradicts Lemma 2.3.34. Next, it is shown that for J sufficiently small, $p \in J$ implies that $\Omega(V_p) \cap (\overline{W}^s(X_J^s) \cap (M \times \{p\})) = \bigcup_{i \in I^+} X^i(V_p)$. This is again proven by obtaining a contradiction using the first claim above, Lemma 2.3.37, and Assumption 2.3.13, using an argument similar to the proof of Lemma 2.3.34.

First we claim that $\Omega(V_{p_0}) \cap (\overline{W}^s(X_J^s) \cap (M \times \{p_0\})) = \bigcup_{i \in I^+} X^i(V_{p_0})$. For, assume towards a contradiction that there exists $x \in \Omega(V_{p_0})$ with $x \notin \bigcup_{i \in I^+} X^i(V_{p_0})$ and such that $(x, p_0) \in \overline{W}^s(X_J^s)$. Then there exists n such that $x \in \operatorname{int} K_n$. Furthermore, by Lemma 2.3.34, $x \notin \overline{W}^s(X^s(V_{p_0}))$ since $\Omega(V_{p_0}) \cap \overline{W}^s(X^s(V_{p_0})) = \bigcup_{i \in I^+} X^i(V_{p_0})$. As $(x, p_0) \in \overline{W}^s(X_J^s)$, there exist $(x_m, p_m) \in W^s(X_J^s)$ with $(x_m, p_m) \to (x, p_0)$. Then m sufficiently large implies $x_m \in$ int K_n . Hence, $x_m \in G_n^1(V_{p_m})$ for m sufficiently large. Since G_n^1 is continuous at V_{p_0} , $V_{p_m} \to V_{p_0}, x_m \to x$, and $x_m \in G_n^1(V_{p_m})$ for m sufficiently large, $x \in G_n^1(V_{p_0}) \cap \operatorname{int} K_n$. Hence, $x \in \overline{W}^s(X^s(V_{p_0}))$, which is a contradiction. So, we must have $\Omega(V_{p_0}) \cap (\overline{W}^s(X_J^s) \cap (M \times \{p_0\})) = \bigcup_{i \in I^+} X^i(V_{p_0})$.

Next we claim that for J sufficiently small, for every $p \in J$, $\Omega(V_p) \cap (\overline{W}^s(X_J^s) \cap (M \times \{p\})) = \bigcup_{i \in I^+} X^i(V_p)$. Assume towards a contradiction that no such open J containing p_0 exists. Then there must exist sequences $p_m \to p_0$ and $(x_m, p_m) \in \overline{W}^s(X_J^s)$ such that $x_m \in \Omega(V_{p_m})$ but $x_m \notin \bigcup_{i \in I^+} X^i(V_{p_m})$. By Lemma 2.3.37, for every $(x, p) \in \overline{W}^s(X_J^s)$, $\omega(x, V_p) \neq \emptyset$. In particular, $\omega(x_m, V_{p_m}) \neq \emptyset$ for all m. As $V_{p_m} \to V_{p_0}$ in the strong C^1

topology, there exists K compact such that for m sufficiently large, V_{p_m} and V_{p_0} agree on M-K. By Assumption 2.3.13, there exists a neighborhood N of ∞ such that $\Omega(V_{p_0}) \cap N = \emptyset$. Then, by the same argument as in the proof of Lemma 2.3.34, $\omega(x_m, V_{p_m}) \cap (K \cup (M-N)) \neq \emptyset$ for all m. As $K \cup (M-N)$ is compact, there exists n such that $K \cup (M-N) \subset \operatorname{int} K_{n-1}$. As $\omega(x_m, V_{p_m}) \cap \operatorname{int} K_{n-1} \neq \emptyset$, int K_{n-1} is open, and $\omega(x_m, V_{p_m})$ is in the closure of the forward orbit of x_m under V_{p_m} , there exists y_m a point in the forward orbit of x_m under V_{p_m} such that $y_m \in \text{int } K_{n-1}$ for each m. Note that $(y_m, p_m) \in \overline{W}^s(X_J^s)$ and is nonwandering under V_{p_m} since x_m is and both $\overline{W}^s(X_J^s)$ and $\Omega(V_{p_m})$ are invariant. Furthermore, $y_m \notin \bigcup_{i \in I^+} X^i(V_{p_m})$ as x_m is not and these sets are also invariant. As K_{n-1} is compact and $y_m \in K_{n-1}$ for all m, passing to a subsequence if necessary we have that $y_m \to y \in K_{n-1} \subset \text{int } K_n$. As $(y_m, p_m) \in \overline{W}^s(X_J^s)$ closed and $(y_m, p_m) \to (y, p_0), (y, p_0) \in \overline{W}^s(X_J^s)$. Note that $V'_m \to V_{p_0}$ with respect to the strong C^1 topology. Then, by the same argument as in the proof of Lemma 2.3.34, there exists a sequence $\{V'_m\}_{m=1}^{\infty}$ of vector fields on M such that V_{p_m} and V'_m agree on $M - (K \cup (M - N))$, are $(1/m) C^1$ -close on $(K \cup (M - N))$, and such that there exist points z_m with distance less than (1/m) from y_m such that z_m is periodic under V'_m and $z_m \notin \bigcup_{i \in I^+} X^i(V'_m)$. Then, by the same arguments as in the proof of Lemma 2.3.34, we have that y is periodic under V_{p_0} . As $(y, p_0) \in \overline{W}^s(X_J^s)$ and since it was shown above that $\Omega(V_{p_0}) \cap (\overline{W}^s(X_J^s) \cap (M \times \{p_0\})) = \bigcup_{i \in I^+} X^i(V_{p_0})$, we must have $y \in X^j(V_{p_0})$ for some $j \in I$. Let O_m denote the orbit of z_m under V'_m . Then, by the same argument as in the proof of Lemma 2.3.34, passing to a subsequence if necessary we have that $O_m \to X^j(V_{p_0})$ with respect to the Hausdorff metric. Finally, by the same argument as in the proof of Lemma 2.3.34, there exists an open neighborhood N_i of $X^j(V_{p_0})$ and an open neighborhood \hat{W} of V_{p_0} in the strong C^1 topology such that for every $V \in \hat{W}$, N_j does not entirely contain any equilibrium points or periodic orbits of V. As $V_{p_0} \in \hat{W}$ and $V'_m \to V_{p_0}$ with respect to the strong C^1 topology, m sufficiently large implies that $V'_m \in \hat{W}$. As $O_m \to X^j(V_0) \subset N_j$, m sufficiently large implies that $O_m \subset N_j$. As O_m is periodic and disjoint from $\bigcup_{i \in I^+} X^i(V'_m)$, this contradicts the choice of \hat{W} and N_i .

Proof of Corollary 2.3.21. The conclusions of the corollary will follow immediately from Theorem 2.2.23 and Corollary 2.2.24 once the assumptions behind them are verified. In particular, Assumptions 2.2.14-2.2.15 and Assumptions 2.2.17-2.2.18 must be verified.

Let U be the neighborhood of Lemma 2.3.18. Then $\Omega(V_{p_0}) \cap U = \bigcup_{i \in I^+} X^i(V_{p_0})$. Let U' be an open neighborhood of $\overline{W}^s(X^s(V_{p_0}))$ in $M \cup \{\infty\}$ such that $\overline{U}' \subset U$. Then by Lemma 2.3.35, for $p \in J$, $\overline{W}^s(X^s(V_p)) \subset U'$. Thus, $W^s(X_J^s) \subset U' \times J$, so $\overline{W}^s(X_J^s) \subset \overline{U}' \times J \subset U \times J$. In particular, U is an open neighborhood of $\partial W^s(X_J^s) \cap (M \times \{p_0\})$ in $(M \times \{p_0\})$.

As $\Omega(V_{p_0}) \cap U$ consists of a finite union of hyperbolic critical elements, Assumptions 2.2.14-2.2.15 are satisfied. By Assumption 2.3.6, the stable and unstable manifolds of each pair of critical elements in $\partial W^s(X_J^s) \cap (M \times \{p_0\})$ are transverse, satisfying Assumption 2.2.18. By Lemma 2.3.38, for every $p \in J$, $\Omega(V_p) \cap (\overline{W}^s(X_J^s) \cap (M \times \{p\})) = \bigcup_{i \in I^+} X^i(V_p)$. By Lemma 2.3.37, for every $(x, p) \in \overline{W}^s(X_J^s)$, $\omega(x, V_p) \neq \emptyset$. As the orbit of x under V_p is contained in $\overline{W}^s(X_J^s)$ closed and invariant, $\omega(x, V_p) \subset \overline{W}^s(X_J^s)$. But, $\omega(x, V_p)$ nonempty and nonwandering implies by the above that $\omega(x, V_p) \subset \bigcup_{i \in I^+} X^i(V_p)$. As $\bigcup_{i \in I^+} X^i(V_p)$ is compact and $\omega(x, V_p)$ is closed, $\omega(x, V_p)$ is compact, so it is also connected. Thus, $\omega(x, V_p) = X^j(V_p)$ for some $j \in I^+$. The above verifies Assumption 2.2.17.

2.3.6 Conclusion

This section completed the final step to the proof of RoA boundary continuity under parameter perturbation for vector fields which are in essence Morse-Smale along the RoA boundary. This step was to show that these assumptions imply that the RoA boundary is Ω -stable. In particular, for compact Riemannian manifolds, it was shown that if the vector field is Morse-Smale along the RoA boundary then under sufficiently small strong C^1 perturbations the RoA boundary is Ω -stable. Initially, Example 2.3.3 showed that in Euclidean space the RoA boundary can vary discontinuously even for a globally Morse-Smale vector field under a strong C^{∞} perturbation. This was a counterexample to [CC95, Theorem 5.4], which claimed that under weaker assumptions the nonwandering set in the RoA boundary could not grow under strong C^1 perturbations. Hence, for Euclidean space it is necessary to make additional generic assumptions, and assumptions regarding the behavior near infinity, along with that the vector field is Morse-Smale along the RoA boundary, to show that the RoA boundary is Ω -stable. These results, together with the results of Section 2.2, imply that under these assumptions, the RoA boundary varies continuously under sufficiently small changes in parameter values.

2.4 Extensions to Vector Fields with Event-Selected Discontinuities

2.4.1 Introduction

Practical models in many application areas, including power systems, involve switching events, which lead to vector fields that are not C^1 . In power systems, these may include

clipping limits on control devices and switching due to tap changing transformers. Furthermore, it has been observed that such switching is fundamentally coupled with loss of system recoverability in practice [His04, VJSZ94]. The purpose of this section is to extend the RoA boundary continuity results of Section 2.2 to the setting of a large class of nonsmooth vector fields, closely related to what have been called vector fields with C^1 event-selected discontinuities [BSKR15], which can exhibit various forms of switching. In turn, this will make it possible to provide theoretical justification for the extension of numerical algorithms for computing critical parameter values to systems exhibiting switching. In particular, Sections 2.5-2.6 provide justification for algorithms for vector fields with C^1 event-selected discontinuities, and the algorithms developed in Chapter III can be applied to these nonsmooth systems as well. Vector fields with event-selected discontinuities have received attention recently due to their applicability to many physical and engineering system models which possess a finite number of triggering hypersurfaces where the vector field is discontinuous.

A sketch of a proof was made in [VJSZ94] for a decomposition of the boundary of the RoA for systems with clipping limits and fixed parameter values. In that work, the vector field was locally Lipschitz. Generalizations to systems exhibiting switching, or any discontinuous vector fields, have not been made even in the case of constant parameters. Therefore, the classification of the boundary of the RoA for vector fields with C^1 event-selected discontinuities, presented here, may be of interest for the fixed parameter case, as well as for its behavior under small variations in parameter. The primary application, though, is the theoretical justification of algorithms for determining the recovery boundary.

The section is organized as follows. Section 2.4.2 gives a motivating example. Section 2.4.3 provides some dynamical systems background. Then Section 2.4.4 discusses the main results. Section 2.4.5 provides a sketch of the key proofs, although some are nearly identical to the corresponding results in Section 2.2 and, therefore, omitted for brevity. Finally, Section 2.4.6 offers some concluding remarks.

2.4.2 Example

Example 2.4.1. The example below serves to illustrate a mechanism whereby the RoA boundary can fail to vary continuously for a parameterized vector field, and how this can lead to situations where no critical parameter value exists. In this case, it will not be possible to determine the recovery boundary by computation of critical parameter values, so the algorithms described in Chapter III would not be successful. The theory developed in subsequent sections will prove that this example is not typical; in particular, a large class of

hybrid dynamical systems show continuity of their RoA boundaries under small changes in parameter values, thereby ensuring that the recovery boundary consists of critical parameter values (see Section 2.5).

Let J = (-1, 1) and for $p \in J$ define V_p as a vector field on \mathbb{R}^2 piecewise as follows. Let $r := r(x, y) = \sqrt{x^2 + y^2}$. Let $\mathcal{S}_{(a,b)} : \mathbb{R} \to \mathbb{R}$ be a smooth bump function with $\mathcal{S}^{-1}(1) = [0, a]$ and $\mathcal{S}^{-1}(0) = [b, \infty)$ (see [Lee13, Lemma 2.21] for a specific example). For $|r| \leq 1$, let $V_p(x, y) = (-x, -y) - p \frac{\mathcal{S}_{(2.5,3)}(r)}{r}(x, y)$, and for |r| > 1, let $V_p(x, y) = \frac{\mathcal{S}_{(1,2)}(r)}{2r}(-x - y, x - y) - p \frac{\mathcal{S}_{(2.5,3)}(r)}{r}(x, y)$. This family of vector fields are each piecewise C^1 with a switching surface on the circle |r| = 1. For $p \in J$, let $s_p(x, y) = (r-1)^2$. Then $s_p^{-1}(0) = \{(x, y) : |r(x, y)| = 1\}$. Thus, for each $p \in J$, $\{V_p, s_p\}$ defines a vector field with event-selected C^1 discontinuities, as will be defined below. Furthermore, $\{V_p, s_p\}_{p \in J}$ is a strong C^1 continuous family of event-selected C^1 vector fields, also defined below.

Figure 2.12 shows the vector field V_p for p = 0.3. It has a stable equilibrium point at the origin whose RoA is the open ball of radius r = 3 with boundary the circle of radius r = 3. This qualitative picture, in particular the RoA and its boundary, remain the same for any p > 0. The red and green curves in the figure show the images of two sets of parameter dependent initial conditions (ICs). The ICs corresponding to p = 0.3 are shown as red and green circles, and the ICs move along the red and green lines as p decreases. Note that both the red and green ICs lie inside the RoA for p = 0.3 (and any p > 0).

Figure 2.13 shows the vector field V_p for p = 0. Note that the RoA of its stable equilibrium point is the open ball of radius r = 2, with boundary the circle of radius r = 2. Here the red circle, denoting the red IC, lies inside the RoA whereas the green circle, denoting the green IC, lies outside the RoA.

Figure 2.14 shows the vector field V_p for p = -0.3. Note that the RoA of its stable equilibrium point is the open ball of radius r = 2, with boundary the circle (now a periodic orbit) of radius r = 2. Here the red circle, denoting the red IC, lies on the boundary of the RoA, and hence p = -0.3 is a critical parameter value for the red ICs, whereas the green circle, denoting the green IC, lies outside the RoA.

In summary, the red ICs intersect the RoA boundary at p = -0.3, so that p = -0.3 is a critical parameter value for them. However, the green ICs pass from inside the RoA (for p > 0) to outside the RoA (for $p \le 0$) without ever lying on the RoA boundary. So, there is no critical parameter value for the green ICs. This is possible because the RoA boundary varies discontinuously at p = 0. The theory developed in this paper will prove that discontinuous variation of the RoA boundary under small parameter variation is not possible for a large



Figure 2.12: The vector field V_p for p = 0.3. The red and green lines show two sets of parameter dependent initial conditions, with the initial conditions for p = 0.3 shown as red and green circles.

class of practical hybrid dynamical systems, that these systems do possess critical parameter values for a particular disturbance, and that a previously designed algorithm can be used to numerically compute those critical values.

Example 2.4.2. The example shown here motivates the need to employ a slightly different definition of vector fields with C^1 event-selected discontinuities than in [BSKR15]. In particular, the definition in [BSKR15] allows switching surfaces to have nontranversal intersection. Under strong C^1 perturbations, as considered here, it is possible for switching surfaces to pull apart, and this could result in a vector field which is no longer event-selected discontinuous. This is problematic for developing the RoA boundary continuity results that are desired, because arbitrarily small variations in parameter values could lead to loss of existence and/or uniqueness of the flow, which complicates the possible behaviors of the RoA boundary. Therefore, to avoid this possibility, we will make the generic assumption below that the switching surfaces intersect transversely.

Figure 2.15 shows an example of a vector field with C^{∞} event-selected discontinuities that has a pair of nontransversal switching surfaces. The example is defined by the following



Figure 2.13: The vector field V_p for p = 0. The red and green lines show two sets of parameter dependent initial conditions, with the initial conditions for p = 0 shown as red and green circles.



Figure 2.14: The vector field V_p for p = -0.3. The red and green lines show two sets of parameter dependent initial conditions, with the initial conditions for p = -0.3 shown as red and green circles.



Figure 2.15: The vector field V of Example 2.4.2 is shown. It shows how a vector field with C^{∞} event-selected discontinuities can become no longer event-selected discontinuous under strong C^{∞} perturbations due to the presence of two nontransversal switching surfaces (black curves).

two smooth vector fields:

$$\omega = 2\pi/6 \tag{2.14}$$

$$V_1(x,y) = (0,1) \tag{2.15}$$

$$V_2(x,y) = (\sin(\omega(x+3)), \cos(\omega(x+3))).$$
(2.16)

(2.17)

There are two switching surfaces, shown as black lines, in the example. Define the rough vector field V as follows. Above the top switching surface and below the bottom switching surface, let $V = V_1$. In between the two switching surfaces, let $V = V_2$. Note that V is a vector field with C^{∞} event-selected discontinuities and, therefore, possess a global flow which exists, is unique, and is piecewise C^{∞} . However, under arbitrarily small C^{∞} perturbations to the top switching surface, the two switching surfaces separate along a portion of their tangency, the resulting vector field is no longer event-selected discontinuous, and it exhibits loss of existence and uniqueness of local flows. This motivates the need to assume below that the switching surfaces are transverse.

2.4.3 Background

We will assume that all vector fields, C^1 or not, are defined on $M := \mathbb{R}^n$ for some n > 0. First we review some terminology of nonlinear systems with a C^1 vector field. Let $J \subset \mathbb{R}$ be an open interval containing p_0 and let $\{V_p\}_{p \in J}$ be a parameterized family of vector fields on M. Define V a vector field on $M \times J$ by $V(x, p) = (V_p(x), 0)$. Let ϕ denote the flow of Vsuch that $\phi(t, x, p)$ is the flow of the system from initial condition x under the vector field V_p for a time t. Let $\phi_{(t,p)} : M \to M$ by $\phi_{(t,p)}(x) = \phi(t, x, p)$. If $X(p) \subset M$ is an equilibrium point or periodic orbit of V_p for some $p \in J$, we call X(p) a critical element of V_p .

An equilibrium point X(p) of V_p is hyperbolic if $d(\phi_{(1,p)})_{X(p)}$ has no imaginary eigenvalues. If X(p) is a periodic orbit and $x \in X(p)$ then there exists a hypersurface $S \supset \{x\}$ and a C^1 map $\hat{\tau} : S \to S$ such that $\hat{\tau}$ is the Poincaré first return map. The periodic orbit X(p) is hyperbolic if $d\hat{\tau}_x$ has no imaginary eigenvalues. For a hyperbolic critical element, let $n^u(X(p_0))$ denote the unstable dimension of $T_{X(p_0)}M$, and let $n^s(X(p_0))$ denote the stable dimension of $T_{X(p_0)}M$. A hyperbolic critical element X(p) possesses local stable and unstable manifolds, denoted $W^s_{\text{loc}}(X(p))$ and $W^u_{\text{loc}}(X(p))$, respectively, such that the flow restricted to $W^s_{\text{loc}}(X(p))$ is a contraction in backwards time. The stable and unstable manifolds, $W^s(X(p))$ and $W^u_{\text{loc}}(X(p))$ and $W^s_{\text{loc}}(X(p))$ and $W^u_{\text{loc}}(X(p))$ and $W^u_{\text{loc}}(X(p))$ and $W^s_{\text{loc}}(X(p))$ and $W^u_{\text{loc}}(X(p))$ and $W^u_{\text{loc}}(X(p))$ and $W^u_{\text{loc}}(X(p))$ and $W^u_{\text{loc}}(X(p))$ and $W^u_{\text{loc}}(X(p))$ and $W^u_{\text{loc}}(X(p))$ and forwards in time, respectively. The RoA of a stable hyperbolic equilibrium point is its stable manifold, and the RoA boundary is the boundary of its stable manifold.

A point $x \in M$ is wandering under V_p if there exists an open neighborhood U of x in M and some T > 0 such that |t| > T implies that $\phi(t, U, p) \cap U = \emptyset$. A point $x \in M$ is nonwandering if it is not wandering. Let $\Omega(V_p)$ denote the set of all nonwandering points of V_p . These include all critical elements of V_p . A pair of C^1 immersed submanifolds X and Y are transverse if for every $q \in X \cap Y$, $T_qX + T_qY = T_qM \cong M$. A Morse-Smale vector field \tilde{V} is a C^1 vector field such that $\Omega(\tilde{V})$ is equal to a finite union of hyperbolic critical elements whose stable and unstable manifolds intersect transversely [Sma60].

Let $C^1(M, \tilde{M})$ denote C^1 maps from M to \tilde{M} C^1 manifolds. The strong C^1 topology on $C^1(M, \tilde{M})$ is defined in [Hir76, Chapter 2]. A parameterized family of vector fields $\{V_p\}_{p \in J}$ is strong C^1 continuous if the map $p \to V_p$ is continuous as a map from J to $C^1(M, TM)$ equipped with the strong C^1 topology. In particular, if $V_p \to \tilde{V}$ as $p \to \hat{p}$ in the strong C^1 topology then there exists $K \subset M$ compact such that for any $\epsilon > 0$, p sufficiently close to \hat{p} implies that V_p and \tilde{V} agree outside of K and that V_p and \tilde{V} along with their first derivatives are ϵ -close on K. A property is generic for vector fields on M if the set of

vector fields possessing this property contains a countable intersection of open, dense sets in $C^1(M, TM)$. A family of critical elements $\{X(p)\}_{p\in J}$ of $\{V_p\}_{p\in J}$ is C^1 continuous if there exists a C^1 function $\tilde{F} : X(p_0) \times J \to M$ such that $\tilde{F}|_{X(p_0) \times \{p\}}$ is injective onto its image X(p) for all $p \in J$. We say $\{E_p\}_{p\in J}$ is a Chabauty continuous family of subsets of \mathbb{R}^n if for every $p_0 \in J$ and every $\{p_n\}_{n=1}^{\infty} \subset J$ such that $p_n \to p_0$, for every $x \in E_{p_0}$ there exists a sequence $\{x_n\}_{n=1}^{\infty}$ with $x_n \in E_{p_n}$ and $x_n \to x$, and every sequence $\{x_n\}_{n=1}^{\infty}$ with $x_n \in E_{p_n}$ has all of its limit points contained in E_{p_0} .

Next we introduce some additional notation. For $Q \subset J$, write $X(Q) = \bigcup_{p \in Q} X(p) \subset M$ and $X_Q = \bigcup_{p \in Q} X_p \subset M \times J$. For any set E, let $\sqcup_{p \in J} E := E \times J$. If $D \subset M$, let $D_{\epsilon} := \{x \in M : d(x, D) < \epsilon\}$ where d is the Euclidean distance.

In order to incorporate the effects of clipping limits and switching, a large class of hybrid dynamical systems, which we will call vector fields with event-selected C^1 discontinuities although their definition in [BSKR15] is slightly more general, is employed. This class of vector fields, which is defined below, has several desirable properties including the existence of a global flow which is piecewise C^1 .

A rough vector field is a vector field that is not assumed to be C^1 or even C^0 ; this includes vector fields exhibiting limits or switching behavior, but also includes C^1 vector fields. The following definition of event functions, whose zero level sets will be the points of discontinuity of the vector field and represent switching surfaces, ensures that the flow of the vector field will cross switching surfaces transversely. This definition rules out grazing of the switching surfaces and Zeno type behavior.

Definition 2.4.3. Let V be a rough vector field on M, $s \in C^1(M, \mathbb{R}^m)$, and U an open set in M. Then define it to be an event or switching function (s) for V on U if for each component s_j of s, either $s_j(U)$ is disjoint from zero or, if not, then there exists c > 0 such that $d(s_j)_x(V(x)) \ge c$ for all $x \in U$. Such a U is referred to as an event neighborhood of V.

Note that the latter part of the definition does not restrict the sign of the event functions in the sense that if an event function s^j satisfied $d(s^j)_y(V(y)) \leq -c^j < 0$ for all $y \in U^j$, then defining the new event function $-s^j$ would satisfy the definition given above.

Let $B_m = \{-1, +1\}^m$ be *m* copies of $\{-1, +1\}$, meant to denote whether each event function is positive or negative to indicate the status of each switching event. For each $j \in \{1, ..., m\}$, define the switching surface corresponding to s^j by letting $Z_j = (s^j)^{-1}(0)$. Let $Z = \bigcup_{j=1}^m Z_j$ be the union of the switching surfaces. Let $\sigma_j : M \to \{-1, +1\}$ that sends x to the sign of $s^j(x)$, or to 1 if $s^j(x) = 0$. Define a function that gives the switching state status (σ), where $\sigma : M \to \{-1, +1\}^m$, by the product of σ_j for $j \in \{1, ..., m\}$. Then $\sigma(x)$ denotes the switching states of the system at x.

The following definition of event-selected C^1 ensures both that the flow encounters switching surfaces transversely and that the flow through points of intersection of multiple switching surfaces is well-defined and piecewise C^1 .

Definition 2.4.4. Let V be a rough vector field on M and $s \in C^1(M, \mathbb{R}^m)$. Then V and s determine an event-selected C^1 vector field if there exists a set of C^1 vector fields $\{V^b\}_{b\in B_m}$ on M, called selection functions, and the following conditions are satisfied:

- 1. (event functions) for every $x \in M$, there exists an open neighborhood $U_x \subset M$ containing x such that s is an event function for V on U_x .
- 2. (C¹ extension) for every $x \in M$, there exists an open neighborhood $U_x \subset M$ containing x such that for every $b \in B_m = \{-1, +1\}^m$, with

$$D^b = \{ y \in U_x : \sigma(y) = b \},\$$

 $V|_{\text{int }D^b} = V^b|_{\text{int }D^b}.$

3. (transversality) for every $j, j' \in \{1, ..., m\}, Z_j$ and $Z_{j'}$ are transverse.

The definition of vector fields with C^1 event-selected discontinuities ensures that orbits of the vector field encounter switching surfaces transversely, and that the vector fields on each side of a switching surface are consistent in such a way that local flows exist and are unique. However, it is important to note that Definition 2.4.4 differs from the original definition in [BSKR15]. Nevertheless, the event functions and C^1 extension parts of Definition 2.4.4 imply that any vector field satisfying Definition 2.4.4 must also satisfy the definition of eventselected discontinuous in [BSKR15]. In particular, the event functions and C^1 extension parts of Definition 2.4.4 require that the vector field is event-selected discontinuous at $x \in M$, and as this holds for all $x \in M$, the definition in [BSKR15] is satisfied. Hence, Definition 2.4.4 is a special case of the definition in [BSKR15], so all the results developed for the class of vector fields considered in [BSKR15] apply to vector fields satisfying Definition 2.4.4.

In addition to the definition in [BSKR15], Definition 2.4.4 includes an additional transversality condition which is required to ensure that the vector field remains event-selected C^1 under perturbations to the vector field. Example 2.4.2 showed that, without assuming transversality of the switching surfaces, arbitrarily small C^{∞} perturbations could cause the vector field to no longer be event-selected C^1 , and to no longer have existence of a local flow. The problem is that, for some $b \in B_m = \{-1, +1\}^m$, it is possible that int D^b is empty, but then, under perturbation, int D^b may suddenly become nonempty, and the selection function V^b may be inconsistent with the selection functions of the neighboring regions. The transversality condition ensures that, under perturbations to the vector field, no region int D^b goes from being empty to being nonempty. It turns out this is the key property required to ensure persistence of C^1 event-selected vector fields under perturbation.

An example of an event-selected C^1 vector field was shown in Section 2.3.3 with event function $s_1(x, y) = (\sqrt{(x^2 + y^2)} - 1)^2$. By [BSKR15, Corollary 1], if V and s determine an event-selected C^1 vector field over M then there exists a piecewise C^1 global flow.

Definition 2.4.5. A piecewise C^1 immersed submanifold is a topological manifold $T \subset M$ together with a C^1 immersed submanifold, denoted \tilde{T} , in $M \setminus Z$ which is an open and dense topological submanifold of T.

Definition 2.4.6. Two piecewise C^1 immersed submanifolds T and T' are transverse if \tilde{T} and \tilde{T}' are transverse in $M \setminus Z$.

If $\{s_p\}_{p\in J}$ is a family of C^1 functions from M to \mathbb{R}^m , for each $p \in J$ we define $\sigma_p : M \to \{0,1\}^m$ such that $(\sigma_p)_j$ sends x to the sign of $(s_p(x))_j$, or to 1 if $(s_p(x))_j = 0$, for all $j \in \{1, ..., m\}$.

Definition 2.4.7. Let $\{V_p\}_{p\in J}$ be a family of rough vector fields on M and let $\{s_p\}_{p\in J}$ be a family of C^1 functions from M to \mathbb{R}^m . We say that $\{V_p, s_p\}_{p\in J}$ is a strong C^1 continuous family of event-selected C^1 vector fields on M if there exists a collection $\{V_p^b\}_{b\in B^m, p\in J}$ such that V_p^b is a C^1 vector field on M for all $b \in B^m$ and $p \in J$, and the following conditions are satisfied:

- 1. (event-selected at one parameter value) there exists $p_0 \in J$ such that V_{p_0} and s_{p_0} determine an event-selected discontinuous C^1 vector field on M with the selection functions $\{V_{p_0}^b\}_{b\in B^m}$.
- 2. (continuity of event functions) $\{s_p\}_{p \in J}$ is a strong C^1 continuous family of functions in $C^1(M, \mathbb{R}^m)$.
- 3. (continuity of selection functions) for each $b \in B^m$, $\{V_p^b\}_{p \in J}$ is a strong C^1 continuous family of C^1 vector fields on M.

4. (consistency of vector fields) for all $p \in J$ and $b \in B_m$, with

$$D_p^b = \{ y \in M : \sigma_p(y) = b \},\$$

 $V|_{\text{int } D^b_p} = V^b_p|_{\text{int } D^b_p}.$

Motivating the terminology of Definition 2.4.7, we will see in Lemma 2.4.15 that if $\{V_p, s_p\}_{p \in J}$ is a strong C^1 continuous family of event-selected C^1 vector fields on M then for J sufficiently small, (V_p, s_p) determines an event-selected C^1 vector field for each $p \in J$. Define V a rough vector field on $M \times J$ by $V(x, p) = (V_p(x), 0)$. Then Lemma 2.4.15 will show that for J sufficiently small, V has a piecewise C^1 flow.

For $j \in \{1, ..., m\}$ and $p \in J$, define the event or switching function j corresponding to parameter value $p(s_p^j)$ and the switching surface j corresponding to parameter value $p(Z_j^p)$, where $Z_j^p = (s_p^j)^{-1}(0)$. For $p \in J$ define $\bigcup_{j=1}^m Z_j^p(Z^p)$. For $j \in \{1, ..., m\}$ define $\sqcup_{p \in J} Z_j^p(Z_j)$. Finally, define $\sqcup_{p \in J} Z^p(Z^J)$ and $(M \times J) \setminus Z^J(C^J)$.

2.4.4 Results

Let $J \subset \mathbb{R}$ be an open interval and let $p_0 \in J$. Let $\{V_p, s_p\}_{p \in J}$ be a strong C^1 continuous family of event-selected C^1 vector fields on M. Let $X^s(p_0)$ be a stable hyperbolic equilibrium point of V_{p_0} which lies in an open neighborhood on which V_{p_0} is C^1 . For J sufficiently small we make the following assumptions:

Assumption 2.4.8. Every equilibrium point of V_{p_0} is disjoint from Z^{p_0} .

Assumption 2.4.9. There exists a neighborhood N of $\pi_M(\partial W^s(X_J^s) \cap M_{p_0})$ such that $\Omega(V) \cap N$ consists of a finite union of critical elements of V_{p_0} ; call them $\{X^i(p_0)\}_{i=1}^k$. Shrink V if necessary so that $X_{p_0}^i \subset \partial W^s(X_J^s)$ for every $i \in I := \{1, ..., k\}$.

Assumption 2.4.10. Let $\gamma \subset \partial W^s(X_J^s)$ be an orbit. Then γ converges to X^i for some $i \in I$.

Assumption 2.4.11. With respect to V_{p_0} , every equilibrium point $X^i(p_0)$ is hyperbolic in the sense that it is hyperbolic in its neighborhood on which V_{p_0} is C^1 . For any $X^i(p_0)$ a periodic orbit, we will see below that it possesses a point $x \in M \setminus Z^{p_0}$ and a C^1 cross section S with a C^1 first return map with x as a fixed point. Then for any periodic orbit $X^i(p_0)$, assume that x is a hyperbolic fixed point of its first return map. **Assumption 2.4.12.** We will see below that every $X^i(p_0)$ possesses stable and unstable manifolds, call them $W^s(X^i(p_0))$ and $W^u(X^i(p_0))$, respectively, which are piecewise C^1 immersed submanifolds. For every $i, j \in I$, $W^u(X^i(p_0))$ and $W^s(X^j(p_0))$ are transversal in M.

Assumptions 2.4.9, 2.4.11, and 2.4.12 ensure that V_{p_0} is Morse-Smale along

 $\partial W^s(X_J^s) \cap M \times \{p_0\}$. Assumpton 2.4.10 ensures that no trajectories in $\partial W^s(X_J^s)$ escape to infinity, which is required since M is not compact, and that no new nonwandering elements enter $\partial W^s(X_J^s)$ for $p \in J$, although we will see that the perturbations of the critical elements $\{X^i(p_0)\}_{i\in I}$ will be contained in the boundary. Assumptions 2.4.8, 2.4.11, and 2.4.12 are generic. For the components on which V_{p_0} is C^1 , it is generically true that $\Omega(V_{p_0})$ is equal to the closure of the union of critical elements of V_{p_0} . However, it is not generic that there exists a neighborhood V containing a finite number of critical elements, so Assumption 2.4.9 is not generic.

Theorem 1 shows that the boundary of the RoA is equal to the union over $p \in J$ of the family of boundaries of the RoAs, and that the decomposition of the boundary of the RoA into a union of stable manifolds persists for $p \in J$. Corollary 1 then states that the family of boundaries varies continuously with $p \in J$.

Theorem 2.4.13. For J sufficiently small, $\partial W^s(X_J^s) = \bigsqcup_{p \in J} \partial W^s(X^s(p)) = \bigcup_{i \in I} W^s(X_J^i)$.

Corollary 2.4.14. $\{\partial W^s(X^s(p))\}_{p\in J}$ is a Chabauty continuous family of subsets of M.

The proofs of these results rely on a large number of technical lemmas. These are provided in the next section, with many proofs omitted because they are nearly identical to prior proofs in Section 2.2.

2.4.5 Proofs

Lemma 2.4.15. If $\{V_p, s_p\}_{p \in J}$ is a strong C^1 continuous family of event-selected vector fields on M, then for J sufficiently small, $p \in J$ implies that V_p and s_p determine an event-selected C^1 vector field over M.

Lemma 2.4.15. We begin by establishing that it suffices to verify the definition of eventselected C^1 on a compact subset of M. For, as $\{V_p, s_p\}_{p \in J}$ is a strong C^1 continuous family of event-selected vector fields on M, by definition for each $j \in I_m := \{1, ..., m\}$ and each $b \in B_m$, $\{s_p^j\}_{p \in J}$ and $\{V_p^b\}_{p \in J}$ are strong C^1 continuous. As I_m and B_m are finite, this implies that there exists a compact set $K \subset M$ such that shrinking J if necessary implies that $(s_p^j)|_{M\setminus K} = (s_{p_0}^j)|_{M\setminus K}$ and $(V_p^b)|_{M\setminus K} = (V_{p_0}^b)|_{M\setminus K}$ for all $p \in J$, $j \in I_m$, and $b \in B_m$. By the consistency of vector fields in Definition 2.4.7, this implies that $(V_p)|_{M\setminus K} = (V_{p_0})|_{M\setminus K}$ for all $p \in J$. As $M\setminus K$ is open and V_{p_0} and s_{p_0} are event-selected C^1 , they satisfy the event function and C^1 extension conditions of Definition 2.4.4 at every $x \in M\setminus K$. This implies that, for every $p \in J$, V_p and s_p satisfy the event function and C^1 extension conditions at every $x \in M\setminus K$ as well. Furthermore, as $M\setminus K$ is open and $Z_{p_0}^j$, $Z_{p_0}^{j'}$ are transverse on $M\setminus K$ for all $j, j' \in I_m$, the above implies that $Z_p^j, Z_p^{j'}$ are transverse on $M\setminus K$ for all $j, j' \in I_m$ and $j \in J$. Let K' be compact such that $K \subset \text{int } K'$. Hence, by the above, to verify the claim it suffices to show that for J sufficiently small, $p \in J$ implies that V_p and s_p satisfy the event function and C^1 extension properties at every $x \in K'$, and that for any $p \in J$ and any $j, j' \in I_m, Z_p^j$ and $Z_p^{j'}$ are transverse over K'.

First we verify the event function condition. As V_{p_0} and s_{p_0} are event-selected C^1 , for each $j \in I_m$ and each $\rho \in Z_{p_0}^j$, $s_{p_0}^j$ is an event function for V_{p_0} over an open neighborhood of ρ . This implies that $d(s_{p_0}^j)_{\rho}$ is nonzero for each $j \in I_m$. As this holds for all $\rho \in Z_{p_0}^j$, zero is a regular value of $s_{p_0}^j$, so $Z_{p_0}^j$ is a properly embedded C^1 submanifold with codimension one for each $j \in I_m$. Fix $\rho \in K'$. For each $j \in I_m$ such that $s_{p_0}^j(\rho) \neq 0$, by continuity of $s_{p_0}^j$ there exists an open neighborhood U_j of ρ in M such that $s_{p_0}^j(U_j)$ has constant sign and is disjoint from zero. For each such $j \in I_m$, as $\{s_p^j\}_{p \in J}$ is strong C^1 continuous, shrinking J if necessary implies that $s_n^j(U_i)$ has constant sign and is disjoint from zero for all $p \in J$. Let $I' \subset I_m$ consist of all j such that $s_{p_0}^j(\rho) = 0$. Let m' be the size of I and relabel I to be $\{1, ..., m'\}$. Let B'_m be the subset of B_m such that for each $b \in B'_m$ and each $j' \in I_m$ such that $s_{p_0}^{j'}(\rho) \neq 0$, $b_{j'} = \sigma_{p_0}^{j'}(\rho)$. Fix $j \in I'$. As V_{p_0} and s_{p_0} are event-selected C^1 , by the event function property of Definition 2.4.4 there exists an open neighborhood U_j of ρ such that $s_{p_0}^j$ is an event function for V_{p_0} over U_j and that $V_{p_0}|_{\text{int } D_{p_0}^b} = V_{p_0}^b|_{\text{int } D_{p_0}^b}$ for each $b \in B'_m$. Hence, there exists a constant c > 0 such that $d(s_{p_0}^j)_x(V_{p_0}(x)) \ge c$ for all $x \in U_j$. As ρ is a point of transversal intersection of C^1 embedded submanifolds $\{Z_{p_0}^j\}_{j \in I'}$, this intersection can be locally brought into a normal form using C^1 adapted coordinates [KH99, Lemma A.3.17] where each embedded submanifold is locally diffeomorphic to a linear subspace of Euclidean space, and all of these linear subspaces are transverse. This implies that for each $b \in B'_m$ and $p \in J$, int $D^b_{p_0}$ is nonempty and that there exists a sequence $x^b_k \in \text{int } D^b_{p_0}$ with $x_k^b \to \rho$ as $k \to \infty$ for each $b \in B'_m$. As mentioned in the text immediately following Definition 2.4.4, this condition is what is required to ensure persistence of C^1 event-selected vector fields under perturbation. Fix $b \in B'_m$. As V_{p_0} and $V^b_{p_0}$ agree on int $D^b_{p_0}$, this implies that $d(s_{p_0}^j)_x(V_{p_0}^b(x)) \ge c$ for all $x \in \text{int } D_{p_0}^b$. As $V_{p_0}^b$ and $d(s_{p_0}^j)$ are continuous, $x_k^b \to \rho$, and $d(s_{p_0}^j)_{x_k^b}(V_{p_0}^b(x_k^b)) \geq c$ for all $k, d(s_{p_0}^j)_{\rho}(V_{p_0}^b(\rho)) \geq c$. As $\{V_p, s_p\}_{p \in J}$ is a strong C^1 continuous family of event-selected vector fields on M, by definition for each $j \in I_m$ and each $b \in B_m$, $\{s_p^j\}_{p \in J}$ and $\{V_p^b\}_{p \in J}$ are strong C^1 continuous. Fix $c' \in (0, c)$. Then this implies that there exists an open neighborhood $U_{(j,b)}$ of ρ in M such that for J sufficiently small, $d(s_p^j)_x(V_p^b(x)) \geq c'$ for all $x \in U_{(j,b)}$ and for all $p \in J$. Shrink U_j from above if necessary so that it is contained in $\bigcap_{b \in B'_m} U_{(j,b)}$. Note that we may still take U_j to be an open neighborhood of ρ because B'_m is finite. Finally, let $U = \bigcap_{j \in I_m} U_j$ and note that it is an open neighborhood of ρ . By construction of U we must have $V_p(x) \in \{V_p^b(x)\}_{b \in B'_m}$ for all $x \in U$ and all $p \in J$ because for each $j \notin I'$, which corresponds to each j such that $s_{p_0}^j(\rho) \neq 0, s_p^j$ is constant in sign and disjoint from zero over U for all $p \in J$. Then, by construction of U, for each $j \in I_m$ and each $p \in J$, either $s_p^j(U)$ is constant in sign and disjoint from zero (true if $j \notin I'$) or $d(s_p^j)_x(V_p^b(x)) \ge c' > 0$ for all $x \in U$ and each $b \in B'_m$ (true if $j \in I'$). The latter case implies, since $V_p(x) \in \{V_p^b(x)\}_{b \in B'_m}$ for all $x \in U$ and all $p \in J$, that $d(s_p^j)_x(V_p(x)) \geq c' > 0$ for all $x \in U$, $p \in J$, and $j \in I'$. Hence, for each $j \in I_m$ and each $p \in J$, s_p is an event function for V_p over U. As such a U exists for every $\rho \in K'$, let $\{U_{\rho}\}_{\rho \in K'}$ be their collection, and let J_{ρ} denote the size of J (possibly shrunk) required for each $\rho \in K'$ to satisfy the event function part of the definition. As K' is compact, there exists a finite subcover of $\{U_{\rho}\}_{\rho \in K'}$, say $\{U_{\rho_k}\}_{k=1}^{k'}$ with k' finite. Let $J = \bigcap_{k=1}^{k'} J_{\rho_k}$ and note that it is open and contains p_0 . Then the event function part of Definition 2.4.4 is satisfied by V_p and s_p at every $\rho \in K'$ and for every $p \in J$.

Next, we show that V_p and s_p satisfy the C^1 extension part of Definition 2.4.4 for every $p \in J$. Fix $\rho \in K'$. As $\{V_p, s_p\}_{p \in J}$ is a strong C^1 continuous family of event-selected vector fields on M, consistency of vector fields implies that for each $b \in B'_m$, $V_p|_{\text{int } D_p^b} = V_p^b|_{\text{int } D_p^b}$. In particular, restricting to U from the proof of satisfying the event function condition above implies that V_p admits the C^1 extension from int $D_p^b \cap U$ to U given by $(V_p^b)|_U$ for each $b \in B'_m$. As $\rho \in K'$ was arbitrary, this implies that the C^1 extension part of Definition 2.4.4 is satisfied by V_p and s_p for every $p \in J$.

For each $p \in J$, to show that V_p and s_p determine an event-selected C^1 vector field it suffices to show that Z_p^j and $Z_p^{j'}$ are transverse over K' for all $j, j' \in I_m$. By the arguments above for the event function condition, for each $j \in I_m$, $p \in J$, and each $\rho \in Z_p^j$, $d(s_p^j)_\rho$ is nonzero. As this holds for all $\rho \in Z_p^j$, zero is a regular value of s_p^j , so $Z_p^j = (s_p^j)^{-1}(0)$ is a properly embedded C^1 submanifold with codimension one for each $j \in I_m$ and $p \in J$. Furthermore, this implies that $T_\rho Z_p^j = \ker d(s_p^j)_\rho$ for each $j \in I_m$, $p \in J$, and $\rho \in Z_p^j$. Additionally, it follows that the span of $d(s_p^j)_\rho$ is the unique one-dimensional linear subspace orthogonal to ker $d(s_p^j)_{\rho}$ for each $j \in I_m$, $p \in J$, and $\rho \in Z_p^j$. Hence, for any $j, j' \in I_m$, $p \in J$, and $\rho \in Z_p^j \cap Z_p^{j'}$, Z_p^j and $Z_p^{j'}$ are transverse at ρ in M if and only if $T_\rho Z_p^j \oplus T_\rho Z_p^{j'} = T_\rho M$ which holds if and only if ker $d(s_p^j)_{\rho} \oplus \ker d(s_p^{j'})_{\rho} = T_{\rho}M$ which holds (by orthogonality) if and only if $d(s_p^j)_{\rho}$ and $d(s_p^{j'})_{\rho}$ are linearly independent. Thus, for any $j, j' \in I_m, p \in J$, and $\rho \in Z_p^j \cap Z_p^{j'}, Z_p^j$ and $Z_p^{j'}$ are transverse at ρ if and only if $d(s_p^j)_{\rho}$ and $d(s_p^{j'})_{\rho}$ are linearly independent. Fix $\rho \in K'$. Construct the open neighborhood U of ρ as in the argument above verifying the event function condition. Let $I' \subset I_m$ be the set of indeces j such that $s_{p_0}^j(\rho) = 0$. As $\{V_p, s_p\}_{p \in J}$ is a strong C^1 continuous family of event-selected vector fields on M, the event surfaces which pass through ρ are pairwise transverse at ρ . So, by the above, this implies that for every $j, j' \in I', d(s_{p_0}^j)_{\rho}$ and $d(s_{p_0}^{j'})_{\rho}$ are linearly independent. As linear independence is an open condition and $s_{p_0}^j$ is C^1 for all $j \in I'$, there exists an open neighborhood $U_{\rho} \subset U$ such that $x \in U_{\rho}$ implies that $d(s_{p_0}^j)_x$ and $d(s_{p_0}^{j'})_x$ are linearly independent for all $j, j' \in I'$. As linear independence is an open condition and $\{s_p^j\}_{p \in J}$ is strong C^1 continuous for any $j \in I'$, shrinking J if necessary implies that $d(s_p^j)_x$ and $d(s_p^{j'})_x$ are linearly independent for all $x \in U_{\rho}$, $p \in J$, and $j, j' \in I'$. By the discussion above, this implies that Z_p^j and $Z_p^{j'}$ are transverse over U_ρ for all $p \in J$ and all $j, j' \in I'$. As $U_\rho \subset U$, by the construction of U we have that Z_p^j is disjoint from U_ρ for each $j \notin I'$, so these are vacuously transverse to $Z_p^{j'}$ over U_ρ for all $j' \in I_m$. Thus, Z_p^j and $Z_p^{j'}$ are transverse over U_{ρ} for all $p \in J$ and all $j, j' \in I_n$. Let $\{U_{\rho}\}_{\rho \in K'}$ be the collection of such neighborhoods, and let J_{ρ} denote the size of J (possibly shrunk) required for each $\rho \in K'$ to satisfy the transversality condition above for all $p \in J$ and all $x \in U_{\rho}$. As K' is compact, there exists a finite subcover of $\{U_{\rho}\}_{\rho \in K'}$, say $\{U_{\rho_k}\}_{k=1}^{k'}$ with k' finite. Let $J = \bigcap_{k=1}^{k'} J_{\rho_k}$ and note that it is open and contains p_0 . Then, for any $p \in J$ and any $j, j' \in I_m, Z_p^j$ and $Z_p^{j'}$ are transverse over K'. Hence, the transversality part of Definition 2.4.4 is satisfied by V_p and s_p at every $\rho \in K'$ and for every $p \in J$. Thus, V_p and s_p are event-selected C^1 for every $p \in J$.

Lemma 2.4.16. For J sufficiently small, V possesses a piecewise C^1 flow $\phi : M \times \mathbb{R} \times J \to M$.

Lemma 2.4.16. We construct a flow analogously to the construction in [BSKR15, Section 3]. The only novel contributions here are the inclusion of the explicit dependence on parameter $p \in J$, and the use of Lemma 2.4.15 which required the additional assumption that switching surfaces are transverse. The idea is to construct a local flow, and then to piece the local flows together in the natural way, as done in [BSKR15, Corollary 1] in an analogous manner to the smooth case [Lee13, Theorem 9.12]. Let $(x, p) \in M \times \mathbb{R} \times J$. If $x \notin Z^p$ then there

exists a neighborhood U^x which is not contained in Z^p (see the proof of Lemma 2.4.15). As $\{V_p^{\sigma(x)}\}_{p\in J}$ is a strong C^1 continuous family of vector fields, shrinking U^x if necessary, the local flow $\phi^{\sigma(x)}$ of $V^{\sigma(x)}$ is C^1 continuous over $U^x \times T^x \times J$ where T^x is an open interval containing zero. Then the local flow of V is given by $(x, t, p) \to (\phi^{\sigma(x)}(x, t, p), p)$, which is C^1 over its domain.

So, suppose $x \in Z^p$. First we construct a local flow candidate, and then show that it is in fact a flow of the vector field V. Let $K' \subset K$ such that $x \in Z_j^p$ if and only if $j \in K'$. Let B^x consist of all $b \in B$ such that for $j \in K - K'$, $b_j = \sigma(s_p^j(x))$. Then B^x consists of all possible combinations of switching states for the switching surfaces that intersect U^x .

By Lemma 2.4.15, V_p is event-selected discontinuous for any $p \in J$. This implies that, if a piecewise C^1 local flow ϕ exists (which we will soon establish), it must satisfy $\frac{d}{dt}s_p^j \circ \phi(t, y, p) =$ $d(s_p^j)_y(V_p^k(y)) \ge c_j > 0$ almost everywhere. In particular, $s_p^j \circ \phi$ is monotonically increasing along the flow in forwards time, and monotonically decreasing along the flow in backwards time, for all $j \in K'$. This implies that, in forwards time, the flow can only cross Z_j^p by passing from $s_p^j \circ \phi$ negative to $s_p^j \circ \phi$ positive. In other words, along an orbit, σ_j can only pass from -1 to +1 for each $j \in K'$, not the other way around. Hence, the only order in which the switching surfaces Z_i^p can be encountered, if they are encountered at all, is in lexicographic order in forwards time, and reverse lexicographic order in backwards time. This observation (due to [BSKR15]) will be crucial in the construction of the local flow below. For each $k \in B^x$, the functions $(x,t,p) \to s_p^j \circ \phi^k(x,t,p)$ are C^1 for all $j \in K$ and $k \in B$ since $\{s_p^j\}_{p \in J}$ is a strong C^1 continuous family and ϕ^k is C^1 . Furthermore, for each $j \in K'$ and $b \in B^x$, $\frac{\partial}{\partial t}|_{t=0}s_p^j \circ \phi^k(x,t,p) = d(s_p^j)_x(V_p^k(x)) \ge c_j > 0$ by Lemma 2.4.15. Hence, as $s_p^j \circ \phi^k(x,0,p) = 0$, since $x \in Z_i^p$, by the implicit function theorem, shrinking U^x and J if necessary, there exist C^1 functions $\tau^{(j,k)}: U^x \times J \to \mathbb{R}$ such that for $(x,p) \in U^x \times J$, $s_p^j \circ \phi^k(x, \tau^{(j,k)}(x, p), p) = 0.$ In other words, $\phi^k(x, \tau^{(j,k)}(x, p), p) \in Z_j^p$. Then $\tau^{(j,k)}(x, p)$ represents the time (positive or negative) for (x, p) to flow under ϕ^k until it hits Z_i^p . For $k \in B^x$, define the budgeted time-to-boundaries by

$$\tau_k^+(t, x, p) = \max\{0, \min(\{t\} \cup \{\tau^{(j,k)}(x, p) : j \in K', k_j < 0\})\},\$$

$$\tau_k^-(t, x, p) = \min\{0, \max(\{t\} \cup \{\tau^{(j,k)}(x, p) : j \in K', k_j > 0\})\}.$$

Then for t > 0, $\tau_k^+(t, x, p)$ is the minimum forward time for ϕ_p^k to flow x to Z^p , or t, whichever is smaller. Similarly, for t < 0, $\tau_k^-(t, x, p)$ is the minimum backward time for ϕ_p^k to flow x to Z^p , or t, whichever is smaller in magnitude. Note that τ_k^+ and τ_k^- are piecewise C^1 because they consist of a finite number of maxima and minima over C^1 functions. Next, for $k \in B^x$, define candidate parts of the local flow by

$$\phi_k^+(t,x,p) = (\phi^k(x,\tau_k^+(t,x,p),p), t - \tau_k^+(t,x,p),p), \phi_k^-(t,x,p) = (\phi^k(x,\tau_k^-(t,x,p),p), t - \tau_k^-(t,x,p),p).$$

Then ϕ_k^+ flows (t, x, p) forwards in time under ϕ^k until its first intersection with Z^p at time $\tau_k^+(t, x, p)$, and deducts the time $\tau_k^+(t, x, p)$ it takes to reach the intersection from t. Similarly, ϕ_k^- flows (t, x, p) backwards in time under ϕ^k until its first intersection with Z^p at time $\tau_k^-(t, x, p)$, and deducts the time $\tau_k^-(t, x, p)$ it takes to reach the intersection from t. Next, define the local flow candidate to be

$$\phi = \pi_1 \circ \prod_{k-} \phi_k^+ \circ \prod_{k+} \phi_k^- \tag{2.18}$$

where π_1 is the projection onto the first coordinate M, $\prod_{k=0} \phi_k^+$ refers to the composition of the functions ϕ_k^+ for $k \in B^x$ in lexicographic order, and $\prod_{k+} \phi_k^-$ refers to the composition of the functions ϕ_k^- for $k \in B^x$ in reverse lexicographic order. Note that ϕ is a finite composition of piecewise C^1 functions, hence piecewise C^1 . Then for t > 0, $\phi(t, x, p)$ first flows x forwards in time along $\phi^{\sigma(x)}$ until it hits Z^p . Then, for each event surface encountered in the intersection, the corresponding entry in $\sigma(x)$ is flipped from -1 to +1, yielding a new k, and then the state flows forward under ϕ^k until the next intersection with Z^p , etc. until the time runs to zero. As discussed above, this process gives the local flow precisely because V_p is event-selected discontinuous for $p \in J$; otherwise, the lexicographic and reverse lexicographic orders would not be guaranteed to capture all possible orbits under V. Furthermore, note that by construction ϕ is C^1 away from switching surfaces. For each $p \in J, V_p$ is event-selected discontinuous and so, by [BSKR15, Corollary 1] it possesses a global flow ϕ_p which satisfies $\frac{\partial}{\partial t}\phi_p(x,t) = V_p(x,t) = V(x,t,p)$ almost everywhere in $M \times \mathbb{R}$ [BSKR15, Lemma 1] (everywhere other than switching surfaces). For fixed $p \in J$, the formula for $\phi(\cdot, \cdot, p)$ in Equation 2.19 agrees with the formula for ϕ_p [BSKR15, Equation 12]. In particular, this implies that $\phi(\cdot, \cdot, p)$ is differentiable with respect to time almost everywhere and $\frac{\partial}{\partial t}\phi(t,x,p) = V_p(x,t) = V(x,t,p)$ almost everywhere (everywhere other than switching surfaces) in an open neighborhood of x and t in $M \times \mathbb{R}$. Hence, $\frac{\partial}{\partial t}\phi(t, x, p) = V(x, t, p)$ on C^J and, thus, is differentiable almost everywhere (everywhere other than switching surfaces) in an open neighborhood of (x, t, p) in $M \times \mathbb{R} \times J$. The verification that ϕ is in fact a local flow of V follows analogously to the proof of [BSKR15, Theorem 1] using the above time derivative $\frac{\partial}{\partial t}\phi$. Finally, the local flows are pieced together to form a global flow in an analogous manner to the fixed parameter case [BSKR15, Corollary 1], which is itself analogous to the construction of global flows for smooth vector fields as in [Lee13, Theorem 9.12]. \Box

Lemma 2.4.17. Z^J is closed and C^J is open in $M \times J$.

Proof of Lemma 2.4.17. For each $j \in K$, since $\{s_p^j\}_{p \in J}$ is strong C^1 continuous, there exists $H: M \times J \to \mathbb{R}$ such that $H(x, p) = s_p^j(x)$ and H is C^1 . Since H is continuous, $Z_j^J = H^{-1}(0)$ is closed in $M \times J$. As K is finite, $Z^J = \bigcup_{j \in K} Z_J^j$ is closed in $M \times J$. Hence, $C^J = (M \times J) - Z^J$ is open in $M \times J$.

Lemma 2.4.18. For any $(x, t, p) \in M \times \mathbb{R} \times J$ such that $x, \phi(t, x, p) \in C^J$, there exist open neighborhoods U^x , T^x , and J^x of x, t, and p, respectively, such that $\phi|_{U^x \times T^x \times J^x}$ is C^1 .

Proof of Lemma 2.4.18. By Lemma 2.4.17, C^J is open in $M \times J$, so there exist $U' \subset M$ and $J' \subset J$ open neighborhoods of $\phi(t, x, p)$ and p, respectively, such that $U' \times J' \subset C^J$. As $\phi(\cdot, x, p)$ is continuous, $T^x = \phi(\cdot, x, p)^{-1}(U' \times J')$ is an open neighborhood of t in \mathbb{R} . As $\phi(t, \cdot, \cdot)$ is continuous, $\hat{U} = (\bigcup_{t \in T^x} \phi(t, \cdot, \cdot)^{-1}(U' \times J')) \cap C^J$ is an open neighborhood of (x, p) in $M \times J$. Let U^x and J^x be open neighborhoods of x and p, respectively, such that $U^x \times J^x \subset \hat{U}$. Then $\phi|_{U^x \times T^x \times J^x}$ maps an open subset of C^J into an open subset of C^J . By the proof of Lemma 2.4.16, the derivative of ϕ exists and is continuous on C^J ; discontinuities are confined to Z^J . Thus, $\phi|_{U^x \times T^x \times J^x}$ is C^1 .

Lemma 2.4.19. For any $p \in J$, $x \in M$, and T > 0 finite, $\phi([0,T], x, p)$ intersects Z^p in only finitely many isolated points.

Proof of Lemma 2.4.19. For any $(x, p) \in M \times J$ and any T > 0 finite, let

 $\gamma = \phi(\cdot, x, p)^{-1}(\phi([0, T], x, p) \cap Z^p)$ denote the intersection times of the flow with the switching surfaces Z^p . First note that γ is closed since $\phi([0, T], x, p)$ is compact, Z^p is closed, and $\phi(\cdot, x, p)$ is continuous. As γ is closed in [0, T] compact, γ itself is compact hence sequentially compact. Assume towards a contradiction that γ has infinite cardinality. Then there exists a monotonically increasing sequence of distinct points $(t_n)_{n=1}^{\infty}$ in γ . Passing to a subsequence if necessary, we may assume $t_n \to \hat{t} \in \gamma$ since γ is sequentially compact. Let $z := \phi(\hat{t}, x, p)$ and let $K' \subset K$ such that $j \in K'$ if and only if $z \in Z_j^p$. As in the proof of Lemma 2.4.15, there exists an open neighborhood U^z of z in M such that U^z is an event neighborhood for each $j \in K'$ and U^z does not intersect $Z_{j'}^p$ for any $j' \in K - K'$. Hence, for $j \in K'$, $\frac{\partial}{\partial t}s_p^j \circ \phi(0, y, p) = d(s_p^j)_z(V_p(z)) \ge c_j > 0$ almost everywhere. Let $T^z = \phi(\cdot, x, p)^{-1}(U^z)$, which is open since U^z is open and $\phi(\cdot, x, p)$ is continuous. Then for each $j \in K'$ and every $t \in T^z - \{\hat{t}\}, \frac{\partial}{\partial t}s_p^j \circ \phi(t, x, p) \ge c_j > 0$. This implies that the restriction of $s_p^j \circ \phi(\cdot, x, p)$ to T^z equals zero if and only if $t = \hat{t}$ for each $j \in K'$. Hence, for $t \in T^z$, along the flow of $\phi(\cdot, x, p)$, Z_j^p for $j \in K'$ is only encountered at $t = \hat{t}$, and Z_j^p for $j \in K - K'$ is not encountered. As T^z is an open neighborhood of \hat{t} such that $\phi(\cdot, x, p)^{-1}(\phi(T^z, x, p) \cap Z^p) = \{\hat{t}\}$, this contradicts that $t_n \to \hat{t} \in \gamma$.

Lemma 2.4.20. Let $X^i(p_0)$ be a periodic orbit for V_{p_0} . Then there exists $x \in X^i(p_0) \cap C^J$ and a C^1 cross section S containing x such that the first return map is well-defined and C^1 on S. Furthermore, for J and S sufficiently small, $p \in J$ implies that the first return map is well-defined and C^1 on S, that it varies C^1 with p, and that V_p is transverse to S.

Proof of Lemma 2.4.20. By Lemma 2.4.19, since $X^i(p_0)$ has finite period T, its intersection with Z^{p_0} is finite. So, there exists (in particular) at least one $x \in X^i(p_0) - Z^{p_0}$. By Lemma 2.4.18, there exist open neighborhoods U^x , T^x , and J^x of x, t, and p, respectively, such that $\phi|_{U^x \times T^x \times J^x}$ is C^1 . Let S be a smooth cross section contained in $Z_i^{p_0} \cap C^J \cap U^x$, transverse to V_{p_0} , and with $x \in S$. Shrink J^x if necessary so that V_p is transverse to S for all $p \in J$. Since S is a smooth embedded submanifold, shrinking S if necessary there exists a smooth submersion $s : M \to \mathbb{R}$ such that $S = s^{-1}(0)$. Hence, $(s \circ \phi)|_{U^x \times T^x \times J^x}$ is C^1 and its derivative with respect to time is full rank since V_{p_0} is transverse to $S = s^{-1}(0)$. Thus, by the implicit function theorem there exist open neighborhoods $U' \subset U^x$, $J' \subset J^x$, and a C^1 function $\tau : U' \times J' \to T^x$ such that $s \circ \phi(tau(z, p), z, p) = 0$ for all $(z, p) \in \phi(\tau(z, p), z, p)$. Then y is C^1 as it is a composition of C^1 maps, and sends $(z, p) \in S \times J^x$ to its first intersection with S.

Corollary 2.4.21. For J sufficiently small and for any $i \in I$ and $p \in J$, there exists a unique critical element $X^{i}(p)$ C^{0} -close to $X^{i}(p_{0})$ such that $X^{i}(p)$ is hyperbolic in the sense described by Assumption 2.4.11.

Proof of Corollary 2.4.21. As I is finite, it suffices to prove the claim for fixed $i \in I$. If $X^i(p_0)$ is an equilibrium point then it lies in C^J , which is open by Lemma 2.4.17. As $X^i(p_0)$ is hyperbolic and $V|_{C^J}$ is C^1 , for J sufficiently small and $p \in J$, there exists a hyperbolic

equilibrium point $X^i(p) \in C^J$ which is C^1 close to $X^i(p_0)$. If $X^i(p_0)$ is a periodic orbit then by Lemma 2.4.20, for J sufficiently small, there exist $x \in X^i(p_0) \cap C^J$ and a smooth cross section S with C^1 first return map $y : S \times J \to S$. Then x is a hyperbolic fixed point of $y|_{S \times \{p_0\}}$ by Assumption 2.4.11. As y is C^1 , for J sufficiently small and $p \in J$, there exists a unique (hyperbolic) fixed point $x_p \in S$ of $y_{S \times \{p\}}$ which is C^1 close to x. Under the flow of V_p this gives rise to a unique periodic orbit $X^i(p)$ which is C^0 close to $X^i(p_0)$.

Lemma 2.4.22. For every $i \in I$ and $p \in J$, $W^s(X^i(p))$ and $W^u(X^i(p))$ are V_p -invariant, piecewise C^1 immersed submanifolds with $\tilde{W}^s(X^i(p)) = W^s(X^i(p)) \cap C^J$ and $\tilde{W}^u(X^i(p)) = W^u(X^i(p)) \cap C^J$.

Proof of Lemma 2.4.22. Fix $i \in I$ and $p \in J$. By Lemma 2.4.15, V_p is event-selected discontinuous. Without loss of generality, consider $W^{s}(X^{i}(p))$; the proof for $W^{u}(X^{i}(p))$ is analogous. If $X^{i}(p)$ is an equilibrium point (periodic orbit) then $X^{i}(p)$ (then there exists a point $x \in X^i(p)$ that) lies in C^J . In the latter case, choose x to be the point that possesses a cross section S with a C^1 first return map by Lemma 2.4.20. By hyperbolicity, if $X^{i}(p)$ is an equilibrium point (periodic orbit) there exists a local stable manifold $W^s_{\text{loc}}(X^i(p))$ $(W^s_{\text{loc}}(X^i(p)) \cap S)$ which is contained in C^J . If $X^i(p)$ is an equilibrium point, define the notation: $W^s_{\text{loc}}(X^i(p))$ $(S^i_s(p))$. If $X^i(p)$ is a periodic orbit, since V_p is transverse to S let $S_s^i(p)$ denote a flowout of $W_{loc}^s(X^i(p)) \cap S$ such that $S_s^i(p) \subset C^J$. Then $W^s(X^i(p)) = \bigcup_{t \leq 0} \phi_{(t,p)}(S^i_s(p))$, so since $S^i_s(p)$ is forward V_p -invariant, $W^s(X^i(p))$ is V_p -invariant. Furthermore, for any point $y \in W^s(X^i(p))$, there exists T > 0 such that $\phi(T, y, p) \in \operatorname{int} S^i_s(p)$. Let $U^y = \phi(T, \cdot, p)^{-1}(\operatorname{int} S^i_s(p))$. Then $\phi(T, \cdot, p)|_{U^y} : U^y \to \operatorname{int} S^i_s(p)$ is a C^0 homeomorphism between a neighborhood of U^y in $W^s(X^i(p))$ and the C^1 embedded submanifold int $S_s^i(p)$. As y was arbitrary, $W^s(X^i(p))$ is a topological manifold. Let $\tilde{W}^{s}(X^{i}(p)) = W^{s}(X^{i}(p)) \cap C^{J}$ and let $y \in \tilde{W}^{s}(X^{i}(p))$. There exists T > 0 such that $\phi(T, y, p) \in \text{int } S^i_s(p) \subset C^J$. By Lemma 2.4.18, $\phi(T, \cdot, p)$ is a C^1 diffeomorphism from an open neighborhood of y in C^J onto an open neighborhood of $\phi(T, y, p)$ in C^J . As int $S^i_s(p)$ is a C^1 embedded submanifold, $\phi(T, \cdot, p)$ restricts to a C^1 diffeomorphism from a neighborhood of y in $W^s(X^i(p)) \cap C^J$ onto a neighborhood of $\phi(T, y, p)$ in $S^i_s(p)$. As y was arbitrary, $\tilde{W}^s(X^i(p))$ is a C^1 immersed submanifold. Since $\tilde{W}^s(X^i(p))$ and $W^s(X^i(p))$ have the same dimension, $\tilde{W}^s(X^i(p))$ is open in $W^s(X^i(p))$. Let $y \in W^s(X^i(p)) - \tilde{W}^s(X^i(p))$. Then $y \in Z^p$. By Lemma 2.4.19, intersections with Z^p are isolated in time, so for some open interval T^y containing zero, $\phi(\cdot, y, p)^{-1}(\phi(T^y, y, p) \cap Z^p) = \{0\}$. Then the sequence $y_n = \phi(\frac{1}{n}, y, p) \in \tilde{W}^s(X^i(p))$ for n sufficiently large and $y_n \to y$ by continuity of ϕ . Hence, $\tilde{W}^{s}(X^{i}(p))$ is dense in $W^{s}(X^{i}(p))$.

Lemma 2.4.23. $W^{s}(X_{I}^{s})$ is open and V-invariant in $M \times J$.

Proof of Lemma 2.4.23. By Lemma 2.4.22, $W^s(X^s(p))$ is V_p -invariant for each $p \in J$, which implies that $W^s(X_J^s)$ is V-invariant. Note that $W^s_{\text{loc}}(X_J^s)$ is open in $M \times J$ by Lemma 2.2.36. As $W^s(X_J^s) = \bigcup_{t \leq 0} \phi_t(W^s_{\text{loc}}(X_J^s))$ is a union of open sets in $M \times J$ - since $\phi(t, \cdot, \cdot)$ is a C^0 homeomorphism hence an open map - $W^s(X_J^s)$ is itself open in $M \times J$.

Lemma 2.4.24. For any $i \in I$ such that $X^i(p_0)$ is an equilibrium point, (periodic orbit with $x \in X^i(p_0)$ possessing a cross section S with a C^1 first return map by Lemma 2.4.20) and for $\epsilon > 0$ sufficiently small, there exists a compact set $D \subset W^u_{loc}(X^i_{J'}) - X^i_{J'}$ ($D \subset$ $W^u_{loc}(X^i_{J'}) \cap S - X^i_{J'}$) and an open neighborhood N of D in $M \times J$ such that $N \subset D_{\epsilon}$, $D_{\epsilon} \cap X^i_J = \emptyset$, and $\bigcup_{t \leq 0} \phi_t(N) \bigcup W^s(X^i_J)$ contains an open neighborhood of $X^i_{p_0}(x)$ in $M \times J$.

Proof of Lemma 2.4.24. As C^J is open by Lemma 2.4.17, and $X^i(p_0)(x)$ is contained in C^J , choose J sufficiently small such that $p \in J$ implies that $X^i(p)(x)$ is contained in C^J . Then since Lemma 2.2.37 is a local result, and $V|_{C^J}$ is C^1 , the claim follows.

Lemma 2.4.25. For any $i \in I$, $(W^u(X^i_{p_0}) - X^i_{p_0}) \cap \overline{W}^s(X^s_J) \neq \emptyset$.

Proof of Lemma 2.4.25. As C^J is open by Lemma 2.4.17, and $X^i(p_0)(x)$ is contained in C^J , choose J sufficiently small such that $p \in J$ implies that $X^i(p)(x)$ is contained in C^J . Then, since the proof of this result in Lemma 2.2.38 for a weakly C^1 continuous family of vector fields followed from Lemma 2.2.37, the same proof carries over without modification to prove the claim here.

Lemma 2.4.26. If $W^s(X^i(p_0)) \cap W^u(X^j(p_0)) \neq \emptyset$ then $n^u(X^j(p_0)) \ge n^u(X^i(p_0))$ for any $i, j \in I$. If $X^j(p_0)$ is an equilibrium point, the inequality is strict.

Proof of Lemma 2.4.26. Since $W^s(X^i(p_0)) \cap W^u(X^j(p_0)) \neq \emptyset$ and they are V_{p_0} -invariant, there exists $q \in S_s^i(p_0) \cap W^u(X^j(p_0))$, where $S_s^i(p)$ is either $W_{\text{loc}}^s(X^i(p)$ if $X^i(p_0)$ is an equilibrium point or a flowout of $W_{\text{loc}}^s(X^i(p)) \cap S$ contained in C^J if $X^i(p_0)$ is a periodic orbit, where in the latter case S is a smooth cross section of $X^i(p_0)$ contained in C^J . As $S_s^i(p_0) \subset C^J$, $q \in$ $C^J \cap W^s(X^i(p_0)) \cap W^u(X^j(p_0))$. As $W^s(X^i(p_0))$ and $W^u(X^j(p_0))$ are V_{p_0} -invariant, and C^J is open by Lemma 2.4.17, there exists an open integral curve γ of V_{p_0} containing q such that $\gamma \subset C^J \cap W^s(X^i(p_0)) \cap W^u(X^j(p_0))$. By Lemma 2.4.22, $\tilde{W}^s(X^i(p_0)) = W^s(X^i(p_0)) \cap C^J$ and $\tilde{W}^u(X^j(p_0)) = W^u(X^j(p_0)) \cap C^J$, so $\gamma \subset C^J \cap \tilde{W}^s(X^i(p_0)) \cap \tilde{W}^u(X^j(p_0))$. As

 $\tilde{W}^{s}(X^{i}(p_{0}))$ and $\tilde{W}^{u}(X^{j}(p_{0}))$ are transverse C^{1} immersed submanifolds, the claim follows from the transversality condition together with the flow dimension, exactly as in the proof of Lemma 2.2.30.

Lemma 2.4.27. If $(W^s(X^i(p_0)) - X^i(p_0)) \cap (W^u(X^j(p_0)) - X^j(p_0)) \neq \emptyset$ and $(W^s(X^j(p_0)) - X^j(p_0)) \cap (W^u(X^k(p_0)) - X^k(p_0)) \neq \emptyset$ then $(\tilde{W}^s(X^i(p_0)) - X^i(p_0)) \cap (\tilde{W}^u(X^k(p_0)) - X^k(p_0)) \neq \emptyset$.

Proof of Lemma 2.4.27. If $X^{j}(p_{0})$ is an equilibrium point, define the notation: $W^{u}_{loc}(X^{j}(p))$ $(S_u^j(p))$. If $X^j(p_0)$ is a periodic orbit, since V_{p_0} is transverse to S let $S_u^j(p_0)$ denote a flowout of $W^u_{\text{loc}}(X^j(p_0)) \cap S$ such that $S^j_u(p_0) \subset C^J$, where S is a smooth cross section of $X^j(p_0)$ as in Lemma 2.4.20. Define $S_s^j(p_0)$ analogously for the stable manifold of $X^j(p_0)$. As $(W^{s}(X^{i}(p_{0})) - X^{i}(p_{0})) \cap (W^{u}(X^{j}(p_{0})) - X^{j}(p_{0})) \neq \emptyset$ and $(W^s(X^j(p_0)) - X^j(p_0)) \cap (W^u(X^k(p_0)) - X^k(p_0)) \neq \emptyset$, and by V_{p_0} -invariance of the stable and unstable manifolds, there exist $y \in S^j_u(p_0) \cap (W^s(X^i(p_0)) - X^i(p_0))$ and $z \in S_s^j(p_0) \cap (W^u(X^k(p_0)) - X^k(p_0))$. As $S_s^j(p_0), S_u^j(p_0) \subset C^J$, by Lemma 2.4.22 $y \in C^J$ $S_{u}^{j}(p_{0}) \cap (\tilde{W}^{s}(X^{i}(p_{0})) - X^{i}(p_{0}))$ and $z \in S_{s}^{j}(p_{0}) \cap (\tilde{W}^{u}(X^{k}(p_{0})) - X^{k}(p_{0}))$. Let $D \subset (\tilde{W}^s(X^i(p_0)) - X^i(p_0))$ be a closed C^1 disk containing y that is transversal to $S^j_u(p_0)$. As D and $S_u^j(p_0)$ are compact C^1 embedded submanifolds with transversal intersection, by [KH99, Corollary A.3.18], there exists $\epsilon > 0$ such that if B' is a compact C^1 embedded submanifold which is ϵC^1 -close to $S^j_u(p_0)$ then it has nonempty, transversal intersection with D. Let $B \subset (\tilde{W}^u(X^k(p_0)) - X^k(p_0))$ be a closed C^1 disk containing z that is transversal to $S_s^j(p_0)$. By Lemma 2.4.26, dim $\tilde{W}^u(X^k(p_0)) \geq \dim S_u^j(p_0)$, so we may choose B such that dim $B = \dim S^j_u(p_0)$. As C^J is open, $V|_{C^J}$ is C^1 , and the Inclination Lemma [Pal69, Lemma 1.1] is a local result for C^1 vector fields, shrinking δ if necessary we may apply the Inclination Lemma to conclude that there exists T' > 0 such that $t \ge T'$ implies $\phi_{(t,p_0)}(B)$ is ϵC^1 -close to $S^j_u(p_0)$. Hence, by the choice of ϵ , $\phi_{(t,p_0)}(B) \cap D \neq \emptyset$. So, let $q \in \phi_{(t,p_0)}(B) \cap D$. As $D \subset (\tilde{W}^s(X^i(p_0)) - X^i(p_0)), q \in (\tilde{W}^s(X^i(p_0)) - X^i(p_0)) \cap C^J$. As $B \subset C^J$. $(W^u(X^k(p_0)) - X^k(p_0))$, and $W^u(X^k(p_0))$ is V_{p_0} -invariant, $q \in (\tilde{W}^u(X^k(p_0)) - X^k(p_0))$. \Box

Corollary 2.4.28. If $W^u(X^i(p_0)) \cap W^s(X^s(p_0)) \neq \emptyset$ and $W^s(X^i(p_0)) \cap W^u(X^j(p_0)) \neq \emptyset$ then $\tilde{W}^u(X^j(p_0)) \cap \tilde{W}^s(X^s(p_0)) \neq \emptyset$.

Proof of Corollary 2.4.28. Follows from Lemma 2.4.27 with $X^i(p_0) = X^s(p_0), X^j(p_0) = X^i(p_0)$, and $X^k(p_0) = X^j(p_0)$.

Lemma 2.4.29. For any $i \in I$, $W^{s}(X^{i}(p_{0})) \cap W^{u}(X^{i}(p_{0})) = X^{i}(p_{0})$.

Proof of Lemma 2.4.29. The proof is identical to the proof of Lemma 2.2.32.

Lemma 2.4.30. There do not exist any transverse heteroclinic cycles of critical elements contained in $\partial W^s(X^s(p_0))$. Hence, every heteroclinic cycle of critical elements contained in $\partial W^s(X^s(p_0))$ has finite length.

Proof of Lemma 2.4.30. Assume towards a contradiction that $\{X^j(p_0)\}_{j=1}^m$ is a heteroclinic cycle contained in $\partial W^s(X^s(p_0))$. By transitivity (Lemma 2.4.27), since $X^n(p_0) = X^1(p_0)$, $(W^s(X^1(p_0)) - X^1(p_0)) \cap (W^u(X^1(p_0)) - X^1(p_0)) \neq \emptyset$. This contradicts Lemma 2.4.29. Since, by Assumption 2.4.9, the intersection of the nonwandering set of V_{p_0} with $\partial W^s(X^s(p_0))$ consists of a finite number of critical elements, and since there are no heteroclinic cycles, every heteroclinic sequence must be finite.

Lemma 2.4.31. For any $i \in I$, $W^u(X^i(p_0)) \cap W^s(X^s(p_0)) \neq \emptyset$.

Proof of Lemma 2.4.31. The proof is identical to the proof of Lemma 2.2.35 using Lemmas 2.4.25-2.4.30 here.

Lemma 2.4.32. If $W^u(X^i(p)) \cap W^s(X^s(p)) \neq \emptyset$ for any $p \in J$ then $W^s(X^i(p)) \subset \partial W^s(X^s(p))$.

Proof of Lemma 2.4.32. If $X^{i}(p)$ is an equilibrium point, let $G^{s} = W^{s}_{loc}(X^{i}(p))$ and let $G^{u} =$ $W^u_{\rm loc}(X^i(p))$. If $X^i(p)$ is a periodic orbit, let S be a smooth cross section with C^1 first return map as in Lemma 2.4.20, let $G^s = W^s_{\text{loc}}(X^i(p)) \cap S$, and let $G^u = W^u_{\text{loc}}(X^i(p)) \cap S$. As $W^s(X^i(p)) = \bigcup_{t \leq 0} \phi_{(t,p)}(G^s)$ and $\partial W^s(X^s(p))$ is V_p -invariant, it suffices to show that $G^s \subset \partial W^s(X^s(p))$. So, let $q \in G^s$. As $W^u(X^i(p)) \cap W^s(X^s(p)) \neq \emptyset$ and $W^s(X^s(p))$ is V_p -invariant, $G^u \cap W^s(X^s(p)) \neq \emptyset$. Hence, there exists $\gamma > 0$ such that if D is γC^1 -close to G^u then $D \cap W^s(X^s(p)) \neq \emptyset$. Let $\epsilon > 0$ and let D_{ϵ} be a closed C^1 disk centered at q, contained in the open ball of radius ϵ centered at q, and transverse to $W^{s}(X^{i}(p))$ such that dim $D_{\epsilon} = \dim G^{u}$. As C^{J} is open, $V|_{C^{J}}$ is C^{1} , and the Inclination Lemma [Pal69, Lemma 1.1] is a local result for C^1 vector fields, shrinking δ if necessary we may apply the Inclination Lemma to conclude that there exists T' > 0 such that $t \ge T'$ implies $\phi_{(t,p)}(D_{\epsilon})$ is γC^1 -close to G^u . Hence, by the choice of γ , $\phi_{(t,p)}(D_{\epsilon}) \cap W^s(X^s(p)) \neq \emptyset$. By V_p -invariance of $W^{s}(X^{s}(p)), D_{\epsilon} \cap W^{s}(X^{s}(p)) \neq \emptyset$. Hence, $d(q, \overline{W}^{s}(X^{s}(p))) \leq \epsilon$. As this holds for all $\epsilon > 0$, $d(D, \overline{W}^s(X^s(p))) = 0$. Since q is compact and $\overline{W}^s(X^s(p))$ is closed, $q \in \overline{W}^s(X^s(p))$. But $q \in G^s \subset W^s(X^i(p))$ implies that $q \in \partial W^s(X^s(p))$.

Proof of Theorem 2.4.13. Let $x \in \partial W^s(X^s(p))$. Then there exists a sequence $x_n \in W^s(X^s(p))$ with $x_n \to x$. Hence, there exists a sequence $(x_n, p) \in W^s(X_J^s)$ with $(x_n, p) \to (x, p)$. Thus, $(x, p) \in \partial W^s(X_J^s)$. So, $\partial W^s(X_J^s) \supset \sqcup_{p \in J} \partial W^s(X^s(p))$. By Assumption 2.4.10, $\partial W^s(X_J^s) \subset \bigcup_{i \in I} W^s(X_J^i)$. Fix $i \in I$. If $X^i(p_0)$ is an equilibrium point, let $S_u^i(p_0) = W_{\text{loc}}^u(X^i(p_0))$. If $X^i(p_0)$ is a periodic orbit, let S be a smooth cross section with C^1 first return map as in Lemma 2.4.20, and let $S_u^i(p_0) = W_{\text{loc}}^u(X^i(p_0)) \cap S$. By Lemma 2.4.31, $W^u(X^i(p_0)) \cap W^s(X^s(p_0)) \neq \emptyset$. As $W^s(X^s(p_0))$ is V_{p_0} -invariant, $S_u^i(p_0) \cap W^s(X^s(p_0)) \neq \emptyset$. As $\{S_u^i(p)\}_{p \in J}$ is C^1 continuous and $W^s(X_J^s)$ is open by Lemma 2.4.23, J sufficiently small implies that for $p \in J$, $S_u^i(p) \cap W^s(X^s(p)) \neq \emptyset$. By Lemma 2.4.32, $W^s(X^i(p)) \subset \partial W^s(X^s(p))$. Hence, as I is finite, for J sufficiently small $\bigcup_{i \in I} W^s(X_J^i) \subset \sqcup_{p \in J} \partial W^s(X^s(p))$. The claim follows by combining with the inclusions above.

Proof of Corollary 2.4.14. The proof is identical to the proof of Corollary 2.2.24 except for $X^i(p_0)$ a periodic orbit, $W^s_{\text{loc}}(X^i(p_0)) \cap S$ is substituted for $W^s_{\text{loc}}(X^i(p_0))$ and the first return map is used in place of the flow.

2.4.6 Conclusion

Theory presented in this section extends the RoA boundary continuity results of Section 2.2 to rough vector fields which exhibit clipping limits and switching, such as power systems. In particular, for vector fields with C^1 event-selected discontinuities which are in essence Morse-Smale along the RoA boundary, it is shown that, for a sufficiently small change in parameter values, the RoA boundary varies Chabauty continuously and it is equal to the union of the stable manifolds of the critical elements it contains. This result will be used to show that the recovery boundary consists of critical parameter values, and that the closest critical parameter value to any recover parameter value lies on the recovery boundary. In turn, this will be crucial in providing theoretical motivation for the algorithms introduced in Chapter III.

2.5 Theoretical Motivation for Maximizing Time in Ball Algorithm

2.5.1 Introduction

The purpose of this section is to provide theoretical motivation for the class of algorithms described in Section 3.2 for computing points on the recovery boundary. The first step is to establish, using the RoA boundary continuity results of Sections 2.2-2.4, that the recovery boundary consists of critical parameter values, and that starting from any recovery value the closest critical parameter value to it is also the closest point to it on the recovery boundary. This provides justification for the computation of the (closest point on the) recovery boundary via the computation of (the closest) critical parameter values.

The remainder of this section provides theoretical motivation for the algorithms of Section 3.2. In particular, it is shown that for each point in the recovery boundary there exists a controlling invariant set possessing the property that the time the trajectory spends in a neighborhood of the controlling invariant set is continuous with respect to parameter values and diverges to infinity as the parameter values approach that point, thereby providing justification for the algorithms.

2.5.2 Results

For each $p \in J$, define initial condition corresponding to parameter value $p(\hat{y}_p)$, where $\hat{y}: J \to M$ sends p to the initial condition of V_p , and we write $\hat{y}_p = \hat{y}(p)$. Assume that \hat{y} is C^1 over J and, as with critical elements above, sometimes consider $\hat{y}_p \in M$ and sometimes $\hat{y}_p \in M \times J$; the distinction should be clear from context. Then a parameter value $p^* \in J$ is a critical parameter value if and only if $\hat{y}_{p^*} \in \partial W^s(X_{p^*}^s)$. Define $R = \{p \in J : \hat{y}_p \in W^s(X^s(p))\}$ to be the set of recovery parameter values (R), define the recovery parameter boundary (∂R) , and let $C = \{p \in J : \hat{y}_p \in \partial W^s(X^s(p))\}$ be the set of critical parameter values (C). Theorem 2.5.1 shows, under the assumptions of Section 2.4.4 (for hybrid systems) or of Section 2.3.4 (for C^1 vector fields), that $\partial R \subset C$ and that the closest critical parameter value.

Theorem 2.5.1. Let M be either a compact Riemannian manifold or Euclidean space, and let $\{V_p\}_{p\in J}$ be a family of vector fields on M satisfying the conclusions of Theorem 2.4.13 (for vector fields with C^1 event-selected discontinuities) or Theorem 2.2.23 (for C^1 vector fields). Let $\hat{y}: J \to M$ be C^1 . Then $\partial R \subset C$. Assume there exist $p_1, p_2 \in J$ such that $\hat{y}_{p_1} \in W^s(X_{p_1}^s)$ and $\hat{y}_{p_2} \notin W^s(X_{p_2}^s)$. Fix any $p_0 \in J$ and let $J_0 = \{p^* \in C : d(p_0, p^*) = d_S(p_0, C)\}$. Then J_0 is nonempty, $J_0 \subset \partial R$, and $J_0 = \{p^* \in \partial R : d(p_0, p^*) = d_S(p_0, \partial R)\}$.

Theorem 2.5.1 justifies the method of determining or finding the closest point on the recovery boundary by computing the closest critical parameter values. Then Corollary 2.5.2 shows that for each point p^* in the recovery boundary, there exists a critical element, called the controlling critical element, such that \hat{y}_{p^*} is in its stable manifold.

Corollary 2.5.2. Assume the conditions of Theorem 2.5.1. Fix any $p^* \in \partial R$. Then there exists a unique critical element $X_J^* \subset \partial W^s(X_J^s)$, called the controlling critical element corresponding to p^* , such that $\hat{y}_{p^*} \in W^s(X_{p^*}^*)$.

Remark 2.5.3. It is straightforward to generalize the notion of controlling critical element to controlling invariant set, where the controlling invariant set would be the ω limit set of \hat{y}_{p^*} for $p^* \in \partial R$ (assuming it is nonempty). This approach is not taken here in order to preserve consistency with the settings of Sections 2.2-2.4, in which the intersection of the nonwandering set with the RoA boundary consists entirely of critical elements, but we will outline the trivial extension to the more general case of controlling invariant sets below.

We make the following assumption:

Assumption 2.5.4. Let γ be a C^1 path in J such that $\gamma([0,1)) \subset R$ and $\gamma(1) \in \partial R$. Let X_J^* be the unique controlling critical element corresponding to $\gamma(1)$, as in Corollary 2.5.2. There exists a compact codimension-zero smooth embedded submanifold with boundary N in M such that for $p \in \gamma([0,1])$, X_p^* is contained in the interior of N, $X^s(p)$ and \hat{y}_p are disjoint from N, and the orbit of \hat{y}_p under V_p has nonempty, transversal intersection with ∂N .

Remark 2.5.5. We modify Assumption 2.5.4 slightly in the case where the vector fields are C^1 event-selected discontinuous. If the controlling critical element is a periodic orbit which has nonempty intersection with at least one switching surface, then let $x \in X^*(p_0) \cap C^J$ as in Lemma 2.4.20, and assume that x is in the interior of N and that N intersects X_p^* for $p \in J$ (instead of assuming that N contains X_p^* for $p \in J$). More generally, assume further that $N \times J \subset C^J$ lies in the same connected component of C^J as X_J^* (in the case where $X_J^* \subset C^J$) or x (in the case where $X_J^* \cap Z^J \neq \emptyset$).

Remark 2.5.6. If it is desirable to consider a more general controlling invariant set, the vector fields have C^1 event-selected discontinuities, and the controlling invariant set intersects at least one switching surface, choose a point x in the controlling invariant set such that $x \in C^J$, and make the assumptions of Remark 2.5.5.

Remark 2.5.7. Unlike Assumption 2.2.18, the transversality condition of Assumption 2.5.4 can be easily checked directly by numerical simulation, and the neighborhood N adjusted accordingly if necessary. In applications, N is typically taken to be a closed ball and its radius is adjusted to ensure the transversality condition of Assumption 2.5.4 holds.

Remark 2.5.8. Assumption 2.5.4 also ensures that the initial conditions and the stable equilibria do not intersect the neighborhood N, and that the controlling critical element X_J^j is contained in N. Let γ and N be as in Assumption 2.5.4. Let $t_{ball} : \gamma([0,1]) \to [0,\infty]$ by $t_{ball}(p) = \int_0^\infty 1_N(\phi(t,\hat{y}_p,p))dt$ where 1_N is the indicator function of N. So, $1_N(x) = 1$ if $x \in N$ and $1_N(x) = 0$ if $x \notin N$. Therefore, $t_{ball}(p)$ measures the length of time the orbit of V_p with initial condition \hat{y}_p spends in N. Theorem 2.5.9 shows that t_{ball} is well-defined and continuous over $\gamma([0,1])$. Since $\hat{y}_{\gamma(1)}$ converges to $X^*_{\gamma(1)} \subset N$, $t_{ball}(\gamma(1)) = \infty$, so continuity implies that $t_{ball}(p)$ diverges to infinity as p approaches $\gamma(1)$ along the path γ .

Theorem 2.5.9. Assume the conditions of Theorem 2.5.1. Fix any $p_0 \in J$ and any $p^* \in \partial R$. By Corollary 2.5.2, there exists a unique controlling critical element $X_J^* \subset \partial W^s(X_J^s)$ such that $\hat{y}_{p^*} \in W^s(X_{p^*}^*)$. Let $\gamma : [0,1] \to J$ be a C^1 path satisfying Assumption 2.5.4 and such that $\gamma(0) = p_0$ and $\gamma(1) = p^*$. Then $t_{ball} : \gamma([0,1]) \to [0,\infty]$ is well-defined and continuous. In particular, $\lim_{p \in \gamma([0,1]), p \to p^*} t_{ball}(p) = t_{ball}(p^*) = \infty$.

Remark 2.5.10. It is straightforward to generalize Theorem 2.5.9 to the case where the orbit of \hat{y}_{p^*} converges to a more general controlling invariant set, rather than a controlling critical element. The proof is analogous because t_{ball} depends only on the choice of neighborhood N, as given in Assumption 2.5.4, and not on the nature of the controlling invariant set.

Remark 2.5.11. It is tempting here to attempt to derive the form of a particular function of $t_{ball}(p)$ with a certain (ex. linear) structure, as in Theorem 2.6.13. In particular, computation of the hitting times in Lemma 2.6.18 indicates that a good choice for this function may be $e^{-\operatorname{Re}(\lambda_1(p))t}$ where $\lambda_1(p)$ is the unstable eigenvalue with largest real part of the linearization of the vector field at $X^*(p)$. Unfortunately, the technique used in the proof of Theorem 2.6.13 to conjugate the nonlinear system with the linear system is merely a diffeomorphism, not an isometry, so it is not time preserving. Consequently, it is not clear how to translate these hitting time results back to the nonlinear system using that method. Therefore, this remains an open problem.

2.5.3 Example

We continue Example 2.2.27 of Section 2.2.4.3. Choose two values of p, call them p_1 and p_2 , such that $\hat{y}_{p_1} \in W^s(X_J^s)$ but $\hat{y}_{p_2} \notin W^s(X_J^s)$. In particular, we may choose $p_1 = p_0 = 1.5$ and $p_2 = 2$. Then $\hat{y}_{p_1} = (1.27, 0.99) \in W^s(X_J^s)$ and $\hat{y}_{p_2} = (2.13, 1.32) \notin W^s(X_J^s)$, as could be verified, for example, by numerical integration. Furthermore, \hat{y} is C^1 since ϕ_d is.

Hence, by Theorem 2.5.2 there must exist a point p^* in the recovery boundary such that $\hat{y}_{p^*} \in \partial W^s(X_{p^*}^s)$ and $d(p_0, p^*) = d_s(p_0, \partial R)$. We will see that $p^* = 1.568$ is the desired closest point on the recovery boundary. Since $\partial W^s(X_{p^*}^s) = W^s(X_{p^*}^1)$, this implies that



Figure 2.16: The transversal intersection of several orbits with the ball N containing the unstable equilibria (red stars). Orbits are shown for parameter values (driving torques) of 1.5 (cyan), 1.516 (yellow), 1.532 (green), 1.55 (magenta), 1.568 (blue), and 1.57 (red), where only the initial condition corresponding to the final parameter value lies outside the regions of attraction of the stable equilibria.

 $\hat{y}_{p^*} \in W^s(X^1_{p^*})$, so $X^*_J = X^1_J$. Let $\gamma : [0,1] \to J$ by $\gamma(s) = (1-s)p_0 + sp^*$. Then γ is C^1 , $\gamma(0) = p_0, \ \gamma(1) = p^*$, and $\gamma(s) \notin \partial R$ for $s \in (0,1)$ because γ is a minimal geodesic and $d(p_0, p^*) = d_S(p_0, C) = d_S(p_0, \partial R)$ by Theorem 2.5.1. As $\gamma([0,1))$ is connected, it does not intersect ∂R , and $\gamma(0) \in R$, we must have $\gamma([0,1)) \subset R$.

Let N be the closed ball centered at $X_{p_1}^* = X_{p_1}^1$ of radius r = 1 in \mathbb{R}^2 . Fig. 2.16 shows X_p^* and the orbit of Eqs. 2.8-2.9 for a range of initial conditions \hat{y}_p for $p \in [p_1, p_2]$. In particular, one can infer that the orbit has nonempty, transversal intersection with ∂N for $p \in [p_1, p_2]$. Furthermore, $X_J^* \subset N$ and $\hat{y}_{[p_1, p_2]}$, X_J^s are disjoint from N. Therefore, the path γ defined above satisfies Assumption 2.5.4 so by Theorem 2.5.9 we must have that the time the orbit spends in the neighborhood N, t_{ball} , is well-defined and continuous over $\gamma([0, 1]) = [p_0, p^*]$. Fig. 2.17 illustrates the dependence of t_{ball} on $p \in \gamma([0, 1]) = [p_0, p^*]$. One observes that t_{ball} is continuous and that t_{ball} diverges to infinity as p converges to a fixed value p^* . For $p = p^*$ Fig. 2.5 shows (solid blue) that $\hat{y}_{p^*} \in \partial W^s(X_J^s)$. Furthermore, $p \in \gamma([0, 1)) = [p_0, p^*]$ implies that $\hat{y}_p \in W^s(X_J^s)$. Although t_{ball} is monotonic in this example, this need not be true in general.

2.5.4 Proofs

Proof of Theorem 2.5.1. Note that since the vector fields satisfy either Theorem 2.4.13 (for vector fields with C^1 event-selected discontinuities) or Theorem 2.2.23 (for C^1 vector fields),



Figure 2.17: Distance from the center of the ball N as a function of time for several orbits. The line r = 1 marks the boundary of the ball ∂N , so the time in the ball equals the difference in time between the intersections of the orbit with this line. The orbits shown correspond to those in Fig. 2.16. As the parameter value approaches its critical value from below, the time in the neighborhood N increases. The final parameter value, which is greater than the critical parameter value, has an orbit (red) which spends less time in N than that corresponding to the critical parameter value (blue).

we have that $\partial W^s(X_J^s) = \bigsqcup_{p \in J} \partial W^s(X^s(p))$, which we will call equality of the boundaries. First we show that R, C, and ∂R are nonempty by connectedness of any path from p_1 to p_2 . Then, we prove that every parameter value p^* in ∂R is a critical parameter value since we will see that $\hat{y}_{p^*} \in \partial W^s(X_J^s)$ which will imply, using equality of the boundaries, that $p^* \in C$. Next it is shown that J_0 is nonempty by considering a compact subset of C and arguing that the minimum distance from C to J_0 is achieved by a point in this compact subset. Finally, we argue that $J_0 = \{p \in \partial R : d(p_0, p) = d_S(p_0, \partial R)\}$ by choosing a minimal geodesic from p_0 to any fixed $p^* \in J_0$, and arguing by connectedness that all points of the geodesic other than p^* must lie in R.

First we show that R, C, and ∂R are nonempty. Since $\hat{y}_{p_1} \in W^s(X_{p_1}^s)$, $p_1 \in R$ so R is nonempty. Let $\delta : [0,1] \to J$ be any continuous path in J from p_1 to p_2 , so $\delta(0) = p_1$ and $\delta(1) = p_2$. Such a path exists because J is a connected manifold, hence pathwise connected. As \hat{y} and δ are continuous and [0,1] is connected, $\hat{y}_{\delta([0,1])}$ is connected. Since $\hat{y}_{\delta([0,1])}$ is connected and intersects both $W^s(X_J^s)$ (at \hat{y}_{p_1}) and $M \times J - W^s(X_J^s)$ (at \hat{y}_{p_2}), it must intersect $\partial W^s(X_J^s)$. Hence, there must exist $p^* \in \delta([0,1]) \subset J$ such that $\hat{y}_{p^*} \in \partial W^s(X_J^s)$. By equality of the boundaries, $\partial W^s(X_J^s) = \sqcup_{p \in J} \partial W^s(X^s(p))$. Hence, $\hat{y}_{p^*} \in \partial W^s(X_{p^*}^s)$, so $p^* \in C$. Thus, C is nonempty. As $C \cap R = \emptyset$ and R is nonempty, this implies that ∂R is nonempty. Next we show that $\partial R \subset C$. So, let $p^* \in \partial R$. Then there exists a sequence $p_n \in R$ with $p_n \to p^*$. Hence, by definition of R, $(\hat{y}_{p_n}, p_n) \in W^s(X_{p_n}^s)$ for all n with $(\hat{y}_{p_n}, p_n) \to (\hat{y}_{p^*}, p^*)$ since \hat{y} is C^1 and $p_n \to p^*$. In particular, $(\hat{y}_{p_n}, p_n) \in W^s(X_{p_n}^s) \subset \overline{W^s}(X_J^s)$ for all n. As $\overline{W^s}(X_J^s)$ is closed and $(\hat{y}_{p_n}, p_n) \to (\hat{y}_{p^*}, p^*)$, this implies that $(\hat{y}_{p^*}, p^*) \in \overline{W^s}(X_J^s)$. By equality of the boundaries, $\overline{W^s}(X_J^s) = \bigsqcup_{p \in J} \overline{W^s}(X^s(p))$. Hence, $\hat{y}_{p^*} \in \overline{W^s}(X_{p^*}^s)$. First assume towards a contradiction that $\hat{y}_{p^*} \in W^s(X_{p^*}^s)$. Let U be an open neighborhood of $X_{p^*}^s$ such that $\overline{U} \subset \operatorname{int} W^s_{\operatorname{loc}}(X_{p^*}^s)$. Then there exists T > 0 such that $\phi_T(\hat{y}_{p^*}) \in U$. As int $W^s_{\operatorname{loc}}(X^s(p))$ varies C^1 with parameter p, there exists an open neighborhood J' of p^* in J such that $p \in J'$ implies that $U \subset \operatorname{int} W^s_{\operatorname{loc}}(X^s(p))$. As U is open in M and both ϕ_T and \hat{y} are C^1 , shrinking J' if necessary implies that for $p \in J', \phi_T(\hat{y}_p) \in U \subset \operatorname{int} W^s_{\operatorname{loc}}(X^s(p))$. Hence, J' is an open neighborhood of p^* in J such that $J' \subset R$. But, this contradicts that $p^* \in \partial R$. So, since $p^* \in \overline{W^s}(X_{p^*}^s)$ but $p^* \notin W^s(X_{p^*}^s)$, we must have $p^* \in \partial W^s(X_{p^*}^s)$. Hence, $p^* \in C$.

We proceed by showing that J_0 is nonempty. So, fix $p_0 \in J$ and let J_0 be the set of critical parameter values $p^* \in C$ such that $d(p_0, p^*) = d_S(p_0, C)$, so $J_0 = \{p^* \in C : d(p_0, p^*) = d_S(p_0, C), s_0\}$ $d_S(p_0, C)$. First we will show that J_0 is nonempty. By the above, C is nonempty, so let $\hat{p} \in C$. Then $d_S(p_0, C) \leq d(p_0, \hat{p}) < \infty$. Shrink J slightly if necessary so that J is contained in what was originally J, and still $p_0, \hat{p} \in J$. Let B be the closed ball centered at p_0 of radius $d(p_0, \hat{p})$ in J, and note that B is compact. As \hat{y} is continuous, \hat{y}_B is compact, so \hat{y}_B is closed in $M \times J$. As $\partial W^s(X_J^s)$ is closed in $M \times J$, $\partial W^s(X_J^s) \cap \hat{y}_B$ is closed in $M \times J$. Let $C' = \hat{y}^{-1}(\partial W^s(X_J^s) \cap \hat{y}_B)$. Since \hat{y} is continuous, C' is closed in J. As $\hat{p} \in C'$, $d_S(p_0, C') \leq d(p_0, \hat{p}) < \infty$. So, since C' is closed and p_0 is compact, there exists $p^* \in C'$ such that $d(p_0, p^*) = d_S(p_0, C')$. However, we claim that $C' = \{p' \in C : d(p_0, p') \le d(p_0, \hat{p})\}$. For, $p' \in C'$ implies that $\hat{y}_{p'} \in \partial W^s(X_J^s)$. But, by equality of the boundaries, $\partial W^s(X_J^s) =$ $\sqcup_{p \in J} \partial W^s(X^s(p))$. So, $\hat{y}_{p'} \in \partial W^s(X^s_{p'})$ and, hence, $p' \in C$. Furthermore, $p' \in C'$ implies that $p' \in B$ and therefore, by definition of B, that $d(p_0, p') \leq d(p_0, \hat{p})$. Similarly, if $p' \in C$ with $d(p_0, p') \leq d(p_0, \hat{p})$, then $p' \in B$ by definition of B and $\hat{y}_{p'} \in \partial W^s(X^s_{p'}) \subset \partial W^s(X^s_J)$. Hence, $\hat{y}_{p'} \in \partial W^s(X_J^s) \cap \hat{y}_B$ so $p' \in C'$. Since $C' = \{p' \in C : d(p_0, p') \le d(p_0, \hat{p})\}$, this implies that $d_S(p_0, C) = d_S(p_0, C') = d(p_0, p^*)$ and that $p^* \in C' \subset C$. Hence, $p^* \in J_0$ so J_0 is nonempty.

Finally, we show that $J_0 = \{p \in \partial R : d(p_0, p) = d_S(p_0, \partial R)\}$. Let $p^* \in J_0$. Let $\gamma : [0,1] \to J$ be a minimal geodesic from p_0 to p^* , so $\gamma(0) = p_0$, $\gamma(1) = p^*$, and the length of γ is equal to $d(p_0, p^*)$. For example, if J was a convex subset of Euclidean space then the image of γ would be the line segment between p_0 and p^* . For every $x \in [0,1)$, by definition of a minimal geodesic, $d(p_0, \gamma(x)) < d(p_0, \gamma(1)) = d(p_0, p^*) = d_S(p_0, C)$, where the last equality follows since $p^* \in J_0$. This implies that for every $x \in [0, 1)$, $\gamma(x) \notin C$, since

otherwise we would have that $d_S(p_0, C) \leq d(p_0, \gamma(x)) < d(p_0, p^*)$, which would contradict that $p^* \in J_0$ (so $d_S(p_0, C) = d(p_0, p^*)$). Hence, $\gamma([0, 1)) \cap C = \emptyset$. Furthermore, since [0, 1)is connected and both γ and \hat{y} are continuous, $\hat{y}_{\gamma([0,1))}$ is connected. Assume towards a contradiction that there exists $x \in [0, 1)$ such that $\hat{y}_{\gamma(x)} \notin W^s(X_{\gamma(x)}^s)$. As $\hat{y}_{p_1} \in W^s(X_J^s)$, $\hat{y}_{\gamma(x)} \notin W^s(X_J^s)$, and $\hat{y}_{p_1}, \hat{y}_{\gamma(x)} \in \hat{y}_{\gamma([0,1))}$ connected, we must have $\hat{y}_{\gamma([0,1))} \cap \partial W^s(X_J^s) \neq \emptyset$. So, there exists $x' \in [0, 1)$ such that $\hat{y}_{\gamma(x')} \in \partial W^s(X_J^s)$. By equality of the boundaries, $\partial W^s(X_J^s) = \bigsqcup_{p \in J} \partial W^s(X^s(p))$. In particular, $\hat{y}_{\gamma(x')} \in \partial W^s(X_{\gamma(x')}^s)$. This implies that $\gamma(x') \in$ C. But, this contradicts that $\gamma([0,1)) \cap C = \emptyset$. So, we must have $\hat{y}_{\gamma(x)} \in W^s(X_{\gamma(x)}^s)$ for all $x \in [0,1)$. Hence, $\gamma([0,1)) \subset R$. Let $p_n = \gamma \left(1 - \frac{1}{n}\right)$. Then $p_n \in R$ with $p_n \to p^*$, so $p^* \in \partial R$. As $\partial R \subset C$ by the above, $d_S(p_0, \partial R) \geq d_S(p_0, C) = d(p_0, p^*)$. As $p^* \in \partial R, d_S(p_0, \partial R) \leq$ $d(p_0, p^*)$. Hence, combining these inequalities we have that $d_S(p_0, \partial R) = d(p_0, p^*)$. As $p^* \in J_0$ was arbitrary, this implies that $J_0 = \{p^* \in \partial R : d(p_0, p^*) = d_S(p_0, \partial R)\}$.

Proof of Corollary 2.5.2. Fix $p^* \in \partial R$. By Theorem 2.5.1, $\partial R \subset C$ so $p^* \in C$. Hence, by definition of C, $\hat{y}_{p^*} \in \partial W^s(X_{p^*}^s)$. By Theorem 2.2.23 (for C^1 vector fields) or Theorem 2.4.13 (for vector fields with C^1 event-selected discontinuities), $\partial W^s(X_{p^*}^s) = \bigcup_{i \in I} W^s(X_{p^*}^i)$. Thus, $\hat{y}_{p^*} \in \partial W^s(X_{p^*}^s)$ implies there exists a unique $j \in I$ such that $\hat{y}_{p^*} \in W^s(X_{p^*}^j)$. Let $X_J^* = X_J^j$ be the controlling critical element. Then $\hat{y}_{p^*} \in W^s(X_{p^*}^s)$.

Lemma 2.5.12. Let $\gamma : [0,1] \to J$ be a path that satisfies Assumption 2.5.4 with embedded submanifold N. For $p \in \gamma([0,1])$ let $T_p \subset [0,\infty)$ denote the set of times $\{t \in [0,\infty) : \phi_t(\hat{y}_p) \in N\}$. Then for $p \in \gamma([0,1])$, T_p consists of a finite union of closed intervals, so $t_{ball}(p)$ is welldefined and finite. For $p = \gamma(1)$, T_p consists of a finite union of closed intervals together with an interval of the form $[t',\infty)$ for some t' > 0, so $t_{ball}(p) = \infty$ is well-defined. For $p = \gamma(1)$, if the vector fields are C^1 event-selected discontinuous and the controlling critical element intersects at least one switching surface, then T_p consists of a countably infinite union of closed intervals, and $t_{ball}(p) = \infty$ is well-defined.

Proof of Lemma 2.5.12. We begin by proving the results in the case that the vector fields are C^1 , and subsequently illustrate the changes necessary in the case where the vector fields are C^1 event-selected discontinuous. First we show that the forward orbit of \hat{y}_p under V_p is a one-dimensional C^1 embedded submanifold. This will imply, since this orbit is transverse to ∂N and N, that its intersection with N is a one-dimensional C^1 embedded submanifold with boundary equal to its intersection with ∂N , which is a zero-dimensional C^1 embedded submanifold. By compactness, this manifold boundary consists of a finite number of points. Then the formulas for T_p are obtained by considering the connected components of a onedimensional manifold in $[0, \infty)$.

First we show that for any $p \in \gamma([0,1]), \phi(\cdot, \hat{y}_p, p)^{-1} \left(\phi_{(0,T)}(\hat{y}_p) \cap N\right)$ is a one-dimensional embedded submanifold with boundary equal to $\phi(\cdot, \hat{y}_p, p)^{-1} \left(\phi_{(0,T)}(\hat{y}_p) \cap \partial N \right)$, which consists of a finite union of points. So, let $p \in \gamma([0,1])$. If $p = \gamma(1)$ then let $X_J = X_J^*$ where X_J^* is the controlling critical element corresponding to $\gamma(1)$ as in Corollary 2.5.2, and choose $W^s_{\rm loc}(X^*_{\gamma(1)})$ sufficiently small so that it is contained in N. Otherwise, let $X_J = X^s_J$ and choose $W^s_{\text{loc}}(X^s_J)$ sufficiently small so that it is disjoint from N. Then the orbit of \hat{y}_p under V_p converges to X_p , so there exists T > 0 such that $\phi_T(\hat{y}_p) \in \operatorname{int} W^s_{\operatorname{loc}}(X_p)$. Hence, by definition of the local stable manifold, $t \geq T$ implies that $\phi_t(\hat{y}_p) \in \operatorname{int} W^s_{\operatorname{loc}}(X_p)$. As $\hat{y}_p \in W^s(X_p)$ but $\hat{y}_p \notin X_p$, the forward orbit of \hat{y}_p under V_p does not contain any critical elements, so $\phi(\cdot, \hat{y}_p, p)$ is an injective C^1 immersion from $[0,\infty)$ into M. As $\phi(\cdot,\hat{y}_p,p)$ is a continuous bijection onto its image, [0,T] is compact, and M is Hausdorff, $\phi(\cdot, \hat{y}_p, p)$ is a homeomorphism from [0,T] onto $\phi_{[0,T]}(\hat{y}_p)$. Hence, $\phi(\cdot, \hat{y}_p, p)$ is a C^1 embedding from [0,T] onto its image, so $\phi_{(0,T)}(\hat{y}_p)$ is a C^1 embedded submanifold in M and $\phi(\cdot, \hat{y}_p, p)$ is a C^1 diffeomorphism from (0,T) onto $\phi_{(0,T)}(\hat{y}_p)$. By Assumption 2.5.4, $\phi_{(0,T)}(\hat{y}_p)$ is transverse to ∂N , and it is trivially transverse to the interior of N since the dimension of N is equal to the dimension of M. Therefore, $\phi_{(0,T)}(\hat{y}_p) \cap N$ is a one dimensional C^1 embedded submanifold with boundary equal to $\phi_{(0,T)}(\hat{y}_p) \cap \partial N$ a zero dimensional C^1 embedded submanifold. Since $\hat{y}_p, \phi_T(\hat{y}_p) \notin$ $\partial N, \phi_{(0,T)}(\hat{y}_p) \cap \partial N = \phi_{[0,T]}(\hat{y}_p) \cap N.$ Hence, N compact and $\phi_{[0,T]}(\hat{y}_p)$ compact implies that their intersection $\phi_{[0,T]}(\hat{y}_p) \cap N$ is compact. So, the above implies that $\phi_{(0,T)}(\hat{y}_p) \cap \partial N$ is a compact zero dimensional embedded submanifold. As zero dimensional manifolds are discrete, this implies that $\phi_{(0,T)}(\hat{y}_p) \cap \partial N$ consists of a finite union of points. As $\phi(\cdot, \hat{y}_p, p)$ is a C^1 diffeomorphism from (0,T) onto $\phi_{(0,T)}(\hat{y}_p), \ \phi(\cdot,\hat{y}_p,p)^{-1}(\phi_{(0,T)}(\hat{y}_p)\cap N)$ is a onedimensional embedded submanifold with boundary equal to $\phi(\cdot, \hat{y}_p, p)^{-1} (\phi_{(0,T)}(\hat{y}_p) \cap \partial N)$, which consists of a finite union of points.

Next, we show that t_{ball} is well-defined and finite for $p \in \gamma([0,1))$. So, suppose $p \in \gamma([0,1))$. Then $\hat{y}_p, \phi_T(\hat{y}_p) \notin N$, so $\phi_{(0,T)}(\hat{y}_p) \cap N = \phi_{[0,T]}(\hat{y}_p) \cap N$ is compact as $\phi_{[0,T]}(\hat{y}_p)$ and N are compact. Thus, as $\phi(\cdot, \hat{y}_p, p)$ is a C^1 diffeomorphism from (0,T) onto $\phi_{(0,T)}(\hat{y}_p)$, this implies that $\phi(\cdot, \hat{y}_p, p)^{-1}(\phi_{(0,T)}(\hat{y}_p) \cap N)$ is a compact one-dimensional embedded submanifold in (0,T) with boundary consisting of a finite number of points. Since it is a compact one-dimensional manifold, it has finitely many connected components and each contains its manifold boundary. Hence, $\phi(\cdot, \hat{y}_p, p)^{-1}(\phi_{(0,T)}(\hat{y}_p) \cap N)$ consists of a finite union of closed intervals. As $\phi_t(\hat{y}_p) \notin N$ for all $t \geq T$, this implies that $T_p = \{t \in [0,\infty) : \phi_t(\hat{y}_p) \in N\}$

consists of this finite union of closed intervals. So, $t_{ball}(p) = \lambda(T_p)$, where λ is the Lebesgue measure, is well-defined and is equal to the sum of the lengths of each such interval, which is finite since the intervals are contained in [0, T] which has finite length T.

Finally, we show that $t_{ball}(\gamma(1)) = \infty$. So, let $p = \gamma(1)$. For $t \geq T$, $\phi_T(\hat{y}_p) \in$ int $W^s_{\text{loc}}(X^*_{\gamma(1)}) \subset \text{int } N$, so $\phi_t(\hat{y}_p) \in \text{int } N$ for all $t \geq T$. In particular, $\phi_t(\hat{y}_p) \notin \partial N$ for all $t \geq T$. As $\phi(\cdot, \hat{y}_p, p)^{-1} \left(\phi_{(0,T)}(\hat{y}_p) \cap \partial N \right)$ consists of a finite number of points, let t' be the largest value in this set. Then t' represents the final intersection of the forward orbit of \hat{y}_p under V_p with ∂N since, by the above, no further interesections occur for $t \geq T$. We claim that for all $t \in [t', T], \phi_t(\hat{y}_p) \in N$. For, assume towards a contradiction that the claim is false. Then there exists $\hat{t} \in (t',T)$ with $\phi_{\hat{t}}(\hat{y}_p) \notin N$. As $\phi_{[\hat{t},T]}(\hat{y}_p)$ is connected with $\phi_{\hat{t}}(\hat{y}_p) \notin N$, $\phi_T(\hat{y}_p) \in N$, and N connected, there must exist $t'' \in [\hat{t}, T)$ such that $\phi_{t''}(\hat{y}_p) \in \partial N$. But, t'' > t' with $\phi_{t''}(\hat{y}_p) \in \partial N$, so this contradicts that t' was the final intersection of the forward orbit of \hat{y}_p under V_p with ∂N . Hence, $[t', \infty) \subset T_p$. As t' is a manifold boundary point for $\phi(\cdot, \hat{y}_p, p)^{-1} \left(\phi_{(0,T)}(\hat{y}_p) \cap N \right)$, there exists $\epsilon > 0$ such that $[t' - \epsilon, t') \cap T_p = \emptyset$. Hence, $\phi_{(0,t')}(\hat{y}_p) \cap N = \phi_{[0,t'-\epsilon]}(\hat{y}_p) \cap N$ is an intersection of two compact sets, hence compact. Thus, $\phi(\cdot, \hat{y}_p, p)^{-1}\left(\phi_{(0,t')}(\hat{y}_p) \cap N\right)$ is a compact one-dimensional embedded submanifold in (0, t')with boundary consisting of a finite number of points. Hence, $\phi(\cdot, \hat{y}_p, p)^{-1} \left(\phi_{(0,t')}(\hat{y}_p) \cap N \right)$ is a finite union of closed intervals. Therefore, T_p is the union of $[t', \infty)$ with a finite union of closed intervals. So, $t_{ball}(p)$ is well-defined with $t_{ball}(p) = \infty$.

Consider now the case where the vector fields are event-selected discontinuous. First suppose that the controlling critical element does not intersect any switching surfaces. Then, by Remark 2.5.5, $N \times J \subset C^J$ is contained in the same connected component of C^J as X_J^* . By Lemma 2.4.16, the flow restricted to this component is C^1 . Thus, by intersecting the orbit with this component of C^J , the proof reduces to the corresponding proof for C^1 vector fields shown above.

Now consider the case where the controlling critical element is a periodic orbit which has nonempty intersection with at least one switching surface. For $p \in \gamma([0, 1))$, the proof proceeds exactly as for the case where $X_J^* \subset C^J$ discussed above. For $p = \gamma(1)$, the argument above shows that for any T > 0, $T_p \cap [0, T]$ consists of a finite union of closed intervals. As the orbit corresponding to $\gamma(1)$ converges to $X_{\gamma(1)}^*$, and since $X_{\gamma(1)}^*$ is not contained in N, for any T > 0 there exists t > T such that $\phi_t(\hat{y}_p) \in \partial N$. Together, the above imply that $T_{\gamma(1)}$ is a countably infinite union of closed intervals. Hence, $t_{ball}(\gamma(1))$ is well-defined and equal to the countable sum of the lengths of these intervals. Let \hat{t} be the length of time the orbit $X_{\gamma(1)}^*$ spends in N during one period and let $\epsilon > 0$ small. For $p = \gamma(1)$, as the orbit of
$\phi_t(\hat{y}_p)$ converges to X_p^* , for T > 0 sufficiently large, the infimum of the lengths of the closed intervals in $T_p \cap [T, \infty)$ is at least $\hat{t} - \epsilon$. Since there are infinitely many such intervals in $[T, \infty), t_{ball}(\gamma(1)) \ge \int_T^\infty \mathbf{1}_N(\phi(t, \hat{y}_p, p)) dt \ge (\hat{t} - \epsilon)n$ for all n > 0. Thus, $t_{ball}(\gamma(1)) = \infty$. \Box

Lemma 2.5.13. Let $\gamma : [0,1] \to J$ be a path that satisfies Assumption 2.5.4 with embedded submanifold N. Then $\lim_{p\to\gamma(1),p\in\gamma([0,1])} t_{ball}(p) = \infty$.

Proof of Lemma 2.5.13. By Lemma 2.5.12, $t_{ball}(\gamma(1)) = \infty$. Let K > 0. First we prove the result here for the case of C^1 vector fields or vector fields with C^1 event-selected discontinuities when the controlling critical element does not intersect any switching surfaces. We then prove the case where the controlling critical element does intersect at least one switching surface. So, assume that $X_J^* \subset C^J$. By the proof of Lemma 2.5.12, there exists a final time $t' \in [0,\infty)$ such that $\phi_{t'}(\hat{y}_{\gamma(1)}) \in \partial N$, and t > t' implies that $\phi_t(\hat{y}_{\gamma(1)}) \in int N$. Choose any $\epsilon > 0$. Then $\phi_{[t'+\epsilon,t'+\epsilon+K]}(\hat{y}_{\gamma(1)}) \subset \text{int } N$. As ∂N and $\phi_{[t'+\epsilon,t'+\epsilon+K]}(\hat{y}_{\gamma(1)})$ are compact and disjoint in M a normal space, there exist disjoint open neighborhoods U and \hat{U} in M such that $\partial N \subset U$ and $\phi_{[t'+\epsilon,t'+\epsilon+K]}(\hat{y}_{\gamma(1)}) \subset \hat{U}$. Shrink \hat{U} if necessary so that $\hat{U} \subset \text{int } N$. As \hat{U} is open in M and $\phi([t'+\epsilon, t'+\epsilon+K], \hat{y}_p, p)$ is compact and C^1 continuous with respect to p, then for $\delta > 0$ sufficiently small, $p \in \gamma((1 - \delta, 1])$ implies that $\phi_{[t'+\epsilon, t'+\epsilon+K]}(\hat{y}_p) \subset \hat{U} \subset \text{int } N$. So, $[t' + \epsilon, t' + \epsilon + K] \subset T_p$. By the proof of Lemma 2.5.12, T_p consists of a finite union of closed intervals, and $t_{ball}(p)$ is equal to the sum of the lengths of these intervals. Hence, $t_{ball}(p)$ is at least as large as the length of the closed interval that contains $[t' + \epsilon, t' + \epsilon + K]$, which is at least length K. As $t_{ball}(p) \ge K$ for all $p \in \gamma((1-\delta), 1]$ and K > 0 was arbitrary, $\lim_{p \in \gamma([0,1]), p \to \gamma(1)} t_{ball}(p) = \infty.$

Suppose now that the controlling critical element intersects at least one switching surface. By Lemma 2.5.12 $T_{\gamma(1)}$ consists of a countably infinite union of closed intervals whose total length is infinite. Choose any $\epsilon > 0$ and let T > 0 such that $\lambda(T_{\gamma(1)} \cap [0,T]) \ge K + \epsilon$, where λ is the Lebesgue measure. By an analogous argument to the proof of Lemma 2.5.14, using Lemma 2.4.18 to ensure the intersection times of the flow with ∂N depend C^1 on parameter, it can be shown that there exists $\delta > 0$ sufficiently small such that the following holds. For $p \in \gamma((1-\delta, 1])$, the intersection times of $\phi_t(\hat{y}_p)$ with ∂N are the same number and sufficiently close to the intersection times of $\phi_t(\hat{y}_{\gamma(1)})$ with ∂N that $\lambda(T_p \cap [0,T]) \ge K$. Hence, $t_{ball}(p) \ge \lambda(T_p \cap [0,T]) \ge K$ for all $p \in \gamma((1-\delta,1))$. As $t_{ball}(p) \ge K$ for all $p \in \gamma((1-\delta), 1]$ and K > 0 was arbitrary, $\lim_{p \in \gamma([0,1]), p \to \gamma(1)} t_{ball}(p) = \infty$.

Lemma 2.5.14. Let $\gamma : [0,1] \to J$ be a path that satisfies Assumption 2.5.4 with embedded submanifold N. Then t_{ball} is continuous over $\gamma([0,1))$



Figure 2.18: For a fixed parameter value $p \in \gamma([0, 1))$, the intersection of the orbit of \hat{y}_p (red and black line segments) with the embedded submanifold with boundary N (cyan ellipse) containing X_p^* (red star), an equilibrium point in this example, is shown. There are a finite number of intersections of the orbit of \hat{y}_p with ∂N (black stars). The orbit is a union of line segments of the form $\phi(T_i, \hat{y}_p, p)$ (black), which contain the intersection points, and line segments of the form $\phi(T'_i, \hat{y}_p, p)$ (red), which are compact, contain no intersection points, and intersect the black line segments on each end (although this intersection is not visible in the figure). This figure originally appeared in [FH17].

Proof of Lemma 2.5.14. We first prove the result in the case where the vector fields are C^1 , and then extend it to vector fields with C^1 event-selected discontinuities below. So, assume the vector fields are C^1 . Fix $s \in [0, 1)$. To show continuity of t_{ball} over $\gamma([0, 1))$ it suffices to show that it is continuous over a neighborhood of $\gamma(s)$ in $\gamma([0,1))$. Let $\epsilon > 0$. The proof proceeds by first showing that there exists T > 0 such that $T_{\gamma(s')} \subset [0, T]$ for s' close to s since $\phi_T(\hat{y}_{\gamma(s')}) \in W^s_{\text{loc}}(X^s_{\gamma(s')})$. Then, by stability of transversal intersections and the implicit function theorem, it is shown that for every intersection point of the orbit of $\hat{y}_{\gamma(s)}$ under $V_{\gamma(s)}$ with ∂N , s' close to s implies that there exists a unique intersection point of the orbit of $\hat{y}_{\gamma(s')}$ under $V_{\gamma(s')}$ with ∂N near the original intersection point. It then argues that for s' close to s, no new intersection points appear beyond the perturbations of the original intersection points. As T_p is equal to a finite union of closed intervals whose boundaries are equal to these intersection points by Lemma 2.5.12, it will be shown that there is a one-to-one correspondence between the closed intervals in $T_{\gamma(s)}$ and the closed intervals in $T_{\gamma(s')}$. As $t_{ball}(\gamma(s'))$ is equal to the sum of the lengths of these closed intervals, there are finitely many of them, and their lengths vary continuously with parameter value since their endpoints (the intersection times) vary continuously with parameter value, it will follow that $|t_{ball}(\gamma(s')) - t_{ball}(\gamma(s))| < \epsilon$ for s' close to s. This proof is illustrated with the aid of Fig. 2.18.

First we show that $T_p \subset [0,T)$ for some T > 0 and p close to $\gamma(s)$. As $\gamma(s) \in R$, the forward orbit of $\hat{y}_{\gamma(s)}$ under $V_{\gamma(s)}$ converges to $X^s_{\gamma(s)}$. So, there exists T > 0 such that $\phi_T(\hat{y}_{\gamma(s)}) \in \operatorname{int} W^s_{\operatorname{loc}}(X^s_{\gamma(s)})$. As $W^s_{\operatorname{loc}}(X^s(p))$ is open and varies C^1 with $p \in J$, and as $\phi_T(\hat{y}_p)$ varies C^1 with $p \in J$, there exists $\delta > 0$ such that $p \in \gamma((s - \delta, s + \delta))$ implies that $\phi_T(\hat{y}_p) \in \operatorname{int} W^s_{\operatorname{loc}}(X^s(p))$. Similarly, choosing $W^s_{\operatorname{loc}}(X^s_{\gamma(s)})$ sufficiently small implies that for $p \in \gamma((s - \delta, s + \delta)), W^s_{\operatorname{loc}}(X^s(p))$ is disjoint from N. Hence, $p \in \gamma((s - \delta, s + \delta))$ implies that for any $t \geq T$, $\phi_t(\hat{y}_p) \in \operatorname{int} W^s_{\operatorname{loc}}(X^s(p))$ which is disjoint from N. Therefore, $T_p \subset [0, T)$.

Next, persistence of the original intersection points is shown under small changes in parameter values. By Lemma 2.5.12, there are a finite number of intersections of $\phi_{[0,T]}(\hat{y}_{\gamma(s)})$ with ∂N . Let t_i denote the *i*th intersection time of the orbit of $\hat{y}_{\gamma(s)}$ under $V_{\gamma(s)}$ with ∂N . Note that $\{\phi_{[0,T]}(\hat{y}_p)\}_{p\in\gamma([0,1])}$ is a C^1 continuous family of compact embedded submanifolds with boundary in M with $\phi_{[0,T]}(\hat{y}_{\gamma(s)})$ transverse to ∂N . From the proof that points of transversal intersection between compact embedded submanifolds with boundary persist under C^1 perturbations [KH99, Proposition A.3.16], it follows by the implicit function theorem that for $\delta > 0$ sufficiently small, there exist open neighborhoods $T_i \subset [0,T]$ and C^1 functions $\tau_i : \gamma((s - \delta, s + \delta)) \to T_i$ such that the following holds. For $p \in \gamma((s - \delta, s + \delta))$, $\phi(\cdot, \hat{y}_p, p)^{-1}(\phi_{T_i}(\hat{y}_p) \cap \partial N) = \{\tau_i(p)\}$ and $\tau_i(\gamma(s)) = t_i$ for each *i*. In other words, for each $p \in \gamma((s - \delta, s + \delta))$ and for each *i*, there exists a unique intersection of $\phi_{[0,T]}(\hat{y}_p)$ with ∂N that occurs in the time interval T_i .

It is shown next that for δ sufficiently small, and for $p \in \gamma((s - \delta, s + \delta))$, the number of intersection times of $\phi_{[0,T]}(\hat{y}_p)$ with ∂N is constant and each intersection time varies continuously with the parameter values. Let m be the number of intersections of $\phi_{[0,T]}(\hat{y}_{\gamma(s)})$ with ∂N . Let $T'_i \subset [0,T]$ be a connected closed interval for each i such that $T'_i \cap T_{i-1} \neq \emptyset$, $T'_i \cap T_i \neq \emptyset$, and T'_i does not contain any times of intersection of $\phi_{[0,T]}(\hat{y}_{\gamma(s)})$ with ∂N , where we let $T_0 = \{0\}$ and $T_{m+1} = \{T\}$. For each i, as $\phi_{T'_i}(\hat{y}_{\gamma(s)})$ and ∂N are compact and disjoint, and since $\phi(T'_i, \hat{y}_p, p)$ varies C^1 with respect to p, shrinking δ further if necessary implies that for $p \in \gamma((s - \delta, s + \delta)), \phi_{T'_i}(\hat{y}_p)$ is disjoint from ∂N for all i. More specifically, $\phi_{T'_i}(\hat{y}_p) \subset N$ if and only if $\phi_{T'_i}(\hat{y}_{\gamma(s)}) \subset N$. We can write $[0,T] = \bigcup_{i=1}^m T_i \bigcup_{i=1}^{m+1} T'_i$, as shown in Fig. 2.18. Hence, for $p \in \gamma((s - \delta, s + \delta))$, the only intersection times of $\phi_{[0,T]}(\hat{y}_p)$ with ∂N occur in $\bigcup_{i=1}^m T_i$. But, by the choice of the T_i above, for $p \in \gamma((s - \delta, s + \delta))$ this implies that the only intersection times of $\phi_{[0,T]}(\hat{y}_p)$ with ∂N are $\bigcup_{i=1}^m \tau_i(p)$. By the choice of T above, this implies that for $p \in \gamma((s - \delta, s + \delta))$, the only intersection times of the orbit of \hat{y}_p under V_p with ∂N are $\bigcup_{i=1}^m \tau_i(p)$. Hence, for $p \in \gamma((s - \delta, s + \delta))$, the number of intersections of the orbit of \hat{y}_p under V_p with ∂N is constant.

Finally, we show that for $p \in \gamma((s - \delta, s + \delta))$, there is a one-to-one correspondence between the closed intervals in T_p and the closed intervals in $T_{\gamma(s)}$, where the interval lengths can be brought arbitrarily close to each other for sufficiently small δ , so we will conclude that $|t_{ball}(p) - t_{ball}(\gamma(s))| < \epsilon$. By Lemma 2.5.12, as $(s - \delta, s + \delta) \subset [0, 1)$, for any p in $\gamma((s-\delta,s+\delta)), T_p$ consists of a finite union of closed intervals whose boundary points are the intersection times. Then for each i, $[\tau_{i-1}(p), \tau_i(p)] \subset T_p$ if and only if $\phi_{T'_i}(\hat{y}_p) \subset N$ since $T'_i \subset [\tau_{i-1}(p), \tau_i(p)], \phi_{[\tau_{i-1}(p), \tau_i(p)]}(\hat{y}_p)$ is connected, and there are no intersections of the orbit of \hat{y}_p under V_p with ∂N in the time interval $(\tau_{i-1}(p), \tau_i(p))$. Hence, for each $i, [\tau_{i-1}(p), \tau_i(p)] \subset T_p$ if and only if $\phi_{T'_i}(\hat{y}_p) \subset N$ if and only if $\phi_{T'_i}(\hat{y}_{\gamma(s)}) \subset N$ if and only if $[\tau_{i-1}(\gamma(s)), \tau_i(\gamma(s))] \subset T_{\gamma(s)}$. Therefore, since the number of intersection times is constant over $p \in \gamma((s - \delta, s + \delta))$, for each $i \in \{1, ..., m\}$, $[\tau_{i-1}(\gamma(s)), \tau_i(\gamma(s))] \subset T_{\gamma(s)}$ if and only if $[\tau_{i-1}(p), \tau_i(p)] \subset T_p$. Therefore, T_p and $T_{\gamma(s)}$ consist of the same finite number of corresponding closed intervals which differ only slightly in their endpoints, which are the intersection times $\{\tau_i(p)\}_{i=1}^m$. Shrink δ such that for $p \in \gamma((s - \delta, s + \delta))$ and each i, $|\tau_i(p) - t_i| < \frac{\epsilon}{2m}$. Since for $p \in \gamma((s - \delta, s + \delta))$ the *i*th boundary point $\tau_i(p)$ is within $\frac{\epsilon}{2m}$ of $t_i = \tau_i(\gamma(s))$ for each *i*, and since T_p consists of the same number of corresponding closed intervals as $T_{\gamma(s)}$, each closed interval in T_p has length within $\frac{\epsilon}{m}$ of the length of the corresponding interval in $T_{\gamma(s)}$. For $p \in \gamma((s - \delta, s + \delta))$, as $t_{ball}(p)$ is equal to the sum of the lengths of the closed intervals in T_p , there are m such closed intervals in T_p , and the length of each closed interval in T_p is within $\frac{\epsilon}{m}$ of the corresponding interval in $T_{\gamma(s)}$, $|t_{ball}(p) - t_{ball}(\gamma(s))| < \epsilon$. Hence, t_{ball} is continuous at $\gamma(s)$.

Next, suppose the vector fields are C^1 event-selected discontinuous. As $N \times J \subset C^J$ is contained in a single connected component of C^J , the flow restricted to C^J is C^1 by Lemma 2.4.16, and by Lemma 2.4.18 the behavior of the flow after a switching surface depends C^1 on the behavior before the switching surface, the implicit function theorem can be applied analogously to the above to conclude that the intersection times $\{\tau_i(p)\}_{i=1}^m$ are C^1 with respect to parameter. The rest of the proof proceeds analogously.

Proof of Theorem 2.5.9. Fix $p_0 \in J$, $p^* \in \partial R$, and let $\gamma : [0,1] \to J$ be a C^1 path satisfying Assumption 2.5.4 and such that $\gamma(0) = p_0$ and $\gamma(1) = p^*$. By Lemma 2.5.12, $t_{ball} : \gamma([0,1]) \to [0,\infty]$ is well-defined and $t_{ball}(\gamma(1)) = \infty$. By Lemma 2.5.13, $\lim_{p \in \gamma([0,1]), p \to p^*} t_{ball}(p) = \infty = t_{ball}(\gamma(1))$, so t_{ball} is continuous at $p^* = \gamma(1)$. By Lemma 2.5.14, t_{ball} is continuous over $\gamma([0,1])$.

2.5.5 Conclusion

This section provided motivation for the algorithms described in Section 3.2 both for the case of C^1 vector fields and for vector fields with C^1 event-selected discontinuities. It was first shown that the recovery boundary consists entirely of critical parameter values, and that the closest critical parameter value to any recovery value lies on the recovery boundary. This serves to justify the approach of computing (often the closest) critical parameter values in order to determine the recovery boundary. Furthermore, it was shown that, under the assumption of transversal intersection, for any point in the recovery boundary, the time the orbit spends in a neighborhood of (at least one point in) the controlling invariant set is continuous over the set of recovery values. This provides motivation for algorithms which compute critical parameter values by varying parameter values so as to maximize the time the orbit spends in a neighborhood of the controlling invariant set.

2.6 Theoretical Motivation for Maximizing Trajectory Sensitivities Algorithm

2.6.1 Introduction

The purpose of this section is to provide theoretical motivation for the algorithms described in Section 3.3. The initial results of Section 2.5 motivate the approach of computing critical parameter values in order to determine the recovery boundary. The class of algorithms mentioned above compute critical parameter values by varying parameter values so as to maximize the sensitivity of the system trajectory with respect to parameters. For numerical purposes, it is advantageous to accomplish this by varying parameter values so as to minimize the inverse trajectory sensitivities. To motivate these algorithms, it is shown here that the infimum over time of the inverse trajectory sensitivities is continuous and strictly positive over the set of recovery values, is identically zero on the recovery boundary, and converges to zero as it approaches the recovery boundary from within the set of recovery values, it is shown that the infimum of the inverse trajectory sensitivities approaches a dot product, motivating algorithms which exploit this linear structure.

2.6.2 Results

Consider the same setting as in Section 2.5, as described in detail in Section 2.5.2. Then Theorem 2.5.1 and Corollary 2.5.2 apply here as well.

Assumption 2.6.1. By Theorem 2.4.13 (for vector fields with C^1 event-selected discontinuities) or Theorem 2.2.23 (for C^1 vector fields), $\partial W^s(X_J^s) = \bigcup_{i \in I} W^s(X_J^i)$. Assume that \hat{y}_J is transverse to $W^s(X_J^i)$ for all $i \in I$.

Remark 2.6.2. Assumption 2.6.1 is generic because C^1 submanifolds are generically transverse [KH99, Theorem A.3.20].

Remark 2.6.3. If it is desirable to consider a more general controlling invariant set, as opposed to the hyperbolic controlling critical elements considered above, we assume that this controlling invariant set is hyperbolic and that \hat{y}_J is transverse to its stable manifold.

By Lemma 2.4.16, V possesses a piecewise C^1 flow ϕ . By [Sch12, Chapter 3], this implies that the flow possesses a first-order approximation called the *Bouligand derivative* which is piecewise linear and continuous, rather than being purely linear like the derivative, at every point in $M \times J$. We write $d\phi$ to represent the Bouligand derivative. This does not conflict with earlier uses of this notaton, because the Bouligand derivative agrees with the (standard) derivative over C^J .

Recall that R refers to the set of recovery values, and ∂R to the recovery boundary. Note that $\frac{d\phi_t(\hat{y}(p))}{dp}$ is the derivative of the flow respect to parameter, and represents the sensitivity of the trajectory with respect to parameter. Section 3.1 discusses trajectory sensitivities in more detail. Define a function $H : [0, \infty) \times \overline{R} \to [0, \infty]$ by $H(t, p) = ||\frac{d\phi_t(\hat{y}(p))}{dp}||^{-1}$ where $|| \cdot ||$ is any norm on M. Then H(t, p) represents the inverse of the norm of the trajectory sensitivity at time t and for parameter value p. Define the function $G : \overline{R} \to [0, \infty)$ by $G(p) = \inf_{t \in [0,\infty)} H(t, p)$. Then G gives the infimum over time of the inverse of the norm of the trajectory sensitivity.

Theorem 2.6.4. Assume the conditions of Theorem 2.5.1 and Assumption 2.6.1. Then G is well-defined and continuous over \overline{R} , strictly positive on R, and identically zero on ∂R . In particular, for any $p^* \in \partial R$, $\lim_{p \in R, p \to p^*} G(p) = 0$.

Remark 2.6.5. This theorem holds as well for a more general hyperbolic controlling invariant set. The proof for this setting is sketched immediately following the main proof of the theorem.

Under additional assumptions it will be possible to show that as $p \to p^*$, G(p) approaches a dot product (in particular, a linear function). To achieve this we will require the following additional assumptions.

Assumption 2.6.6. The vector field V_{p^*} is C^{∞} , $\{V_p\}_{p \in J}$ is strong C^2 continuous, and \hat{y} is C^2 .

Assumption 2.6.7. The matrix $\frac{\partial V_{p^*}(X^*(p^*))}{\partial x}$ has distinct eigenvalues and satisfies a nonresonance condition - known as the strong Sternberg condition for Q with Q sufficiently large - which will be discussed in more detail in Section 2.6.3.

Assumption 2.6.8. The unstable manifold $W^u(X^*(p^*))$ is one dimensional.

Assumption 2.6.9. By the proof of Theorem 2.2.23, $W^u(X^*(p^*)) \cap W^s(X^s(p^*)) \neq \emptyset$. So, let $\hat{x} \in W^u_{loc}(X^*(p^*)) \cap W^s(X^s(p^*))$. By the proof of Theorem 2.6.4, the function $t \rightarrow ||d(\phi_{(t,p^*)})_{\hat{x}}V_{p^*}(\hat{x})||_1$ achieves a maximum on $\{t : t \ge 0\}$. Assume that this function achieves a unique maximum on $\{t : t \ge 0\}$.

Remark 2.6.10. The C^{∞} smoothness of Assumption 2.6.6 is more than is needed - in particular, C^k for k sufficiently large would suffice - but is adapted for simplicity of presentation.

Remark 2.6.11. The strong Sternberg condition for Q of Assumption 2.6.7 will be discussed in detail in Section 2.6.3. It is generically true that the eigenvalues of a matrix are distinct.

Remark 2.6.12. By the generic Assumption 2.6.1, $W^s(X^*(p^*))$ has codimension one. If $X^*(p^*)$ is an equilibrium point, this then implies Assumption 2.6.8. Also note that under Assumption 2.6.8, $X^*(p^*)$ must be an equilibrium point.

Theorem 2.6.13. Assume the conditions of Theorem 2.5.1 and that Assumptions 2.6.6-2.6.9 are satisfied. Then for any $p^* \in \partial R$, there exists a constant nonzero vector c such that for $p \in R$, as $p \to p^*$, $G(p) \to |c^T(p - p^*)|$.

Remark 2.6.14. Theorem 2.6.13 can also be extended to vector fields with C^{∞} event-selected discontinuities. A brief description of this extension is included in a remark following the main proof.

2.6.3 Proofs

Lemma 2.6.15. G is well-defined, strictly positive, and continuous on R.

Proof of Lemma 2.6.15. Define the functions $\mathcal{F} : [0, \infty) \times \overline{R} \to [0, \infty)$ by $\mathcal{F}(t, p) = ||\frac{d\phi_t(\hat{y}(p))}{dp}||$ and $\hat{F} : \overline{R} \to [0, \infty]$ by $\hat{F}(p) = \sup_{t \in [0, \infty)} \mathcal{F}(t, p)$. Then $G = \frac{1}{\hat{F}}$, so to prove the claim it suffices to show that \hat{F} is well-defined, finite, and continuous over R. Note that, by the chain rule, $\frac{d\phi_t(\hat{y}(p))}{dp} = d(\phi_t)_{\hat{y}(p)}(\hat{y}'(p))$, where $\hat{y}'(p) = \frac{d\hat{y}(p)}{dp}$.

Fix $\hat{p} \in R$. Since $\hat{y}(\hat{p}) \in W^s(X^s(\hat{p}))$, there exists T > 0 such that $\phi_T(\hat{y}_{\hat{p}}) \in \operatorname{int} W^s_{\operatorname{loc}}(X^s_J)$ (note that $W^s_{\operatorname{loc}}(X^s_J) \subset C^J$). Since $W^s_{\operatorname{loc}}(X^s_J)$ is open, there exists $\delta > 0$ such that $|p - \hat{p}| < \delta$ implies $\phi_T(\hat{y}(p)) \in \operatorname{int} W^s_{\operatorname{loc}}(X^s_J)$. As $W^s_{\operatorname{loc}}(X^s_J)$ is forward invariant, $\phi_{t+T}(\hat{y}(p)) \in W^s_{\operatorname{loc}}(X^s_J)$ for all $t \geq 0$. This implies that for any t > 0,

$$||d(\phi_{t+T})_{\hat{y}(p)}(\hat{y}'(p))|| = ||d(\phi_t)_{\phi_T(\hat{y}(p))} \circ d(\phi_T)_{\hat{y}(p)}(\hat{y}'(p))||$$

$$\leq ||d(\phi_t)_{\phi_T(\hat{y}(p))}|| ||d(\phi_T)_{\hat{y}(p)}(\hat{y}'(p))|| < ||d(\phi_T)_{\hat{y}(p)}(\hat{y}'(p))||$$

since $\phi_t|_{W^s_{\text{loc}}(X^s(p))}$ is a contraction so $||d(\phi_t)_{\phi_T(\hat{y}(p))}|| < 1$. Thus, for $p \in B_{\delta}(\hat{p})$, $\hat{F}(p) = \sup_{t \in [0,\infty)} ||d(\phi_t)_{\hat{y}(p)}(\hat{y}'(p))|| = \sup_{t \in [0,T]} \mathcal{F}(t,p)$. Since \hat{y}' is continuous and the Bouligand derivative is piecewise continuous, $\mathcal{F}(t,p)$ is piecewise continuous over $[0,T] \times \overline{R}$. As [0,T] is compact, this implies that, for $p \in B_{\delta}(\hat{p})$, $\hat{F}(p) = \sup_{t \in [0,T]} \mathcal{F}(t,p) = \max_{t \in [0,T]} \mathcal{F}(t,p) < \infty$. Hence, as $\hat{p} \in R$ was arbitrary, \hat{F} is well-defined and finite over R.

It suffices to show that \hat{F} is continuous at \hat{p} . As \mathcal{F} is piecewise continuous over $[0,T] \times$ $B_{\delta}(\hat{p})$, there exists a finite set of continuous selection functions $H^{j}:[0,T]\times B_{\delta}(\hat{p})\to [0,\infty)$ for $j \in I_m$ with m finite such that $\mathcal{F}(t,p) \in \{H^j(t,p)\}_{j \in I_m}$ for all $(t,p) \in [0,T] \times B_{\delta}(\hat{p})$. By Lemma 2.4.19, since [0,T] is compact, $\phi(\hat{y}(p), [0,T], p)$ intersects Z^p in only finitely many isolated points. In particular, there are only finitely many intersection times at which discontinuities of $\mathcal{F}(\cdot, p)$ can occur. Using the implicit function theorem as in the proof of Lemma 2.5.14, shrinking δ if necessary implies that for $p \in B_{\delta}(\hat{p})$, the number of intersection times of the orbit of $\hat{y}(p)$ with Z^p is constant, and that the intersection times vary C^1 with parameter over $B_{\delta}(\hat{p})$. This implies that for any $p \in B_{\delta}(\hat{p})$, we can write [0,T] as a finite union of closed intervals on which $\mathcal{F}(\cdot, p)$ is continuous, that the number of such intervals is constant over $B_{\delta}(\hat{p})$, and that the endpoints of the intervals vary continuously over $B_{\delta}(\hat{p})$. For $p \in B_{\delta}(\hat{p})$, as $\mathcal{F}(\cdot, p)$ is continuous over each such closed interval and the intervals are compact, it achieves a maximum on each closed interval. Moreover, since the endpoints of each interval vary continuously over $B_{\delta}(\hat{p})$, and \mathcal{F} is equal to a continuous selection function on each interval, the maximum of \mathcal{F} over each interval varies continuously over $B_{\delta}(\hat{p})$. Note that, for $p \in B_{\delta}(\hat{p}), \hat{F}(p)$ is given by the maximum over each interval of the maximum of $\mathcal{F}(\cdot, p)$ on that interval. Hence, over $B_{\delta}(\hat{p})$, \hat{F} is the maximum of a finite number of continuous functions, so \hat{F} is continuous.

Lemma 2.6.16. *G* is identically zero on ∂R and for any $p^* \in \partial R$, $\lim_{p \to p^*, p \in R} G(p) = 0$.

Proof of Lemma 2.6.16. Define the functions $\mathcal{F}: [0,\infty) \times \overline{R} \to [0,\infty)$ by $\mathcal{F}(t,p) = ||\frac{d\phi_t(\hat{y}(p))}{dp}||$ and $\hat{F}: \overline{R} \to [0,\infty]$ by $\hat{F}(p) = \sup_{t \in [0,\infty)} \mathcal{F}(t,p)$. Then $G = \frac{1}{\hat{F}}$, so to prove the claim it suffices to show that \hat{F} is identically infinite over ∂R , and that for any $p^* \in \partial R$, $\lim_{p \to p^*, p \in R} \hat{F}(p) = \infty$. Note that, by the chain rule, $\frac{d\phi_t(\hat{y}(p))}{dp} = d(\phi_t)_{\hat{y}(p)}(\hat{y}'(p))$, where $\hat{y}'(p) = \frac{d\hat{y}(p)}{dp}$.

Let $p^* \in \partial R$. By Theorem 2.5.1, p^* is a critical parameter value, and so by Corollary 2.5.2 there exists a unique controlling critical element X_J^* corresponding to p^* . First we show that $\hat{F}(p^*) = \infty$. Let K > 0. It suffices to show that there exists T > 0 such that $\mathcal{F}(T, p^*) > K$. If $X^*(p^*)$ is an equilibrium point (periodic orbit) let $x = X^*(p^*)$ ($x \in X^*(p^*)$ such that $x \notin Z^{p^*}$), let S be a neighborhood of x (S be a C^1 cross section containing x) such that $S \cap Z^{p^*} = \emptyset$, and let $g = \phi(\cdot, 1, p^*)|_S$ be the C^1 time-one map $(g: S \to S$ be a C^1 first return map as in Lemma 2.4.20). Then x is a hyperbolic fixed point of $q C^1$ so it possesses a local stable manifold $W^s_{\text{loc}}(x)$ contained in M(S). As $\hat{y}(p^*) \in W^s(X^*(p^*))$, there exists t' such that $x_0 := \phi(\hat{y}(p^*), t', p^*) \in W^s_{\text{loc}}(x)$. By Lemma 2.4.18, the flow is C^1 on a neighborhood of $\hat{y}(p^*)$ which implies, since by Assumption 2.6.1 $\hat{y}'(p^*)$ is transversal to $W^s(X^*(p^*))$ at $\hat{y}(p^*)$, and $W^s(X^*(p^*))$ is invariant under the flow, that $v_0 := d(\phi(\cdot, t', p^*))_{\hat{y}(p^*)}(\hat{y}'(p^*))$ is transversal to $W^s_{\text{loc}}(x)$ at $\phi(\hat{y}(p^*), t', p^*)$ in M(S). If $X^*(p^*)$ is a periodic orbit, let $\hat{\tau} : S \to [0, \infty)$ send x to the time it takes for the flow from initial condition x to intersect S again. Let $v_n = dg_{g^n(x_0)} \circ dg_{g^{n-1}(x_0)} \circ \ldots \circ dg_{x_0}(v_0)$. If $X^*(p^*)$ is an equilibrium point (periodic orbit) then $||v_n|| = \mathcal{F}(t'+n,p^*)$ $(||v_n|| = \mathcal{F}(t'+\sum_{i=0}^{n-1}\hat{\tau}(g^i(x_0)),p^*))$. So it suffices to show using T = t' + n $(T = t' + \sum_{i=0}^{n-1} \hat{\tau}(g^i(x_0)))$ - that there exists n such that $||v_n|| > K$. By the Inclination Lemma [Pal69, Lemma 1.1], $\frac{||v_{n+1}||}{||v_n||} \rightarrow b' > 1$, where b' is a constant, and $||v_n|| \neq 0$ for all n. Hence, for N sufficiently large, $n \geq N$ implies that $\frac{||v_{n+1}||}{||v_n||} \geq b > 1$. Thus,

$$||v_n|| = \frac{||v_n||}{||v_{n-1}||} \frac{||v_{n-1}||}{||v_{n-2}||} \dots \frac{||v_{N+1}||}{||v_N||} ||v_N|| \ge b^{n-N} ||v_N||.$$

So, taking n sufficiently large implies that $||v_n|| > K$.

Next we show that $\lim_{p\to p^*, p\in R} \hat{F}(p) = \infty$. Let K > 0. It suffices to show that there exists $\delta > 0$ such that $p \in B_{\delta}(\hat{p}) \cap R$ implies that $\hat{F}(p) > K$. By the argument above, there exists T > 0 such that $\mathcal{F}(T, p^*) > K$ and $\phi(\hat{y}(p^*), T, p^*) \notin Z^{p^*}$. As $\phi(\hat{y}(p^*), T, p^*) \in C^J$, by Lemma 2.4.17 C^J is open, and \hat{y} is C^1 , there exists $\delta > 0$ such that $p \in B_{\delta}(p^*)$ implies that $\phi(\hat{y}(p), T, p) \in C^J$. In particular, this implies that $\mathcal{F}(T, \cdot)$ is continuous over $B_{\delta}(p^*)$ since the

flow is C^1 over C^J by Lemma 2.4.16. As $\mathcal{F}(T, p^*) > K$, shrinking δ if necessary implies, by continuity of $\mathcal{F}(T, \cdot)$ over $B_{\delta}(p^*)$, that for $p \in B_{\delta}(p^*)$, $\mathcal{F}(T, p) > K$. Hence, for $p \in B_{\delta}(p^*)$, $\hat{F}(p) \geq \mathcal{F}(T, p) > K$.

Proof of Theorem 2.6.4. Follows immediately from the conclusions of Lemma 2.6.15 and Lemma 2.6.16. $\hfill \Box$

Remark 2.6.17. We sketch a brief summary of the proof of Theorem 2.6.4 in the case of a more general controlling invariant set, under the assumption that this controlling invariant set is hyperbolic and that \hat{y}_J is transverse to its stable manifold (see Remark 2.6.5). First, we note that the proof of Lemma 2.6.15 proceeds identically to the earlier case. So, it suffices to prove Lemma 2.6.16. Since the controlling invariant set is hyperbolic, there exists a continuous splitting of the ambient manifold's tangent bundle into stable and unstable bundles such that the bundles are invariant under $d\phi_1$ and the norm of $d\phi_1$ on the unstable bundle is bounded below by some constant b > 1. As the controlling invariant set is closed, these bundles extend to stable and unstable bundles over an open neighborhood N of the controlling invariant set such that, possibly shrinking b slightly, the norm of $d\phi_1$ on the unstable bundle is bounded below by b > 1. As $\hat{y}(p^*)$ converges to the controlling invariant set, after some finite time t' it enters the local stable manifold of the controlling invariant set and remains there for all times $t \ge t'$. In particular, we may assume that $\phi(\hat{y}(p^*), t, p^*) \in N$ for $t \ge t'$. Let $g = \phi_1$. Construct x_0 and v_0 as in the proof of Lemma 2.6.16, and note that v_0 is transversal to the stable bundle, so it must have a nonzero component in the unstable direction. Define v_n analogously to the proof of Lemma 2.6.16 as well. Then $\frac{||v_{n+1}||}{||v_n||} \ge b$ for sufficiently large n (since the component in the stable direction will converge to zero), so by an analogous argument to that in the proof of Lemma 2.6.16, $\hat{F}(p^*) = \infty$. That $\lim_{p \to p^*, p \in \mathbb{R}} \hat{F}(p) = \infty$ follows by an analogous argument to that in the proof of Lemma 2.6.16.

Let A be a square matrix, and let $\Sigma(A) = \{\lambda_1, ..., \lambda_n\}$ be its eigenvalues. Let m be any vector whose entries are nonnegative integers. Define

$$\gamma(\lambda, m) = \lambda - \sum_{i=1}^{n} m_i \lambda_i.$$

Let $|m| = \sum_{i=1}^{n} m_i$. Then A satisfies the Sternberg condition of order N if $\gamma(\lambda, m) \neq 0$ for all $\lambda \in \Sigma(A)$ and all m such that $2 \leq |m| \leq N$. Furthermore, A satisfies the strong Sternberg condition of order N if A satisfies the Sternberg condition of order N and Re $\gamma(\lambda, m) \neq 0$

for all $\lambda \in \Sigma(A)$ and all m such that |m| = N. Note that if A satisfies the strong Sternberg condition then A is hyperbolic.

If A is hyperbolic, let $\Sigma^+(A)$ denote its eigenvalues with positive real part, and let $\Sigma^-(A)$ denote its eigenvalues with negative real part. For $i \in \{+, -\}$, let $\rho^i = \frac{\max\{|\text{Re }\lambda|:\lambda \in \Sigma^i(A)\}}{\min\{|\text{Re }\lambda|:\lambda \in \Sigma^i(A)\}}$. Let Q be a positive integer. Then the Q-smoothness of A is the largest integer $K \ge 0$ such that there exist positive integers Q^+ and Q^- with $Q = Q^+ + Q^-$, $Q^+ - K\rho^+ \ge 0$, and $Q^- - K\rho^- \ge 0$. In particular, the Q-smoothness of A approaches infinity as $Q \to \infty$, so for Q sufficiently large we may assume it is at least two. Assumption 2.6.7 can be stated as assuming that the matrix $\frac{\partial V_{p^*}(X^*(p^*))}{\partial x}$ satisfies the strong Sternberg condition for Q with Q sufficiently large such that the Q-smoothness of $\frac{\partial V_{p^*}(X^*(p^*))}{\partial x}$ is at least two.

Lemma 2.6.18. Consider the vector field $\dot{x} = A(p)x$ for $p \in J$ where A(p) varies C^1 with respect to p and is linear for each p. Assume that A(0) is hyperbolic, has distinct eigenvalues, and has exactly one unstable eigenvalue. Let $\hat{y}(p)$ be a C^1 initial condition for this linear vector field, assume that $\hat{y}(0)$ is in the stable eigenspace of A(0), and that $\hat{y}(J)$ is transverse to the disjoint union of the stable eigenspaces of A(p) for $p \in J$. Let u > 0 small and let $t_{hit} = t_{hit}(p)$ be the time at which the projection of the forward orbit of $\hat{y}(p)$ onto the unstable eigenspace has magnitude u (note this is defined only for a connected component $J^T \subset J$ where $\hat{y}(J^T)$ is disjoint from the stable manifolds of A(p)). Let $\hat{G}(p) =$ $\left(\sum_{j=1}^n \left\| \frac{\partial x(t_{hit}(p))}{\partial p_j} \right\|_1 \right)^{-1}$. Then there exists a constant nonzero vector c such that for $p \in J^T$, as $p \to p^*$, $\hat{G}(p) \to |c^T p|$.

Proof of Lemma 2.6.18. The proof proceeds primarily by computation. The main intuition is that the trajectory sensitivities in the direction of the (one dimensional) unstable eigenspace dominate the other sensitivities in the directions of the stable eigenspace. As $p \to p^*$, this will imply that the structure of $\hat{G}(p)$ is dominated by this one dimensional behavior, which in turn is governed by a dot product. The dot product structure will follow from taking the explicit solution of the linear system, computing $t_{\rm hit}(p)$ directly, computing the trajectory sensitivities explicitly, and then deriving a formula for $\hat{G}(p)$.

Let $x \in \mathbb{R}^n$ and $p \in J$. Consider the vector field $\dot{x} = A(p)x$ where A(p) is linear for each $p \in J$. We have that for all $p \in J$, A(p) is hyperbolic, all its eigenvalues are distinct, and its unstable eigenspace is one dimensional. Let $\{\lambda_1(p), ..., \lambda_n(p)\}$ be the eigenvalues of A(p), and order them by their real component in descending order so that λ_1 is the unstable eigenvalue. As the eigenvalues of A are distinct, A is diagonalizable with a basis of eigenvectors; denote them by $\{v_1, ..., v_n\}$ and assume they are normalized so that $\sum_j |v_i|_j = 1$ for all i. Let $\Lambda(p)$

be the diagonal matrix of the eigenvalues of A(p) and let V(p) be the matrix whose columns are the eigenvectors of A(p). Note that $A(p) = V(p)\Lambda(p)V(p)^{-1}$. Let $z(t) = V(p)^{-1}x(t)$ so that

$$\dot{z} = \Lambda(p)z. \tag{2.20}$$

Then the solution to Eq. 2.20 is given by

$$z(t) = e^{\Lambda(p)t} z(0)$$

$$z_i(t) = e^{\lambda_i(p)t} z_i(0).$$

We have that the initial condition and eigenvalues are at least C^1 so that we may write $z(0) = Bp + b + o(||p||_1)$, which implies that $z_i(0) = B_{in}p_n + b_i + o(||p||_1)_i$, and $\lambda_i(p) = \lambda_i^0 + c_{in}p_n + o(||p||_1)_i$ for all *i*. We have that $z_i(0)$ at p = 0 is in the stable eigenspace of the origin, which implies that $b_1 = 0$ since the stable eigenspace is given by $\{z : z_1 = 0\}$. Furthermore, we have that $z_0(J)$ is transverse to the stable eigenspace at p = 0. The solution to Eq. 2.20 can be written as

$$z_i(t) = e^{(\lambda_i^0 + c_{in}p_n + o(||p||_1)_i)t} (B_{in}p_n + b_i + o(||p||_1)_i).$$

Let u > 0 small. We wish to compute the time t_{hit} at which the first coordinate (the unstable coordinate) reaches u, This is merely the hitting time of the hypersurface $\{z : z_1 = u\}$. So, we wish to solve $|z_1(t_{\text{hit}}(p))| = u$. This gives

$$t_{\rm hit}(p) = \frac{1}{\text{Re }\lambda_1(p)}\log\frac{u}{|z_1(0)|} = \frac{1}{\text{Re }(\lambda_i^0) + \text{Re }(c_{in})p_n + o(||p||_1)_i}\log\frac{u}{|B_{1n}p_n + o(||p||_1)_1|}$$

which is well-defined for all $p \in J$ such that z(0) is not in the stable eigenspace of the origin since the stable eigenspace is given by $\{z : z_1 = 0\}$. Note that the set on which t_{hit} is well-defined is nonempty because $z_0(J)$ is transverse to the stable eigenspace at p = 0. We compute

$$\left|\frac{\partial z_i(t)}{\partial p_j}\right| = e^{\operatorname{Re}(\lambda_i(p))t} \left| (c_{ij} + o(1)_{ij})t(B_{in}p_n + b_i + o(||p||_1)_i) + (B_{ij} + o(1)_{ij}) \right|$$

since $\frac{\partial o(||p||_1)}{\partial p_j} = o(1)$. Evaluating at $t = t_{hit}(p)$ gives

$$\left|\frac{\partial z_i(t_{\rm hit}(p))}{\partial p_j}\right| = \left(\frac{u}{|B_{1n}p_n + o(||p||_1)_1|}\right)^{\frac{\operatorname{Re}(\lambda_i(p))}{\operatorname{Re}(\lambda_1(p))}} \\ * \left|(c_{ij} + o(1)_{ij})\frac{B_{in}p_n + b_i + o(||p||_1)_i}{\operatorname{Re}(\lambda_1(p))}\log\frac{u}{|B_{1n}p_n + o(||p||_1)_1|} + B_{ij} + o(1)_{ij}\right|.$$

Note that

$$\lim_{||p||_1 \to 0} \left(B_{1n} p_n + o(||p||_1)_1 \right) \log \frac{u}{|B_{1n} p_n + o(||p||_1)_1|} = 0$$

as $\lim_{x\to 0} x \log \frac{u}{x} = 0$. So,

$$(B_{1n}p_n + o(||p||_1)_1) \log \frac{u}{|B_{1n}p_n + o(||p||_1)_1|} = o(1).$$
(2.21)

In particular, this implies that at i = 1, and since $b_1 = 0$, we have

$$\begin{aligned} \left| \frac{\partial z_1(t_{\text{hit}}(p))}{\partial p_j} \right| &= \frac{u}{|B_{1n}p_n + o(||p||_1)_1|} \\ & * \left| \left(\sum_j c_{1j} + o(1)_{1j} \right) \frac{B_{1n}p_n + o(||p||_1)_1}{\operatorname{Re}(\lambda_1(p))} \log \frac{u}{|B_{1n}p_n + o(||p||_1)_1|} + \left(\sum_j B_{1j} + o(1)_{1j} \right) \right| \\ &= \frac{u}{|B_{1n}p_n + o(||p||_1)_1|} \left| \left(\sum_j c_{1j} + o(1)_{1j} \right) \frac{o(1)}{\operatorname{Re}(\lambda_1(p))} + \left(\sum_j B_{1j} + o(1)_{1j} \right) \right| \\ &= \frac{u}{|B_{1n}p_n + o(||p||_1)_1|} \left| \sum_j B_{1j} + o(1) \right| \end{aligned}$$

where we use that $\lambda_1^0 \neq 0$. As $\lim_{||p||_1 \to 0} \frac{|B_{1n}p_n + o(||p||_1)_1|}{u} = 0$ and $\frac{\operatorname{Re}\lambda_i(p)}{\operatorname{Re}\lambda_1(p)} < 0$ for all $i \geq 2$ and $p \in J$,

$$\lim_{||p||_1 \to 0} \left(\frac{u}{|B_{1n}p_n + o(||p||_1)_1|} \right)^{\frac{\operatorname{Re}\lambda_i(p)}{\operatorname{Re}\lambda_1(p)}} = 0$$

so
$$\left(\frac{u}{|B_{1n}p_n+o(||p||_1)_1|}\right)^{\frac{\operatorname{Re}\lambda_i(p)}{\operatorname{Re}\lambda_1(p)}} = o(1)_i$$
. Hence, for $i \ge 2$ we have
 $\left|\frac{\partial z_i(t_{\operatorname{hit}}(p))}{\partial p_j}\right| = o(1)_i \left|(c_{ij}+o(1)_{ij})\frac{B_{in}p_n+b_i+o(||p||_1)_i}{\operatorname{Re}(\lambda_1(p))}\log\frac{u}{|B_{1n}p_n+o(||p||_1)_1|} + B_{ij}+o(1)_{ij}\right|.$

Define

$$\hat{G}^{z}(p) = \sum_{i} \left\| \frac{\partial z_{i}(t_{\text{hit}}(p))}{\partial p} \right\|_{1}$$
$$= \sum_{i,j} \left| \frac{\partial z_{i}(t_{\text{hit}}(p))}{\partial p_{j}} \right|.$$

We compute

$$\begin{split} \hat{G}^{z}(p) &= \frac{u}{|B_{1n}p_{n} + o(||p||_{1})_{1}|} \left| \sum_{j} B_{1j} + o(1) \right| \\ &+ \sum_{i=2}^{n} o(1)_{i} \left| \left(\sum_{j} c_{ij} + o(1)_{ij} \right) \frac{B_{in}p_{n} + b_{i} + o(||p||_{1})_{i}}{\operatorname{Re}(\lambda_{1}(p))} \log \frac{u}{|B_{1n}p_{n} + o(||p||_{1})_{1}|} + \left(\sum_{j} B_{ij} + o(1)_{ij} \right) \right. \\ &= \frac{u}{|B_{1n}p_{n} + o(||p||_{1})_{1}|} \left| \sum_{j} B_{1j} + o(1) \right| \\ &+ \frac{u}{|B_{1n}p_{n} + o(||p||_{1})_{1}|} \sum_{i=2}^{n} o(1)_{i} \\ &+ \left| \left(\sum_{j} c_{ij} + o(1)_{ij} \right) \frac{B_{in}p_{n} + b_{i} + o(||p||_{1})_{i}}{\operatorname{Re}(\lambda_{1}(p))} \frac{|B_{1n}p_{n} + o(||p||_{1})_{1}|}{u} \log \frac{u}{|B_{1n}p_{n} + o(||p||_{1})_{1}|} \right. \\ &+ \left. \frac{|B_{1n}p_{n} + o(||p||_{1})_{1}|}{u} \left(\sum_{j} B_{ij} + o(1)_{ij} \right) \right| \\ &= \frac{u}{|B_{1n}p_{n} + o(||p||_{1})_{1}|} \left(\left| \sum_{j} B_{1j} + o(1) \right| + o(1) \right) \end{split}$$

by Eq. 2.21 and standard rules for arithmetic with o(1) functions.

We have that x(t) = V(p)z(t), so that $x_i(t) = V(p)_{ik}z_k(t)$. This implies that

$$\frac{\partial x_i(t)}{\partial p_j} = \frac{\partial V(p)_{ik}}{\partial p_j} z_k(t) + V(p)_{ik} \frac{\partial z_k(t)}{\partial p_j}$$
$$\frac{\partial x_i(t_{\rm hit}(p))}{\partial p_j} = \frac{\partial V(p)_{ik}}{\partial p_j} z_k(t_{\rm hit}(p)) + V(p)_{ik} \frac{\partial z_k(t_{\rm hit}(p))}{\partial p_j}$$

Let $V(p)_{ik} = d_{ik} + D_{ikn}p_n + o(||p||_1)_{ik}$. Then $\frac{\partial V(p)_{ik}}{\partial p_j} = D_{ikj} + o(1)_{ikj}$ is bounded over $p \in J$. Furthermore, as $z_k(t_{hit}(p))$ is contained in a small neighborhood of the origin for all $p \in J$ along the target surface $\{z : z_1 = u\}$, $z_k(t_{hit}(p))$ is uniformly bounded over $p \in J$. Hence, $\frac{\partial V(p)_{ik}}{\partial p_j} z_k(t_{hit}(p))$ is uniformly bounded over $p \in J$. By an analogous computation to the derivation of $\hat{G}^z(p)$ above, we can show that for any fixed *i* such that $|V(p)_{i1}| \neq 0$,

$$\begin{split} \sum_{k} \left| V(p)_{ik} \frac{\partial z_k(t_{\text{hit}}(p))}{\partial p_j} \right| &= \sum_{k} \left| V(p)_{ik} \right| \left| \frac{\partial z_k(t_{\text{hit}}(p))}{\partial p_j} \right| \\ &= \left| V(p)_{i1} \right| \frac{u}{|B_{1n}p_n + o(||p||_1)_1|} \left(\left| \sum_{j} B_{1j} + o(1) \right| + o(1) \right). \end{split}$$

As $\frac{\partial V(p)_{ik}}{\partial p_j} z_k(t_{\text{hit}}(p))$ is uniformly bounded over $p \in J$, we have

$$\frac{\partial V(p)_{ik}}{\partial p_j} z_k(t_{\rm hit}(p)) = \frac{u}{|B_{1n}p_n + o(||p||_1)_1|} \left(\frac{\partial V(p)_{ik}}{\partial p_j} z_k(t_{\rm hit}(p)) \frac{|B_{1n}p_n + o(||p||_1)_1|}{u}\right)$$
$$= \frac{u}{|B_{1n}p_n + o(||p||_1)_1|} (o(1)).$$
(2.22)

Define

$$\hat{G}(p) = \sum_{i} \left\| \frac{\partial x_i(t_{\text{hit}(p)})}{\partial p} \right\|_1$$
$$= \sum_{i,j} \left| \frac{\partial x_i(t_{\text{hit}}(p))}{\partial p_j} \right|.$$

Then

$$\hat{G}(p) = \sum_{i} \sum_{j} \left| \frac{\partial x_{i}(t_{\text{hit}}(p))}{\partial p_{j}} \right|$$

$$= \left(\sum_{i} |V(p)_{i1}| \right) \frac{u}{|B_{1n}p_{n} + o(||p||_{1})_{1}|} \left(\left| \sum_{j} B_{1j} + o(1) \right| + o(1) \right)$$

$$= \frac{u}{|B_{1n}p_{n} + o(||p||_{1})_{1}|} \left(\left| \sum_{j} B_{1j} + o(1) \right| + o(1) \right)$$

since by Eq. 2.22 the terms $\frac{\partial V(p)_{ik}}{\partial p_j} z_k(t_{\text{hit}}(p))$ can be incorporated into the additive o(1) term in the parentheses, as can the contributions from the eigenvectors for $i \geq 2$ (as in the derivation of $\hat{G}^z(p)$ above), and since $\sum_i |V(p)_{i1}| = 1$ by the normalization above. Thus, we have

$$\frac{1}{\hat{G}(p)} = \frac{|B_{1n}p_n + o(||p||_1)_1|}{u\left(\left|\sum_j B_{1j} + o(1)\right| + o(1)\right)}$$

which approaches a dot product - an affine function - as $||p||_1 \rightarrow 0$.

Remark 2.6.19. It will be useful for Chapter III to note the relationship between $t_{\rm hit}(p)$ and $\hat{G}(p)$ in the proof of Lemma 2.6.18. We note that as $p \to p^* = 0$, $t_{\rm hit}(p) \to c_1 \log \frac{1}{c_2^T p}$ while $\hat{G}(p) \to \frac{1}{c_3^T p}$ for nonzero constant vectors c_2 and c_3 , and for real nonzero constant c_1 . In particular, this implies that $\hat{G}(p)$ grows exponentially faster than $t_{\rm hit}(p)$ as $p \to p^* = 0$.

Proof of Theorem 2.6.13. For the intuition behind this proof, consider partitioning each trajectory of the system into four parts: the initial segment from $\hat{y}(p)$ into a neighborhood of $X^*(p)$, the behavior in this neighborhood of $X^*(p)$, a segment from the neighborhood of $X^*(p)$ to $W^s_{loc}(X^s(p))$, and then the forward orbit inside $W^s_{loc}(X^s(p))$. As the flow restricted to $W^s_{\text{loc}}(X^s(p))$ is a contraction, trajectory sensitivities must achieve their maximum before entering $W^s_{\text{loc}}(X^s(p))$, so we can restrict our attention to the first three parts mentioned above. For the initial segment from $\hat{y}(p)$ into a neighborhood of $X^*(p)$, we may neglect this segment and instead absorb this behavior into new parameter dependent initial conditions in the neighborhood of $X^*(p)$. Taking the neighborhood of $X^*(p)$ to be sufficiently small, we may apply a Theorem of George Sell which is similar to Sternberg's Theorem to obtain a smooth (for our purposes C^2 will suffice) conjugacy between the nonlinear vector field V_p and its linearization at $X^*(p)$ for all $p \in J$. This will allow us to translate the results from the linear dynamics of Lemma 2.6.18 into the nonlinear setting here. Finally, we show that G achieves its minimum in the segment from the neighborhood of $X^*(p)$ to $W^s_{\text{loc}}(X^s(p))$ and that the time at which this minimum is achieved approaches a constant as $p \to p^*$ for $p \in R$. From this, it is shown that the map from the trajectory sensitivities as the trajectory is leaving the neighborhood of $X^*(p)$ to the trajectory sensitivities when G achieves its minimum is nearly constant. Consequently, the dominant effect in the structure of G under the nonlinear vector field comes from the linear dynamics near $X^*(p)$, and this dot product structure then follows from Lemma 2.6.18.

Fix $p^* \in \partial R$. By Theorem 2.5.1, $p^* \in C$. So, there exists a critical element X^* such that $\hat{y}(p^*) \in W^s(X^*(p^*))$. We have that $X^*(p^*)$ is an equilibrium point and $W^u(X^*(p^*))$ is one dimensional. Furthermore, we assume that $\frac{\partial \hat{y}}{\partial p_j}$ is transverse to $W^s(X^*(p^*))$ at $\hat{y}(p^*)$ for all j.

For $p \in J$ let $A(p) = \frac{\partial V_p(X^*(p))}{\partial x}$. Assume that $A(p^*)$ satisfies the strong Sternberg condition for Q sufficiently large such that the Q-smoothness of $A(p^*)$ is at least two. Then by [Sel85, Theorem 7], for J sufficiently small there exists a neighborhood U of $X^*(p^*)$ in M and a conjugacy Θ between $\dot{x} = V_p(x)$ on U and $\dot{x} = A(p)x$ on a neighborhood of the origin in \mathbb{R}^n such that the conjugacy is at least C^2 in both p and x. Let T > 0 such that $\phi(\hat{y}(p^*), T, p^*) \in U$. Shrink J if necessary so that $\phi(\hat{y}(p), T, p) \in U$ for $p \in J$. As in the proof of Lemma 2.6.18, let $z(t) = V(p)^{-1}x(t)$, and let $z(0) = V(p)^{-1}\Theta(\phi(\hat{y}(p), T, p), p)$. By the proof of Theorem 2.2.23, $W^u(X^*(p^*)) \cap W^s(X^s(p^*)) \neq \emptyset$. As $W^u(X^*(p^*))$ is one dimensional, this implies that one connected component of $W^u(X^*(p^*))$ lies inside $W^s(X^s(p^*))$. Let u > 0 small, let s be either +1 or -1, and let $\hat{x} = \Theta^{-1}(V(p) * (su, 0), p^*)$. Note that $\hat{x} \in W^u(X^*(p^*))$ and, by the proper choice of s, we may have that $\hat{x} \in W^s(X^s(p^*))$.

Hence, there exists T' > 0 such that $\phi(\hat{x}, p^*, T') \in W^s_{\text{loc}}(X^s(p^*))$. Let N be a neighborhood of \hat{x} in M and shrink J if necessary such that $p \in J$ and $x \in N$ implies that $\phi(x, p, T') \in W^s_{\text{loc}}(X^s(p))$. Shrink J further if necessary so that $p \in J$ implies that $\Theta^{-1}(x(t_{\text{hit}}(p)), p) \in N$. For each \hat{i} define

For each j, define

$$G^{j}(t,p) = \left\| d(\phi_{t})_{\Theta^{-1}(x(t_{\mathrm{hit}}(p)),p)} d\Theta_{x(t_{\mathrm{hit}}(p)),p}^{-1} \frac{\partial x(t_{\mathrm{hit}}(p))}{\partial p_{j}} \right\|_{1}.$$

As the flow restricted to $W^s_{\text{loc}}(X^s(p^*))$ is a contraction, for any $p \in J$, $G^j(\cdot, p)$ must achieve its maximum on $[0, t_{\text{hit}}(p) + T']$. As $G^j(\cdot, p)$ achieves its maximum over $[0, t_{\text{hit}}(p)]$ at $t = t_{\text{hit}}(p)$, $G^j(\cdot, p)$ must achieve its maximum on $[t_{\text{hit}}(p), t_{\text{hit}}(p) + T']$. Define

$$G^{\lim}(t) = ||d(\phi_t)_{\hat{x}} V_{p^*}(\hat{x})||_1.$$

As the forward orbit of \hat{x} under V_{p^*} eventually enters $W^s_{\text{loc}}(X^s(p^*))$ on which the flow of V_{p^*} is a contraction, $G^{\text{lim}}(t)$ must attain a maximum on [0, T']. Assume that $G^{\text{lim}}(t)$ attains a unique global maximum for $t \geq 0$. Let $\delta > 0$. Note that $G^j(t + t_{\text{hit}}(p), p) \rightarrow c^j(p)G^{\text{lim}}(t)$ as $p \rightarrow p^*$ since $d(\phi_t)_{\Theta^{-1}(x(t_{\text{hit}}(p)),p)} \rightarrow d(\phi_t)_{\hat{x}}, d\Theta^{-1}_{x(t_{\text{hit}}(p)),p} \rightarrow d\Theta^{-1}_{(u,0),p^*}$, and $d\Theta^{-1}_{(u,0),p^*} \frac{\partial x(t_{\text{hit}}(p))}{\partial p_j} \rightarrow c^j(p)V_{p^*}(\hat{x})$ by the Inclination Lemma. As [0, T'] is compact, $G^j(t + t_{\text{hit}}(p), p)$ can be brought arbitrarily close to $c^j(p)G^{\text{lim}}(t)$ on this time interval. In particular, let $\epsilon > 0$ sufficiently small such that if d is the largest local maximum of $G^{\text{lim}}(t)$ other than $G^{\text{lim}}(t_{\text{max}})$ (if such d exists) then $G^{\text{lim}}(t_{\text{max}}) - \epsilon > d$. Then there exists h(t) such that $G^j(t + t_{\text{hit}}(p), p) - c^j(p)G^{\text{lim}}(t) = c^j(p)h(t)$ and $|h(t)| < \epsilon$ for all $t \in [0, T']$. We wish to determine $t^j_{\text{max}}(p) = \operatorname{argmax}_t G^j(t + t_{\text{hit}}(p), p)$. Note that $t_{\text{max}}(p) \in [0, T']$ by the argument above. As $c^j(p)$ is constant in time, we have

$$\begin{split} t^{j}_{\max}(p) &= \operatorname{argmax}_{t \in [0,T']} G^{j}(t+t_{\operatorname{hit}}(p),p) \\ &= \operatorname{argmax}_{t \in [0,T']} \frac{1}{c^{j}(p)} G^{j}(t+t_{\operatorname{hit}}(p),p) \\ &= \operatorname{argmax}_{t \in [0,T']} G^{\lim}(t) + h(t). \end{split}$$

By construction, shrinking ϵ further if desired implies that $t_{\max}^j(p)$ can be brought arbitrarily close to t_{\max} for p sufficiently close to p^* . In particular, we have shown that $\lim_{p\to p^*} t_{\max}^j(p) = t_{\max}$ for all j.

Define

$$\hat{G}(t,p) = \sum_{j} G^{j}(t,p)$$

and let $t_{\max}(p) = \operatorname{argmax}_{t \in [0,T']} \hat{G}(t + t_{\operatorname{hit}}(p), p)$. Then $\lim_{p \to p^*} t_{\max}^j(p) = t_{\max}$ for all j implies that $\lim_{p \to p^*} t_{\max}(p) = t_{\max}$.

As $\hat{G}(t,p)$ converges to $\sum_{j} c^{j}(p) G^{\lim}(t)$ uniformly on [0,T'] as $p \to p^{*}$, and as $\lim_{p \to p^{*}} t_{\max}(p) = t_{\max}$, we have

$$\lim_{p \to p^*} \hat{G}(t_{\max}(p), p) = \lim_{p \to p^*} \hat{G}(t_{\max}, p)$$
$$= \sum_j c^j(p) G^{\lim}(t_{\max}).$$

Hence,

$$\hat{G}(t_{\max}(p), p) = \left(\sum_{j} c^{j}(p)\right) G^{\lim}(t_{\max}) + o(1).$$

In the notation of the proof of Lemma 2.6.18, $\sum_j c^j(p) = \hat{G}(p)$, so substituting the expression for $\hat{G}(p)$ in the proof of Lemma 2.6.18 for $\sum_j c^j(p)$ here gives

$$\hat{G}(t_{\max}(p), p) = \frac{G^{\lim}(t_{\max})}{|V_{p^*}(\hat{x})|} \frac{u}{|B_{1n}(p_n - p_n^*) + o(||p - p^*||_1)_1|} \left(\left| \sum_j B_{1j} + o(1) \right| + o(1) \right) + o(1).$$

Thus,

$$\begin{aligned} G(p) &= \frac{1}{\hat{G}(t_{\max}(p), p)} \\ &= \frac{|V_{p^*}(\hat{x})| |B_{1n}(p_n - p_n^*) + o(||p - p^*||_1)_1|}{uG^{\lim}(t_{\max}) \left(\left| \sum_j B_{1j} + o(1) \right| + o(1) \right) + |V_{p^*}(\hat{x})| |B_{1n}(p_n - p_n^*) + o(||p - p^*||_1)_1| o(1)} \\ &= \frac{|V_{p^*}(\hat{x})| |B_{1n}(p_n - p_n^*) + o(||p - p^*||_1)_1|}{uG^{\lim}(t_{\max}) \left(\left| \sum_j B_{1j} + o(1) \right| + o(1) \right) + o(||p - p^*||_1)} \end{aligned}$$

which approaches a dot product - an affine function - as $||p - p^*||_1 \to 0$, which corresponds to $p \to p^*$.

Remark 2.6.20. The proof of Theorem 2.6.13 reveals that the result can be extended to vector fields with C^{∞} event-selected discontinuities (or C^k with k sufficiently large). In particular, close examination of the above proof indicates that while smoothness was required on an arbitrarily small neighborhood of $X^*(p^*)$ in $M \times J$, a piecewise C^k would be sufficient to complete the remainder of the proof. As equilibria are disjoint from switching surfaces for vector fields with event-selected discontinuities, and as the flow is piecewise C^k (k equal to the smoothness of the vector field selection functions) for vector fields with event-selected discontinuities, the proof of Theorem 2.6.13 for vector fields with C^{∞} event-selected discontinuities proceeds analogously to the above

2.6.4 Conclusions

This section developed theoretical motivation for the algorithms of Section 3.3 for vector fields with C^1 event-selected discontinuities. It was shown that the infimum over time of the inverse trajectory sensitivity is continuous and strictly positive over the set of recovery values, and converges to zero as it approaches the recovery boundary. This provides justification for algorithms which determine the recovery boundary by varying parameter values so as to minimize the infimum over time of the inverse trajectory sensitivities. Furthermore, it was shown that as parameter values approach the recovery boundary, the infimum over time of the inverse trajectory sensitivity approaches a dot product - a linear structure - motivating algorithms in Section 3.3.7.

CHAPTER III

Algorithms

This chapter develops the algorithms which were theoretically motivated by Chapter II. The numerical backbones of these algorithms are trajectory sensitivities, which are first and higher order derivatives of the system trajectory with respect to parameters. Therefore, to apply the algorithms it is important that these be computed efficiently. Section 3.1 introduces modular techniques for numerical integration of first and higher order trajectory sensitivities for hybrid dynamical systems, which include switching and limit behavior, that exploit sparse network structure to reduce computation time and storage requirements. In particular, the necessary equations for numerical integration of first, second, and third order trajectory sensitivities, including derivations of initial conditions and behavior at discrete switching events, are presented. Although first and second order trajectory sensitivity integration for hybrid systems has been developed previously [HP00, GH19], the use of modularity to improve computational efficiency is novel, as are the formulation introduced here for computation of sensitivities at discrete events and the extension of the above to third order sensitivities.

Two classes of algorithms are designed for numerical computation of critical parameter values based on the theory of Chapter II. Section 3.2 presents algorithms which vary parameter values so as to maximize the time the system trajectory spends inside a ball around the controlling invariant set (typically an equilibrium point), thereby driving them to their critical values. Section 3.3 presents algorithms which vary parameter values so as to maximize the norm of the trajectory sensitivities with respect to parameters, thereby driving the parameter values to their critical values. In the latter case, the maximization is performed by minimizing inverse trajectory sensitivities for numerical reasons. Within each class of algorithms, several algorithm types are developed. In the case of one-dimensional parameter space, the unique closest critical parameter value is nu-

merically computed. For two-dimensional parameter space, the recovery boundary is traced numerically via continuation methods. And for arbitrary dimensional parameter space, algorithms are introduced to find the nearest point on the recovery boundary to some initial set of recovery values.

3.1 Efficient Computation of Nonsmooth, Higher Order Trajectory Sensitivities

3.1.1 Introduction

The algorithms described in Sections 3.2-3.3 require trajectory sensitivities for implementation. Trajectory sensitivities are partial derivatives of the flow with respect to components of the initial condition or parameters. This section develops methods for efficient numerical integration of trajectory sensitivities that exploit sparse network structure to reduce computational cost and storage requirements. In particular, efficient integration schemes for computation of first, second, and third order trajectory sensitivities are presented for systems described by differential algebraic equations (DAEs) that can also exhibit discrete impulse and switching behavior. First [HP00] and second [GH19] order trajectory sensitivity integration schemes have been previously developed for this setting. The novel contributions here are the exploitation of network structure to improve efficiency and reduce memory usage, the integration of third order trajectory sensitivities, and a novel formulation for sensitivity computation at discrete events.

This section is organized as follows. Section 3.1.2 provides background and notation for the model of interest and trajectory sensitivities. Section 3.1.3 develops the time derivatives of the trajectory sensitivities away from discrete events. Section 3.1.4 provides the equations for numerical integration of the trajectory sensivities away from discrete events. Section 3.1.5 gives the initial conditions for trajectory sensitivities with respect to parameters. Section 3.1.6 develops the equations for trajectory sensitivities at discrete impulse and switching events. Section 3.1.7 introduces techniques to exploit sparse network topology to reduce computational cost and storage requirements. Finally, Section 3.1.8 offers some concluding remarks.

3.1.2 Background

The model of interest away from discrete events is a coupled set of C^1 differential and algebraic equations (DAEs). Let $x \in \mathbb{R}^n$ be dynamic states, $y \in \mathbb{R}^m$ be algebraic states, and write

$$\dot{x} = f(x, y)$$

$$0 = g(x, y)$$
(3.1)

where $f : \mathbb{R}^{n+m} \to \mathbb{R}^n$ and $g : \mathbb{R}^{n+m} \to \mathbb{R}^m$ are both C^1 , f represents the differential equations, and g represents the algebraic equations. In general, we let dots denote derivatives (typically of states) with respect to time. Under reasonable assumptions [BBCK08] this system of DAEs can be treated as an equivalent system of ordinary differential equations (ODEs), so that the theory developed in Chapter II may be applied. Let ν denote the current time step of numerical integration, and let Δ denote the step size of each integration time step. Let the superscript of ν on a term denote the corresponding value of that term evaluated at $(x(\nu), y(\nu))$, and similarly for superscripts of $\nu + 1$. Away from switching events, Eq. 3.1 is numerically integrated by the following trapezoidal integration scheme:

$$x^{\nu+1} = x^{\nu} + \frac{\Delta}{2} \left(f^{\nu+1} + f^{\nu} \right)$$

$$0 = g(x^{\nu+1}, y^{\nu+1}).$$
(3.2)

This system of nonlinear equations is typically solved numerically for the unknown variables $(x^{\nu+1}, y^{\nu+1})$ via Newton-Raphson.

Triggering of each discrete event is defined by an associated switching function s: $\mathbb{R}^{n+m} \to \mathbb{R}$ such that a switch occurs precisely when s(x, y) = 0. Let (x^-, y^-) denote the states at the time instant just before the switch, and let (x^+, y^+) denote the states at the time instant just after the switch. Let τ denote the intersection time of the orbit with the hypersurface s(x, y) = 0, which is also the time of the event, and is an implicit function of initial conditions. The dynamics of the switch are given by

$$\begin{aligned} x^{-}(x(0), y(0)) &= \phi(x(0), y(0), \tau(x(0), y(0))) \\ 0 &= s(x^{-}, y^{-}) \\ 0 &= g^{-}(x^{-}, y^{-}) \\ x^{+} &= h(x^{-}, y^{-}) \\ 0 &= g^{+}(x^{+}, y^{+}) \end{aligned}$$
(3.3)

where $h : \mathbb{R}^{n+m} \to \mathbb{R}^n$ is a C^1 impulse function, $g^- : \mathbb{R}^{n+m} \to \mathbb{R}^n$ are the set of algebraic equations at the time instant before the switch, and $g^+ : \mathbb{R}^{n+m} \to \mathbb{R}^n$ are the set of algebraic equations at the time instant just after the switch. Hence, in this model formulation, switching of the vector field occurs indirectly through the switching of the algebraic equations from g^- to g^+ . Let $\phi(x, y, t)$ denote the flow of Eqs. 3.1 3.3 at time t from initial condition (x, y). Let $\hat{\phi}(x, y, t)$ denote the dynamic states of $\phi(x, y, t)$, and let $\tilde{\phi}(x, y, t)$ denote the algebraic states of $\phi(x, y, t)$.

Trajectory sensitivities are partial derivatives of the flow $\phi(x, t)$ with respect to components of the initial condition (x, y). Trajectory sensitivities are therefore functions of both time and initial condition. The order of a trajectory sensitivity is the number of partial derivatives of the flow that it involves. To consider trajectory sensitivities with respect to parameters, we introduce each parameter p as a dynamic state with trivial dynamics given by $\dot{p} = 0$. Then the trajectory sensitivity with respect to parameter p is simply the trajectory sensitivity with respect to the initial condition of the dynamic state p.

We introduce the following notation for first-order trajectory sensitivities, where $i, j \in \{1, ..., n\}$ and $\alpha, \beta \in \{1, ..., m\}$.

$$\Phi_{ij}(t) := \frac{\partial \hat{\phi}_i(x, y, t)}{\partial x_j}, \qquad \Phi_{i\alpha}(t) := \frac{\partial \hat{\phi}_i(x, y, t)}{\partial y_\alpha}, \\ \Psi_{\alpha j}(t) := \frac{\partial \tilde{\phi}_\alpha(x, y, t)}{\partial x_j}, \qquad \Psi_{\alpha \beta}(t) := \frac{\partial \tilde{\phi}_\alpha(x, y, t)}{\partial y_\beta}.$$

We often suppress the dependence on time for notational convenience, and write Φ_{ij} , $\Phi_{i\alpha}$, $\Psi_{\alpha j}$, and $\Psi_{\alpha \beta}$ to refer implicitly to the trajectory sensitivities as functions of time. More generally, let $i, j, k, l, m, n, o \in \{1, ..., n\}$ and let $\alpha, \beta, \gamma, \delta \in \{1, ..., m\}$. For higher order trajectory sensitivities of ϕ , given a sequence of subscripts we let the first subscript denote the component of ϕ , the remaining English letter subscripts denote partial derivatives with

respect to components of the dynamic states x, and the remaining Greek letter subscripts denote partial derivatives with respect to components of the algebraic states y. For example,

$$\Phi_{i\alpha j\beta} := \frac{\partial^3 \hat{\phi}_i(x, y, t)}{\partial y_\alpha \partial x_j \partial y_\beta}.$$

Higher order trajectory sensitivities of $\tilde{\phi}$ are defined analogously, for example:

$$\Psi_{\alpha i j \beta} := \frac{\partial^3 \tilde{\phi}_{\alpha}(x, y, t)}{\partial x_i \partial x_j \partial y_{\beta}}.$$

Finally, the partial derivatives of the functions f, g, and h are defined analogously, with the first subscript denoting the component of the function, and the remaining subscripts denoting partial derivatives with respect to dynamic variables (English letters) and algebraic variables (Greek letters).

For clarity and brevity in presenting the equations of the following sections, we use Einstein summation notation, which holds that for each subscript index that appears on the right hand side of an equation but not on the left, that index should be summed over on the right hand side. For example, the equation $a_i = b_{ij} + c_{ijk}$ in Einstein summation notation means $a_i = \sum_j b_{ij} + \sum_{j,k} c_{ijk}$. Many of the subsequent equations will involve tensor contractions, which are a generalization of matrix multiplication. For example, the following is a tensor contraction: $a_{ijk} = \sum_l b_{ijl}c_{lk}$, which we write using the above notation as $a_{ijk} = b_{ijl}c_{lk}$. While matrix multiplication is linear, tensor contractions are multilinear (linear in each dimension, known as modes). This property is used to justify the following procedure for performing tensor contractions in MATLAB: reshaping tensors into matrices, performing matrix multiplication along the desired mode of contraction, and then reshaping the result back into a tensor. Similarly, when solving a system of multilinear equations for a tensor, multilinearity implies that the tensor can be reshaped into a matrix, the ensuing linear system solved in the standard way by MATLAB for a matrix solution, and then the result reshaped back into a tensor.

3.1.3 Time Derivatives

Trajectory sensitivities are computed in practice through numerical integration, as shown in Section 3.1.4. To perform the integration, the variational equations describing their evolution away from discrete events are required. This section is devoted to their presentation. The first- and second-order sensitivity variational equations have been found previously [HP00, GH19], while the third-order sensitivity variational equations are novel. They are obtained by repeated differentiation of Eq. 3.1 with respect to components of the initial condition of the dynamic states, and application of the chain rule.

The first-order variational equations are:

$$\dot{\Phi}_{ij} = f_{im} \Phi_{mj} + f_{i\alpha} \Psi_{\alpha j}$$

$$0 = g_{\alpha m} \Phi_{mj} + g_{\alpha \beta} \Psi_{\beta j}.$$
 (3.4)

The second-order variational equations are given by:

$$\dot{\Phi}_{ijk} = f_{imn}\Phi_{nk}\Phi_{mj} + f_{im\alpha}\Psi_{\alpha k}\Phi_{mj} + f_{im}\Phi_{mjk} + f_{i\alpha m}\Phi_{mk}\Psi_{\alpha j} + f_{i\alpha\beta}\Psi_{\beta k}\Psi_{\alpha j} + f_{i\alpha}\Psi_{\alpha jk}$$
$$0 = g_{\alpha mn}\Phi_{nk}\Phi_{mj} + g_{\alpha m\beta}\Psi_{\beta k}\Phi_{mj} + g_{\alpha m}\Phi_{mjk} + g_{\alpha\beta m}\Phi_{mk}\Psi_{\beta j} + g_{\alpha\beta\gamma}\Psi_{\gamma k}\Psi_{\beta j} + g_{\alpha\beta}\Psi_{\beta jk}.$$
(3.5)

The third-order variational equations are:

$$\begin{split} \dot{\Phi}_{ijkl} &= f_{imno} \Phi_{ol} \Phi_{nk} \Phi_{mj} + f_{imn\alpha} \Psi_{\alpha l} \Phi_{nk} \Phi_{mj} + f_{imn} \Phi_{nkl} \Phi_{mj} + f_{imn\alpha} \Phi_{nk} \Phi_{mjl} + f_{im\alpha\alpha} \Phi_{nl} \Psi_{\alpha k} \Phi_{mjl} \\ &+ f_{im\alpha\beta} \Psi_{\beta l} \Psi_{\alpha k} \Phi_{mj} + f_{im\alpha} \Psi_{\alpha kl} \Phi_{mj} + f_{im\alpha} \Psi_{\alpha k} \Phi_{mjl} + f_{imn\alpha} \Phi_{nl} \Phi_{mjk} + f_{im\alpha} \Psi_{\alpha l} \Phi_{mjk} \\ &+ f_{im} \Phi_{mjkl} + f_{i\alpha mn} \Phi_{nl} \Phi_{mk} \Psi_{\alpha j} + f_{i\alpha m\beta} \Psi_{\beta l} \Phi_{mk} \Psi_{\alpha j} + f_{i\alpha m} \Phi_{mkl} \Psi_{\alpha j} + f_{i\alpha m} \Phi_{mkl} \Psi_{\alpha j} \\ &+ f_{i\alpha \beta m} \Phi_{ml} \Psi_{\beta k} \Psi_{\alpha j} + f_{i\alpha \beta \gamma} \Psi_{\gamma l} \Psi_{\beta k} \Psi_{\alpha j} + f_{i\alpha \beta} \Psi_{\beta kl} \Psi_{\alpha j} + f_{i\alpha \beta} \Psi_{\beta k} \Psi_{\alpha jl} + f_{i\alpha m} \Phi_{ml} \Psi_{\alpha jk} \\ &+ f_{i\alpha \beta} \Psi_{\beta l} \Psi_{\alpha jk} + f_{i\alpha} \Psi_{\alpha jk} \\ 0 &= g_{\alpha mno} \Phi_{ol} \Phi_{nk} \Phi_{mj} + g_{\alpha mn\beta} \Psi_{\beta l} \Phi_{nk} \Phi_{mj} + g_{\alpha mn} \Phi_{nkl} \Phi_{mj} + g_{\alpha mn\beta} \Phi_{nk} \Phi_{mjl} + g_{\alpha m\beta n} \Phi_{nl} \Psi_{\beta k} \Phi_{mj} \\ &+ g_{\alpha m\beta \gamma} \Psi_{\gamma l} \Psi_{\beta k} \Phi_{mj} + g_{\alpha m\beta \beta} \Psi_{\beta kl} \Phi_{mj} + g_{\alpha m\beta \gamma} \Psi_{\beta l} \Phi_{mk} \Psi_{\beta j} + g_{\alpha \beta m} \Phi_{mkl} \Psi_{\beta j} + g_{\alpha \beta \gamma} \Psi_{\gamma kl} \Psi_{\beta j} + g_{\alpha \beta \gamma} \Psi_{\gamma kl} \Psi_{\beta jk} + g_{\alpha \beta \gamma} \Psi_{\beta jkl}. \end{split}$$

An additional set of time derivatives that will be useful for the algorithms described in Section 3.3 are given here. They are obtained by differentiating Eq. 3.1 with respect to time,

and then with respect to components of the initial condition of the dynamic states.

$$g_{\alpha\beta}\dot{y}_{\beta} = -g_{\alpha m}f_{m}$$

$$g_{\alpha\beta}\dot{\Psi}_{\beta j} = -\left(g_{\alpha m n}\Phi_{n j}f_{m} + g_{\alpha m\beta}\Psi_{\beta j}f_{m} + g_{\alpha m}f_{m n}\Phi_{n j} + g_{\alpha m}f_{m\beta}\Psi_{\beta j} + g_{\alpha\beta m}\Phi_{m j}\dot{y}_{\beta} + g_{\alpha\beta\gamma}\Psi_{\gamma j}\dot{y}_{\beta}\right)$$

$$\ddot{\Phi}_{ij} = f_{imn}f_{n}\Phi_{m j} + f_{im\beta}\dot{y}_{\beta}\Phi_{m j} + f_{im}\dot{\Phi}_{m j} + f_{i\betam}f_{m}\Psi_{\beta j} + f_{i\beta\gamma}\dot{y}_{\gamma}\Psi_{\beta j} + f_{i\beta}\dot{y}_{\beta j}, \qquad (3.7)$$

where

$$\dot{y}_{\beta} = rac{\partial \tilde{\phi}_{\beta}(x, y, t)}{\partial t},$$

dots denote first derivatives with respect to time, and double dots denote second derivatives with respect to time. Note that under reasonable assumptions [BBCK08], \dot{y}_{β} and $\dot{\Psi}_{\beta j}$ exist away from discrete events and are given by the equations above.

3.1.4 Numerical Integration

Earlier work has shown that, using a trapezoidal integration scheme to numerically integrate the underlying dynamics of Eq. 3.1, the first [HP00] and second [GH19] order trajectory sensitivities can be efficiently computed as a byproduct of this underlying integration. Here, this efficient computation is extended to third order trajectory sensitivities. Let ν denote the current time step of numerical integration, and let Δ denote the step size of each integration time step. Given the trajectory sensitivities at ν , we will compute their values at $\nu + 1$. For each term in the equations, let the superscript of ν denote the corresponding value of that term evaluated at $(x(\nu), y(\nu), t(\nu))$, and similarly for superscripts of $\nu + 1$. The trajectory sensitivity variational equations of Eqs. 3.4-(3.6) can be numerically integrated using trapezoidal integration as in Eq. 3.2. We formulate this below for first, second, and third order trajectory sensitivities. For the first order trajectory sensitivities, this gives:

$$\begin{bmatrix} \frac{\Delta}{2} f_{im}^{\nu+1} - I_{im} & \frac{\Delta}{2} f_{i\beta}^{\nu+1} \\ g_{\alpha m}^{\nu+1} & g_{\alpha\beta}^{\nu+1} \end{bmatrix} \begin{bmatrix} \Phi_{mj}^{\nu+1} \\ \Psi_{\beta j}^{\nu+1} \end{bmatrix} = -\begin{bmatrix} \Phi_{ij}^{\nu} \\ 0 \end{bmatrix} - \frac{\Delta}{2} \begin{bmatrix} f_{im}^{\nu} \Phi_{mj}^{\nu} + f_{i\alpha}^{\nu} \Psi_{\alpha j}^{\nu} \\ 0 \end{bmatrix}.$$
 (3.8)

Trapezoidal integration of the second order sensitivities is given by:

$$\begin{bmatrix} \frac{\Delta}{2} f_{im}^{\nu+1} - I_{im} & \frac{\Delta}{2} f_{i\beta}^{\nu+1} \\ g_{\alpha m}^{\nu+1} & g_{\alpha \beta}^{\nu+1} \end{bmatrix} \begin{bmatrix} \Phi_{mjk}^{\nu+1} \\ \Psi_{\beta jk}^{\nu+1} \end{bmatrix} = -\begin{bmatrix} \Phi_{ijk}^{\nu} \\ 0 \end{bmatrix} - \begin{bmatrix} \frac{\Delta}{2} \left(f_{imn}^{\nu} \Phi_{nk}^{\nu} \Phi_{mj}^{\nu} + f_{im\alpha}^{\nu} \Phi_{\alpha k}^{\nu} \Phi_{mj}^{\nu} \right) \\ g_{\alpha mn}^{\nu+1} \Phi_{nk}^{\nu+1} \Phi_{mj}^{\nu+1} + g_{\alpha m\beta}^{\nu+1} \Psi_{\beta k}^{\nu+1} \Phi_{mj}^{\nu+1} \end{bmatrix} \\ -\begin{bmatrix} \frac{\Delta}{2} \left(f_{im}^{\nu} \Phi_{mjk}^{\nu} + f_{i\alpha m}^{\nu} \Phi_{mk}^{\nu} \Psi_{\alpha j}^{\nu} + f_{i\alpha \beta}^{\nu} \Psi_{\beta k}^{\nu} \Psi_{\alpha j}^{\nu} + f_{i\alpha \gamma}^{\nu} \Psi_{\alpha jk}^{\nu} + f_{imn}^{\nu+1} \Phi_{nk}^{\nu+1} \Phi_{mj}^{\nu+1} \right) \\ g_{\alpha \beta m}^{\nu+1} \Phi_{mk}^{\nu+1} \Psi_{\beta j}^{\nu+1} + g_{\alpha \beta \gamma}^{\nu+1} \Psi_{\gamma k}^{\nu+1} \Psi_{\beta j}^{\nu+1} \end{bmatrix} \\ -\begin{bmatrix} \frac{\Delta}{2} \left(f_{im\alpha}^{\nu+1} \Psi_{\alpha k}^{\nu+1} \Phi_{mj}^{\nu+1} + f_{i\alpha m}^{\nu+1} \Phi_{mk}^{\nu+1} + f_{i\alpha \beta}^{\nu+1} \Psi_{\beta k}^{\nu+1} \Psi_{\beta j}^{\nu+1} \right) \\ 0 \end{bmatrix}$$
(3.9)

The third order sensitivity integration is given by:

$$\begin{bmatrix} \frac{\lambda}{2} f_{imn}^{\nu+1} - I_{im} & \frac{\lambda}{2} f_{ijn}^{\nu+1} \\ g_{\alpha mn}^{\nu+1} & g_{\alpha m}^{\nu+1} \end{bmatrix} \begin{bmatrix} \Phi_{njkl}^{\nu+1} \\ \Phi_{njkl}^{\nu+1} \end{bmatrix} = - \begin{bmatrix} \Phi_{ijkl}^{\nu} \\ 0 \end{bmatrix} - \begin{bmatrix} \frac{\lambda}{2} \left(f_{imn}^{\nu} \Phi_{nk}^{\nu} \Phi_{mj}^{\nu+1} + f_{imn}^{\nu+1} \Phi_{mk}^{\nu+1} \Phi_{mj}^{\nu+1} + g_{\alpha mnn}^{\nu+1} \Phi_{kl}^{\nu+1} \Phi_{mj}^{\nu+1} + g_{\alpha mnn}^{\nu+1} \Phi_{kl}^{\nu+1} \Phi_{mj}^{\nu+1} \end{bmatrix} \\ - \begin{bmatrix} \frac{\lambda}{2} \left(f_{imn}^{\nu} \Phi_{nkl}^{\nu} \Phi_{mj}^{\nu} + f_{imn}^{\nu} \Phi_{nk}^{\nu} \Phi_{mj}^{\nu} + f_{imnn}^{\nu} \Phi_{nk}^{\nu} \Phi_{mj}^{\nu} + f_{imnn}^{\nu} \Phi_{nk}^{\nu} \Phi_{mj}^{\nu} + f_{imnn}^{\nu+1} \Phi_{mj}^{\nu+1} + g_{mnn}^{\nu+1} \Phi_{mj}^{\nu+1} + g_{mj}^{\nu+1} + g_{mj}^{\nu+1$$

Efficient methods to perform the integration will be discussed in Section 3.1.7.

3.1.5 Initial Conditions

To perform the trajectory sensitivity integration, it is necessary to find the proper initial conditions. As the application of trajectory sensitivities here is to study the response of a system to large disturbances, it is natural to consider systems which are initially at a stable equilibrium point (before being subjected to the disturbance). Typically such equilibrium points are functions of parameter. Therefore, we assume that $(x, y) = (x_e(p), y_e(p))$ is an equilibrium point and a function of parameter. Hence,

$$f(x_e(p), y_e(p)) = 0$$

$$g(x_e(p), y_e(p)) = 0.$$
(3.11)

In the equations below, all trajectory sensitivities shown are assumed to be the values at time t = 0. The initial conditions for the trajectory sensitivities are obtained by repeated differentiation of Eq. 3.11 with respect to parameters. For first order trajectory sensitivity initial conditions this yields:

$$\begin{bmatrix} f_{im} & f_{i\beta} \\ g_{\alpha m} & g_{\alpha \beta} \end{bmatrix} \begin{bmatrix} \Phi_{mj} \\ \Psi_{\beta j} \end{bmatrix} = \begin{bmatrix} 0 \\ 0 \end{bmatrix}$$
(3.12)

which can be rewritten as:

$$\begin{bmatrix} f_x & f_y \\ g_x & g_y \end{bmatrix} \begin{bmatrix} \tilde{\Phi}_{mj} \\ \Psi_{\beta j} \end{bmatrix} = - \begin{bmatrix} f_{param} \\ g_{param} \end{bmatrix}$$
(3.13)

where f_x and g_x are derivatives of f and g with respect to the dynamic states with nontrivial dynamics (the dynamic states which are not parameters), f_{param} and g_{param} are derivatives of f and g with respect to the parameters p, and $\tilde{\Phi}_{mj}$ consists only of the trajectory sensitivities of the dynamic states which are not parameters.

The second order trajectory sensitivity initial conditions are given by:

$$\begin{bmatrix} f_{im} & f_{i\beta} \\ g_{\alpha m} & g_{\alpha\beta} \end{bmatrix} \begin{bmatrix} \Phi_{mjk} \\ \Psi_{\beta jk} \end{bmatrix} = -\begin{bmatrix} f_{imn}\Phi_{nk}\Phi_{mj} + f_{im\alpha}\Psi_{\alpha k}\Phi_{mj} + f_{i\alpha m}\Phi_{mk}\Psi_{\alpha j} + f_{i\alpha\beta}\Psi_{\beta k}\Psi_{\alpha j} \\ g_{\alpha mn}\Phi_{nk}\Phi_{mj} + g_{\alpha m\beta}\Psi_{\beta k}\Phi_{mj} + g_{\alpha\beta m}\Phi_{mk}\Psi_{\beta j} + g_{\alpha\beta\gamma}\Psi_{\gamma k}\Psi_{\beta j} \end{bmatrix}$$
(3.14)

The third order trajectory sensitivity initial conditions are given by:

$$\begin{bmatrix} f_{im} & f_{i\beta} \\ g_{\alpha m} & g_{\alpha\beta} \end{bmatrix} \begin{bmatrix} \Phi_{mjkl} \\ \Psi_{\beta jkl} \end{bmatrix} = -\begin{bmatrix} f_{imno}\Phi_{ol}\Phi_{nk}\Phi_{mj} + f_{imn\alpha}\Psi_{al}\Phi_{nk}\Phi_{mj} + f_{imn\alpha}\Phi_{nkl}\Phi_{mj} \\ g_{\alpha mno}\Phi_{ol}\Phi_{nk}\Phi_{mj} + g_{\alpha mn\beta}\Psi_{\beta l}\Phi_{nk}\Phi_{mj} + g_{\alpha mn\beta}\Psi_{\beta l}\Phi_{nk}\Phi_{mj} + g_{\alpha mn}\Phi_{nkl}\Phi_{mj} \end{bmatrix} \\ -\begin{bmatrix} f_{imn}\Phi_{nk}\Phi_{mjl} + f_{im\alpha n}\Phi_{nl}\Psi_{\alpha k}\Phi_{mj} + f_{im\alpha\beta}\Psi_{\beta l}\Psi_{\alpha k}\Phi_{mj} + f_{im\alpha}\Psi_{\alpha kl}\Phi_{mj} + f_{im\alpha}\Psi_{\alpha k}\Phi_{mjl} \\ g_{\alpha mn}\Phi_{nk}\Phi_{mjl} + g_{\alpha m\beta n}\Phi_{nl}\Psi_{\beta k}\Phi_{mj} + g_{\alpha m\beta\gamma}\Psi_{\gamma l}\Psi_{\beta k}\Phi_{mj} + g_{\alpha m\beta}\Psi_{\beta kl}\Phi_{mj} + g_{\alpha m\beta}\Psi_{\beta k}\Phi_{mjl} \end{bmatrix} \\ -\begin{bmatrix} f_{imn}\Phi_{nl}\Phi_{mjk} + f_{im\alpha}\Psi_{\alpha l}\Phi_{mjk} + f_{i\alpha mn}\Phi_{nl}\Phi_{mk}\Psi_{\alpha j} + f_{i\alpha m\beta}\Psi_{\beta l}\Phi_{mk}\Psi_{\alpha j} + f_{i\alpha m}\Phi_{mkl}\Psi_{\alpha j} \\ g_{\alpha mn}\Phi_{nl}\Phi_{mjk} + g_{\alpha m\beta}\Psi_{\beta l}\Phi_{mjk} + g_{\alpha \beta mn}\Phi_{nl}\Phi_{mk}\Psi_{\beta j} + g_{\alpha \beta m\gamma}\Psi_{\gamma l}\Phi_{mk}\Psi_{\beta j} + g_{\alpha \beta m}\Phi_{mkl}\Psi_{\beta j} \end{bmatrix} \\ -\begin{bmatrix} f_{i\alpha m}\Phi_{mk}\Psi_{\alpha jl} + f_{i\alpha \beta m}\Phi_{ml}\Psi_{\beta k}\Psi_{\alpha j} + f_{i\alpha \beta}\Psi_{\gamma l}\Psi_{\beta k}\Psi_{\alpha j} + f_{i\alpha \beta}\Psi_{\beta k}\Psi_{\alpha j} \\ g_{\alpha \beta m}\Phi_{mk}\Psi_{\beta jl} + g_{\alpha \beta \gamma m}\Phi_{ml}\Psi_{\gamma k}\Psi_{\beta j} + g_{\alpha \beta \gamma}\Psi_{\gamma l}\Psi_{\beta jk} \end{bmatrix} \\ -\begin{bmatrix} f_{i\alpha \beta}\Psi_{\beta k}\Psi_{\alpha jl} + f_{i\alpha m}\Phi_{ml}\Psi_{\alpha jk} + f_{i\alpha \beta}\Psi_{\beta l}\Psi_{\alpha jk} \\ g_{\alpha \beta \gamma}\Psi_{\gamma k}\Psi_{\beta jl} + g_{\alpha \beta m}\Phi_{ml}\Psi_{\beta jk} + g_{\alpha \beta \gamma}\Psi_{\gamma l}\Psi_{\beta jk} \end{bmatrix}$$

$$(3.15)$$

3.1.6 Discrete Events

This section presents a novel formulation of trajectory sensitivity computations at discrete impulse and switching events. The dynamics at the discrete event are given by Eq. 3.3. Appendix D shows the derivation of the computation of trajectory sensitivities at the discrete event. The final results are summarized below. Let τ_j , τ_{jk} , and τ_{jkl} denote derivatives of the intersection time τ with respect to components of the initial condition of the dynamic states. Let terms with minus superscripts denote the value of that term just before the discrete event, and let terms with plus superscripts denote the value of that term just after the discrete event.

The equations for computation of first order trajectory sensitivities after a discrete event are:

$$\begin{bmatrix} s_{im}f_m^- & s_{i\beta} \\ g_{\alpha m}^-f_m^- & g_{\alpha\beta}^- \end{bmatrix} \begin{bmatrix} \tau_j \\ \Psi_{\beta j}^- \end{bmatrix} = -\begin{bmatrix} s_{im}\Phi_{mj} \\ g_{\alpha m}^-\Phi_{mj} \end{bmatrix}$$
(3.16)

$$\Phi_{ij}^- = \Phi_{ij} + f_i^- \tau_j \tag{3.17}$$

$$\begin{bmatrix} -I_{im} & 0\\ g^+_{\alpha m} & g^+_{\alpha \beta} \end{bmatrix} \begin{bmatrix} \Phi^+_{mj}\\ \Psi^+_{\beta j} \end{bmatrix} = -\begin{bmatrix} h_{im}\Phi^-_{mj} + h_{i\alpha}\Psi^-_{\alpha j}\\ 0 \end{bmatrix}.$$
 (3.18)

The second order trajectory sensitivity equations at a discrete event are given by:

$$\begin{bmatrix} s_{im}f_{m}^{-} & s_{i\beta} \\ g_{\alpha m}^{-}f_{m}^{-} & g_{\alpha\beta}^{-} \end{bmatrix} \begin{bmatrix} \tau_{jk} \\ \Psi_{\beta jk}^{-} \end{bmatrix} = -\begin{bmatrix} s_{imn}\Phi_{nk}^{-}\Phi_{mj}^{-} + s_{im\alpha}\Psi_{\alpha k}^{-}\Phi_{mj}^{-} + s_{i\alpha m}\Phi_{mk}^{-}\Psi_{\alpha j}^{-} + s_{i\alpha\beta}\Psi_{\beta k}^{-}\Psi_{\alpha j}^{-} \\ g_{\alpha mn}^{-}\Phi_{nk}^{-}\Phi_{mj}^{-} + g_{\alpha m\beta}^{-}\Psi_{\beta k}^{-}\Phi_{mj}^{-} + g_{\alpha\beta m}^{-}\Phi_{mk}^{-}\Psi_{\beta j}^{-} \end{bmatrix} - \begin{bmatrix} s_{im}\left(\Phi_{mjk} + f_{mn}^{-}\Phi_{nk}\tau_{j} + f_{m\alpha}^{-}\Psi_{\alpha k}\tau_{j}\right) \\ g_{\alpha m}^{-}\left(\Phi_{mjk} + f_{mn}^{-}\Phi_{nk}\tau_{j} + f_{m\beta}^{-}\Psi_{\beta k}\tau_{j}\right) \end{bmatrix}$$
(3.19)
$$\Phi^{-} = \Phi_{m} + f_{m}^{-}\Phi_{mk}\tau_{j} + f_{m\beta}^{-}\Psi_{\beta k}\tau_{j} + f_{m\beta}^{-}\Psi_{\beta k}\tau_{j} + f_{m\beta}^{-}\Psi_{\beta k}\tau_{j} \end{bmatrix}$$

$$\Phi_{ijk} = \Phi_{ijk} + f_{im} \Phi_{mk} \tau_j + f_{i\alpha} \Psi_{\alpha k} \tau_j + f_i^{-} \tau_{jk}$$
(3.20)
$$\begin{bmatrix} -I_{im} & 0\\ g^+_{\alpha m} & g^+_{\alpha \beta} \end{bmatrix} \begin{bmatrix} \Phi^+_{mjk}\\ \Psi^+_{\beta jk} \end{bmatrix} = -\begin{bmatrix} h_{imn} \Phi^-_{nk} \Phi^-_{mj} + h_{im\alpha} \Psi^-_{\alpha k} \Phi^-_{mj} + h_{im} \Phi^-_{mjk} + h_{i\alpha m} \Phi^-_{mk} \Psi^-_{\alpha j}\\ g^+_{\alpha mn} \Phi^+_{nk} \Phi^+_{mj} + g^+_{\alpha m\beta} \Psi^+_{\beta k} \Phi^+_{mj} + g^+_{\alpha \beta m} \Phi^+_{mk} \Psi^+_{\beta j} + g^+_{\alpha \beta \gamma} \Psi^+_{\gamma k} \Psi^+_{\beta j} \end{bmatrix}$$
$$-\begin{bmatrix} h_{i\alpha\beta} \Psi^-_{\beta k} \Psi^-_{\alpha j} + h_{i\alpha} \Psi^-_{\alpha jk}\\ 0 \end{bmatrix}.$$
(3.21)

The third order trajectory sensitivity equations at a discrete event are:

$$\begin{split} \begin{bmatrix} s_{im}f_m^- & s_{i\beta} \\ g_{\alpha}^-mf_m^- & g_{\alpha}^-\beta \end{bmatrix} \begin{bmatrix} \tau_{jkl} \\ \Psi_{\beta jkl} \end{bmatrix} &= -\begin{bmatrix} s_{imno}\Phi_{\alpha}^-\Phi_{nk}^-\Phi_{mj}^- + s_{imn\alpha}\Psi_{\alpha}^-\Phi_{nk}^-\Phi_{mj}^- + s_{imn\alpha}\Phi_{nkl}^-\Phi_{mj}^- \end{bmatrix} \\ &- \begin{bmatrix} s_{imn}\Phi_{nk}^-\Phi_{mjl}^- + s_{im\alpha\alpha}\Phi_{nl}^-\Psi_{\alphak}^-\Phi_{mj}^- + s_{im\alpha\beta}\Psi_{\beta}^-\Psi_{\alpha}^-\Phi_{mj}^- + s_{im\alpha}\Psi_{\alpha}^-\Phi_{nkl}^-\Phi_{mj}^- \end{bmatrix} \\ &- \begin{bmatrix} s_{imn}\Phi_{nk}^-\Phi_{mjl}^- + s_{im\alpha\alpha}\Phi_{nl}^-\Psi_{\betak}^-\Phi_{mj}^- + s_{im\alpha\beta}\Psi_{\beta}^-\Psi_{mj}^- + s_{im\alpha\beta}\Psi_{\alpha}^-\Phi_{mjl}^- + s_{im\alpha\beta}\Psi_{\alpha}^-\Phi_{mjl}^- \end{bmatrix} \\ &- \begin{bmatrix} s_{imn}\Phi_{nk}^-\Phi_{mjl}^- + s_{im\alpha\beta}^-\Phi_{nl}^-\Psi_{\betak}^-\Phi_{mj}^- + s_{i\alpham\beta}^-\Psi_{\gamma}^-\Psi_{\gamma}^-\Psi_{\betak}^-\Phi_{mj}^- + s_{i\alpham\beta}^-\Psi_{\betak}^-\Phi_{mjl}^- + s_{i\alpham\beta}^-\Phi_{mkl}^-\Phi_{mjl}^- \end{bmatrix} \\ &- \begin{bmatrix} s_{imn}\Phi_{nl}^-\Phi_{mjk}^- + s_{i\alpham\beta}^-\Phi_{nl}^-\Phi_{mjk}^- + s_{i\alpham\beta}^-\Phi_{ml}^-\Phi_{mk}^-\Psi_{\alphaj}^- + s_{i\alpham\beta}^-\Phi_{mkl}^-\Psi_{\alphaj}^- + s_{i\alpham\beta}^-\Phi_{mkl}^-\Psi_{\alphaj}^- \end{bmatrix} \\ &- \begin{bmatrix} s_{i\alpham}\Phi_{ml}^-\Phi_{mjk}^- + s_{i\alpha\beta}^-\Phi_{ml}^-\Psi_{\betak}^- \Psi_{\alphaj}^- + s_{i\alpha\beta\beta}^-\Psi_{\betak}^- \Psi_{\alphaj}^- + s_{i\alpha\beta\beta}^-\Psi_{\alphajk}^- \Psi_{\alphaj}^- + s_{i\alpha\beta\beta}^-\Psi_{\alphajk}^- \Psi_{\alphaj}^- \end{bmatrix} \\ &- \begin{bmatrix} s_{i\alpham}\Phi_{mk}^-\Psi_{\alphajl}^- + s_{i\alpha\beta}^- \Phi_{ml}^- \Psi_{\betaj}^- + s_{i\alpha\beta\gamma}^- \Psi_{\alphal}^- \Psi_{\betaj}^- + s_{i\alpha\beta\gamma}^- \Psi_{\alphajk}^- \Psi_{\alphajl}^- + s_{i\alpha\beta\beta}^- \Psi_{\alphajk}^- \Psi_{\alphajl}^- \end{bmatrix} \\ &- \begin{bmatrix} s_{i\alpham}\Phi_{mk}^-\Psi_{\alphajl}^- + s_{i\alpha\beta}^- \Psi_{ml}^- \Psi_{\betajk}^- + s_{\alpha\beta\gamma}^- \Psi_{\alphal}^- \Psi_{\betaj}^- + s_{\alpha\beta\gamma}^- \Psi_{\alphajk}^- \Psi_{\betaj}^- + s_{\alpha\beta\gamma}^- \Psi_{\alphajk}^- \Psi_{\betaj}^- + s_{\alpha\beta\gamma}^- \Psi_{\alphajk}^- \Psi_{\betaj}^- + s_{\alpha\beta\gamma}^- \Psi_{\alpha\beta\gamma}^- \Psi$$

$$\begin{bmatrix} -I_{im} & 0\\ g_{\alpha m}^{+} & g_{\alpha \beta}^{+} \end{bmatrix} \begin{bmatrix} \Phi_{mjkl}^{+}\\ \Psi_{jjkl}^{+} \end{bmatrix} = -\begin{bmatrix} h_{imno}\Phi_{ol}^{-}\Phi_{nk}^{-}\Phi_{mj}^{-} + h_{imn\alpha}\Psi_{al}^{-}\Phi_{nk}^{-}\Phi_{mj}^{-} + h_{imn\alpha}\Phi_{nkl}^{-}\Phi_{mj}^{-} \\ g_{\alpha mno}^{+}\Phi_{ol}^{+}\Phi_{nk}^{+}\Phi_{mj}^{+} + g_{\alpha mn\beta}^{+}\Psi_{\beta l}^{+}\Phi_{nk}^{+}\Phi_{mj}^{+} + g_{\alpha mn\alpha}^{+}\Phi_{nkl}^{+}\Phi_{mj}^{+} \\ g_{\alpha mn}^{+}\Phi_{nk}^{-}\Phi_{mjl}^{-} + h_{im\alpha n}\Phi_{nl}^{-}\Psi_{\alpha k}^{-}\Phi_{mj}^{-} + h_{im\alpha \beta}\Psi_{\beta l}^{-}\Psi_{\alpha k}^{-}\Phi_{mj}^{-} + h_{im\alpha \gamma}\Psi_{\alpha kl}^{-}\Phi_{mj}^{-} + h_{im\alpha \gamma}\Psi_{\alpha kl}^{-}\Phi_{mj}^{-} \\ g_{\alpha mn}^{+}\Phi_{nk}^{+}\Phi_{mjl}^{+} + g_{\alpha m\beta n}^{+}\Phi_{nl}^{+}\Psi_{\beta k}^{+}\Phi_{mj}^{+} + g_{\alpha m\beta \gamma}^{+}\Psi_{\gamma l}^{+}\Psi_{\beta k}^{+}\Phi_{mj}^{+} + g_{\alpha m\beta \gamma}^{+}\Psi_{\beta kl}^{+}\Phi_{mj}^{+} \\ = \begin{bmatrix} h_{imn}\Phi_{nl}^{-}\Phi_{mjk}^{-} + h_{im\alpha}\Psi_{al}^{-}\Phi_{mjk}^{-} + h_{im}\Phi_{mjkl}^{-} + h_{i\alpha mn}\Phi_{nl}^{-}\Phi_{mk}^{-}\Psi_{\alpha j}^{-} + h_{i\alpha m\beta}\Psi_{\beta l}^{-}\Phi_{mk}^{-}\Psi_{\alpha j}^{-} \\ g_{\alpha mn}^{+}\Phi_{nl}^{+}\Phi_{mjk}^{+} + g_{\alpha m\beta \gamma}^{+}\Psi_{\beta l}^{+} + g_{\alpha \beta mn}^{-}\Phi_{ml}^{-}\Phi_{mk}^{-}\Psi_{\alpha j}^{-} + h_{i\alpha \beta m}\Phi_{ml}^{-}\Psi_{\beta k}^{-}\Psi_{\alpha j}^{-} + h_{i\alpha \beta m}\Phi_{mkl}^{-}\Psi_{\beta j}^{-} \\ = \begin{bmatrix} h_{i\alpha m}\Phi_{mkl}^{-}\Psi_{\alpha j}^{-} + h_{i\alpha m}\Phi_{mk}^{-}\Psi_{\alpha jl}^{-} + h_{i\alpha \beta m}\Phi_{ml}^{-}\Psi_{\beta k}^{-}\Psi_{\alpha j}^{-} + h_{i\alpha \beta \gamma}\Psi_{\gamma l}^{-}\Psi_{\beta k}^{-}\Psi_{\gamma l}^{-} + h_{i\alpha \beta \gamma}\Psi_{\gamma l}^{-}\Psi_{\beta j}^{+} \\ g_{\alpha \beta m}^{+}\Phi_{mk}^{+}\Psi_{\beta jl}^{+} + g_{\alpha \beta \gamma m}^{+}\Phi_{ml}^{+}\Psi_{\beta j}^{+} + g_{\alpha \beta \gamma \gamma}^{+}\Psi_{\gamma l}^{+}\Psi_{\beta j}^{+} + g_{\alpha \beta \gamma \gamma}^{+}\Psi_{\gamma l}^{+}\Psi_{\beta j}^{-} \\ = \begin{bmatrix} h_{i\alpha \beta}\Psi_{\beta k}^{-}\Psi_{\alpha jl}^{-} + h_{i\alpha m}\Phi_{ml}^{-}\Psi_{\gamma k}^{-}\Psi_{\beta j}^{+} + g_{\alpha \beta \gamma \gamma}^{-}\Psi_{\gamma l}^{+}\Psi_{\beta j}^{+} + g_{\alpha \beta \gamma \gamma}^{-}\Psi_{\gamma l}^{+}\Psi_{\beta j}^{+} + g_{\alpha \beta \gamma \gamma}^{-}\Psi_{\gamma l}^{+}\Psi_{\beta j}^{+} \\ g_{\alpha \beta m}^{+}\Phi_{mk}^{+}\Psi_{\beta jl}^{-} + g_{\alpha \beta \gamma m}^{+}\Phi_{ml}^{+}\Psi_{\beta j}^{-} + g_{\alpha \beta \gamma \gamma}^{-}\Psi_{\gamma l}^{+}\Psi_{\beta j}^{+} \\ \end{bmatrix} \end{bmatrix}$$

$$(3.24)$$

3.1.7 Efficient Computation and Storage

Improving computational efficiency and reducing memory requirements are important for facilitating trajectory sensitivity integration in practice. The key property we wish to exploit is that for many practical applications that consider dynamics over a sparse network, including power systems, the model given by Eq. 3.1 admits a decomposition as follows into a modular structure. Suppose there are $\mathbb{N} > 0$ components that are connected through the network. For each $i \in \mathbb{N}$, let x_i and y_i refer to the dynamic states and algebraic states associated with that component, respectively. Then assume that Eq. 3.1 can be written in the form:

$$f(x,y) = \begin{bmatrix} f_1(x_1, y_1) \\ f_2(x_2, y_2) \\ \dots \\ f_N(x_N, y_N) \end{bmatrix}$$
$$g(x,y) = \begin{bmatrix} g_1(x_1, y_1) \\ g_2(x_2, y_2) \\ \dots \\ g_N(x_N, y_N) \\ Ly \end{bmatrix}$$
(3.25)

where L is a constant matrix. This structure implies that while the individual components may have complicated, nonlinear and nonsmooth dynamics, their coupling through the network (via the algebraic variables) is linear. For example, this is true in electrical networks such as power systems in which the network coupling equations are given by Kirchoff's Laws, which are linear in currents and voltages.

Assuming the structure of Eq. 3.25, consider first the integration of trajectory sensitivities away from discrete events. Inspection of the right hand sides of Eqs. 3.8-3.10, used for numerical integration of the sensitivities, reveals that every term either consists of the contraction of derivatives of f against sensitivities, or of the contraction of at least second order derivatives of g against sensitivities. However, taking derivatives of f preserves its modular structure, while taking at least second (partial) derivatives of g both preserves the modular part of its structure and cancels out the network part since the second derivative of a linear function is zero. In particular, this implies that the computations required to construct the right hand sides of Eqs. 3.8-3.10 can be carried out in a modular fashion. Hence, both runtimes and storage requirements for these computations grow linearly with the number of network components \mathbb{N} (assuming a bound on the dimension of the network components, which is typical in practice). Furthermore, as a result of this modularity, these computations are easily parallelizable, and could take advantage of parallel computing resources.

In fact, the discussion above also applies to the computation of trajectory sensitivities at discrete events. Each term in the right hand side of Eqs. 3.16-3.24 consists either of the contraction of f or its derivatives against sensitivities, the tensor product of f or its derivatives with derivatives of the intersection time, or of the contraction of at least second order derivatives of g against sensitivities. In all of these cases, the computation can be carried out in a modular fashion. As in the case of integration away from discrete events, this significantly decreases computation time but, perhaps more importantly, it dramatically reduces the memory usage required for the computations. Similarly, these computations could take advantage of parallel computing as well.

Once the computations of the right hand side of Eqs. 3.8-3.10 are complete, it is necessary in each case to solve a system of linear equations. Note that all three linear systems have the same matrix on the left hand side. It was observed [HP00] that this matrix is merely the Jacobian of the system, and is obtained as a byproduct of the underlying trapezoidal integration. In particular, an LU decomposition of this matrix is performed during the underlying integration of the system dynamics so that, as pointed out by [HP00], it is only necessary to solve triangular linear systems for the trajectory sensitivity integration steps, which can be solved very efficiently. Furthermore, at discrete events, the two pairs of linear systems for each order of trajectory sensitivity in Eqs. 3.16-3.24 have the same matrices on the left hand side. As was observed by [GH19], this implies that once the LU decomposition is performed for the first order sensitivity computation, this can be used to solve triangular linear systems for the second (and, we add, third) order sensitivity integration as well.

3.1.8 Conclusion

This section developed trapezoidal integration schemes for first, second, and third order trajectory sensitivities for differential algebraic equations with discrete impulse and switching events. These trajectory sensitivities will be crucial in implementing the algorithms described in Sections 3.2-3.3. The novel contributions presented here are the exploitation of sparse network structure to develop modular and parallel algorithms to drastically reduce computational cost and storage requirements, an integration scheme for third order trajectory sensitivities, and a novel formulation of trajectory sensitivity computation at discrete events.

3.2 Maximizing Time in Ball Around Controlling Equilibrium Point Algorithms

3.2.1 Introduction

Algorithms are developed based on the theoretical motivation of Section 2.5 for determining the recovery boundary via computation of critical parameter values. The key idea is to vary parameter values so as to maximize the time the system trajectory spending inside a ball around the controlling UEP, thereby driving parameter values to their critical values. For the purposes of this discussion, it is assumed that the controlling UEP has already been found by some other method, such as those of [TVK96, Chi11]. Algorithms are developed to find the unique closest critical parameter value in one dimensional parameter space, trace the recovery boundary in two dimensional parameter space, and find the nearest point on the recovery boundary to some initial set of recovery values in higher dimensional parameter space. The algorithms are illustrated on a test case of a network of three synchronous machines.

3.2.2 One Dimensional Parameter Space

The first step of the algorithm is to find the parameter value corresponding to the system spending a fixed target time inside the ball. Let \hat{x} denote the controlling UEP, r the radius

of the ball around it, t_{ball} the target time inside the ball, and $\phi(t, p)$ the system state at time t for parameter value p. Furthermore, let t_1, t_2 denote the initial and final crossings of the boundary of the ball by the system trajectory, and let x_1, x_2 denote the system dynamic states at times t_1, t_2 , respectively. Similarly, let y_1, y_2 denote the system algebraic states at times t_1, t_2 , respectively. Letting $z = [x_1 \ y_1 \ t_1 \ x_2 \ y_2 \ t_2 \ p]^{\intercal}$, these conditions can be described algebraically by,

$$F(z) = \begin{bmatrix} \phi(t_1, p) - x_1 \\ g(x_1, y_1) \\ ||x_1 - \hat{x}||_2^2 - r^2 \\ \phi(t_2, p) - x_2 \\ g(x_2, y_2) \\ ||x_2 - \hat{x}||_2^2 - r^2 \\ t_2 - t_1 - t_{ball} \end{bmatrix}.$$
(3.26)

The desired conditions are obtained by solving,

$$F(z) = 0.$$
 (3.27)

A solution implies that $||x_1 - \hat{x}||_2 = ||x_2 - \hat{x}||_2 = r$, so x_1, x_2 are on the boundary of the ball around the controlling UEP. Also, $\phi(t_1, p) = x_1$ and likewise for t_2 ensures that t_1, t_2 are the initial and final times at which the trajectory intersects the boundary of the ball. Furthermore, $t_{ball} = t_2 - t_1$ enforces the time constraint between the first and final intersections. Finally, $g(x_1, y_1) = g(x_2, y_2) = 0$ ensures that y_1, y_2 are the algebraic states corresponding to the dynamic states x_1, x_2 .

This system of equations can be solved using Newton-Raphson, which requires the cor-

responding Jacobian matrix,

$$J(z) = \frac{\partial F(z)}{\partial z} = \begin{bmatrix} -I & 0 & f|_{1} & 0 & 0 & 0 & \frac{\partial \phi(t_{1},p)}{\partial p} \\ \frac{\partial g}{\partial x}|_{1} & \frac{\partial g}{\partial y}|_{1} & 0 & 0 & 0 & 0 & 0 \\ (x_{1} - \hat{x})^{\mathsf{T}} & 0 & 0 & 0 & 0 & 0 & 0 \\ 0 & 0 & 0 & -I & 0 & f|_{2} & \frac{\partial \phi(t_{2},p)}{\partial p} \\ 0 & 0 & 0 & \frac{\partial g}{\partial x}|_{2} & \frac{\partial g}{\partial y}|_{2} & 0 & 0 \\ 0 & 0 & 0 & (x_{2} - \hat{x})^{\mathsf{T}} & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & -1 & 0 \end{bmatrix}$$
(3.28)

where $f|_1 := f(x_1, y_1)$ and $f|_2 := f(x_2, y_2)$. To form this Jacobian matrix, it is necessary to compute the partial derivatives $\frac{\partial \phi(t_1, p)}{\partial p}$ and $\frac{\partial \phi(t_2, p)}{\partial p}$. They can be obtained from the numerical integration of the trajectory sensitivities, as described in Section 3.1.

The Newton-Raphson update has the usual form,

$$z^{\nu+1} = z^{\nu} - J(z^{\nu})^{-1} F(z^{\nu}) \tag{3.29}$$

where ν is the iteration number. For purposes of computational efficiency, this system is usually solved using either LU or QR decomposition of $J(z^{\nu})$. The result of this Newton-Raphson algorithm is a parameter value p which drives the system to spend a target time t_{ball} inside the ball around the controlling UEP.

The next step is to vary a parameter so as to maximize the time spent by the trajectory in the ball around the controlling UEP. This is accomplished by now treating t_{ball} as a variable in (3.26), giving $z_{cont} = [x_1 \ y_1 \ t_1 \ x_2 \ y_2 \ t_2 \ p \ t_{ball}]^{\intercal}$. Then z_{cont} , together with the constraints $F(z_{cont}) = 0$, locally defines a curve since there is one more variable in z_{cont} than the number of equations in $F(z_{cont}) = 0$. A continuation method, similar in concept to those in [DH06], can be employed to track the solution as t_{ball} is increased. In general, $F(z_{cont}) = 0$ may not be a smooth curve, but empirical observations indicate that it typically is smooth in practice. Further study is required to determine the reasons for this smoothness.

The key idea of this continuation method is to iteratively trace a sequence of points along a curve defined by a set of algebraic equations, as illustrated in Figure 3.1. At each step a prediction is made by moving along the tangent to the curve. Then the predicted point is corrected by projecting back onto the curve. This process repeats until the curve is complete. The unit tangent vector η to the curve can be found by letting

$$J_{cont}(z_{cont}) = \begin{bmatrix} J(z_{cont}) & \frac{\partial F}{\partial t_{ball}}(z_{cont}) \end{bmatrix}.$$
 (3.30)

Then η satisfies

$$J_{cont}(z_{cont})\eta = 0 \tag{3.31}$$

$$||\eta||_2^2 = 1. \tag{3.32}$$

The solution is given by taking a QR decomposition of $J_{cont}(z_{cont})^{\intercal}$, and setting η equal to the final column of the orthogonal matrix in the decomposition. Starting from the ν -th point on the curve, the next predicted point is,

$$z_p = z_{cont}^{\nu} + \kappa \eta \tag{3.33}$$

where κ is the step size. For the correction step, which involves projecting back onto the curve, we require that $z_{cont}^{\nu+1} - z_p$ be orthogonal to η , so

$$\eta^{\mathsf{T}}(z_{cont}^{\nu+1} - z_p) = 0. \tag{3.34}$$

From (3.33), this implies that,

$$\eta^{\mathsf{T}}(z_{cont}^{\nu+1} - z_{cont}^{\nu}) - \kappa = 0.$$
(3.35)

This corrector step is accomplished by setting,

$$F_{proj}(z_{cont}^{\nu+1}) = \begin{bmatrix} F(z_{cont}^{\nu+1}) \\ \eta^{\mathsf{T}}(z_{cont}^{\nu+1} - z_{cont}^{\nu}) - \kappa \end{bmatrix}$$
(3.36)

with the corresponding Jacobian,

$$J_{proj}(z_{cont}^{\nu+1}) = \begin{bmatrix} J_{cont}(z_{cont}^{\nu+1}) \\ \eta^{\mathsf{T}}. \end{bmatrix}$$
(3.37)

Then we can obtain a solution to $F_{proj}(z_{cont}^{\nu+1}) = 0$ using the standard Newton-Raphson update. This method varies the parameter value p so as to move in the direction of maximizing


Figure 3.1: A continuation method for iteratively tracing a sequence of points along a curve defined by a set of algebraic equations. At each step a prediction is made by moving along the tangent to the curve. Then the predicted point is projected back onto the curve. This process repeats until the curve is traced.

 t_{ball} . As a consequence, p is driven to its critical value.

3.2.3 Two Dimensional Parameter Space

Once a critical parameter value has been found, a pair of parameters can be chosen, and their recovery boundary computed. Fix t_{ball} to its value from the end of the continuation process described above. Let $z_{bound} = [z^{\intercal} p_1 p_2]^{\intercal}$ for a pair of parameters p_1, p_2 . Then $F(z_{bound}) = 0$ locally describes a curve in parameter space which lies arbitrarily close to the desired recovery boundary. Using a continuation method like the one described in the previous step of Section 3.2.2, this curve can be traced progressively to obtain the desired recovery boundary. Note that the recovery boundary may not be a smooth curve in general, but numerical experiments have suggested that it is smooth in applications of intersest. The reasons for this remain unknown.

3.2.4 Higher Dimensional Parameter Space

Given a set of P > 0 parameters, let $z = [x_1^{\mathsf{T}} y_1^{\mathsf{T}} t_1 x_2^{\mathsf{T}} y_2^{\mathsf{T}} t_2]^{\mathsf{T}}$ and let $p \in \mathbb{R}^P$ be parameter values. Then F(z, p) = 0, where F is given by Eq. 3.26, locally defines a (P - 1)-manifold. Fix any recover parameter value $p_0 \in \mathbb{R}^P$. The goal is to find the closest point p^* (it may not be unique) on the recovery boundary to p_0 . This can be expressed by the following optimization problem:

$$\min_{z,p} \frac{1}{2} ||p - p_0||_2^2$$
s.t. $F(z,p) = 0.$
(3.38)

The Lagrangian for this problem is:

$$L(z, p, \lambda) = ||p - p_0||_2^2 + \lambda^{\mathsf{T}} F(z, p)$$

where λ is a vector of Lagrange multipliers. Any stationary point, such as a local minimum, must satisfy:

$$\nabla L(z, p, \lambda) = 0 = \begin{bmatrix} \left(\frac{\partial F}{\partial z}\right)^{\mathsf{T}} \lambda\\ \left(p - p_0\right) + \left(\frac{\partial F}{\partial p}\right)^{\mathsf{T}} \lambda\\ F(z, p) \end{bmatrix}.$$
(3.39)

The matrices $\frac{\partial F}{\partial z}$ and $\frac{\partial F}{\partial p}$ can be computed analogously to the computation of J in Eq. 3.28, except that $\frac{\partial F}{\partial p}$ has P columns corresponding to the P dimensions of parameter space instead of just the one column devoted to the one dimensional parameter space in Eq. 3.28.

To solve Eq. 3.39, we use an optimization algorithm similar to [DH06] which was adapted from [Dob03]. This algorithm typically requires local convexity of the recovery boundary in order to converge, which does not hold on general. On the Modified IEEE 37-bus feeder of Section 4.1 the method converges rapidly, but this may not be true for other test cases, especially in the absence of local convexity. We begin each iteration ν with a unit vector ρ^{ν} in parameter space \mathbb{R}^{P} . Let $p = p_{0} + \gamma \rho^{\nu}$, where γ is a scalar and is initialized to zero. Then a continuation process is used to vary γ from zero until the recovery boundary is encountered at $p^{\nu} = p_{0} + \gamma^{\nu} \rho^{\nu}$. Let z^{ν} be the values of z obtained at p^{ν} . As p^{ν} lies on the recovery boundary, and hence is a critical parameter value, this implies that $F(z^{\nu}, p^{\nu}) =$ 0. The Lagrange multipliers λ^{ν} at each iteration are chosen to satisfy the first line of Eq. 3.39, so that $\left(\frac{\partial F(z^{\nu}, p^{\nu})}{\partial z}\right)^{\mathsf{T}} \lambda^{\nu} = 0$. Hence, λ^{ν} can be computed by performing a QR decomposition of $\frac{\partial F(z^{\nu}, p^{\nu})}{\partial z}$ and setting λ^{ν} equal to the last column of the orthogonal matrix in the decomposition. Then let $\rho^{\nu+1} = \left(\frac{\partial F(z^{\nu}, p^{\nu})}{\partial p}\right)^{\mathsf{T}} \lambda^{\nu}$ and normalize.

The iterations described above are repeated until ρ^{ν} converges, say to ρ^* . This implies that γ^{ν} , $p^{\nu} = p_0 + \gamma^{\nu} \rho^{\nu}$, z^{ν} , and λ^{ν} must also converge, say to γ^* , p^* , z^* , and λ^* , respectively. As p^* is a critical parameter value, $F(z^*, p^*) = 0$, so the last line of Eq. 3.39 is satisfied. Note that the method above for choosing λ^{ν} immediately implies that $\left(\frac{\partial F(z^*, p^*)}{\partial z}\right)^{\mathsf{T}} \lambda^* = 0$, so the first line of Eq. 3.39 is satisfied. Furthermore, multiplying λ by any constant implies this equation is still satisfied. The method of updating ρ implies that $\left(\frac{\partial F(z^*, p^*)}{\partial p}\right)^{\mathsf{T}} \lambda^* = c\rho^*$ where c is a nonzero constant. Choose $\hat{\lambda}^* = -\frac{\gamma^*}{c}\lambda^*$ and note that the first and third lines of Eq. 3.39 are still satisfied by $\hat{\lambda}^*$. However, the second line of Eq. 3.39 now reads:

$$(p^* - p_0) + \left(\frac{\partial F(z^*, p^*)}{\partial p}\right)^{\mathsf{T}} \hat{\lambda}^* = (p^* - p_0) - \frac{\gamma^*}{c} \left(\frac{\partial F(z^*, p^*)}{\partial p}\right)^{\mathsf{T}} \lambda^*$$
$$= (p^* - p_0) - \frac{\gamma^*}{c} (c\rho^*) = (p^* - p_0) - \gamma^* \rho^* = 0$$

since $p^* = p_0 + \gamma^* \rho^*$. Thus, convergence of the above method implies that Eq. 3.39 is satisfied, and at least a local solution of Eq. 3.38 has been obtained.

3.2.5 Choice of Norms for Measuring Distance

The minimization of Section 3.2.4 was performed using the norm $||p - p_0||_2^2$ in parameter space. However, more generally the optimization could be performed by minimizing the objective function $(p - p_0)^T A(p - p_0)$ where A is a positive semidefinite matrix. This would introduce only minor changes to the disucssion and derivation of Section 3.2.4, but could have a large impact on the obtained solutions. A natural choice for the weighting matrix A would be so as to reflect the uncertainty in measurements or estimation of system parameter values. In particular, using a Gaussian model for parameter uncertainty, the covariance matrix for the uncertainty would be a reasonable choice for the weighting matrix. Future work will consider this choice of weighting matrix.

3.2.6 Example: Three Synchronous Machines

The algorithms presented above were applied to a test case of three synchronous machines arranged in a loop. The machines are described by two-axis fourth-order synchronous machine models, with complete details given by equations (6.110)-(6.113) in [SP97]. All three machines have constant field voltage, constant mechanical torque, and mechanical damping $D\omega$ where D is a parameter implicitly representing the damping due to the governor. The parameters of the generators are provided in Table 3.1. The moments of inertia for the three generators are 0.077 pu, 0.105 pu, and 0.084 pu, respectively. The constant field voltages were chosen to ensure that the terminal bus voltages are all at 1.05 pu in steady state. The generator active powers were initialized with $P_1 = 0.4$ pu, $P_2 = 0.5$ pu, and $P_3 = -0.9$ pu. The line resistances are assumed to be zero, and the reactances are $X_{13} = 1.0000$ pu, $X_{12} = 1.2500$ pu and $X_{23} = 0.8333$ pu.

The disturbance considered is a fault applied at the terminals of generator one. The fault is modelled as a switched, constant shunt reactance with $X_{fault} = 0.001$ pu. The

Table 3.1: Synchronous machine parameters.

| r_d | 0 | x_d | 0.5880 | x'_d | 0.0913 | T'_{do} | 6.59 |
|-------|--------|--------|--------|-----------|--------|-----------|--------|
| x_q | 0.5880 | x'_q | 0.1 | T'_{qo} | 1.0 | D | 0.2653 |

main difference between the three generators in this example is their moments of inertia. Accordingly, the fault was located at generator one because it has the lowest inertia, and hence is most vulnerable to the fault. A fault duration time of 1.0 sec has been used throughout the example. Whilst this is an unrealistically long fault clearing time, it is a useful choice for illustrating algorithm characteristics.

The generator moments of inertia are parameters of interest in this system as they play an important role in recovery from the fault. We introduce a scaling factor which multiplies the moments of inertia of all three synchronous machines simultaneously. Lower values of the scaling factor lead to less inertia in the system, which reduces its ability to recover from faults.

The controlling UEP for this test case was identified through a two step process. First, a continuation method was used to identify four power flow solutions. Then, for each of those solutions, all possible choices of machine states were determined. Once those equilibria were known, it was straightforward to determine the controlling UEP by observing the trajectory as the system was driven to instability.

The first step is to drive the trajectory to spend a fixed length of time t_{ball} inside the ball around the controlling UEP. This is illustrated in Figure 3.2, which depicts the distance from a point on the trajectory to the controlling UEP as a function of time. Given a ball of radius r = 10.5, the moment of inertia scaling factor was determined using Eq. 3.27 such that the trajectory spent the target time $t_{ball} = 6.5$ sec inside the ball.

Next, the time t_{ball} in the ball was increased to drive the moment of inertia scaling factor to its critical value. This effectively moved the post-fault initial conditions onto the postfault RoA boundary. This was accomplished via the predictor-corrector continuation process presented in Section 3.2.2. Figure 3.3 shows the moment of inertia scaling factor as the time in the ball is increased. Note that the scaling factor converges to its critical value quite quickly as the time inside the ball increases, as suggested by the theory. This process yields a critical moment of inertia scaling factor whose associated post-fault initial conditions lie on the RoA boundary.

Now that a critical value of the parameter has been found, a pair of parameters can be treated as free variables, and their recovery boundary computed. For this test case, the mo-



Figure 3.2: The distance of the trajectory from the controlling unstable equilibrium point as a function of time. The radius of the ball around it is r = 10.5. The moment of inertia scaling factor was varied to drive the trajectory to spend a fixed time $t_{ball} = 6.5$ sec inside the ball.

ments of inertia of generators one and two were selected as the two free parameters. Using an analogous continuation method, as described in Section 3.2.3, the recovery boundary is numerically computed. Figure 3.4 depicts this boundary. For any parameter values above the boundary, the system will recover from the fault, returning to the original SEP. Conversely, for any parameter values below the boundary, the system will not recover from the specified disturbance. Although the recovery boundary turns out to be almost linear for this example, the continuation method is capable of tracing curves that have high curvature [FH16]. Further investigations are required to determine the reason for the apparent linearity in this example.

3.2.7 Conclusion

Algorithms are developed to find the unique closest point on the recovery boundary for one dimensional parameter space, trace the recovery boundary for two dimensional parameter space, and find the closest point on the recovery boundary to any initial recovery value in arbitrary dimensional parameter space. The main idea behind the approaches is to vary parameter values so as to maximize the time the system trajectory spends in a ball around the controlling UEP, thereby driving parameter values to the recovery boundary. Continuation methods are used to perform the maximizations and to trace the recovery boundary. The algorithms are illustrated on a network of three synchronous machines, and are applied to a large power system in Section 4.1.



Figure 3.3: The moment of inertia scaling factor is varied to maximize the time spent by the trajectory inside the ball around the controlling unstable equilibrium point. As this time increases, the moment of inertia scaling factor converges to its critical value.



Figure 3.4: The recovery boundary for the moments of inertia of generators one and two. All parameter values above the boundary lead to fault recovery, whereas all parameter values below the boundary lead to instability.

3.3 Maximizing Trajectory Sensitivities Algorithms

3.3.1 Introduction

The purpose of this section is to develop methods for determining the recovery boundary based on the theoretical motivation of Section 2.6. Unlike the techniques presented in Section 3.2, the algorithms here do not require prior knowledge of the controlling UEP, the determination of which is computationally intractable for realistic system models. Furthermore, unlike those in Section 3.2, the algorithms here do not require the assumption of transversal intersection of the system trajectory with a ball in state space, which was often violated in practice, and do not have to worry about solution degeneracy resulting from many intersections of the system trajectory with that ball, both of which led to challenges in practical implementations of those algorithms. The approach behind the algorithms here is that parameter values are varied so as to minimize the infimum over time of the inverse of the norm of the trajectory sensitivities, thereby driving parameter values onto the recovery value. Algorithms are presented for numerically computing the recovery boundary in two parameter dimensions, and for finding the nearest point on the recovery boundary from any initial set of recovery values in higher parameter dimensions. The algorithms are illustrated on the simple test case of two classical machines. The algorithms are then modified so as to exploit the linear structure discussed in Section 2.6.

3.3.2 One Dimensional Parameter Space

Let p be a one-dimensional real-valued parameter, let j be the index of the dynamic state corresponding to p, and let i be an index that runs through the set of dynamic states. We use the notation of Section 3.1 for trajectory sensitivities and related computations. Initially, it was observed through numerical experiments that minimizing inverse trajectory sensitivities instead of directly maximizing them led to faster convergence rates, and that root finding methods such as Newton-Raphson converged rapidly for these inverse sensitivities. More recently, it was shown (see Theorem 2.6.13) that the inverse trajectory sensitivities approach a dot product as parameter values approach the recovery boundary. This indicates that inverse sensitivities possess an approximately linear structure, which implies they can be rapidly solved by root finder methods such as Newton-Raphson. See Section 3.3.7 for further discussion. Define

$$H(t,p) = \frac{1}{\Phi_{ij}(t,p)\Phi_{ij}(t,p)}$$
$$G(p) = \inf_{t \ge 0} H(t,p).$$

Then by Theorem 2.6.4, p^* in the recovery boundary implies that it satisfies:

$$G(p^*) = 0. (3.40)$$

To find the critical parameter p^* , Eq. 3.40 will be solved using Newton-Raphson, which will require the derivative $\frac{dG(p)}{dp}$. To compute this, first note that

$$\frac{\partial H(t,p)}{\partial p} = -2 \frac{\Phi_{ijj} \Phi_{ij}}{\left(\Phi_{ij} \Phi_{ij}\right)^2} \tag{3.41}$$

where the dependence of the trajectory sensitivities on (t, p) is understood, but has been supressed in the notation for clarity of presentation. Let $\hat{t}(p) = \operatorname{argmin}_{t\geq 0} H(t, p)$. Then, by the proof of Lemma 2.6.15, if p is a recovery value then $\hat{t}(p)$ is finite so $G(p) = H(\hat{t}(p), p)$. Differentiating with respect to p yields

$$DG(p) = \frac{dG(p)}{dp} = \frac{\partial}{\partial p} H(\hat{t}(p), p)$$

$$= \frac{\partial H}{\partial t}(\hat{t}(p), p) \frac{d\hat{t}(p)}{dp}(p) + \frac{\partial H}{\partial p}(\hat{t}(p), p)$$

$$= \frac{\partial H}{\partial p}(\hat{t}(p), p)$$
(3.42)

where the final equality follows since $\frac{\partial H}{\partial t}(\hat{t}(p),p) = 0$ because $\hat{t}(p)$ is the time when H(t,p) achieves a maximum in time, and $\frac{\partial H}{\partial t} = 0$ at an extremum point. Therefore, Eq. 3.40 is solved iteratively by the following standard Newton-Raphson update, where p^{ν} denotes the parameter value of the current iteration, and $p^{\nu+1}$ the value of the next iteration:

$$p^{\nu+1} = p^{\nu} - DG(p^{\nu})^{-1}G(p^{\nu}).$$
(3.43)

As $DG(p) = \frac{\partial H}{\partial p}(\hat{t}(p), p)$ and $G(p) = H(\hat{t}(p), p)$, by the formulas for H and $\frac{\partial H}{\partial p}$, computation of G and DG requires only knowledge of trajectory sensitivities, which can be computed efficiently by the methods of Section 3.1. Then $\hat{t}(p)$ is observed from the output of the integration, and the values of G(p) and DG(p) can be computed. As the algorithm proceeds, $G(p^{\nu})$ approaches zero, causing p^{ν} to approach the recovery boundary.

3.3.3 Two Dimensional Parameter Space

Let p be a parameter that takes values in \mathbb{R}^2 , let j, k run through the two indeces of the two dynamic states corresponding to the parameter p, and let i be an index that runs through all the dynamic states. Define

$$H(t,p) = \frac{1}{\Phi_{ij}\Phi_{ij}} \tag{3.44}$$

$$G(p) = \inf_{t \ge 0} H(t, p).$$
(3.45)

By Theorem 2.6.4, the recovery boundary for p is given by $\{p : G(p) = 0\}$, which typically describes a curve in practice. However, it may be nonsmooth in general, and determining why it is typically smooth in practice is a subject of future work. Let $\epsilon > 0$ small. To establish consistent precision, we choose to approximate the recovery boundary with the curve $\{p : G(p) = \epsilon\}$. The goal is to use a continuation method to iteratively construct a sequence of points that traces this manifold. At each step a prediction is made by moving along the tangent to the curve. Then the predicted point is corrected by projecting back onto the curve. This process repeats until the curve is complete. Let p^{ν} denote the current point on the curve. The unit tangent vector η to the curve at p^{ν} is given by $\eta = DG(p^{\nu})^{\intercal}/||DG(p^{\nu})||_2$ where, by an analogous argument to the derivation of Eq. 3.42, we have

$$DG(p) = \frac{\partial H}{\partial p}(\hat{t}(p), p)$$
(3.46)

$$\frac{\partial H(t,p)}{\partial p_j} = -2 \frac{\Phi_{ijk} \Phi_{ik}}{\left(\Phi_{ik} \Phi_{ik}\right)^2}.$$
(3.47)

In the predictor step, for some choice of $\kappa > 0$ we let

$$p_{pred}^{\nu+1} = p^{\nu} + \kappa \eta$$

In the corrector step we solve

$$0 = (p_{pred}^{\nu+1} - p^{\nu})^{\mathsf{T}} (p^{\nu+1} - p_{pred}^{\nu+1})$$

which implies, after substituting for $p_{pred}^{\nu+1}$ above,

$$0 = \eta^T (p^{\nu+1} - p^{\nu}) - \kappa.$$
(3.48)

Hence, we obtain

$$F_{cont}(p^{\nu+1}) = \begin{bmatrix} G(p^{\nu+1}) - \epsilon \\ \eta^T(p^{\nu+1} - p^{\nu}) - \kappa. \end{bmatrix}$$

The corrector step must solve:

$$F_{cont}(p^{\nu+1}) = 0. (3.49)$$

To do so, we require the derivative DF which is given by

$$DF_{cont}(p^{\nu+1}) = \begin{bmatrix} DG(p^{\nu+1}) \\ \eta^T \end{bmatrix}.$$
(3.50)

We then solve Eq. 3.49 iteratively using Newton-Raphson, analogously to Eq. 3.43. This process of prediction and correction is then applied iteratively to generate a sequence of points along $\{p : G(p) = \epsilon\}$, thereby computing the recovery boundary to arbitrarily close precision.

3.3.4 Higher Dimensional Parameter Space: Newton's Method

Let p be a parameter that takes values in \mathbb{R}^{P} for $P \geq 1$. Let j, k, l run through the indeces of the P dynamic states corresponding to the parameter p, and let i be an index that runs through all the dynamic states. Define

$$H(t,p) = \frac{1}{\Phi_{il}\Phi_{il}}$$
$$G(p) = \inf_{t \ge 0} H(t,p) = H(\hat{t}(p),p)$$

where $\hat{t}(p) = \operatorname{argmin}_{t\geq 0} H(t, p)$. Fix any $p_0 \in \mathbb{R}^P$ a recover parameter value. As the dimension P is arbitrary, there may be many points in the recovery boundary that could be found by varying p. Therefore, we seek the point on the recovery boundary that is closest to p_0 , which represents the smallest change in parameter space that could lead to failure to recover from

the disturbance. So, we wish to solve

$$\min_{p \in \mathbb{R}^P} \frac{1}{2} (p - p_0)^{\mathsf{T}} A(p - p_0)$$
(3.51)

s.t.
$$G(p) - \epsilon = 0$$
 (3.52)

where A is a positive semidefinite weighting matrix (often set to the identity matrix) and $\epsilon > 0$ is small and will be required to ensure feasibility. The weighting matrix could be chosen as in Section 3.2.5. In general, the solution to this problem does not necessarily depend smoothly (or even continuously) with respect to p_0 . Nevetheless, when the technique for solving this problem, as described below, was applied to the simple test case of Section 3.3.6 it converged rapidly to the desired minimum. However, in general this problem is very difficult and the method offers no guarantees of convergence or that convergence will be to the global (or even a local) minimum. Further experimentation is required to assess its potential.

To solve Eqs. 3.51-3.52, we form the Lagrangian $\mathcal{L}(p,\lambda) = \frac{1}{2}(p-p_0)^{\mathsf{T}}A(p-p_0) + \lambda(G(p)-\epsilon)$ where λ is a Lagrange multiplier. Let $F(p,\lambda) := D\mathcal{L}(p,\lambda)$. Any stationary point of $\mathcal{L}(p,\lambda)$, such as a local minimum, must satisfy:

$$0 = D\mathcal{L}(p,\lambda) = \begin{bmatrix} A(p-p_0) + \lambda DG(p)^{\mathsf{T}} \\ G(p) - \epsilon \end{bmatrix} =: F(p,\lambda).$$
(3.53)

Eq. 3.53 is solved iteratively by the following standard Newton-Raphson update, where (p^{ν}, λ^{ν}) denote the corresponding values at the current iteration, and $(p^{\nu+1}, \lambda^{\nu+1})$ the values at the next iteration:

$$\begin{bmatrix} p^{\nu+1} \\ \lambda^{\nu+1} \end{bmatrix} = \begin{bmatrix} p^{\nu} \\ \lambda^{\nu} \end{bmatrix} - DF(p^{\nu}, \lambda^{\nu})^{-1}F(p^{\nu}, \lambda^{\nu})$$
(3.54)

where $F(p^{\nu}, \lambda^{\nu}) = D\mathcal{L}(p, \lambda)$ is given in Eq. 3.53 and DF is given by:

$$DF(p,\lambda) = \begin{bmatrix} A + \lambda D(DG(p)^{\intercal}) & DG(p)^{\intercal} \\ DG(p) & 0 \end{bmatrix}$$
(3.55)

Computation of DG(p) and $D(DG(p)^{\intercal})$ first requires several additional derivatives, which are obtained via repeated differentiation of H(t, p) with respect to parameter components and time:

$$\frac{\partial H}{\partial p_j}(t,p) = -2 \frac{\Phi_{ijl} \Phi_{il}}{\left(\Phi_{il} \Phi_{il}\right)^2} \tag{3.56}$$

$$\frac{\partial^2 H}{\partial p_j \partial p_k}(t, p) = -2H^2 \Big(\Phi_{ijkl} \Phi_{il} + \Phi_{ijl} \Phi_{ikl} \Big) + 8H^3 \Big(\Phi_{ikl} \Phi_{il} \Big) \Big(\Phi_{ijl} \Phi_{il} \Big)$$
(3.57)

$$\frac{\partial^2 H}{\partial p_j \partial t}(t, p) = -2H^2 \left(\dot{\Phi}_{ijl} \Phi_{il} + \Phi_{ijl} \dot{\Phi}_{il} \right) + 8H^3 \left(\dot{\Phi}_{il} \Phi_{il} \right) \left(\Phi_{ijl} \Phi_{il} \right) \frac{\partial^2 H}{\partial t}(t, p) = 8H^3 \left(\dot{\Phi}_{il} \Phi_{il} \right)^2$$
(3.58)

$$\frac{\partial^2 H}{\partial t^2}(t,p) = 8H^3 \left(\dot{\Phi}_{il} \Phi_{il}\right)^2 - 2H^2 \left(\Phi_{il} \ddot{\Phi}_{il} + \dot{\Phi}_{il} \dot{\Phi}_{il}\right)$$
(3.59)

where H = H(t, p) and $\dot{\Phi}_{il}$, $\dot{\Phi}_{ijl}$ are given by Eqs. 3.4-3.5 in Section 3.1.3, and $\ddot{\Phi}_{il}$ is given by Eq. 3.7 of that section. Next we compute the derivatives DG(p) and $D(DG(p)^{\dagger})$. First, note that $(DG(p))_j = \frac{\partial G(p)}{\partial p_j}$. By an analogous argument as in the derivation of Eq. 3.42 for the one dimensional parameter case in Section 3.3.2, $(DG(p))_j = \frac{\partial H}{\partial p_j}(\hat{t}(p), p)$. Next, differentiating this equation with respect to p_k gives

$$(D(DG(p)^{\mathsf{T}}))_{jk} = \frac{\partial^2 G(p)}{\partial p_k \partial p_j}$$

= $\frac{\partial}{\partial p_k} (DG(p)^{\mathsf{T}})_j = \frac{\partial}{\partial p_k} \frac{\partial H}{\partial p_j} (\hat{t}(p), p)$
= $\frac{\partial^2 H}{\partial p_j \partial t} (\hat{t}(p), p) \frac{\partial \hat{t}(p)}{\partial p_k} + \frac{\partial^2 H}{\partial p_k \partial p_j} (\hat{t}(p), p).$

Note that computation of $D(DG(p)^{\intercal})$ requires $\frac{\partial \hat{t}(p)}{\partial p_k}$. We derive this as follows. Since $\hat{p} = \operatorname{argmin}_{t \geq 0} H(t, p), \ \frac{\partial}{\partial t} H(\hat{t}(p), p) = 0$. Hence, as this function of p is identically zero,

differentiating with respect to p_k gives:

$$\begin{split} 0 &= \frac{\partial}{\partial p_k} \frac{\partial}{\partial t} H(\hat{t}(p), p) = \frac{\partial}{\partial t} \frac{\partial}{\partial p_k} H(\hat{t}(p), p) \\ &= \frac{\partial}{\partial t} \left(\frac{\partial H}{\partial t}(\hat{t}(p), p) \frac{\partial \hat{t}(p)}{\partial p_k} + \frac{\partial H}{\partial p_k}(\hat{t}(p), p) \right) \\ &= \frac{\partial^2 H}{\partial t^2}(\hat{t}(p), p) \frac{\partial \hat{t}(p)}{\partial p_k} + \frac{\partial^2 H}{\partial p_k \partial t}(\hat{t}(p), p). \end{split}$$

Hence, solving for $\frac{\partial \hat{t}(p)}{\partial p_k}$ yields

$$\frac{\partial \hat{t}(p)}{\partial p_k} = -\frac{\partial^2 H}{\partial p_k \partial t}(\hat{t}(p), p) \left(\frac{\partial^2 H}{\partial t^2}(\hat{t}(p), p)\right)^{-1}.$$

Substituting this back into the expression for $D(DG(p)^{\intercal})_{jk}$, we obtain

$$(DG(p))_{j} = \frac{\partial H}{\partial p_{j}}(\hat{t}(p), p)$$
$$(D(DG(p)^{\intercal}))_{jk} = \frac{\partial^{2} H}{\partial p_{k} \partial p_{j}}(\hat{t}(p), p)$$
$$- \frac{\partial^{2} H}{\partial p_{j} \partial t}(\hat{t}(p), p) \frac{\partial^{2} H}{\partial p_{k} \partial t}(\hat{t}(p), p) \left(\frac{\partial^{2} H}{\partial t^{2}}(\hat{t}(p), p)\right)^{-1}$$

which can be computed from Eqs. 3.56-3.59. In turn, the expressions for DG(p) and $D(DG(p)^{\intercal})$ given here can be used to compute F and DF in Eq. 3.55 and Eq. 3.53, respectively. Finally, F and DF are used to perform the Newton-Raphson updates of Eq. 3.54, which drive p^{ν} towards one of the (possibly many locally) closest points on the recovery boundary to p_0 .

3.3.5 Higher Dimensional Parameter Space: Alternate Method

Consider the setting of Section 3.3.4. The goal is to solve Eq. 3.53 to obtain the (possibly one of many) closest point on the recovery boundary to some initial recovery value p_0 . We present an alternative solution strategy to that of Section 3.3.4 which does not require third order trajectory sensitivities and is similar to that of Section 3.2.4. As remarked in Section 3.2.4, this method requires local convexity of the recovery boundary to guarantee convergence, but the recovery boundary is not in general convex. So, further testing is required to determine the practicality of this technique.

| P_2 | P_3 | V_2 | V_3 | D_2 | D_3 | H_2 | H_3 | X_{12} | X_{23} |
|-------|-------|-------|-------|-------|-------|-------|-------|----------|----------|
| 1 | 0.5 | 1 | 0.9 | 0.3 | 0.2 | 0.3 | 0.2 | 0.4 | 0.5 |

 Table 3.2:
 Test Case Parameter Values

Define an iterative algorithm as follows. Let ν denote the current iteration. Given a unit vector ρ^{ν} in \mathbb{R}^{P} , find γ^{ν} such that $G(p_{0} + A^{-1}\gamma^{\nu}\rho^{\nu}) = \epsilon$. Then set $p^{\nu} = p_{0} + A^{-1}\gamma^{\nu}\rho^{\nu}$. This can be accomplished by using the one dimensional method of Section 3.3.2 to solve for γ^{ν} using Newton-Raphson and noting that

$$\Phi_{i\gamma} = \Phi_{ij}\rho_j^{\nu}$$

$$\Phi_{i\gamma\gamma} = \Phi_{ijk}\rho_j^{\nu}\rho_k^{\nu}$$
(3.60)

where $\Phi_{i\gamma} = \frac{d\phi_i}{d\gamma}$ and $\Phi_{i\gamma\gamma} = \frac{d^2\phi_i}{d\gamma^2}$. Once γ^{ν} , and hence p^{ν} , is obtained, set $\rho^{\nu+1} = DG(p^{\nu})^T/||DG(p^{\nu})||_2$ where $DG = \frac{\partial H}{\partial p}(\hat{t}(p), p)$ and $\frac{\partial H}{\partial p}$ is given by substituting Eq. 3.60 into Eq. 3.41 of Section 3.3.2.

The process described above is then repeated iteratively until ρ^{ν} converges, say to ρ^* . Then γ^{ν} and p^{ν} also converge, say to γ^* and p^* , respectively. Let $\lambda = -\gamma^*/||DG(p^*)||_2$. By the method for computing γ^{ν} , we must have that $G(p^*) = G(p_0 + A^{-1}\gamma^*p^*) = \epsilon$ and $p^* = p_0 + A^{-1}\gamma^*\rho^*$. Furthermore, by the method for determining $\rho^{\nu+1}$, we have that $\rho^* = DG(p^*)^T/||DG(p^*)||_2$. Hence,

$$A(p^* - p_0) + \lambda DG(p^*)^T = A(p^* - p_0) + (-\gamma^* / ||DG(p^*)||_2)(||DG(p^*)||_2\rho^*)$$

= $A(p^* - p_0) - \gamma^* \rho^*$
= $A(p^* - p_0) - A(p^* - p_0)$
= 0.

Hence, convergence of the algorithm implies that Eq. 3.53 is satisfied, so a local minimum of Eqs. 3.51-3.52 has been found.

3.3.6 Example: Two Classical Machines

The test case considered is a simple model of a power system which consists of two synchronous generators connected to two buses arranged radially, with an infinite bus at the head of the feeder, as shown in Fig. 3.5. The infinite bus is held at constant voltage



Figure 3.5: Power system model used as the test case in Section 2.3.3. An infinite bus, which is held at constant voltage and can draw arbitrarily high power, is connected to two synchronous generators.

of 1 p.u. (per unit) and can draw arbitrarily high power to match the power generation of the two generators. Let the infinite bus be connected to bus one, generator 2 to bus 2, and generator 3 to bus 3. Let δ_2 and ω_2 be the rotor angle and its angular velocity for generator 2, and define δ_3 and ω_3 analogously for generator 3. Let $y = [\delta_2 \ \omega_2 \ \delta_3 \ \omega_3]^{\mathsf{T}}$ be the vector of state variables. Let P_2 , D_2 , H_2 , and V_2 be the active power, damping coefficient, moment of inertia, and voltage magnitude at generator 2, and define P_3 , D_3 , H_3 , and V_3 analogously for generator 3. Let X_{ij} denote the impedance on the line from bus *i* to bus *j*. Parameter values used are given in Table 3.2. Let $p = [P_2 \ P_3 \ V_2 \ V_3 \ D_2 \ D_3 \ H_2 \ H_3]^{\mathsf{T}}$ be the vector of chosen parameters, and let $x = [y^{\mathsf{T}} \ p^{\mathsf{T}}]^{\mathsf{T}}$. Then the dynamics of this system are given by:

$$\begin{aligned} x_1 &= x_2 \\ H_2 \dot{x}_2 &= P_2 - D_2 x_2 - \sin(x_1) \frac{V_2}{X_{12}} - (\sin(x_1)\cos(x_3) - \cos(x_1)\sin(x_3)) \frac{V_2 V_3}{X_{23}} \\ \dot{x}_3 &= x_4 \\ H_3 \dot{x}_4 &= P_3 - D_3 x_4 - (\sin(x_3)\cos(x_1) - \cos(x_3)\sin(x_1)) \frac{V_2 V_3}{X_{23}}, \end{aligned}$$

which can be written more succinctly as

$$\dot{x} = V(x) \tag{3.61}$$

Let p_0 be the parameter values given in Table 3.2. There exists a stable equilibrium point of Eq. 3.61 of $y^0(p_0) = [0.6435 \ 0 \ 0.9250 \ 0]^{\intercal}$. For p reasonably near p_0 there exists a stable equilibrium point $y^0(p)$ of Eq. 3.61 close to $y^0(p_0)$ which can be found, for example, by solving $V(y^0(p)) = 0$ using Newton-Raphson with initial condition p_0 . After the system begins at the stable equilibrium point $x^0(p)$, a fault is applied at the head of the feeder, lasts for a



Figure 3.6: For the test case of Section 2.3.3, $x_3 = \delta_3$ is shown as a function of time for (a) the initial value $P_3 = 0.5$ (blue line) and (b) the critical value $P_3 = 0.7309$ (red line).

duration of 0.8 s, and then is cleared. The fault is modeled by setting the term $-\sin(x_1)\frac{V_2}{X_{12}}$ to zero in equation V_2 , leaving the other components of V unchanged, and integrating this modified system for 0.8 s. Before and after the fault, the system dynamics are given by Eq. 3.61.

Fig. 3.6 shows $x_3 = \delta_3$ as a function of time for $p = p_0$. Note that the system is initially at a stable equilibrium point. Then, a fault is applied, and the angle δ_3 deviates far from equilibrium. After 0.8 s the fault is cleared, normal system dynamics are restored, and the system evolves until δ_3 returns to its prior equilibrium value. This picture is typical for cases in which the system is able to recover from the fault.

For each $p_i \in p$, the algorithm detailed in Section 3.3.2 was applied to find the nearest point on the recovery boundary of p_i . There were some parameters, namely H_2 , H_3 , and D_3 , where varying just one of them was not sufficient to make the system fail to recover from the fault (for positive values of the parameters, as is required physically). In these cases, the algorithm attempted to send the parameter values towards zero or negative, so it was quickly clear that they were unable to make the system fail to recover.

Among the remaining parameters, $p = P_3$ is representative of the observed behavior of the algorithm. Fig. 3.7 shows the convergence of the one-dimensional parameter space algorithm for $p = P_3$. Although the critical value of P_3 is about 50% larger than its initial value, convergence to its critical value is rapid and monotonic. At the final iterate, $G(P_3) < 10^{-5}$. More generally, for each $p_i \in p$ (other than those mentioned above which could not drive the system to non-recovery), rapid, monotonic convergence to the corresponding critical value was observed.



Figure 3.7: Computation of critical parameter value for P_3 using the one-dimensional parameter space algorithm of Section 3.3.2. Iterations begin in the top left and proceed towards the bottom right. Rapid convergence to the critical parameter value is observed.

Several choices of multi-dimensional parameter spaces were tested. The weighting matrix A was set to the identity matrix in all cases. The initial parameter values p_0 were taken from Table 3.2. First, the choice $p = p_i$ for each i was made to confirm that the higher dimensional parameter space algorithm of Section 3.3.4 gave the same critical parameter values as the one-dimensional parameter space algorithm of Section 3.3.2. Next, the following three sets of parameters were considered:

$$P_4 := \{P_2, P_3, V_2, V_3\}$$
$$P_6 := \{P_2, P_3, V_2, V_3, D_2, D_3\}$$
$$P_8 := \{P_2, P_3, V_2, V_3, D_2, D_3, H_2, H_3\}$$

Note that P_4 consists of all the parameter values that power system operators can select during real-time operation, P_6 consists of P_4 together with the parameter values that can be tuned by control engineers (namely, the damping that is a result of controller design), and P_8 consists of P_6 together with the remaining generator parameter values.

We chose $\epsilon = 10^{-5}$ for the constraint of Eq. 3.52. Then, the higher dimensional parameter space algorithm of Section 3.2.4 converged for P_4 in 9 iterations, for P_6 in 9 iterations, and for P_8 in 8-iterations. Let p_4^* , p_6^* , and p_8^* denote the critical parameter values obtained via

| Parameter Set | Objective Function Value |
|---------------|--------------------------|
| P_4 | 0.0075 |
| P_6 | 0.0062 |
| P_8 | 0.0053 |

Table 3.3: Values of the objective function of (3.51) for the critical parameter values obtained by the higher dimensional optimization algorithm for the parameter sets P_4 , P_6 , and P_8 .

this algorithm for P_4 , P_6 , and P_8 , respectively. Then

$$p_4^* = \begin{bmatrix} 1.0657 & 0.5651 & 0.9198 & 0.8986 \end{bmatrix}^{\mathsf{T}}$$
$$p_6^*(1:3) = \begin{bmatrix} 1.0555 & 0.5549 & 0.9352 \end{bmatrix}^{\mathsf{T}}$$
$$p_6^*(4:6) = \begin{bmatrix} 0.8991 & 0.2618 & 0.1736 \end{bmatrix}^{\mathsf{T}}$$
$$p_8^*(1:4) = \begin{bmatrix} 1.0466 & 0.5462 & 0.9489 & 0.8992 \end{bmatrix}^{\mathsf{T}}$$
$$p_8^*(5:8) = \begin{bmatrix} 0.2648 & 0.1751 & 0.2767 & 0.1647 \end{bmatrix}^{\mathsf{T}}$$

The parameter values satisfy $G(p_4^*) = G(p_6^*) = G(p_8^*) = \epsilon$, so they all lie on their respective recovery boundarys.

The goal was to obtain parameter values on the recovery boundary which minimized the objective function of Eq. 3.51: $\frac{1}{2}(p-p_0)^{\intercal}A(p-p_0)$. Table 3.3 shows the values of the objective functions at p_4^* , p_6^* and p_8^* . As additional parameters are introduced, more options for values on the recovery boundary become available, some of which may be closer to p_0 than the previous options. Consequently, the objective function value, which measures the distance of the chosen value on the recovery boundary from p_0 , decreased from P_4 to P_6 , and again from P_6 to P_8 . This suggests that the result of the higher dimensional parameter space algorithm for all of the cases above is indeed the value on the recovery boundary at a minimum distance from p_0 , as desired. Overall, for all the combinations of parameters chosen above, the higher dimensional parameter space algorithm converged rapidly to a value on the recovery boundary which appears to be the closest such value to p_0 in each parameter space.

3.3.7 Algorithm Adjustments Based on Linearization Theory

Empirical observations [NPH02b] and analysis for parameter-independent linear vector fields [RH12] indicated that, in the case of one dimensional parameter space, the infimum over time of the inverse trajectory sensitivities is approximately linear in parameter value. Section 2.6 has extended the analysis to parameter-dependent nonlinear vector fields with higher dimensional parameter spaces. In particular, Theorem 2.6.13 shows that the infimum of the one norm of inverse trajectory sensitivities approaches a dot product as parameter values approach the recovery boundary. Motivated by this theory, the algorithms of this section, which were developed using the two norm (Euclidean norm) above, are modified for use with the one norm instead. By doing so, the algorithms are then able to exploit the linear structure revealed by the theory.

3.3.7.1 One Dimensional Parameter Space

We preserve the notational conventions used previously in Section 3.3. Let p be a onedimensional real-valued parameter, let j be the index of the dynamic state corresponding to p, and let i be an index that runs through the set of dynamic states. Motivated by Theorem 2.6.13 of Section 2.6, we use inverse trajectory sensitivities with one norms to obtain a function that is approximately linear. Consequently, we solve for G(p) = 0 using Newton-Raphson because of its ability to rapidly solve linear and approximately linear systems. Define

$$H(t,p) = \frac{1}{|\Phi_{ij}(t,p)|}$$
$$G(p) = \inf_{t \ge 0} H(t,p).$$

The goal is to solve Eq. 3.40: $G(p^*) = 0$. This requires the derivative

$$\frac{\partial H(t,p)}{\partial p} = -H^2(t,p)\Phi_{ijj}\operatorname{sign}(\Phi_{ij})$$

where for any vector v, $(\operatorname{sign}(v))_i = 1$ if $v_i \ge 0$ and $(\operatorname{sign}(v))_i = -1$ if $v_i < 0$ for all components *i*. Then, as in Section 3.3.2, we have

$$DG(p) = \frac{\partial H}{\partial p}(\hat{t}(p), p)$$

and solve Eq. 3.40 iteratively using Newton-Raphson. Newton-Raphson is used because it efficiently solves linear or approximately linear systems, and G(p) approaches a dot product as $p \to p^*$.

3.3.7.2 Two Dimensional Parameter Space

Let p be a parameter that takes values in \mathbb{R}^2 , let j, k run through the two indeces of the two dynamic states corresponding to the parameter p, and let i be an index that runs through all the dynamic states. Define

$$H(t,p) = \frac{1}{|\Phi_{ij}(t,p)|}$$
$$G(p) = \inf_{t \ge 0} H(t,p)$$

as above (note that now j runs through two values, not just one as in Section 3.3.7.1). We require the derivative

$$\frac{\partial H(t,p)}{\partial p_k} = -H^2(t,p)\Phi_{ijk}\mathrm{sign}(\Phi_{ij}).$$

Then

$$DG(p) = \frac{\partial H}{\partial p}(\hat{p}, p)$$

has two components corresponding to the derivatives with respect to p_1 and p_2 . A predictorcorrector continuation method can then be formulated analogously to that in Section 3.3.3.

3.3.7.3 Higher Dimensional Parameter Space

Let p be a parameter that takes values in \mathbb{R}^P for $P \ge 1$. Let j, k run through the indeces of the P dynamic states corresponding to the parameter p, and let i be an index that runs through all the dynamic states. Define

$$H(t,p) = \frac{1}{|\Phi_{ij}(t,p)|}$$
$$G(p) = \inf_{t \ge 0} H(t,p) = H(\hat{t}(p),p)$$

as above (note that now j runs through P values). We will require the derivatives

$$\frac{\partial H(t,p)}{\partial p_k} = -H^2(t,p)\Phi_{ijk}\operatorname{sign}(\Phi_{ij})$$
$$DG(p) = \frac{\partial H}{\partial p}(\hat{p},p).$$

As in Section 3.3.4, for fixed $\epsilon > 0$ small and A positive semidefinite (where the weighting matrix A could be chosen as in Section 3.2.5), we wish to solve

$$\min_{p \in \mathbb{R}^P} \frac{1}{2} (p - p_0)^{\mathsf{T}} A(p - p_0)$$
(3.62)

s.t.
$$G(p) - \epsilon = 0.$$
 (3.63)

Motivated by the linearity theory of Section 2.6 (especially Theorem 2.6.13) we approximate $G(p) - \epsilon$ by its linearization and solve a succession of quadratic programs. In particular, at each iteration ν we replace the nonlinear constraint $G(p^{\nu+1}) - \epsilon = 0$ with the corresponding linear constraint $DG(p^{\nu})(p^{\nu+1} - p^{\nu}) + G(p^{\nu}) - \epsilon = 0$. This gives the quadratic program

$$\min_{p^{\nu+1} \in \mathbb{R}^P} \frac{1}{2} (p^{\nu+1} - p_0)^{\mathsf{T}} A (p^{\nu+1} - p_0)$$
(3.64)

s.t.
$$DG(p^{\nu})(p^{\nu+1}-p^{\nu})+G(p^{\nu})-\epsilon=0.$$
 (3.65)

The solution is given by solving the following linear system

$$\begin{bmatrix} A & DG(p^{\nu})^T \\ DG(p^{\nu}) & 0 \end{bmatrix} \begin{bmatrix} p^{\nu+1} \\ \lambda^{\nu+1} \end{bmatrix} = \begin{bmatrix} Ap_0 \\ DG(p^{\nu})p^{\nu} - G(p^{\nu}) + \epsilon \end{bmatrix}$$

where $\lambda^{\nu+1}$ is a Lagrange multiplier but is not needed by the algorithm. If $p^{\nu+1}$ lies on the other side of the recovery boundary, which is determined as a byproduct of the simulation required to compute $G(p^{\nu+1})$ and $DG(p^{\nu+1})$, then we still move in the direction $p^{\nu+1}-p^{\nu}$ but only half the distance as before. This distance is repeatedly halved until a value for $p^{\nu+1}$ is found such that $p^{\nu+1}$ is a recovery value. This process is then repeated iteratively until the algorithm converges. As the solution to Eqs. 3.64-3.65 (in which G(p) is replaced by a linear constraint) exists and is unique, and since G(p) approaches a linear function as p approaches the recovery boundary by Theorem 2.6.13, p_0 sufficiently close to the recovery boundary implies the existence of a unique solution to Eqs. 3.62-3.63. So, unlike the methods of

Section 3.2.4, Section 3.3.4, and Section 3.3.5, the techniques of this section have local guarantees (for p_0 sufficiently close to the recovery boundary) of existence and uniqueness of a solution, and of converging to that solution rapidly, due to the linear structure discussed in Section 2.6.

3.3.7.4 Implementation Concern: Divergence of Runtimes

From an implementation perspective, it is important that the runtimes for algorithms which compute critical parameter values do not diverge rapidly to infinity as the critical parameter values are approached. Otherwise, computing these critical parameter values to desired accuracy quickly becomes infeasible as the accuracy is increased due to unreasonably high runtimes. Several results from Chapter II suggest that rapid divergence of runtimes as critical parameter values are approached may be a serious concern. Theorem 2.5.9 shows that as a critical parameter value is approached, the time to escape a ball of fixed radius around the controlling critical element diverges to infinity. By the proof of Lemma 2.6.18, for a parameter-dependent linear system, as a critical parameter value is approached, the time to escape to a fixed distance in the unstable direction is approximately logarithmic in the inverse of the distance to the critical parameter value. In particular, this implies that it diverges to infinity at a logarithmic rate as the parameter approaches its critical value, and one would expect (although it remains to be proven) similar behavior for the nonlinear system as well.

Fortunately, Theorem 2.6.13 shows that as any critical parameter value is approached, the norm of the trajectory sensitivities is approximately linear in the inverse of the distance to the critical parameter value. Thus, as discussed in Remark 2.6.19, as parameters approach a critical value the norm of the trajectory sensitivities is expected to grow exponentially faster than the escape time. Since the algorithms of this section proceed by maximizing the norm of the trajectory sensitivities (or, equivalently, minimizing their inverse), critical parameter values are computed to sufficient accuracy for these algorithms once the trajectory sensitivity norms are sufficiently large. The above then implies that, as parameters approach a critical value, it is expected that the norm of the trajectory sensitivity will converge to a sufficiently large size exponentially faster than the rate at which the escape times diverge towards infinity. From the proof of Theorem 2.6.13 it is clear that as parameters approach a critical value, the runtime is dominated by the time the system spends in a neighborhood of the controlling critical element: the escape time. Consequently, it is expected, and numerical experiments confirm, that even for high levels of accuracy in computing critical parameter

values, runtimes do not rapidly diverge towards infinity, and practical implementation of the algorithms remains feasible.

3.3.8 Conclusion

Algorithms are developed to find the unique closest point on the recovery boundary in one parameter dimension, compute the recovery boundary in two parameter dimensions, and find the nearest point on the recovery boundary to any initial set of recover parameter values in arbitrary parameter dimensions. Unlike the algorithms of Section 3.2, the algorithms presented here do not require prior knowledge of the controlling UEP. The algorithms proceed by varying parameter values so as to minimize the infimum over time of the norm of the inverse trajectory sensitivities. They are illustrated on the test case of two classical machines, and are applied to a larger power system in Section 4.2. Recently, the algorithms have been updated to exploit the linear structure discussed in Section 2.6.

CHAPTER IV

Application to Power Systems

The algorithms developed in Chapter III are applied in this chapter to assess fault vulnerability in power systems. There are two main power system test cases of interest. The first, as discussed in Section 4.1, takes the IEEE 37-bus feeder, which is a model power distribution system, and modifies it by replacing 60% of the background load with induction motors. The purpose of this test case is a first step towards the study of the onset of Fault Induced Delayed Voltage Recovery (FIDVR), which is a phenomenon, likely driven by induction motors in residential air conditioners, in which voltages are slow or fail to recover in distribution grids following a fault. Results are presented for analyzing the onset of induction motor stalling. The second test case, as discussed in Section 4.2, is the IEEE 39-bus power system, which is a network of synchronous machines and includes the nonsmooth controller limits on the Active Voltage Regulators (AVRs) and Power System Stabilizers (PSSs). The purpose of this test case is to analyze the stability of a network of synchronous machines with uncertain background load and load dynamics. Results reveal an unexpected negative influence of constant impedance load on system recovery that would not have been identified otherwise.

4.1 Modified IEEE 37-Bus Feeder

4.1.1 Introduction

Distribution networks that serve large numbers of air conditioners are vulnerable to Fault Induced Delayed Voltage Recovery (FIDVR), [NER09]. FIDVR events are typically initiated by a fault near a substation, triggering a temporary drop in voltage that affects the associated distribution grid. When the local voltage drops, the induction motors driving air-conditioner compressors are affected; the electrical torque drops without a corresponding decrease in mechanical torque, causing the induction motor to decelerate. Induction motors draw more current when they reaccelerate, leading to a further reduction in local voltage. As a result, restoration of pre-fault voltages can be significantly delayed. Furthermore, if the fault is not cleared quickly enough, some motors will stall and the system may not be able to recover to pre-fault conditions.

Although the root cause for FIDVR events is well understood in a general sense [NER09], rigorous analysis of the instability mechanisms is still required. Whether induction motors stall or recover depends on system parameters, such as motor moments of inertia and fault clearing times. We consider a fixed disturbance, such as a three-phase fault on a particular line, and a particular set of parameters that affect induction motor stalling. The algorithms of Section 3.2 are applied to determine the set of recovery values for which all induction motors recover to pre-fault conditions. This provides a first step towards understanding the parameter dependence of cascading induction motor stalling during FIDVR events by characterizing the onset of induction motor stalling in a distribution network.

4.1.2 Numerical Experiments

The algorithms presented above were applied to a modified version of the IEEE 37-bus test feeder shown in Figure 4.1. The dimension of the dynamic states for this system is 100. In order to induce stressed conditions, the total load on the system was doubled. Then 60% of the load at each bus was replaced by an induction motor with the equivalent real power, leaving 40% background load. This is consistent with the peak air-conditioning load penetration of Southern California Edison inland load, [NER09]. The moments of inertia of the induction motors were chosen to be proportional to their real power, with a common proportionality constant used for all motors. This proportionality constant will be refer to as the moment of inertia scaling factor, SF, and forms one of the parameters of interest in the following studies, along with the fault duration and the voltage at the infinite bus during the fault.

We consider a nominal steady state in which no motors are stalled, apply a fault on the infinite bus at the head of the feeder, and then clear the fault after a specified duration. Stressed conditions led to cascaded stalling indicative of FIDVR events. Also note that the test cases below share a common controlling UEP, and that they appear to satisfy all the technical assumptions discussed in Chapter II.

A full 3-phase model could be used for the algorithm, but results shown here accurately



Figure 4.1: IEEE 37-bus test feeder.

capture the qualitative tradeoffs inherent in distribution grids with high penetration of AC load.

4.1.3 Results

Each of the algorithms presented above was applied to the modified 37-bus IEEE test feeder. First, Figure 4.2 depicts the (Euclidean) distance from the trajectory to the controlling UEP over time. The dashed line indicates a fixed distance of r = 190 from the UEP. Using the Newton-Raphson algorithm described earlier, the moment of inertia scaling factor SF was varied to ensure that the trajectory spent a fixed amount of time $t_{ball} = 0.2$ sec inside the specified ball. Close inspection of the figure shows that the intersections of the trajectory with the ball occur exactly 0.2 sec apart.

The next step was to run a continuation process, by treating t_{ball} as a second free parameter, in order to reach a parameter value that was arbitrarily close to the recovery boundary. Figure 4.3 illustrates this continuation process. Here, t_{ball} was increased from 0.2 sec inside the ball to 1.5 sec, leading to a decrease in SF of just over 0.02. When the slope of the curve approaches zero it implies that further increasing t_{ball} will have negligible effect on the value of the free parameter SF. Practically, this implies that SF is near its critical value SF^* , indicating that the algorithm has succeeded in approaching arbitrarily closely to the recovery boundary.

After finding a point arbitrarily close to the recovery boundary, t_{ball} is fixed and a pair of parameters can be freed, often resulting in a 1-manifold in parameter space. Figure 4.4



Figure 4.2: Distance from the trajectory to the unstable equilibrium point as a function of time, for a trajectory that has been driven inside a ball of radius 190 for a time $t_{ball} = 0.2$ s. The radius 190 is depicted by the dotted line.

depicts the recovery boundary for bus 740 motor moment of inertia and bus 741 motor moment of inertia, the two motors at the end of the feeder. Points above the curve correspond to parameter sets that lead to recovery of all motors after the fault clears while points below the curve are parameter sets that cause one or more motors to stall after the fault.

In contrast, Figure 4.5 shows close to the recovery boundary for bus 740 motor moment of inertia and bus 725 motor moment of inertia. Note that bus 740 is at the end of the feeder whereas bus 725 is on another branch of the feeder with a tolerance of less than 5e - 4 from the recovery boundary in their parameter space.

Comparison of the figures illustrates that, for a fixed bus 740 motor moment of inertia, a much larger increase in motor moment of inertia is required for the distant motor at bus 725 than for the near motor at bus 741 to lead to recover of all motors after the fault clears. So, in the parameter space of the pairs of motor moments of inertia, the near motor has a much stronger influence on motor recovery than the one further away. This is representative of typical behavior observed in the network, where stalling cascades from the ends of the feeder upstream towards the substation.

Finally, we considered an arbitrary number P of free parameters p, specified values p_0 , and seek the point \hat{p} on the recovery boundary that is closest to p_0 . For this example, the free parameters used were fault duration time and voltage at the infinite bus during the fault. The specified values p_0 are 0.35 sec and 0.55 pu, respectively. Figure 4.6 shows the iterations of the optimization algorithm. Note that convergence was achieved in just three iterations, and that the points generated at each iteration lie on the recovery boundary in



Figure 4.3: Time t_{ball} spent within a fixed radius of the unstable equilibrium point versus moment of inertia scaling factor SF.



Figure 4.4: Recovery boundary for the moments of inertia at buses 740 and 741.

the directions successively estimated by the algorithm.

We also considered a case where the background active and reactive power load at every bus was freed, a total of 50 free parameters. This gave the smallest change in background load that would lead the system to be arbitrarily close to the recovery boundary. The optimization algorithm in this case converged in two iterations.

Explicit computation of recovery boundarys and knowledge of the point on the border of minimum distance from a given initial parameter set can be used to better understand the dependence of FIDVR events on parameters, and motivate approaches for reducing the likelihood and severity of these events in practice.



Figure 4.5: Recovery boundary for the moments of inertia at buses 740 and 725.



Figure 4.6: Optimization algorithm to find the nearest point on the recovery boundary. The specified value p_0 , first iteration, second iteration, and third iteration are given by the black star, green plus, cyan circle, and red star, respectively. The black dashed line connects p_0 with the final solution, and is orthogonal to the recovery boundary.

4.1.4 Conclusion

The algorithms of Section 3.2 are applied to analyze the onset of induction motor stalling in a distribution network. The recovery boundary was computed in term of pairs of moments of inertia of various motors. The optimization algorithm was used to compute the nearest point on the recovery boundary in the parameter space of fault duration time and fault voltage dip. The optimization algorithm was also applied to find the smallest change in background load that resulted in instability.

4.2 IEEE 39-Bus New England Power System

4.2.1 Introduction

The algorithms of Section 3.3 are applied to the IEEE 39-bus New England power system. This is a 39-bus network containing 10 synchronous machines, and subject to a three-phase fault at the terminals of one of the generators. The purpose of this test case is to demonstrates the effectiveness of the algorithms on a more realistic network of synchronous machines with fourth order machine models, and including the nonsmooth PSS and AVR controller limits. Whereas the algorithms of Section 3.2 have already been demonstrated on a system of induction motors in Section 4.2 which exhibit dynamics similar to a gradient system, synchronous machines behave more like Hamiltonian systems. The algorithms of interest work best for systems which exhibit strong damping and minimal oscillations. As induction motors contribute real eigenvalues to their linearization and resemble a gradient system, they represent an easier test case for these algorithms. In contrast, a network of synchronous machines is typically weakly damped and possesses complex eigenvalues, hence oscillations, which represents a challenge to these algorithms. The aim of this work is to illustrate that the algorithms of Section 3.3 can overcome these challenges and, furthermore, can succeed without knowledge of the controlling UEP. Parameters of interest for assessing fault vulnerability are primarily background real and reactive power load, as well as the exponents governing the voltage dependence of these loads, as these are typically uncertain and timevarying in realistic power systems.

4.2.2 Numerical Experiments

The IEEE 39-bus New England power system test case shown in Fig. 4.7 will be used to illustrate the algorithms of Section 3.3. Generators are modeled using a 4th order machine



Figure 4.7: IEEE 39-bus New England power system.

model as given in [SP97], AVR and PSS models are based on the IEEE standard models PSS 1A and Exciter ST1 [IEE82], and the full set of dynamic equations and system parameters are given in [fSCTF15]. These models include the nonsmooth limits on AVR and PSS controller states. The dimensional of the dynamic states for this system is 111. A fault occurs at bus 16, and is cleared after 0.2s. The fault is modeled as a switched, constant shunt reactance with $X_{fault} = 0.001$ p.u.

Many model parameters of the system are of interest for recovery considerations. A background load scaling factor (SF) is introduced which multiplies the active and reactive power background loads at every bus in the network. As background load is time-varying and uncertain, it is a natural choice for assessing system recoverability. An AVR gain SF multiplies the AVR gain for every generator, and helps to capture the impact of controller tuning on system stability. The load active and reactive power are represented by the standard exponential voltage load model. The voltage exponents are set equal for all background loads, with one set of exponents for active power and another set for reactive power. As load dynamics are notoriously difficult to model, these parameters serve to quantize the impact of uncertain load behavior on system recoverability.

4.2.3 Results

The algorithms of Section 3.3 were demonstrated on the test case of Section 4.2.2. The dynamic states appearing in the first and second order trajectory sensitivities used in Section 3.3 were restricted to the generator dynamic states to avoid the possibility for the sensitivities of internal controller states to overshadow the sensitivities of the physical gen-



Figure 4.8: G as a function of background load scaling factor (blue line). The iterations of the one dimensional parameter space algorithm are labeled in order (circles).

erator states. The algorithm for one dimensional parameter space, which finds the closest point on the recovery boundary to an initial given parameter value, was applied with the parameter of interest being the background load SF.

Fig. 4.8 illustrates that the algorithm converged to the recovery boundary in just 2 iterations (3 simulations total). Furthermore, it shows that G is approximately linear near the recovery boundary, both when approaching the recovery boundary for high load and for low load, as predicted by the theory in Section 2.6. Hence, this algorithm rapidly and accurately determines the nearest point on the recovery boundary in one dimensional parameter space.

To observe the approach to instability, note that under stressed conditions the first generator to go unstable is generator 5. Fig. 4.9 shows the relative angle of generator 5 as a function of time for all values of the background load SF which were attained during the algorithm of Fig. 4.8. Observe that as the background load SF approaches the recovery boundary, the initial fluctuations in the relative angle of generator 5 grow larger, indicative of proximity to instability.

Once a point on the recovery boundary has been identified, the continuation method described in Section 3.3.3 can be applied to numerically trace the recovery boundary in two dimensional parameter space. This algorithm was applied for the two dimensional parameter space consisting of the AVR gain SF and the reactive load voltage exponent. The tolerance was set to $\epsilon = 10^{-5}$. Fig. 4.10 shows the set of recovery values, which we term the recovery region (RR), in this parameter space. If the AVR gains are reduced about 20% less than nominal (SF=1.0), the system will recover regardless of the reactive load exponent. For



Figure 4.9: The angle of generator 5 relative to the angle of generator 2 as a function of time for the values of the background load scaling factor that correspond to the iterations of the one dimensional algorithm of Fig. 4.8.

slightly higher AVR gain SFs, a nonempty region outside the set of recovery values emerges. In this regime of the AVR gain SFs, there is a wide range of reactive load voltage exponent values which lie outside the set of recovery values. In particular, as the loads approach constant current and constant impedance loads, for nominal and slightly above nominal AVR gains the system will not be able to recover from the fault. In contrast, as the loads approach constant power loads, the system becomes more resilient and is able to recover. These observations run counter to standard intuition that constant power loads have a more detrimental impact on system stability than constant impedance loads. This behavior is counterintuitive, and serves as an example of how these algorithms have the potential to reveal unexpected dynamic behaviors which would not have been observed otherwise.

To observe the influence of variations in reactive load voltage exponents on system recovery, recall that generator 5 is the first generator to go unstable as a result of the fault. Fig. 4.11 shows the relative angle of generator 5 as a function of time for a fixed AVR gain SF and for several reactive load voltage exponent values. For sufficiently high or sufficiently low exponents, the angle shows smaller initial fluctuations and a more rapid return to synchrony. For exponents near the recovery boundary, initial fluctuations are larger and take a longer time to damp before resynchronization. For exponent values between these recovery boundary values, the system goes unstable and is unable to recover from the fault. Hence, for intermediary voltage exponents, angle fluctuations grow and lead to instability, whereas for high or low load voltage exponents the angles are able to recover from the transient disturbance.



Figure 4.10: The recovery boundary and the set of recovery values (recovery region) in the two dimensional parameter space of AVR gain SF and reactive load voltage exponent. Colored circles indicate the parameter values whose corresponding dynamic behavior is shown in Fig. 4.11.



Figure 4.11: The angle of generator five relative to the angle of generator two as a function of time for an AVR gain SF of 1.054 and a range of reactive load voltage dynamics exponents.



Figure 4.12: The angle of generator five relative to the angle of generator two as a function of time for the solutions of the optimization algorithm applied to the parameter sets S_1 , S_2 , and S_3 , as well as for the nominal parameter values S_0 .

The optimization algorithm described in Section 3.3.7.3 was applied to find the smallest changes in background load, load voltage exponents, and AVR gain SFs that would result in inability to recover from the disturbance. In particular, the following sets of parameters were considered. Let S_1 denote the set of real and reactive background loads, representing uncertainty in background load. Let S_2 consist of S_1 together with the set of real and reactive load voltage exponents, which captures the uncertainty not only in the level of background load but also its dynamics. Define S_3 to be the union of S_2 and the set of AVR gain SFs, for the purpose of illustrating the influence of controller setpoints on system recoverability. Note that S_1 consists of 38 parameters, S_2 of 76 parameters, and S_3 of 86 parameters. The optimization method was applied to each of these sets of parameter values, with $\epsilon = 10^{-5}$ chosen for the constraint $G(p) = \epsilon$. The algorithm converged on all of the parameter sets to within a tolerance of 10^{-6} . Fig. 4.12 shows the dynamic response of the relative angle of generator 5 for the trajectories corresponding to these optimal solutions, as well as for the set of nominal parameter values (denoted by S_0). Note that the relative angles for the optimal solutions are in close agreement with each other, and show larger initial fluctuations than for the nominal parameter values, indicative of proximity to instability.

The goal was to obtain parameter values which minimized the (square of the) Euclidean distance from the nominal parameter values p_0 , while satisfying the constraint $G(p) = \epsilon$. In other words, these are the closest parameter values to p_0 which lie on the recovery boundary. Table 4.1 shows the value of this objective function - the Euclidean distance from p_0 squared - for the optimal solution of each set of parameter values. As additional parameters are

| Parameter Set | Objective Function Value |
|---------------|--------------------------|
| S_1 | 0.0107 |
| S_2 | 0.0079 |
| S_3 | 0.0071 |

Table 4.1: Values of the objective function of Eq. 3.64 for the parameter values on the recovery boundary obtained as the optimal solutions to the higher dimensional parameter space optimization algorithm of Section 3.3.7.3 for the parameter sets S_1 , S_2 , and S_3 .

introduced, more options for parameter values on the recovery boundary become available, some of which may be closer to p_0 than the previous options. Consequently, the objective function value, which measures the distance of the chosen optimal parameter value from p_0 , decreased from S_1 to S_2 , and again from S_2 to S_3 . Overall, for all the chosen combinations of parameters, the higher dimensional parameter space algorithm converged to a parameter value on the recovery boundary which appears to be the closest such value to p_0 .

4.2.4 Conclusion

The algorithms of Section 3.3 were successfully applied to the IEEE 39-bus New England power system to find the closest value of the background load SF that lies on the recovery boundary, to numerically trace the recovery boundary for reactive load voltage exponent and AVR gain SF, and to find the smallest change in background loads, load voltage exponents, and AVR gain SFs that lies on the recovery boundary. Unlike current industry practice, these algorithms are not approximate or conservative, and are able to efficiently compute the recovery boundary to arbitrary precision, or to find the closest point on it to some nominal set of parameter values. Furthermore, these algorithms do not require prior knowledge of the controlling unstable equilibrium point, unlike those applied in Section 4.1. The algorithms reveal an unexpected negative influence of constant impedance loads, and positive influence of constant power loads, on system recovery that would not have been identified otherwise.
CHAPTER V

Conclusions and Future Work

This work introduces a new method for assessing stability of nonlinear systems in the presence of parameter uncertainty. The approach is to consider a parameter dependent family of deterministic vector fields possessing parameter dependent ICs. The set of recovery values consists of the parameter values whose corresponding trajectories converge to a particular SEP representing desired operation. The objective is to determine the recovery boundary: the boundary in parameter space of the set of recovery values. For many applications, such as power systems, this provides a quantitative measure of the margins for safe operation.

Critical parameter values, which are the parameter values whose corresponding ICs lie on the boundary of the RoA of the corresponding SEP, provide an explicit link between parameter space and state space behavior. If the recovery boundary consisted of critical parameter values, this could be exploited for algorithm design. Unfortunately, Example 2.2.1 illustrated that, even when the IC is continuous with respect to parameter, it is possible that there are no critical parameter values on the recovery boundary because the RoA boundary could vary discontinuously with parameter. It is shown that, for a C^1 vector field on a compact Riemannian manifold, if the vector field is Morse-Smale along the RoA boundary, then for sufficiently small changes in parameter values, the RoA boundary varies continuously with respect to parameter and permits a decomposition into a union of stable manifolds of the critical elements it contains. Example 2.3.3 gives a globally Morse-Smale vector field on Euclidean space in which the RoA varies discontinuously under arbitrarily small parameter variations. Hence, the above continuity and decomposition result is obtained for Euclidean space under additional generic assumptions and assumptions about the nonwandering set in a neighborhood of infinity. Using these results, it is shown that the closest point on the recovery boundary to an initial set of recovery values is a critical parameter value. The above results are extended to a class of hybrid systems which are vector fields with C^1 event-selected discontinuities that also have transversal intersection of switching surfaces.

Two collections of algorithms are devised to numerically compute critical parameter values, thereby computing points on the recovery boundary. To motivate the first group of algorithms, it is shown that, under transversality assumptions, for any critical parameter value there exists a controlling invariant set in the RoA boundary and a neighborhood of it such that the time the system trajectory spends inside that neighborhood is continuous over the set of recovery values, and diverges to infinity as the critical parameter value is approached from within the set of recovery values. To motivate the second collection of algorithms, it is shown that, assuming the controlling invariant set is hyperbolic, the supremum over time of the norm of the trajectory sensitivities with respect to parameters is continuous over the set of recovery values, and diverges to infinity as any critical parameter value is approached from within the set of recovery values. Hence, the first class of algorithms computes critical parameter values by maximizing the time the system trajectory spends in the neighborhood of the controlling invariant set, and the second class of algorithms compute critical parameter values by maximizing the supremum over time of the norm of the trajectory sensitivities. Thus, the abstract problem of computing critical parameter values is reduced in both cases to a concrete optimization problem which can be solved efficiently. Moreover, in the latter case it is shown that as parameter values approach the recovery boundary, the infimum over time of the inverse trajectory sensitivities approaches a dot product. This linearity is exploited to improve algorithm design and theoretical guarantees.

Both collections of algorithms require trajectory sensitivities for implementation. Therefore, modular methods are introduced to numerically integrate first, second, and third order trajectory sensitivities for a hybrid dynamical system model by exploiting sparse network structure to reduce computational time and storage requirements. These are used to develop algorithms for finding the unique point on the recovery boundary in the case of one parameter dimension, numerically computing the recovery boundary for two-dimensional parameter space, and finding the nearest point on the recovery boundary to some initial set of recover parameter values for arbitrary parameter dimensions. The algorithms represent significant improvements over current industry state-of-the-art by providing computationally efficient means of considering high dimensional parameter spaces for determining fault recoverability, computing points on the recovery boundary to arbitrary precision (without any conservativeness), and offer the potential for real-time fault vulnerability assessment during power system operations. The algorithms address major drawbacks of historical methods by not requiring energy functions, eliminating the conservativeness of earlier estimation methods, working for a collection of vector fields which exhibit restricted switching and limit behavior, incorporating parameters which influence post-disturbance dynamics, handling cases where the controlling invariant set is not an equilibrium point, and, for the second collection of algorithms, not requiring computation of the controlling invariant set.

The algorithms are applied to determine the margins for recovery of power systems from faults. For the test case of an IEEE 37-bus feeder modified to include induction motor load, the algorithm traced the recovery boundary for motor moment of inertia scaling factor and fault clearing time, and found the nearest point on the recovery boundary in the parameter space of background load. For the case of a modified IEEE 39-bus system of synchronous machines, the algorithm traced the recovery boundary for AVR gain scaling factor and reactive load voltage exponents, uncovering the counterintuitive result that constant power loads improve system resiliency while constant impedance loads reduce it for this case. The nearest point on the recovery boundary was found for parameter spaces including background loads, load voltage exponents, and AVR gain scaling factors, indicating the margins for safe operation.

Considerable challenges remain before industrial use of these algorithms is possible. Of these, one is the scaling of the computational difficulties as the size of the network approaches realistic industrial systems. In particular, computing trajectory sensitivities increases both the runtime and the storage requirements of a single time domain simulation compared to merely integrating the system dynamics. The idea is that, while each individual simulation takes longer when computing trajectory sensitivities, the algorithms presented here can converge in a small number of iterations to the desired recovery boundary, which dramatically reduces the total number of simulations required, thereby reducing the total computational cost. While work here has begun to exploit the structure of the differential-algebraic equations to improve efficiency (see Section 3.1.7), there is more work to be done in this direction. Furthermore, future work will involve demonstrating the ability to apply these algorithms to larger, more realistic power systems. In the process, further testing and refinement of the algorithms on a diverse set of cases should be undertaken.

Several additional questions have arisen as a result of this work. A common theme is attempting to provide theoretical understanding or justification for observations from numerical experiments. Future work should investigate the causes behind the empirical result that the recovery boundary has been smooth in all cases explored thus far, when one might expect to observe nonsmoothness. Another question is related to the linearity of the inverse trajectory sensitivities. Although the theory of Section 2.6 shows that they resemble a dot product for parameter values sufficiently close to the recovery boundary, numerical experiments have indicated that this linearity holds for large ranges in parameter values. Further work is required to elucidate the explanation for this behavior. Finally, the empirical effectiveness of the higher dimensional parameter space optimization methods should be investigated in greater detail.

While the focus of this thesis was on assessing vulnerability of the system to one particular disturbance, such as one particular fault, it is a natural next step to consider vulnerability to a collection of potential disturbances. In theory one could, for a fixed parameter space, perform the algorithms described in this thesis for every possible disturbance in the collection, identifying for each disturbance the closest point on the recovery boundary for that disturbance. This would give, for each disturbance, the radius of a ball in parameter space around the initial parameter values that is entirely contained in the set of recovery values for that disturbance. Choosing the smallest such radius (and recall that the norm used to measure parameter space distance can be defined with any positive semidefinite weighting matrix) yields a ball in parameter space in which recovery is guaranteed from all disturbances in the collection. Unfortunately, large collections of disturbances, as expected in practice, could lead to prohibitively large computation times. One possibility for mitigating this computational cost would be to attempt to identify offline a relatively small subset of disturbances to which the system is most vulnerable. Then, once these most dangerous disturbances have been identified, the above technique could be applied to this small subset during operational time scales to obtain more accurate vulnerability assessment. The most dangerous disturbances could be identified by applying the above technique offline and identifying the disturbances with the smallest radii for which recovery was guaranteed. Another potential means of reducing computational effort in the case of multiple disturbances would be to employ approximation techniques, such as approximating the derivative of the function G using low rank updates, a generalization of the secant approximation in one dimension. For the case of one dimensional parameter space, historical applications of this technique have shown strong performance [NPH02b, NPH02a] and extensions to arbitrary dimensional parameter space may be possible.

There are many other avenues for potential future work as well. In terms of dynamics, these techniques can be used to quantify the impact of nonsmooth controller limits and switching behavior on fault recovery, as well as the impact of replacing conventional generation with distributed resources interfaced via power converters. It appears that, as a consequence of convergence, the trajectory sensitivity method should reveal the unstable eigenvalue of the controlling UEP. In turn, this provides information about the primary mode of instability for a particular fault, and could be useful both for vulnerability assessment and control design. Finally, the methods used here could be applied to develop fault recovery-constrained optimal power flow algorithms, and to aid in controller design by varying controller parameters so as to enlarge the region of attraction in the directions of the most dangerous faults.

APPENDICES

APPENDIX A

Proof of Lemma 2.2.28

Figure 2.10 shows a visual depiction of the disk family in the proof of Lemma 2.3.24, which is an extension of Lemma 2.2.28 to incorporate perturbations to the vector field. Although the notation is slightly different, the figure is useful for following the proof of Lemma 2.2.28 here as well.

The claim is simple in the case that $W^s(X^i)$ has dimension zero. If the dimension of $W^s(X^i)$ is zero, let $\hat{U} = W^s_{\text{loc}}(X^i)$. This satisfies the claim, so assume that the dimension of $W^s(X^i)$ is greater than zero.

We begin by constructing a C^1 continuous disk family along D using the vector field V. Below this disk family will be extended to all of $W^u_{\text{loc}}(X^i)$ using the diffeomorphism \tilde{f} . Let $A = \partial W^u_{\text{loc}}(X^i)$. Then A is a C^1 immersed submanifold of W^u_{loc} of codimension one, and $\bigcup_{t<0} \phi_t(A) = W^u_{\text{loc}}(X^i)$. As the time-t flow restricted to $W^u_{\text{loc}}(X^i)$ is a contraction for any t < 0, for each $y \in D$ there exists a unique $x = x(y) \in A$ and t = t(y) > 0 such that $\phi(t(y), y) = x(y) \in A$. By the tubular neighborhood theorem [Lee13, Theorem 6.24], as discussed in Section 2.2.2, there exists a C^1 continuous family of pairwise disjoint disks $\{D(x)\}_{x\in A}$ centered along A and transverse to $W^u(X^i)$. For each $y \in D$, let D(y) = $\phi_{-t(y)}(D(x(y)))$. Since A is a C^1 immersed submanifold of $W^u(X^i)$ of codimension one, there exists a real vector-valued function s defined on a neighborhood of $W^u_{\text{loc}}(X^i)$ in $W^u(X^i)$ such that s is a C^1 submersion, $A = s^{-1}(0)$, and the codomain of s has dimension one less than the dimension of $W^u(X^i)$. Let $y \in D$ and choose t such that $\phi(t, y) \in A$. Then $s \circ \phi(t, y) = 0$. Furthermore, $\frac{\partial}{\partial t} s \circ \phi(t, y) = ds_{\phi(t,y)}(V(\phi(t, y)))$. But, since s is constant on A, and by dimensionality the kernel of $ds_{\phi(t,y)}$ has dimension one, the kernel of $ds_{\phi(t,y)}$ is exactly $T_{\phi(t,y)}A$. However, since the time-t flow restricted to $W^u_{\text{loc}}(X^i)$ is a contraction, V is transverse to A, so $V_{\phi(t,y)} \notin T_{\phi(t,y)}A$. Thus, $V_{\phi(t,y)}$ is not in the kernel of $ds_{\phi(t,y)}$, so $\frac{\partial}{\partial t}s \circ \phi(t,y) = ds_{\phi(t,y)}(V(\phi(t,y))) \neq 0$. Hence, by the implicit function theorem there exists a neighborhood N' of y in $W_{\text{loc}}^u(X^i)$ and a C^1 function $t : N' \to \mathbb{R}$ such that $y' \in N'$ implies that $s \circ \phi(t(y'), y') = 0$ or, equivalently, $\phi(t(y'), y') \in A$. Thus, the function t = t(y) is C^1 , and $x = x(y) = \phi(t(y), y)$ is also C^1 since ϕ and t(y) are. Therefore, by the construction of the disk family above, the disk family $\{D(y)\}_{y\in D}$ is C^1 continuous. As D is compact, we may shrink $W_{\text{loc}}^u(X^i)$, shrink the disk family $\{D(x)\}_{x\in A}$, and choose N an open neighborhood in M such that $D \subset N \subset D_{\epsilon}$ and for each $y \in D$, $D(y) \subset N$.

Next the C^1 continuous disk family along $W^u_{\text{loc}}(X^i)$ is constructed for \tilde{f} by backward iteration of the disk family above, and it is shown to contain an open neighborhood of X^i . If X^i is an equilibrium point, let $x_0 = X^i$ and let S = M. If X^i is a periodic orbit, let $x_0 \in X^i$ such that \tilde{f} is the first return map for a cross section S centered at x_0 . Let Y be a neighborhood of x_0 in S such that Y is diffeomorphic to the C^1 local coordinates of [JPdM82, p. 80-81] so that we can write $Y = W_{loc}^s(x_0) \times W_{loc}^u(x_0)$. In these coordinates, we can have $Y \subset \mathbb{R}^n = E^s \oplus E^u$, where E^s and E^u are the stable and unstable eigenspaces of $d\tilde{f}_{x_0}$, respectively, $W^s_{\text{loc}}(x_0) \subset E^s$, and $W^u_{\text{loc}}(x_0) \subset E^u$. Next we extend the disk family ${D(x)}_{x\in D}$ to all of $W^u_{\text{loc}}(x_0)$. For each $x \in W^u_{\text{loc}}(x_0) - {x_0}$, let n(x) be the smallest n > 0such that $\tilde{f}^{n(x)}(x) \in D$. Note that n(x) is finite since such an intersection always exists by the construction of D. Let $x \in W^u_{\text{loc}}(x_0) - \{x_0\}$. Let $D^0(x) = D(\tilde{f}^{n(x)}(x))$ and for each $m \in$ 1,2,..., n(x), let $D^m(x)$ be the connected component of $(\tilde{f})^{-1}(D^{m-1}(x)) \cap Y$ that contains $\tilde{f}^{n(x)-m}(x)$ where we set $\tilde{f}^{0}(x) = x$. Then let $D(x) = D^{n(x)}(x)$. Let $D(x_{0}) = W^{s}_{loc}(x_{0})$. This gives a family of connected C^1 disks centered along $W^u_{\text{loc}}(x_0)$. Let B_0 and B_I be the outer and inner topological boundaries of D, respectively. Let $y \in W^u_{\text{loc}}(x_0) - \{x_0\}$. Then $\tilde{f}^{n(y)} \in D$. Since the family $\{D(x)\}_{x\in D}$ is C^1 continuous, $(\tilde{f})^{-1}$ maps B_0 onto B_I , and for $x \in B_0(\tilde{f})^{-1}(D(x)) = D((\tilde{f})^{-1}(x))$, the disk family $\{D(x)\}_{x \in W^u_{loc}(x_0)}$ is C^1 continuous on a neighborhood of $\tilde{f}^{n(y)}$. Hence, applying the diffeomorphism $\tilde{f}^{-n(y)}$, by the construction of the disk family above this implies that $\{D(x)\}_{x \in W^u_{loc}(x_0)}$ is C^1 continuous on a neighborhood of y. Since the family $\{D(x)\}_{x\in D}$ is C^1 continuous and transverse to $W^u_{\text{loc}}(x_0)$, by the Inclination Lemma [Pal69], the family $\{D(\tilde{f}^n(x))\}_{x\in D}$ converges uniformly in the C^1 topology to $W^s_{\rm loc}(x_0)$ as $n \to \infty$. Hence, $\{D(x)\}_{x \in W^u_{\rm loc}(x_0) - \{x_0\}}$ converges uniformly to $W^s_{\rm loc}(x_0)$ as $x \to x_0$. This shows that the family $\{D(x)\}_{x \in W^u_{\text{loc}}(x_0)}$ defined above is C^1 continuous at x_0 . Thus, $\{D(x)\}_{x\in W^u_{loc}(x_0)}$ is a C^1 continuous family of disks transverse to $W^u_{loc}(x_0)$ and such that $x \neq y$ with $x, y \in W^u_{\text{loc}}(x_0)$ implies that $D(x) \cap D(y) = \emptyset$. Hence, there exists a C^1 injective function $\tilde{F}: B^s_r(0) \times W^u_{\text{loc}}(x_0) \to M$ such that $\tilde{F}(B^s_r(0) \times \{x\}) = D(x)$ for every $x \in W^u_{\text{loc}}(x_0)$. Thus, since \tilde{F} is a continuous injection between manifolds of the same dimension, by invariance of domain [Hat01, Theorem 2B.3] $\hat{U} = \tilde{F}(\text{int } B^s_r(0) \times \text{int } W^u_{\text{loc}}(x_0))$ is an open neighborhood of int $W^u_{\text{loc}}(x_0)$ in M. By construction, for every $x \in \hat{U} - W^s_{\text{loc}}(x_0)$, its forward orbit intersects D(y) for some $y \in D$ in finite positive time. But by construction, $D(y) \subset N$ for all $y \in D$. Hence, $\bigcup_{t<0} \phi_t(N) \bigcup W^s_{\text{loc}}(x_0)$ contains \hat{U} an open neighborhood of X^i .

APPENDIX B

Proof of Lemma 2.2.37

Figure 2.10 shows a visual depiction of the disk family in the proof of Lemma 2.3.24, which is a slight extension of Lemma 2.2.37. Although the notation is slightly different, the figure is useful for following the proof of Lemma 2.2.37 here as well.

Let \tilde{f}_p denote the time-one flow (if $X_{p_0}^i$ is an equilibrium point) or a C^1 first return map (if $X_{p_0}^i$ is a periodic orbit) corresponding to parameter value p. First note that $W_{loc}^s(X^i(p))$, $W^{u}_{\text{loc}}(X^{i}(p))$, the exponential maps, and the flows are C^{1} continuous with respect to parameter value, so that p close to p_0 implies that the disk family $\{D(x)\}_{x\in D_p}$ of the perturbed vector field will be uniformly C^1 -close to that of the original. Hence, $(D_p)_{\epsilon}$ will be C^1 -close to $(D_{p_0})_{\epsilon}$. So, J sufficiently small implies that there exists N' open such that for $p \in J$, $D_p \subset N' \subset (D_p)_{\epsilon}$. Let $x_0(p)$ be defined in the natural way so that it is C^1 with respect to parameter, $x_0(p) = X^i(p)$ if $X^i(p)$ is an equilibrium point, and if $X^i(p)$ is a periodic orbit then $x_0(p) \in X^i(p)$. Construct the full disk family $\{D(x)\}_{x \in W^u_{loc}(x_0(p))}$ for the perturbed vector field as in the proof of Lemma 2.2.28. In the proof of the Inclination Lemma, \hat{g} sufficiently C^1 -close to \hat{f} implies that the same bounds on \hat{g} and its partial derivatives will hold as for f. Hence, the uniform bounds on the inclinations (slopes) obtained for the transverse disk family $\{D(x)\}_{x\in D_{p_0}}$ can also be taken to apply to $\{D(x)\}_{x\in D_p}$ for p sufficiently close to p_0 . In particular, there exists $Z_N > 0$ such that $n \ge Z_N$ implies that for any $x \in D_{p_0}$, $D(\tilde{f}_{p_0}^{-n}(x))$ is δC^1 -close to $W^s_{\text{loc}}(x_0(p_0))$ and for any $x \in D_p$, $D(\tilde{f}_p^{-n}(x))$ is δC^1 -close to $W^s_{\text{loc}}(x_0(p))$. This implies that for every $x \in D_{p_0}$ and $x' \in D_p$, for every $n \ge Z_N$, $D(\tilde{f}_{p_0}^{-n}(x))$ is $3\delta C^1$ close to $D(\tilde{f}_p^{-n}(x'))$. As $\{D(x)\}_{x\in D_{p_0}}$ and $\{D(x)\}_{x\in D_p}$ are uniformly C^1 -close and Z_N is finite, for p sufficiently close to p_0 we may have that the two disk families $\{D(\tilde{f}_{p_0}^{-n}(x))\}_{x\in D_{p_0}}$ and $\{D(\tilde{f}_p^{-n}(x))\}_{x\in D_p}$ are uniformly $3\delta C^1$ -close for $n \leq Z_N$. Hence, combining the above, we have that the disk families $\{D(x)\}_{x \in W^u_{\text{loc}}(x_0(p_0))}$ and $\{D(x)\}_{x \in W^u_{\text{loc}}(x_0(p))}$ are uniformly $3\delta C^1$ -close. Constructing \tilde{F}_p for \tilde{f}_p analogously to the construction in Lemma 2.2.28 of \tilde{F} , this implies that $\tilde{F}_p(\text{int } B^s_r(0) \times \text{int } W^u_{\text{loc}}(x_0(p)))$ has Hausdorff distance no greater than 3δ from $\tilde{F}_{p_0}(\text{int } B^s_r(0) \times \text{int } W^u_{\text{loc}}(x_0(p_0)))$. In particular, this implies that $\delta > 0$ and \hat{U} can be chosen sufficiently small, such that for J sufficiently small, $p \in J$ implies that $\tilde{F}_p(\text{int } B^s_r(0) \times \text{int } W^u_{\text{loc}}(x_0(p))) \supset \hat{U}$. Hence, for every $x \in \hat{U} - W^s_{\text{loc}}(x_0(p))$, the forward orbit of x intersects N' in finite time.

APPENDIX C

Proof of Lemma 2.3.24

Figure 2.10 shows a visual depiction of the disk family in the proof of Lemma 2.3.24 here. The majority of the proof, including the construction of a disk family and its persistence under perturbation, follows from the proofs of Lemma 2.2.28 and Lemma 2.2.37 in Appendices A-B.

We construct $\hat{U} \subset U'_i$ such that the backwards orbit of any point in \hat{U} under \hat{g} sufficiently close to \hat{f} remains in \hat{U} at least until it enters N_i . Choose $r' \in (0, r)$ such that the closure of $\hat{F}(\operatorname{int} B^u_{r'}(0) \times \operatorname{int} W^s_{\operatorname{loc}}(x_0))$ is contained in U'_i . Then there exists $\delta > 0$ such that the distance between their boundaries is greater than δ . The argument of Appendix B shows that for γ sufficiently small, $\hat{F}'(\operatorname{int} B^u_{r'}(0) \times \operatorname{int} W^{s'}_{\operatorname{loc}}(x'_0)) \subset U'_i$ since its Hausdorff distance from $\hat{F}(\operatorname{int} B^u_{r'}(0) \times \operatorname{int} W^s_{\operatorname{loc}}(x_0))$ can be made less than δ . Repeating that argument again, one can choose $\hat{U} \subset \hat{F}(\operatorname{int} B^u_{r'}(0) \times \operatorname{int} W^s_{\operatorname{loc}}(x_0))$ open such that $x'_0 \in \hat{U}$ for γ sufficiently small. As $\hat{U} \subset \hat{F}'(\operatorname{int} B^u_{r'}(0) \times \operatorname{int} W^{s'}_{\operatorname{loc}}(x'_0)) \subset U'_i$, this implies that for every point in \hat{U} its backwards orbit is contained in U'_i at least until it enters N_i .

Finally, we return from diffeomorphisms to vector fields. Let W be an open neighborhood of V_0 such that $V \in W$ implies that g, constructed for V analogously to the construction of \hat{f} for V_0 above, is γC^1 -close to \hat{f} . If $X^i(V_0)$ is an equilibrium point, this completes the proof. If $X^i(V_0)$ is a periodic orbit, let T > 0 finite be such that for every point $x \in S$, the first return time of x is less than T. Redefine $U_i := \bigcup_{t \in (-T,T)} \phi_t(U_i)$ and $\hat{U} := \bigcup_{t \in (-T,T)} \phi_t(\hat{U})$. Then U_i is open in M and contains $W^s_{\text{loc}}(X^i(V_0))$.

APPENDIX D

Derivation of Trajectory Sensitivities at Discrete Events

Differentiate Eqs. 3.3 with respect to the components of the initial condition of the dynamic states (components of x(0)). For first derivatives, differentiate with respect to x_j to obtain

$$\Phi_{ij}^- = \Phi_{ij} + f_i^- \tau_j \tag{D.1}$$

$$0 = s_{im}\Phi_{mj}^- + s_{i\beta}\Psi_{\beta j}^- \tag{D.2}$$

$$0 = g_{\alpha m}^{-} \Phi_{mj}^{-} + g_{\alpha \beta}^{-} \Psi_{\beta j}^{-}$$
(D.3)

$$\Phi_{ij}^+ = h_{im}\Phi_{mj}^- + h_{i\beta}\Psi_{\beta j}^- \tag{D.4}$$

$$0 = g_{\alpha m}^{+} \Phi_{mj}^{+} + g_{\alpha \beta}^{+} \Psi_{\beta j}^{+}.$$
 (D.5)

Then Eq. D.1 is equal to Eq. 3.17. Substituting the expression for Φ_{ij}^- from Eq. D.1 into Eqs. D.2-D.3 and rearranging yields Eq. 3.16. Combining Eqs. D.4-D.5 and rearranging yields Eq. 3.18. This gives the first order trajectory sensitivity equations.

For second order, differentiate Eqs. D.1-D.5 with respect to x_k to get

$$\Phi_{ijk}^{-} = \Phi_{ijk} + f_{im}^{-} \Phi_{mk} \tau_j + f_{i\alpha}^{-} \Psi_{\alpha k} \tau_j + f_i^{-} \tau_{jk}$$
(D.6)
$$0 = s_{imn} \Phi_{nk}^{-} \Phi_{mj}^{-} + s_{im\alpha} \Psi_{\alpha k}^{-} \Phi_{mj}^{-} + s_{im} \Phi_{mjk}^{-} + s_{i\alpha m} \Phi_{mk}^{-} \Psi_{\alpha j}^{-} + s_{i\alpha \beta} \Psi_{\beta k}^{-} \Psi_{\alpha j}^{-} + s_{i\alpha} \Psi_{\alpha jk}^{-}$$
(D.7)

$$0 = g_{\alpha m n}^{-} \Phi_{n k}^{-} \Phi_{m j}^{-} + g_{\alpha m \beta}^{-} \Psi_{\beta k}^{-} \Phi_{m j}^{-} + g_{\alpha m}^{-} \Phi_{m j k}^{-} + g_{\alpha \beta m}^{-} \Phi_{m k}^{-} \Psi_{\beta j}^{-} + g_{\alpha \beta \gamma}^{-} \Psi_{\gamma k}^{-} \Psi_{\beta j}^{-} + g_{\alpha \beta}^{-} \Psi_{\beta j k}^{-}$$
(D.8)

$$\Phi_{ijk}^{+} = h_{imn}\Phi_{nk}^{-}\Phi_{mj}^{-} + h_{im\alpha}\Psi_{\alpha k}^{-}\Phi_{mj}^{-} + h_{im}\Phi_{mjk}^{-} + h_{i\alpha m}\Phi_{mk}^{-}\Psi_{\alpha j}^{-} + h_{i\alpha\beta}\Psi_{\beta k}^{-}\Psi_{\alpha j}^{-} + h_{i\alpha}\Psi_{\alpha jk}^{-}$$
(D.9)

$$0 = g_{\alpha m n}^{+} \Phi_{nk}^{+} \Phi_{mj}^{+} + g_{\alpha m \beta}^{+} \Psi_{\beta k}^{+} \Phi_{mj}^{+} + g_{\alpha m}^{+} \Phi_{mjk}^{+} + g_{\alpha \beta m}^{+} \Phi_{mk}^{+} \Psi_{\beta j}^{+} + g_{\alpha \beta \gamma}^{+} \Psi_{\gamma k}^{+} \Psi_{\beta j}^{+} + g_{\alpha \beta}^{+} \Psi_{\beta jk}^{+}.$$
(D.10)

Then Eq. D.6 is equal to Eq. 3.20. Substituting the expression for Φ_{ijk}^- from Eq. D.6 into Eqs. D.7-D.8 and rearranging yields Eq. 3.19. Combining Eqs. D.9-D.10 and rearranging yields Eq. 3.21. This gives the second order trajectory sensitivity equations.

For third order, differentiate Eqs. D.6-D.10 with respect to x_k to get

$$\begin{split} \Phi_{ijkl}^{-} &= \Phi_{ijkl} + f_{imn}^{-} \Phi_{nl} \Phi_{mk} \tau_{j} + f_{im\alpha}^{-} \Psi_{\alpha l} \Phi_{mk} \tau_{j} + f_{im}^{-} \Phi_{mkl} \tau_{j} + f_{im}^{-} \Phi_{mk} \tau_{jl} + f_{i\alpha}^{-} \Phi_{mk} \tau_{jl} + f_{i\alpha}^{-} \Phi_{mk} \tau_{jl} + f_{i\alpha}^{-} \Phi_{mk} \tau_{jl} + f_{i\alpha}^{-} \Phi_{mk} \tau_{jk} + f_{i\alpha}^{-} \Psi_{\alpha l} \tau_{jk} + f_{i}^{-} \tau_{jkl} \qquad (D.11) \\ 0 &= s_{imno} \Phi_{ol}^{-} \Phi_{nk}^{-} \Phi_{mj}^{-} + s_{imn\alpha} \Psi_{\alpha l}^{-} \Phi_{nk}^{-} \Phi_{mj}^{-} + s_{imn\alpha} \Phi_{nkl}^{-} \Phi_{mj}^{-} + s_{imn\alpha} \Phi_{nk}^{-} \Phi_{mj}^{-} \\ &+ s_{im\alpha n} \Phi_{nl}^{-} \Psi_{\alpha k}^{-} \Phi_{mj}^{-} + s_{im\alpha \beta} \Psi_{\beta l}^{-} \Psi_{\alpha k}^{-} \Phi_{mj}^{-} + s_{im\alpha \beta} \Psi_{\alpha k}^{-} \Phi_{mj}^{-} + s_{im\alpha \beta} \Psi_{\alpha l}^{-} \Phi_{mj}^{-} + s_{im\alpha \beta} \Psi_{\alpha l}^{-} \Phi_{mj}^{-} \\ &+ s_{im\alpha} \Phi_{nl}^{-} \Psi_{\alpha k}^{-} \Phi_{mj}^{-} + s_{i\alpha m\alpha \beta} \Psi_{\beta l}^{-} \Psi_{\alpha k}^{-} \Phi_{mj}^{-} + s_{i\alpha m\beta \beta} \Psi_{\beta l}^{-} \Phi_{mk}^{-} \Psi_{\alpha j}^{-} + s_{i\alpha m\beta \beta} \Psi_{\beta l}^{-} \Phi_{mk}^{-} \Psi_{\alpha j}^{-} \\ &+ s_{i\alpha m} \Phi_{mk}^{-} \Psi_{\alpha jl}^{-} + s_{i\alpha \beta m} \Phi_{ml}^{-} \Psi_{\beta k}^{-} \Psi_{\alpha j}^{-} + s_{i\alpha \beta \beta} \Psi_{\beta l}^{-} \Psi_{\alpha j}^{-} + s_{i\alpha \beta \beta} \Psi_{\beta k}^{-} \Psi_{\alpha j}^{-} \\ &+ s_{i\alpha \beta} \Psi_{\beta k}^{-} \Psi_{\alpha jl}^{-} + s_{i\alpha m} \Phi_{ml}^{-} \Psi_{\alpha jk}^{-} + s_{i\alpha \beta \beta} \Psi_{\beta l}^{-} \Psi_{\alpha jk}^{-} + s_{i\alpha \beta} \Psi_{\beta l}^{-} \Psi_{\alpha jk}^{-} \\ &+ s_{i\alpha \beta} \Psi_{\beta k}^{-} \Psi_{\alpha jl}^{-} + s_{i\alpha m} \Phi_{ml}^{-} \Psi_{\alpha jk}^{-} + s_{i\alpha \beta} \Psi_{\beta l}^{-} \Psi_{\alpha jk}^{-} + s_{i\alpha \beta} \Psi_{\beta l}^{-} \Psi_{\alpha jk}^{-} \\ &+ s_{i\alpha \beta} \Psi_{\beta k}^{-} \Psi_{\alpha jl}^{-} + s_{i\alpha m} \Phi_{ml}^{-} \Psi_{\alpha jk}^{-} + s_{i\alpha \beta} \Psi_{\beta l}^{-} \Psi_{\alpha jk}^{-} + s_{i\alpha \beta} \Psi_{\beta l}^{-} \Psi_{\alpha jk}^{-} \\ &+ s_{i\alpha \beta} \Psi_{\beta k}^{-} \Psi_{\alpha jl}^{-} + s_{i\alpha m} \Phi_{ml}^{-} \Psi_{\alpha jk}^{-} + s_{i\alpha \beta} \Psi_{\beta l}^{-} \Psi_{\alpha jk}^{-} + s_{i\alpha \beta} \Psi_{\beta k}^{-} \Psi_{\alpha jk}^{-} \\ &+ s_{i\alpha \beta} \Psi_{\beta k}^{-} \Psi_{\alpha jl}^{-} + s_{i\alpha \beta} \Psi_{\beta l}^{-} \Psi_{\alpha jk}^{-} + s_{i\alpha \beta} \Psi_{\alpha jk}^{-} \\ &+ s_{i\alpha \beta} \Psi_{\beta k}^{-} \Psi_{\alpha jl}^{-} + s_{i\alpha \beta} \Psi_{\beta l}^{-} \Psi_{\alpha jk}^{-} + s_{i\alpha \beta} \Psi_{\beta l}^{-} \Psi_{\alpha jk}^{-} \\ &+ s_{i\alpha \beta} \Psi_{\beta k}^{-} \Psi_{\alpha jl}^{-} + s_{i\alpha \beta} \Psi_{\beta l}^{-} \Psi_{\alpha jk}^{-} + s_{i\alpha \beta} \Psi_{\beta l}^{-} \Psi_{\alpha jk}^{-} \\ &+ s_{i\alpha \beta} \Psi_{\beta k}^{-} \Psi_{\alpha jl}^{-} + s_{i\alpha \beta} \Psi_{\beta l}^{-} \Psi_{\alpha jk}^{-} + s_{i\alpha \beta} \Psi_{\beta l}^{-} \Psi_{\alpha jk}$$

Then Eq. D.11 is equal to Eq. 3.23. Substituting the expression for Φ_{ijkl}^- from Eq. D.11 into Eqs. D.12-D.13 and rearranging yields Eq. 3.22. Combining Eqs. D.14-D.15 and rearranging yields Eq. 3.24. This gives the third order trajectory sensitivity equations.

BIBLIOGRAPHY

BIBLIOGRAPHY

- [APV79] T. Athay, R. Podmore, and S. Virmani, A practical method for the direct analysis of transient stability, IEEE Transactions on Power Apparatus and Systems 98 (1979), no. 2, 573–584.
- [Ayl58] P. D. Aylett, The energy-integral criterion of transient stability limits of power systems, Proceedings of the IEEE **105** (1958), no. 8, 527–536.
- [BBCK08] M. Di Bernardo, C. Budd, A. Champneys, and P. Kowalczyk, *Piecewise-smooth dynamical systems, theory, and applications*, Springer-Verlag, 2008.
- [BH81] A. R. Bergen and D. J. Hill, A structure preserving model for power system stability analysis, IEEE Transactions on Power Apparatus and Systems **100** (1981), no. 1, 25–35.
- [BHD86] A. R. Bergen, D. J. Hill, and C. L. DeMarco, Lyapunov function for multimachine power systems with generator flux decay and voltage dependent loads, Electrical Power and Energy Systems 8 (1986), no. 1, 2–10.
- [BPRP+84] Anjan Bose, M. A. Pai, M. Ribbens-Pavella, A. A. Fouad, Tom Athay, P. Kundur, and J. V. Mitsche, *Application of direct methods to transient stability anal*ysis of power systems, IEEE Transactions on Power Apparatus and Systems 103 (1984), no. 7, 1629–1636.
- [BSKR15] Samuel A. Burden, S. Shankar Sastry, Daniel E. Koditschek, and Shai Revzen, Event-selected vector field discontinuities yield piecewise-differentiable flows, arXiv: hep-th/1407.1775.
- [CA15] Hsiao-Dong Chiang and Luís F. C. Alberto, *Stability regions of nonlinear dynamical systems: Theory, estimation, and applications*, Cambridge University Press, 2015.
- [CC95] Hsiao-Dong Chiang and Chia-Chi Chu, Theoretical foundation of the bcu method for direct stability analysis of network-reduction power system models with small transfer conductances, IEEE Transactions on Circuits and Systems - I: Fundamental Theory and Applications 42 (1995), no. 5, 252–265.

- [Chi11] Hsiao-Dong Chiang, Direct methods for stability analysis of electric power systems, John Wiley & Sons, Inc., 2011.
- [CHW88] Hsiao-Dong Chiang, Morris W. Hirsch, and Felix F. Wu, Stability regions of nonlinear autonomous dynamical systems, IEEE Transactions on Automatic Control 33 (1988), no. 1, 16–27.
- [CWV88] Hsiao-Dong Chiang, Felix Wu, and Pravin P. Varaiya, Foundations of the potential energy boundary surface method for power system transient stability analysis, IEEE Transactions on Circuits and Systems 35 (1988), no. 6, 712–728.
- [DH06] V. Donde and I. A. Hiskens, Shooting methods for locating grazing phenomena in hybrid systems, International Journal of Bifurcation and Chaos 16 (2006), no. 3, 671–692.
- [Dob03] I. Dobson, Distance to bifurcation in multidimensional parameter space: Margin sensitivity and closest bifurcations, in Bifurcation Control, Theory and Application **293** (2003), 49–66.
- [EAN66] A. H. El-Abiad and K. Nagappan, Transient stability regions of multimachine power systems, IEEE Transactions on Power Apparatus and Systems 85 (1966), no. 2, 169–178.
- [Eng89] Ryszard Engelking, *General topology*, Sigma Series in Pure Mathematics, vol. 6, Heldermann Verlag, 1989.
- [FH16] Michael W. Fisher and Ian A. Hiskens, Numerical computation of parameterspace stability/instability partitions for induction motor stalling, International Federation of Automatic Control - PapersOnLine **49** (2016), no. 27, 250–255.
- [FH17] _____, Parametric dependence of large disturbance response and relationship to stability boundary, Proc. IEEE 54th Conference on Decision and Control, 2017, pp. 1821–1827.
- [fSCTF15] Benchmark Systems for Stability Controls Task Force, *Benchmark systems for small-signal stability analysis and control*, Tech. report, IEEE Power and Energy Society, 2015.
- [FVO84] A. A. Fouad, Vijay Vittal, and Tae Kyoo Oh, Critical energy for direct transient stability assessment of a multimachine power system, IEEE Transactions on Power Apparatus and Systems 103 (1984), no. 8, 2199–2206.
- [GH83] John Guckenheimer and Philip Holmes, Nonlinear oscillations, dynamical systems, and bifurcations of vector fields, Applied Mathematical Sciences, vol. 42, Springer, 1983.

| [GH19] | Sijia Geng and Ian A. Hiskens, Second-order trajectory sensitivity analysis of hybrid systems, IEEE Transactions on Circuits and Systems I 66 (2019), no. 5, 1922–1934. |
|----------|---|
| [Gle66] | G. E. Gless, Direct method of lyapunov applied to transient power system stability, IEEE Transactions on Power Apparatus and Systems 85 (1966), no. 2, 159–168. |
| [Hat01] | Allen Hatcher, Algebraic topology, Cambridge University Press, 2001. |
| [Hau57] | Felix Hausdorff, Set theory, Chelsea Publishing Company, 1957. |
| [Hir76] | Morris W. Hirsch, <i>Differential topology</i> , Graduate Texts in Mathematics, vol. 33, Springer-Verlag, 1976. |
| [His04] | Ian A. Hiskens, <i>Power system modeling for inverse problems</i> , IEEE Transactions on Circuits and Systems I 51 (2004), no. 3, 539–551. |
| [HP00] | Ian A. Hiskens and M. A. Pai, <i>Trajectory sensitivity analysis of hybrid systems</i> , IEEE Transactions on Circuits and Systems - I: Fundamental Theory and Applications 47 (2000), no. 2, 204–220. |
| [HPS77] | M. W. Hirsch, C. C. Pugh, and M. Shub, <i>Invariant manifolds</i> , Lecture Notes in Mathematics, vol. 583, Springer-Verlag, 1977. |
| [HS74] | Morris W. Hirsch and Stephen Smale, <i>Differential equations, dynamical systems, and linear algebra</i> , Pure and Applied Mathematics: A Series of Monographs and Textbooks, Academic Press, Inc., 1974. |
| [IEE82] | Tech. report, IEEE Standard 421.5, 1982. |
| [JPdM82] | Jr. Jacob Palis and Welington de Melo, <i>Geometric theory of dynamical systems</i> , Springer-Verlag, 1982. |
| [KH99] | Anatole Katok and Boris Hasselblatt, Introduction to the modern theory of dy- namical systems, Encyclopedia of Mathematics and its Applications, vol. 54, Cambridge University Press, 1999. |
| [KOH78] | N. Kakimoto, Y. Ohsawa, and M. Hayashi, <i>Transient stability analysis of electric power systems via lure type lyapunov function, part i. new critical value for transient stability</i> , Transactions IEE of Japan 98 (1978), no. 5, 62–71. |
| [Kup63] | Ivan Kupka, <i>Contribution à la théorie des champs génériques</i> , Contributions to Differential Equations (1963), no. 2, 457–484. |
| [Kur68] | K. Kuratowski, Topology, volume ii, Academic Press, Inc., 1968. |

- [Lee13] John M. Lee, *Introduction to smooth manifolds*, 2 ed., Graduate Texts in Mathematics, Springer, 2013.
- [LLM⁺95] A. Llamas, J. De La Ree Lopez, L. Mili, A. G. Phadke, and J. S. Thorp, *Clar-ifications of the bcu method for transient stability analysis*, IEEE Transactions on Power Systems 10 (1995), no. 1, 210–219.
- [Mag47] P. C. Magnusson, *The transient-energy method of calculating stability*, Transactions of the American Institute of Electrical Engineers **66** (1947), no. 1, 747–755.
- [MDL⁺12] Yuri V. Makarov, Pengwei Du, Shuai Lu, Tony B. Nguyen, Xinxin Guo, J. W. Burns, Jim F. Gronquist, and M. A. Pai, *PMU-based wide-area security assessment: Concept, method, and implementation*, IEEE Transactions on Smart Grid 3 (2012), no. 3, 1325–1332.
- [MWTP97] James D. McCalley, Shimo Wang, Roger T. Treinen, and Alex D. Papalexopoulos, Security boundary visualization for systems operation, IEEE Transactions on Power Systems 12 (1997), no. 2, 940–947.
- [NER09] NERC, Fault-induced delayed voltage recovery (FIDVR), Tech. report, NERC, 2009.
- [NM84] N. Narasimhamurthi and M. R. Musavi, A general energy function for transient stability analysis of power systems, IEEE Transactions on Circuits and Systems **31** (1984), no. 7, 637–645.
- [NPH02a] T. B. Nguyen, M. A. Pai, and I. A. Hiskens, Sensitivity approaches for direct computation of critical parameters in a power system, Electrical Power and Energy Systems 24 (2002), 337–343.
- [NPH02b] Tony B. Nguyen, M. A. Pai, and I. A. Hiskens, *Computation of critical values of* parameters in power systems using trajectory sensitivities, 14th Power Systems Computation Conference (PSCC) (2002), 1–6.
- [Pal69] Jacob Palis, On morse-smale dynamical systems, Topology 8 (1969), no. 4, 385– 405.
- [PEA75] F. S. Prabhakara and A. H. El-Abiad, A simplified determination of transient stability regions for lyapunov methods, IEEE Transactions on Power Apparatus and Systems 94 (1975), no. 2, 672–689.
- [Pei66] M. M. Peixoto, On an approximation theorem of kupka and smale, Journal of Differential Equations (1966), no. 3, 214–227.
- [PL99] Fernando Paganini and Bernard C. Lesieutre, Generic properties, one-parameter deformations, and the bcu method, IEEE Transactions on Circuits and Systems I 46 (1999), no. 6, 760–763.

| [PR83] | Charles C. Pugh and Clark Robinson, The c^1 closing lemma, including hamil- tonians, Ergodic Theory and Dynamical Systems (1983), no. 3, 311–312. |
|---------|---|
| [PS68] | Jacob Palis and Stephen Smale, <i>Structural stability theorems</i> , Global Analysis (Shiing-Shen Chern and Stephen Smale, eds.), vol. 14, American Mathematical Society, 1968, pp. 223–231. |
| [Pug67] | Charles C. Pugh, An improved closing lemma and a general density theorem, American Journal of Mathematics 89 (1967), no. 4, 1010–1021. |
| [RH12] | Sandip Roy and Ian Hiskens, Inverse-affine dependence of recovery-time sensi- tivities on critical disturbance parameters: A nonlinear dynamics explanation, American Control Conference (ACC) (2012), 4452–4456. |
| [Sch12] | S. Scholtes, Introduction to piecewise differentiable equations, Springer-Verlag, 2012. |
| [Sel85] | George R. Sell, <i>Smooth linearization near a fixed point</i> , American Journal of Mathematics 107 (1985), no. 5, 1035–1091. |
| [Sma60] | Stephen Smale, <i>Morse inequalities for a dynamical system</i> , Bulletin of the American Mathematical Society 66 (1960), no. 1, 43–49. |
| [Sma62] | , Stable manifolds for differential equations and diffeomorphisms, Topolo- gia differenziale (E. Vesentini, ed.), 1962, pp. 93–126. |
| [Sma63] | Steven Smale, A structurally stable differentiable homeomorphism with an in- finite number of periodic points, Proc. of the Internation Symposium on Non- Linear Oscillators, 1963, pp. 365–366. |
| [Sma67] | Stephen Smale, <i>Differentiable dynamical systems</i> , Bulletin of the American Mathematical Society 73 (1967), no. 6, 747–817. |
| [SP97] | Peter W. Sauer and M. A. Pai, Power system dynamics and stability, 1997. |
| [TAV85] | N. Tsolas, A. Arapostathis, and P. Varaiya, A structure preserving energy func- tion for power system transient stability analysis, IEEE Transactions on Circuits and Systems 32 (1985), no. 10, 1041–1049. |
| [TS72] | Carlos J. Tavora and Otto J. M. Smith, <i>Stability analysis of power systems</i> , IEEE Transactions on Power Apparatus and Systems 91 (1972), no. 3, 1138–1144. |
| [TVK96] | Roger T. Treinen, Vijay Vittal, and Wolfgang Kliemann, An improved technique to determine the controlling unstable equilibrium point in a power system, IEEE Transactions on Circuits and Systems - I: Fundamental Theory and Applications 43 (1996), no. 4, 313–323. |
| | |

- [VJSZ94] V. Venkatasubramanian, X. Jiang, H. Schattler, and J. Zaborszky, Current status of the taxonomy theory of large power system dynamics: Dae systems with hard limits, Proc. Bulk Power System Voltage Phenomena III, 1994, pp. 15–103.
- [ZHZL88] John Zaborszky, Garng Huang, Baohua Zheng, and Tin-Chi Leung, On the phase portrait of a class of large nonlinear dynamic systems such as the power system, IEEE Transactions on Automatic Control **33** (1988), no. 1, 4–15.